

Title	The design, development, implementation and evaluation of the Gaelic4Girls intervention
Authors	Farmer, Orlagh
Publication date	2020-06-26
Original Citation	Farmer, O. 2020. The design, development, implementation and evaluation of the Gaelic4Girls intervention. PhD Thesis, University College Cork.
Type of publication	Doctoral thesis
Rights	© 2020, Orlagh Farmer https://creativecommons.org/ publicdomain/zero/1.0/
Download date	2024-12-25 03:13:10
Item downloaded from	https://hdl.handle.net/10468/11919



University College Cork, Ireland Coláiste na hOllscoile Corcaigh Ollscoil na hÉireann, Corcaigh National University of Ireland, Cork



# The design, development, implementation and evaluation of the Gaelic4Girls intervention

Thesis presented by

## **Orlagh Farmer**

B.Ed.

https://orcid.org/0000-0002-6970-4060

for the degree of

Doctor of Philosophy,

University College Cork,

School of Education

Sports Studies and Physical Education Programme

Head of School/Department: Dr. Fiona Chambers Supervisor(s): Dr. Wesley O'Brien, Dr. Kevin Cahill

2020

#### **AUTHOR DECLARATION:**

This is to certify that the work I am submitting is my own and has not been submitted for another degree, either at University College Cork or elsewhere. All external references and sources are clearly acknowledged and identified within the contents. I have read and understood the regulations of University College Cork concerning plagiarism.

Signed: Orlagh Farmer

Orlagh Farmer (Candidate)

Student ID: 111524727

Date: 26<sup>th</sup> June 2020

#### **ACKNOWLEDGEMENTS**

"Non nobis solum nati sumus. (Not for ourselves alone are we born.)"

#### - Marcus Tullius Cicero

The path toward PhD completion has been circuitous. Its completion is thanks in large part to special people who challenged, supported, and stuck with me throughout my research journey. To those people, I am extremely fortunate and thankful.

- Foremost, I wish to express my sincere gratitude to my supervisors, Dr Wesley O'Brien and Dr Kevin Cahill: for your dedication and time to supervise me; your support, constant encouragement, and unwavering belief in me; for your immense knowledge and research experience you shared with me, and your friendship. Thank you for instilling confidence in me over the years, especially when I needed it most. Not only have you both helped me complete this research journey but allowed me to fruitfully grow both personally and professionally, for that I will always be grateful.
- This study would not have been possible without the continued support and assistance from the National Ladies Gaelic Football Association (LGFA) in Ireland. A special thanks to Lyn Savage and William Harmon who have supported and encouraged me over the last few years.
- To my fellow postgraduate students, Diarmuid Lester and Niamh O'Brien, we have stood by each other throughout our PhD journeys, thanks for your genuine friendship and continual support. All those reassuring phone-calls got us over the line!
- To all the study participants, without your interest and dedication, none of this research would have been possible. Thank you for your willing participation and for your honesty in sharing your thoughts and opinions. Sincere thanks also to the parents, teachers, coaches, and principals from all the primary schools and LGFA clubs involved.
- I would also like to acknowledge the assistance of all the undergraduate students from the Sports Studies and Physical Education degree programme involved in the data collection stages, thank you for your time and dedication.

- To my co-authors, Sarahjane Belton in Dublin City University and Donna Duffy in University of North Carolina at Greensboro in America, thank you for your time and insightful feedback in helping me to prepare the published articles which form two chapters in this thesis.
- To all '*the gang*' in Midleton and Cork, your constant support and endless encouragement has been treasured. Laura, Emma, Sheila, Orla, Juls, Eoin, and Cian – thank you for your friendship and lifting my spirits at times when I struggled the most. Thank you also for providing me with valuable study breaks, particularly the encouraging phone calls and trips to '*Monty*'s' for a coffee! To my teammates and management on the Cork Ladies Football and UCC O'Connor Cup teams, taking time out to train and play matches with you all was always a welcome break for the OAP of UCC!

# Surround yourself with people who will lift you higher'Oprah Winfrey

- To Sinéad, you have been with me since day one on this PhD journey and your continuous encouragement and unshakeable belief in me has been invaluable. Thank you for listening to all my rants, dealing with my mini meltdowns and for picking my spirits up when I needed it most. You always push me to be the best version of myself and your selfless support means the world to me.
- A special thanks to my Granny 'Sheila Bells' for always believing in me and for constantly keeping me entertained. Your 'surprise snacks' and 'motivational quotes/letters' left at my study desk in your house haven't gone unnoticed. Thank you for simply brightening up my days!
- To Sr. Helen ('Aunty Nell'), you have kept me on track with your daily
  phone calls from the Religious Order in Baton Rouge Louisiana. In moments
  of distress, you always reminded me to '*take it one day at a time*'. I am
  forever grateful for your genuine interest, optimism and kind words of
  encouragement.
- Finally, and most importantly, to my family. To my brother Kian and sister Sinead, thank you for your support and hearty encouragement.

*To my parents, Mairéad and Seánie – I dedicate this thesis to you.* I am forever indebted to you for your moral encouragement, endless patience, and unconditional love, not only over these past four years but throughout my whole life. You have always provided me with every opportunity I could ever wish for. Your unbounded support has been the source of my motivation and determination in everything that I do in both my sporting and academic endeavours. Thank you for believing in me.

Mo bhuíochas leo siúd a thug misneach dom.

Funding sources: This research programme was funded by the Irish Research Council (Government of Ireland Postgraduate Scholarship) from October 2018-October 2019.

### TABLE OF CONTENTS

Author Declaration:	ii
Acknowledgements	iii
List of Tables	xiii
List of Figures	XV
List of Abbreviations	xvi
List of Appendices	xvii
Abstract	xviii
Publications and Conference Proceedings Related to the Research	XX
Chapter One Introduction to the Thesis	
1.1 Introduction	2
1.1.1 Rationale for thesis	2
1.1.2 Background to Gaelic4Girls in an Irish context	
1.1.3 Summary and significance of G4G study	
1.2 Aims and objectives	
1.3 Research questions	14
1.4 Structure of the thesis	14
1.6 Definition of terms	
1.7 G4G schematic overview	
Chapter Two Literature Review	
2.1 Introduction	
2.2 Physical Activity and Health	
2.2.1 The importance of physical activity	
2.2.2 Physical activity (PA) guidelines	
2.2.3 Global prevalence of PA in children and youth	
2.2.4 Gender differentials in PA	
2.2.4.1 Declining levels of PA from childhood to adolescence	
2.2.4.2 Gender division during primary and post-primary school years	
2.2.5 Physical inactivity	
2.2.5.1 Sedentary Behaviour	
2.2.6 Participation in Organised Youth Sport (OYS) – an avenue to increa	ase youth
PA	
2.2.6.1 Community club sport and PA	
2.2.6.2 OYS participatory benefits	
2.2.6.3 Declining participation rates among youth	
2.2.6.4 Gender differences in OYS among youth	
2.2.7 Factors influencing 'drop-out' in OYS	

2.2.7.1 The 'problem' with girls' - motivators and barriers to sport participation	37
2.3 Correlates of PA (Personal and Social Factors)	39
2.3.1 An introduction to PA correlates	39
2.3.2 Psychological correlates of PA - factors influencing girl's PA engagement a experiences	nd 40
2.3.3 Social influences on PA	44
2.3.3.1 Parental PA influences	45
2.3.3.2 Peer PA influence	47
2.3.3.3 Coach influences	51
2.3.4 Fundamental Movement Skill (FMS) influences on PA	53
2.4 Fundamental Movement Skills (FMS)	55
2.4.1 Introduction and description of FMS	55
2.4.2 Benefits of FMS	57
2.4.3 The relationship between FMS and PA	58
2.4.4 FMS proficiency levels among youth	59
2.4.5 Gender differences in youth FMS	61
2.4.5.1 Gender differences in object control (OC) skills	62
2.4.5.2 Gender differences in locomotor (LOM) skills	63
2.4.5.3 Causes for gender differences in FMS proficiency	64
2.4.6 Perceived motor competence and confidence	66
2.4.7 FMS assessment in youth	68
2.4.8 Advancing towards specialised movement skills in a youth sporting context the relationship between sports participation and the level of motor coordination childhood	t – n in 71
2.5 Bridging the Gap: Promoting Girls' PA and Sport Participation	76
2.5.1 Call for multi-component community sports-based PA interventions for give	rls 76
2.5.2 PA and sport interventions - evidence for successful implementation	77
2.5.2.1 School-based PA interventions	77
2.5.2.2 Community-based PA interventions for girls	79
2.5.2.3 Modified sports programmes - an avenue to increase PA participation	81
2.5.3 Effective PA intervention components	82
2.5.4 Research considerations and frameworks for designing and evaluating PA sport intervention studies	and 86
2.5.5 Theoretical considerations for successful PA and sportinterventions	89
2.5.5.1 Self-Determination Theory (SDT)	89
2.5.5.2 Social Ecological Model (SEM)	92
2.5.5.3 Theory of Planned Behaviour (TPB)	94

**Chapter Three Methodology** 

3.1 Overview of chapter	
3.2 Research philosophy and paradigm	
3.2.1 Justification of the pragmatic approach	
3.3 Research design	
3.3.1 Mixed methods	
3.3.2 Justification of a mixed-methods approach	
3.3.3 Cross-sectional design	
3.3.4 Intervention protocol study	
3.3.5 Quasi-experimental non-randomised controlled trial, three-armed e pre-post design	xploratory 105
3.4 Theoretical framework	
3.5 Ethical considerations	
3.5.1 Informed consent	
3.5.2 Inclusion criteria:	
3.5.3 Exclusion criteria:	
3.5.4 Confidentiality and anonymity	
3.5.5 Other ethical considerations	
3.6 Participants and setting	
3.6.1 Sampling	110
3.7 Data collection procedure	111
3.7.1 Recruitment	111
3.7.2 Field training	112
3.7.3 Procedure	112
3.8 Measures/Instruments	114
3.8.1 Self-reported PA	115
3.8.2 Objective measures of PA	117
3.8.3 Psychological correlates of PA	121
3.8.4 FMS assessment	124
3.8.5 Focus groups	124
3.8.6 Body mass index (BMI)	126
3.9 Intervention design and components	
3.10 Data analysis	
3.11 Research rigour, validity, and reliability of data	131
3.11.1 Cronbach's alpha co-efficient	
3.11.2 FMS inter-reliability scoring	
3.11.3 Trustworthiness of data	
3.11.4 Reporting of data	
3.11.5 Peer-review process	

3.12 Role of researcher	
3.13 Chapter conclusion	
Chapter Four The Relationship between Actual Fundamental M Proficiency, Perceived Motor Skill Confidence and Competence Activity in 8–12-Year-Old Irish Female Youth	Aotor Skill e, and Physical
4.1 Purpose of chapter	
4.1.1 Rationale	
4.1.2 Contribution to the field	
4.2 Abstract	
4.3 Introduction	
4.4 Methods	
4.4.1 Participants and study design	
4.4.2. Recruitment and data collection	142
4.4.3 PA self-report assessment	
4.4.4 Measures: fundamental movement skills (FMS)	
4.4.5 Perceived physical self-confidence	
4.4.6 Physical performance self-concept	146
4.4.7 Data analysis	
4.5 Results	147
4.5.1. Physical activity	147
4.5.2 Fundamental movement skills	147
4.5.3 Perceived physical self-confidence	
4.5.4. Physical performance self-concept	
4.5.5 Interrealtionship between variables	
4.6 Discussion	
4.6.1 Future practical implications	154
4.7 Conclusions	
4.8 Link to chapter 5	
Chapter Five Enhancing the Evidence Base for Irish Female Yo Participation in Physical Activity—The Development of the Gas Programme	outh elic4Girls
5 1 Duenose of aboutor	150
5.1.1 Detionale	
5.1.1 Rationale	
5.1.2 Contribution to the netu	
5.2 Introduction	
5.4 Methods	

5.4.1 Overview of the study 1	164
5.4.2 Participants and setting1	165
5.4.3 Measures/Instruments 1	167
5.4.3.1 PA self-report assessment 1	167
5.4.3.2 Anthropometric measurements 1	168
5.4.3.3 Fundamental movement skills (FMS) assessment 1	168
5.4.3.4 Focus groups 1	169
5.4.4 Procedures (field training and data collection)1	170
5.4.5 Data analysis1	175
5.5 Results 1	175
5.5.1 Physical activity1	176
5.5.2 Fundamental movement skills1	176
5.5.3 Psychosocial correlates of PA1	177
5.5.4 Focus groups1	181
5.5.4.1 PA motivators: fun, enjoyment, and friends1	181
5.5.4.2 PA barriers: lack of time, fear of injury, weather, and friend- related factors . I	181
5.5.4.3 Factors influencing PA/sport engagement and disengagement (dropouts) 1	182
5.6 Discussion 1	183
5.8 Link to chapter 6	199
	100
Chapter Six Rationale and Study Protocol for the 'Gaelic4Girls' Project: A Novel Multi-Component Community Sports-Based Coaching Intervention for Increased Physical Activity Participation of Pre-Adolescent Girls	
Chapter Six Rationale and Study Protocol for the 'Gaelic4Girls' Project: A Novel Multi-Component Community Sports-Based Coaching Intervention for Increased Physical Activity Participation of Pre-Adolescent Girls 6.1 Purpose of chapter	191
Chapter Six Rationale and Study Protocol for the 'Gaelic4Girls' Project: A Novel Multi-Component Community Sports-Based Coaching Intervention for Increased Physical Activity Participation of Pre-Adolescent Girls 6.1 Purpose of chapter	191 191
Chapter Six Rationale and Study Protocol for the 'Gaelic4Girls' Project: A Novel Multi-Component Community Sports-Based Coaching Intervention for Increased Physical Activity Participation of Pre-Adolescent Girls 6.1 Purpose of chapter	191 191 191
Chapter Six Rationale and Study Protocol for the 'Gaelic4Girls' Project: A Novel Multi-Component Community Sports-Based Coaching Intervention for Increased Physical Activity Participation of Pre-Adolescent Girls 6.1 Purpose of chapter	191 191 191 192
Chapter Six Rationale and Study Protocol for the 'Gaelic4Girls' Project: A         Novel Multi-Component Community Sports-Based Coaching Intervention for         Increased Physical Activity Participation of Pre-Adolescent Girls         6.1 Purpose of chapter         6.1.1 Rationale         1         6.1.2 Contribution to the field         1         6.3 Introduction	191 191 191 192 193
Chapter Six Rationale and Study Protocol for the 'Gaelic4Girls' Project: A         Novel Multi-Component Community Sports-Based Coaching Intervention for         Increased Physical Activity Participation of Pre-Adolescent Girls         6.1 Purpose of chapter         6.1.1 Rationale         1         6.1.2 Contribution to the field         1         6.3 Introduction         1         6.4 Methods	191 191 191 192 193 196
Chapter Six Rationale and Study Protocol for the 'Gaelic4Girls' Project: A         Novel Multi-Component Community Sports-Based Coaching Intervention for         Increased Physical Activity Participation of Pre-Adolescent Girls         6.1 Purpose of chapter         1         6.1.1 Rationale         1         6.1.2 Contribution to the field         1         6.3 Introduction         1         6.4 Methods         1         6.4.1 Study design	191 191 192 193 196
Chapter Six Rationale and Study Protocol for the 'Gaelic4Girls' Project: A         Novel Multi-Component Community Sports-Based Coaching Intervention for         Increased Physical Activity Participation of Pre-Adolescent Girls         6.1 Purpose of chapter         6.1.1 Rationale         1         6.1.2 Contribution to the field         1         6.2 Abstract         1         6.3 Introduction         1         6.4.1 Study design         1         6.4.2 Recruitment, setting, and participants	191 191 192 193 196 196
Chapter Six Rationale and Study Protocol for the 'Gaelic4Girls' Project: A         Novel Multi-Component Community Sports-Based Coaching Intervention for         Increased Physical Activity Participation of Pre-Adolescent Girls         6.1 Purpose of chapter         6.1.1 Rationale         1         6.1.2 Contribution to the field         1         6.2 Abstract         1         6.3 Introduction         1         6.4.1 Study design         1         6.4.2 Recruitment, setting, and participants	191 191 192 193 196 196
Chapter Six Rationale and Study Protocol for the 'Gaelic4Girls' Project: A         Novel Multi-Component Community Sports-Based Coaching Intervention for         Increased Physical Activity Participation of Pre-Adolescent Girls         6.1 Purpose of chapter         6.1.1 Rationale         1         6.1.2 Contribution to the field         1         6.2 Abstract         1         6.3 Introduction         1         6.4.1 Study design         1         6.4.2 Recruitment, setting, and participants         1         6.5.1 Theoretical framework	191 191 192 193 196 196 196 197
Chapter Six Rationale and Study Protocol for the 'Gaelic4Girls' Project: A         Novel Multi-Component Community Sports-Based Coaching Intervention for         Increased Physical Activity Participation of Pre-Adolescent Girls         6.1 Purpose of chapter         6.1.1 Rationale         1         6.1.2 Contribution to the field         1         6.2 Abstract         1         6.3 Introduction         1         6.4.1 Study design         1         6.5 Gaelic4Girls intervention         1         6.5.1 Theoretical framework         1         6.5.2 G4G intervention structure and components	191 191 192 193 196 196 196 197 197
Chapter Six Rationale and Study Protocol for the 'Gaelic4Girls' Project: A         Novel Multi-Component Community Sports-Based Coaching Intervention for         Increased Physical Activity Participation of Pre-Adolescent Girls         6.1 Purpose of chapter         6.1.1 Rationale         6.1.2 Contribution to the field         6.2 Abstract         1         6.3 Introduction         1         6.4.1 Study design         1         6.5 Gaelic4Girls intervention         1         6.5.1 Theoretical framework         1         6.5.2 G4G intervention structure and components         2         6.5.2.1 G4G participant components	191 191 192 193 196 196 196 197 197 200
Chapter Six Rationale and Study Protocol for the 'Gaelic4Girls' Project: A         Novel Multi-Component Community Sports-Based Coaching Intervention for         Increased Physical Activity Participation of Pre-Adolescent Girls         6.1 Purpose of chapter         6.1.1 Rationale         1         6.1.2 Contribution to the field         1         6.2 Abstract         1         6.3 Introduction         1         6.4.1 Study design         1         6.5 Gaelic4Girls intervention         1         6.5.1 Theoretical framework         1         6.5.2.1 G4G participant components         2         6.5.2.2 G4G coach components	191 191 192 193 196 196 196 197 200 200 202
Chapter Six Rationale and Study Protocol for the 'Gaelic4Girls' Project: A         Novel Multi-Component Community Sports-Based Coaching Intervention for         Increased Physical Activity Participation of Pre-Adolescent Girls         6.1 Purpose of chapter         6.1.1 Rationale         1         6.1.2 Contribution to the field         1         6.2 Abstract         1         6.3 Introduction         1         6.4 Methods         1         6.5 Gaelic4Girls intervention         1         6.5.1 Theoretical framework         1         6.5.2.1 G4G participant components         2         6.5.2.3 G4G parental and guardian component.	191 191 192 193 196 196 196 197 200 200 200 202 204

6.6.1 Data collection	
6.6.1.1 Physical activity self-report assessment	
6.6.1.2 Fundamental movement skills	
6.6.1.3 Psychological correlates of PA	
6.6.2 Data analysis	
6.7 Discussion	
6.8 Conclusion	
6.9 Link to chapter 7	
Chapter 7 'Gaelic4Girls' – The Effectiveness of a 10-Week	x Multi-Component
<b>Community-Sports Based Physical Activity Intervention f</b>	or 8-12-year-old Girls
7.1 Purpose of chapter	
7.1.1 Rationale	
7.1.2 Contribution to the field	
7.2 Abstract	
7.3 Introduction	
7.4 Materials and methods	
7.4.1 Study design	
7.4.2 Recruitment, setting and participants	
7.4.3 Outcomes	
7.4.3.1 Data collection	
7.4.3.2 Physical activity self-report assessment	
7.4.3.3 Fundamental movement skills	
7.4.3.4 Psychological correlates of PA	
7.4.3.5 Focus group Interview	
7.4.4 Data analysis	
7.5 Results	
7.5.1 Descriptive statistics	
7.5.2 Physical activity	
7.5.3 Fundamental movement skills	
7.5.4 Psychological correlates of PA	
7.5.5 Focus group	
7.6 Discussion	
7.7 Conclusion	
Chapter Eight Discussion, Conclusion, and Future Direction	on of 'Gaelic4Girls'
8.1 Introduction	
8.2 Thesis overview	
8.3 Contribution of thesis to the literature	

8.3.1 Pedagogical coaching frameworks: an avenue for increased physical activity and motor competence in 8 to 12-year-old Irish girls
8.3.2 Psychological correlates of PA – knowing your target audience for intervention design
8.3.2.1 'Girls just wanna have fun': capitalising on PA enjoyment
8.3.2.2 'I won't because I think I can't': Building confidence and competence in 8 to 12-year-old girls
8.3.3 The impact of social support from 'significant others' on 8 to 12-year-old girls' PA: peers, parents, and positive role-models
8.3.3.1 Peer and parental social support
8.3.3.2 Role-models as significant influencers for increased participation in OYS267
8.4 Strengths and limitations
8.5 Research, policy, and practice implications
8.5.1 Future directions of G4G
8.5.2 Implications for policy
8.5.3 Implications for practice
8.5.3.1 A mentor-based approach building role models within the G4G and LGF community
8.5.3.2 The school-community link – educating and empowering teachers and students
8.5.3.3 Enhancing LGFA coach education practices for increased PA levels, movement confidence, competence, and psychological wellbeing
8.6 Concluding thoughts
References
Appendices
Appendix A: Ethical Approval Letter
Appendix B: G4G Information Sheet and Consent Forms
Appendix C: G4G Questionnaire
Appendix D: Fundamental Movement Skill Performance Criteria (TGMD, TGMD-2, Get Skilled Get Active
Appendix E: G4G Data Collection Protocol Manual
Appendix F: Baseline Study Focus Group Questions and Drawings (Chapter 5) 411
Appendix G: G4G Intervention Study Focus Group Questions and Drawings (Chapter 7)
Appendix H: Editor Acceptance Notification of Published Studies
Appendix I: G4G Coaching Manual (weekly sessions (x3), workshop overview, and electronic LGF/FMS video clips
Appendix J: G4G Skill Cards, Poster and Certificate
Appendix K: G4G Parents Evening Content and Presentation Evening Flyer

# LIST OF TABLES

#### CHAPTER 1

Table 1.1: Differences between the existing G4G programme and revised	
programme	11
CHAPTER 3	
Table 3.1: Characteristics of studies conducted as part of the thesis	100
Table 3.2: G4G psychological correlates of PA assessment instruments and	122
scales, with Alpha Cronbach Coefficients	
Table 3.3: Summary of study objectives and statistical analysis across all	129
studies	
CHAPTER 4	
Table 4. 1: Descriptive data for each variable (frequency, means, standard	148
deviation), and physical activity (PA) grouping (low, moderately, and highly	
active)	
Table 4. 2: Mean scores for fundamental movement skills (FMSs), perceived	149
physical self-confidence (PSC), and physical performance self-concept	
(PPSC) by physical activity (PA) grouping (low, moderately, and highly	
active)	
CHAPTER 5	
Table 5.1: Procedure & questions used for the G4G baseline focus groups	173
(n=6)	
Table 5.2 Mean (SD) across physical activity grouping with significant main	179
effects and effect size of one-way ANOVAs	
CHAPTER 6	
Table 6.1: Gaelic4Girls (G4G) intervention components and theoretical	208
constructs	
CHAPTER 7	
Table 7.1: Overview of the research informed G4G intervention group 1	229
programme components with associated pedagogical considerations	
Table 7.2: G4G psychological correlates of PA assessment instruments and	234
scales, with Alpha Cronbach Coefficients	

<b>Table 7.3:</b> Descriptive statistics [means and standard deviations (M±SD)] of	236
PA, FMS and psychological measurements, stratified by group and time-	
period	
Table 7.4: Mean change (MC) differences and significant interaction effects	241
(within time and between-group) for the psychological	
variables	
<b>Table 7.5:</b> G4G Research-informed intervention group 1 focus group thematic	246
analysis findings (N=6)	

# LIST OF FIGURES

CHAPTER 1	
Figure 1.1: G4G schematic overview	21
CHAPTER 2	
Figure 2.1: The phases and stages of motor development (taken from Gallahue	56
and Ozmun, 2012)	
Figure 2.2: Medical Research Council (2000) Framework- Phases of	87
investigation in the evaluation of a complex intervention	
Figure 2.3: Social-ecological model of children's participation in physical	92
activity programmes - adapted from Sallis, Owen, & Fisher	
(2008)	
Figure 2.4: Theory of Planned Behaviour. Source: Ajzen (2005)	95
p.135	
CHAPTER 4	
Figure 4.1: Study design and flow of data collection ( $n = 160$ )	142
CHAPTER 5	
Figure 5.1: Study design and flow of baseline data collection (N=344)	166
Figure 5.2: Percentage of advanced skill proficiency of 8-12-year-old	177
participants (N=228)	
CHAPTER 6	
Figure 6. 1: Theoretical underpinnings of the revised G4G	199
intervention	
Figure 6. 2: Gaelic4Girls weekly session station-based approach pitch	206
layout	
CHAPTER 7	
Figure 7.1. Comparison of the mean self-reported minutes of MVPA over time	237
(pre to post) by group	
Figure 7.2. Mean locomotor, object-control, and overall FMS proficiency	238
differences among groups over time (pre to post stages; $n = 120$ )	

#### LIST OF ABBREVIATIONS

BMI = body mass index

- CPD = continuous professional development
- DEIS = delivering equality of opportunity in schools

FG = focus group

G4G = Gaelic4Girls

HSD = honest significance difference

ICC = intra-class correlation coefficient

LGFA = Ladies Gaelic Football Association

LOM = locomotor

M = mean

MNM = mastery and near mastery

MVPA = moderate-to-vigorous-physical activity

OC = object-control

OYS = organised youth sport

PA = physical activity

PACE = Physician-based Assessment and Counseling for Exercise

PAQ-C = Physical Activity Questionnaire for Older Children

PE = physical education

PPSC = physical performance self-concept

PSC = perceived physical self-confidence

SD = standard deviation

SDT = self-determination theory

SE = self-efficacy

SEM = social ecological model

SREC = social research ethical committee

TGMD-2 = test of gross motor development-2

UCC = University College Cork

### LIST OF APPENDICES

Appendices	378
Appendix A: Ethical approval letter	378
Appendix B: G4G information sheet and informed consent form	381
Appendix C: G4G questionnaire	385
Appendix D: Fundamental movement skill performance criteria	401
Appendix E: G4G data collection protocol manual	404
Appendix F: Baseline study focus group questions and drawings (chapter 5)	
Appendix G: G4G intervention study focus group questions and drawings	411
(chapter 7)	
Appendix H: Editor acceptance notification of baseline studies	417
Appendix I: G4G Coaching Manual (weekly sessions (x3), workshop	422
overview, and electronic LGF/FMS video clips)	
Appendix J: G4G Skill Cards and Certificate	453
Appendix K: G4G Parents Evening content and Presentation Evening flyer	457

#### ABSTRACT

**Background:** It is widely reported that girls are less physically active than boys throughout childhood, and the age-related decline in physical activity (PA) participation, particularly from early adolescence onwards, is steeper for girls than for boys. Correlates of PA, such as fundamental movement skills (FMS), club-based participation in organised youth sport (OYS), psychological correlates (self-efficacy, enjoyment, PA attitudes), and social support structures (family and peer support) during childhood and adolescence contributes considerably to leisure-time PA for health-enhancing benefits in young girls.

**Purpose:** The purpose of this study was to design, develop, implement and evaluate a multi-component community sports-based PA intervention, specifically tailored for 8- to 12- year old girls in Ladies Gaelic Football (LGF) clubs in Ireland. The existing programme, known as Gaelic4Girls (G4G), was re-designed and revised using the theoretical underpinnings of the Self-Determination Theory (SDT) and elements of the Social Ecological Model (SEM).

**Methodology:** Data for this PhD thesis were gathered from participants (n= 568), using a mixed-methods research design. At baseline, information was gathered on participants (n = 331) levels of PA (self-report questionnaire), FMS proficiency (live assessment of motor skills), and psychological correlates of PA (self-report questionnaire), using validated and reliable protocol. A sub-sample of participants (n = 37) also participated in focus group interviews to explore their perceptions of PA and sport participation. Based on this data, and an exploration of the literature, a revised G4G intervention was developed. A quasi-experimental, non-randomised controlled trial involving three community sports clubs (group 1 - revised G4G intervention) was then implemented to evaluate the revised G4G intervention's efficacy. Participants' data (n=120) was collected at pre and post time points on the following variables; PA levels, FMS proficiency, and psychological correlates of PA. Focus group data (n = 6) was collected at post-intervention to explore perceptions of the revised G4G intervention.

**Results:** Following a 2 (pre to post) by 3 (group 1, 2 and 3) mixed-model ANOVA, it was highlighted over time that the revised G4G intervention group 1 significantly increased in PA (mean change = 39.7, SD = 81.66, p=.003), FMS proficiency (mean

xviii

change = 1.86, SD = 4.78, p=.005) and their associated psychological correlates of PA (namely self-efficacy (p<.002), perceived self-confidence (p<.002), enjoyment (p<.003), attitudes towards PA (p<.003), and family social support (p<.002). FG findings reported positive findings for the revised G4G intervention group 1 participants, specifically the heightened psychological wellbeing for girls, and the establishment of emerging friendships between peers.

**Conclusion:** The findings demonstrate that the 10-week specifically tailored, research-informed and revised G4G intervention is a feasible and efficacious programme, leading to a positive effect on the physical and psychological wellbeing of pre-adolescent Irish girls, as relative to the traditionally delivered existing G4G comparative programme, and control group conditions. Further research involving a randomised controlled trial, with a larger sample size is warranted.

# PUBLICATIONS AND CONFERENCE PROCEEDINGS Related to the Research

#### **Published Journal Articles:**

- Farmer, O., Belton, S., & O'Brien, W. (2017). The Relationship between Actual Fundamental Motor Skill Proficiency, Perceived Motor Skill Confidence and Competence, and Physical Activity in 8–12-Year-Old Irish Female Youth. Sports, 5(74). <u>https://doi.org/10.3390/sports5040074</u>
- O'Brien, W., Duncan, M. J., Farmer, O., & Lester, D. (2018). Do Irish adolescents have adequate functional movement skill and confidence? *Journal of Motor Learning and Development*, 6(s2), S301–S319. https://doi.org/10.1123/jmld.2016-0067
- Farmer, O., Duffy, D., Cahill, K., Lester, D., Belton, S., & O'Brien, W. (2018). Enhancing the evidence base for Irish female youth participation in physical activity - the development of the Gaelic4Girls program. *Women in Sport and Physical Activity Journal*, 26(2), 111–123. <u>https://doi.org/10.1123/wspaj.2017-0046</u>
- Connolly, S., Carlin, A., Johnston, A., Woods, C., Powell, C., Belton, S., O'brien,
  W., Saunders, J., Duff, C., Farmer, O., & Murphy, M. (2020). Physical Activity,
  Sport and Physical Education in Northern Ireland School Children: A crossSectional Study. *International Journal of Environmental Research and Public Health*, 17(6849). https://doi.org/10.3390/ijerph17186849
- Farmer, O., Cahill, K., & O'Brien, W. (2020). Gaelic4Girls—The Effectiveness of a 10-Week Multicomponent Community Sports-Based Physical Activity Intervention for 8 to 12-Year-Old Girls. *International Journal of Environmental Research and Public Health*, 17(6928). <u>https://doi.org/10.3390/ijerph17186928</u>

#### **Published Report:**

Woods, C. B., Powelll, C., Saunders, J. A., O'Brien, W., Murphy, M. H., Duff, C., Farmer, O., Johnston, A., Connolly, S., & Belton, S. (2018). *The Children's Sport Participation and Physical Activity Study (CSPPA 2018)*. Department of Physical Education and Sport Sciences, University of Limerick, Limerick, Ireland, Sport Ireland, and Healthy Ireland, Dublin, Ireland and Sport Northern Ireland, Belfast, Northern Ireland.

https://doi.org/10.13140/RG.2.2.22091.95521

#### Journal Articles Under Review:

Farmer, O., Cahill, K., & O'Brien, W. (2020). Rationale and Study Protocol for the 'Gaelic4Girls' Project: A Novel Multi-Component Community Sports-Based Coaching Intervention for Increased Physical Activity Participation of Pre-Adolescent Girls. Sports Coaching Review.

#### **Poster Presentations:**

- O' Brien, W. (2019) 'Gaelic4Girls' (G4G) evaluating the effectiveness of a 10week multi-component community-sports based physical activity intervention for 8-12-year-old girls'. *International Motor Development Research Consortium (I-MDRC) The Healthy and Active Children Symposium*, Verona Italy (September 2019).
- Farmer, O. (2018) 'Gaelic4Girls' for increased physical activity participation: A multi-component, pilot intervention: study design and protocol. *Physical Education, Physical Activity and Youth Sport (PEPAYS) Forum,* University of Limerick Ireland (November 2018).
- Farmer, O. (2018) 'Gaelic4Girls' for increased physical activity participation: A multi-component, pilot intervention: study design and protocol. *International Society for Physical Activity and Health (ISPAH)*, London (October 2018).
- O'Brien, W., & Farmer, O (2018). 'Gaelic4Girls' for increased physical activity participation – a multi-component, pilot intervention: Study design and protocol. North American Society for the Psychology of Sport and Physical

Activity' (NASPSPA) International Conference, Denver USA (June 2018).

- Farmer, O. (2017) 'The Design, Development, Implementation & Evaluation of the Gaelic4Girl's Programme'. 16th Meeting of The International Society for Behavioural Nutrition and Physical Activity (ISBNPA) Conference, Victoria, Canada (June 2017).
- Farmer, O. (2017) 'Gaelic for Girls': The Design, Development, Implementation and Evaluation of the G4G's Programme'. GAA Researches Exhibition – GAA Games Development Conference, Croke Park Dublin (January 2017).
- Farmer, O. (2016) 'Gaelic for Girls': Rationale, Study Protocol & Methodological Considerations'. 15<sup>th</sup> Meeting of The International Society of Behavioural Nutrition & Physical Activity (ISBNPA) Conference, Cape Town South Africa (June 2016).
- Farmer, O. (2016) 'Gaelic for Girls': Rationale, Study Protocol & Methodological Considerations'. *Psychology, Health & Medicine Conference*, University College Cork, Ireland (May 2016).
- Farmer, O. (2016) 'Gaelic for Girls': Rationale, Study Protocol & Methodological Considerations'. *PEPAYS Forum*, Tralee Institute of Technology, Ireland (May 2016).

#### **Oral Presentations:**

- Farmer, O. (2019) 'Gaelic4Girls' (G4G) evaluating the effectiveness of a 10-week multi-component community-sports based physical activity intervention for 8-12-year-old girls. *International Physical Literacy (IPLC)* Conference, Winnipeg, Canada (May 2019).
- Farmer, O. (2019) 'Gaelic4Girls' (G4G) evaluating the effectiveness of a 10-week multi-component community-sports based physical activity intervention for 8-12-year-old girls. *Physical and Health Education (PHE) Conference*, Montreal (May 2019).
- Farmer, O. (2019). 'The design, development, implementation and evaluation of the 'Gaelic4Girls' Intervention'. *Motor Competence: The Forgotten Pathway to Childhood Health Symposium*, University College Cork Ireland (January 2019).

- Farmer, O. (2018). 'The design, development, implementation and evaluation of the 'Gaelic4Girls' Intervention'. *Children's Motor Competence and its Impact on Physical Activity, Exercise, Sport and Health Symposium*, Coventry University UK (January 2018).
- Farmer, O. (2017). Enhancing the evidence-base for Irish female youth participation in physical activity – the development of the 'Gaelic for Girls' programme. *Physical Education, Physical Activity and Youth Sport (PEPAYS) Research Forum*, Limerick, Ireland (June 2017).
- Farmer, O. (2016) 'The design, development, implementation, and evaluation of the Gaelic4Girls intervention'. PhD Symposium 2016 – At the forefront of Contemporary Research with Children and Young People in Ireland. Maynooth University (September 2016).
- Farmer, O. (2015) Rationale and proposed study protocol of the 'Gaelic4Girls' (G4G) programme. *The Graduate School, College of Arts, Celtic Studies & Social Sciences Postgraduate Research Conference,* University College Cork (December 2015).

#### **Keynote Presentations:**

- Farmer, O. (2020) 'Gaelic Games for physical activity and sport promotion in Ireland for at-risk, drop-out girls'. *National GAA Games Development Conference*, Croke Park, Dublin (January 2020).
- Farmer, O. (2019) 'Gaelic4Girls an avenue to increase physical activity participation and psychological wellbeing in young girls'. *Sligo GAA Health & Wellbeing Conference*, Sligo Institute of Technology (March 2019).

#### Webinars:

Farmer, O. (2020). iCoach Kids Europe Shares Webinar Series 'To Play or Not to Play'. April 2020

#### Workshops:

Farmer, O. (2017-2019) 'Gaelic4Teens' National workshops,

- 'Coach the Teenager Roadshow'
- 'Lifestyle planning for teenage girls'
- 'The 'importance of the psychosocial element of female youth physical activity participation'.
- Maynooth University (November 2019), University College Limerick (May 2019), Kildare (2018), Croke Park (April 2017).

#### Awards:

Government of Ireland Postgraduate Irish Research Council Scholarship awardee (October 2018 – October 2019).

The Irish Times 'Sportswoman of the Month Award' (May 2020). Recognised and awarded for the innovative creation of research-informed videos on social media platforms targeting young girls (March – May during Covid19 lockdown). The online #DrilloftheDay video's, as part of my '5 Day Ladies Gaelic Football Challenge', garnered attention nationally and internationally.

# CHAPTER 1 INTRODUCTION TO THE THESIS



#### **1.1 INTRODUCTION**

This chapter provides the context, rationale and justification for the design, development, implementation and evaluation of the 'Gaelic4Girls' (G4G) intervention - a multi-component physical activity (PA) community sports-based intervention for 8 to 12-year-old girls. This includes the wide range of benefits gained from engaging in PA and organised youth sport (OYS), the current global physical inactivity levels of pre-adolescent girls, interventions designed to increase PA and sport participation levels, fundamental movement skill (FMS) proficiency and psychological wellbeing in this cohort and the current gaps in the literature. This introductory chapter presents an overview of the dissertation. Initially, the chapter outlines the focus of the study, by drawing on salient themes within the field, and provides a background to the existing Ladies Gaelic Football (LGF) G4G programme. In doing so, this chapter identifies gaps in the literature, and documents the key research considerations that underpin the basis of the study. Thereafter, the significance and originality of the study is highlighted. Finally, the chapter concludes by presenting the research aims, objectives, research questions and structure of the thesis to follow.

#### 1.1.1 Rationale for thesis

It is well established that regular participation in PA is imperative for good health (Eime et al., 2013), and is associated with an array of physical, psychological and social health benefits (Biddle et al., 2019; Janssen & Leblanc, 2010) for children and young people aged 5 to 18 years old. Despite the widely-known health benefits of PA participation, studies continue to show that lack of PA participation among children and adolescents has become a global health concern (Guthold et al., 2019; Tremblay et al., 2016). Most recent nationally representative Irish data (n = 6,651; mean age=13.8  $\pm$  2 years) from the 'Children's Sport Participation and Physical Activity Study' (CSPPA) (Woods et al., 2018) found that only 17% of children aged 9- to 11- years old, and 10% of adolescents aged 12- to 17- years old self-reported meeting the recommended 60-minute moderate-to-vigorous PA (MVPA) daily. These low levels of PA among children and adolescents are in line with global trends (Sallis et al., 2016).

It is widely reported that girls are less physically active than boys throughout childhood (Sallis et al., 2016; Van Hecke et al., 2016), and the age-related decline in PA participation, particularly from early adolescence onwards, is steeper for girls when compared to boys (Beltran-Valls et al., 2019; Dumith et al., 2011; Farooq et al., 2018; Owen et al., 2017). Within the general population, pre-adolescent children are a sub-group of the general population that warrant particular focus (Biddle et al., 2014). This cohort of children are considered the most active segment of society, however, there remains concern that even for this age group, many children, particularly girls, have PA levels lower than those recommended for good health (Jekauc et al., 2013; Kalman et al., 2015; Van Hecke et al., 2016). In an Irish context, the Health Behaviour in School-Aged Children (HBSC) main study (Költő et al., 2020), children from 5th class to 5th year who were aged 10 to 17 years found that 32% of females aged 10- to 11- years, and 42% of males the same age (N= 12,002) self-reported accumulating at least 60 minutes of MVPA daily. Most recent research shows that fewer Irish girls aged 9- to 11- years old meet the MVPA recommended guidelines when compared to boys (13% vs. 23%) (Woods et al., 2018). These differences are meaningful across child and pre-adolescent age groups for girls in nearly all countries and regions (Demetriou et al., 2019; Inchley et al., 2020; Ishii et al., 2015). Based on the consistently reported research-informed literature, 8 to 12-year-old girls have been identified as a high priority population for childhood PA promotion (Biddle et al., 2014).

To combat the high physical inactivity levels of this population, several interventions have been designed and implemented, particularly in school settings, in an effort to increase young girls PA levels (Robbins et al., 2019; Sutherland et al., 2016). The effects of these interventions, however, have remained modest (Pearson et al., 2015) and the quantification of intervention effectiveness for this age group of girls have not been adequately reported (Biddle et al., 2014). Besides, there is a lack of effective intervention strategies to promote sustainable PA participation in female youth (Anokye et al., 2018), particularly within a sporting context (Whitley et al., 2019). OYS participation is one such strategy with the existing potential to increase overall PA levels in young people (Hebert et al., 2015), and is positively associated with an increased likelihood of complying with national PA and sedentary behaviour guidelines (Vella et al., 2013). Sports club membership predicts higher levels of

leisure-time PA engagement, specifically among girls (Marques et al., 2016). Participation in club-based OYS during childhood and adolescence contributes considerably to leisure-time PA for health-enhancing benefits (Eime et al., 2013), and has the potential to increase overall PA levels (Dalene et al., 2018; Fröberg et al., 2020; Lee et al., 2018), physical fitness (Drenowatz et al., 2019), and improve FMS competence (Holfelder & Schott, 2014; O' Connor et al., 2018) in young people. The International Society for PA and Health (Trost et al., 2014) has identified participation in sports as one of the seven worldwide "investments that work" for improving youth PA levels, and for these reasons, community-sports based settings are recognised as key health settings in promoting knowledge of PA (Beets et al., 2016), and healthy lifestyle behaviours (Finch & Donaldson, 2010; Kokko, 2014).

Encouragingly, sport occupies a prominent place in Ireland's national cultural identity (Federation of Irish Sport, 2013), which is reflected in the high prevalence of primary school children (75%) aged 10- to 12- years old who reported being members of at least one community sports club (Woods et al., 2018). A further 80% of primary school children reported participating in community sport at least once a week (Woods et al., 2018). Ireland has a very specific sporting culture, with the Gaelic Athletic Association (GAA), a sporting organisation which promotes Gaelic games (such as Hurling (Camogie), Gaelic Football, Rounders, and Handball), dominating a large portion of our PA culture in school and community settings. LGF, the women's equivalent of Gaelic football is a team sport organised by the Ladies Gaelic Football Association (LGFA). Founded in 1974, LGF is recognised as one of the fastest-growing female sports in Europe, with most recent Irish research revealing LGF as the most popular sport among Irish 9- to 11-year- old primary school girls (Woods et al., 2018). It is also important to note that LGF (like hurling and camogie), has garnered massive interest and reach worldwide with Ireland's Gaelic games played in Africa, Asia, Great Britain, Canada, Europe, South America, the United States, New Zealand, and Australia. Gaelic games are high-intensity sports that require players to use sport-specific and specialised FMS movement sequences, such as running, jumping, catching, turning, kicking, tackling, and soloing (kick the ball to yourself while running).

FMS are the building blocks for movement and form the foundation for many of the specialised movement skills needed to participate successfully in sport and PA (Goodway et al., 2020). There is evidence emerging that developing FMS during childhood may be a significant step towards establishing a lifelong commitment to PA (Burns et al., 2017; Logan et al., 2015; Nilsen et al., 2020). As well as sustained engagement in PA, FMS proficiency is also positively associated with a range of health, fitness and academic outcomes, and participation in organised sports (Bolger et al., 2019; Jaakkola et al., 2015; Jones et al., 2020; Robinson et al., 2015; Stodden et al., 2014). Despite the benefits, a growing body of literature illustrates that children and adolescent youth do not obtain proficiency in FMS development (Bardid et al., 2016; Lester et al., 2017; Mukherjee et al., 2017), with current FMS proficiency levels of children worldwide being reported as low (Behan et al., 2019; Bryant et al., 2014; Rainer & Jarvis, 2015). Females have been found to consistently underperform in FMS, when compared to males (Eather et al., 2018; Kelly et al., 2019; O' Brien et al., 2016; Spessato et al., 2013).

In Ireland, at primary school level, recent studies (Behan et al., 2019; Bolger et al., 2018; Kelly et al., 2019) reported that children, particularly girls, are performing below their developmental potential. At secondary school level, O' Brien et al. (2016a) revealed that only 11% of adolescents achieved advanced FMS proficiency across a range of basic FMS, with gender differences present. These findings specifically suggest that in the context of FMS, Irish 8 to 12-year-old girls are performing below their expected developmental capability and are transitioning to adolescence without prior acquisition of basic FMS. There is a distinct lack of data published regarding FMS proficiency levels of Irish children, particularly among girls, and therefore it may not be appropriate to generalise international findings to an Irish context due to different cultural, educational, and traditional sporting pursuits in Ireland (O' Connor et al., 2018).

Based on the results of previous literature, the potential impact of developing FMS through OYS may be important for promoting long-term health-enhancing PA across childhood (Clarke & Metcalfe, 2002; Lubans et al., 2010). In Ireland, there is a noticeable absence of data concerning 8 to 12-year-old girl's FMS proficiency within an OYS setting, clearly indicating a gap within the literature. Findings from a recent Irish study found that Irish juvenile Gaelic game players ( $9.9 \pm 1.3$  years)

5

display higher FMS mastery than their age-matched, general population peers (O' Connor et al., 2018). Despite the low numbers in this study (N = 63), this suggests that participation in Gaelic games facilitates FMS development in children. Therefore, low FMS proficiency levels among 8 to 12-year-old girls highlight the need for coaches to incorporate FMS development in coaching sessions (Morgan et al., 2013; Tompsett et al., 2017).

Empirical studies have found that OYS coaches who participated in nonformal coach training programmes enhanced their ability to improve their team's environment, build positive coach-athlete relationships, and foster positive outcomes in their athletes (Koh et al., 2014; Mansfield et al., 2018). There are currently no evidence-based studies regarding coach education and the provision of coach Continuous Professional Development (CPD) opportunities within the existing LGFA organisational context in Ireland. Due to the amateur nature of Ireland's national game (GAA and LGFA), many coaches, parents, and guardians are voluntarily coaching, and this may lead to a lack of confidence for effective coach pedagogy (Gearity, 2012). The potential for LGF and the G4G programme, therefore, to develop children's FMS skills may offer an important opportunity for enhancing multiple aspects of children's movement skill competence, as well as developing the complex skills which are essential for ongoing participation in PA.

In an Irish context, despite the national policies such as i) the National Physical Activity Plan (Department of Health, 2016), ii) the National Sports Policy 2018 – 2027 (Department of Transport, Tourism, and Sport, 2018), and iii) Sport Ireland's (2019) Women in Sport policy, specifically to promote PA participation through the medium of sport, minimal evaluations of 8 to 12-year old girls within a community sports-based setting exist. In order to develop and implement targeted strategies to increase team sport participation and, consequently, PA levels among 8 to 12-year-old girls, there is a need to understand trends in, and influences on, girls' participation in a team sport setting (Allison et al., 2017). Equally, researchers and practitioners must be aware of the factors that discourage or prevent their participation in PA (Coleman et al., 2008). Factors influencing girls' PA levels and participation include psychological correlates, such as perceived competence and confidence, self-efficacy motivation, attitude and enjoyment of PA, as well as external socioecological factors such as family, peer and coach support (Howie et al.,

6

2018; Laird et al., 2018; Sheridan et al., 2014; Sterdt et al., 2014). As illustrated by socioecological model approaches (Sallis et al., 2006), it is, therefore, important to consider a wide range of correlates from multiple domains, given that focusing on a single domain may give rise to inaccurate conclusions (Brodersen et al., 2005). Despite this, there are a lack of studies simultaneously exploring a range of potential correlates from multiple domains (Van Sluijs et al., 2013), specifically among girls in an OYS setting.

The majority of research conducted to-date has taken place in school settings, with little attention given to multi-component OYS interventions, incorporating psychological and socioecological factors in the development and reinforcement of healthy PA behaviours (Morgan et al., 2019). This is an important consideration, as community sports-based interventions among girls (Pearson et al., 2015) found that greater intervention effectiveness appears to result from interventions that are

- 1) multi-component, theory-based, (Van Sluijs et al., 2007),
- 2) focused on girls only (Camacho-Minano et al., 2011),
- facilitated to ensure they are meeting the needs of this age group (James et al., 2018),
- 4) include combinations of support components (i.e. coaches, family, friends, etc.) (Laird et al., 2018; Sheridan et al., 2014), and
- include a non-formal coach education element (Koh et al., 2014) including FMS coach support strategies (Lubans et al., 2017).

Considering the dearth of research in an Irish OYS setting, specifically multicomponent interventions targeting 8 to 12-year-old girls in the Gaelic Games context, this is an area that merits exploration. As far as the researchers are aware, this is the first study that has been conducted in Ireland aiming to increase PA levels, FMS proficiency and psychological wellbeing of 8- to 12-year- old girls within an OYS setting. The existing G4G programme, as run by the LGFA, has no evidence-base behind their programme and has potential to be tailored to increase participation levels among 8 to 12-year-old Irish girls. With the recently documented low levels of PA and FMS proficiency in young girls in Ireland, and the high drop-out rate from sport approaching the adolescent years, the initiation of the revised research-informed G4G intervention is warranted, innovative and indeed novel.

As previously highlighted, childhood and particularly pre-adolescence is a critical period of change in the PA participation levels of girls (Biddle et al., 2014), and finding ways to help girls to become more physically active at this age is important for short and long-term health. It is during these years that children are developmentally ready to learn motor skill competence, confidence and PA behaviours, that will give them the best chance of engaging in lifelong, health-enhancing PA (Balyi et al., 2013). The question about effective strategies to address and increase young girls' PA is an important public health topic that has yet to be adequately explored and it remains unclear how-to best address low levels and declines in PA among 8 to 12-year-old girls (Biddle et al., 2014).

#### 1.1.2 Background to Gaelic4Girls in an Irish context

G4G is an existing nationally run LGFA 12-week programme consisting of fun, noncompetitive coaching sessions aimed at increasing participation in LGF. The programme, which runs annually (usually between March and May) targets girls aged between 8- to 12- years old who are not currently registered with a LGF club. In essence, G4G is a tool to attract new players in a club, specifically those who have never played before, supporting a club in increasing their number of players (<u>https://ladiesgaelic.ie/learn/games-development-initiatives/gaelic4girls/</u>). The programme is run on an application basis, and any interested LGF club in Ireland can apply, and clubs which meet the criterion, as outlined on the LGFA annual application form, are selected (e.g. established clubs with low numbers, and/or newly established clubs).

• The first 8 weeks of the programme is aimed at introducing the girls to the skills of LGF through a fun, non-competitive game-based approach. After the 8 weeks, the integration of new players into existing teams within the specific age-groups in the club takes place. By the end of the 12 weeks, the G4G programme aims to register all new players to the club.

- Upon completion of the programme, the new players get an opportunity to execute their newly-acquired skills in fun cluster blitzes with G4G participants from other surrounding Provisional counties.
- G4G coaches are also supported throughout the 12-week programme with a Coach Manual resource booklet and are required to attend one Provisional training day, prior to the start of the programme. The coach component of the programme helps to reinvigorate, grow and sustain the clubs well into the future.

The G4G programme is an example of what Cote et al. (2009) describe as a modified sports 'entry-level' programme for sports organisations (clubs and governing bodies), offered to engage children in play activities designed, among other things, to develop FMS and sport-specific skills for future participation. These programmes are modified to match the developmental capacity of children and are aimed at the development of FMS and sport-specific skills, rather than competition (Eime et al., 2018). This is the essence of the G4G programme, whereby young girls are introduced to the skills of LGF through a fun, non-competitive games-based approach to increase participation in LGF. Essentially, the sport is modified to match the developmental capabilities of children, by adapting games and activities through changes to the rules, equipment, and/or physical space to encourage inclusion and maximise participation (Eime et al., 2018). One example of a recently developed modified sports programme in Ireland within the GAA context is the 'GAA Super Games Centre' initiative, a theoretically-grounded social support intervention delivered and evaluated for youth participants (N = 103) aged 12 to 16 years over a 24-week period (Lavallee et al., 2019). The Super Game Centres has introduced planned modifications to the standard rules and were informed from six values (i.e., positive feedback, empowerment, belonging, effort, respect, and enjoyment) that have been shown to positively impact youth engagement in sport (Sheridan et al., 2014).

Modified sports programmes should be encouraged in an Irish youth sporting context, as recent findings reveal a high proportion of Irish children participate in organised sport. LGF has recently been recognised as the most attractive sport for 9-11-year-old girls in Ireland (Woods et al., 2018) and the G4G programme which has been running with 12 years, has proven hugely successful in recruiting new young

LGF players since its inception in 2008. To date, the G4G programme has reached 401 clubs with over 12,000 8- to 12- year-old participants (LGFA, 2020). In 2019, 38 LGF clubs representing 22 counties across Ireland and Britain took part in the G4G programme, with over 1600 girls aged between 8- to 12- years old involved. The documented G4G figures are a testimony to the success of the programme, exposing a significant number of young players to LGF and a club setting for the first time. It is, however, important to note a high dropout rate exists with increasing age among young Irish girls (Harrington et al., 2016; Woods et al., 2018). Furthermore, it is not yet fully understood how OYS in female youth could be optimised to facilitate continued participation and increased PA, specifically in a female sport context such as LGF. With no current evidence-base behind the existing G4G programme, coupled with the sharp decline in sport participation evident among young Irish girls, calls for further investigation are warranted.

It is important to note that the existing G4G programme, as run by the LGFA since 2008, is not theoretically designed using research-informed coaching principles. As there is no previous evidence-base behind the existing G4G programme, the Gaelic4Girls intervention study is the first of its kind adding significance to the field. Differences between the existing G4G programme and the revised research-informed G4G intervention study are presented in Table 1.1.

G4G	Existing G4G programme	<b>Revised Research informed G4G study</b>
Components	12 (0 1 1 1	
Player	• 12 x 60-minute weekly sessions	• 10 x 60-minute weekly sessions
	• 6 rotatory LGF skills stations	• 6 rotatory LGF skills stations (i.e.
	(no research-informed sport	adopting a TGfU sport pedagogy
	pedagogy approach)	approach)
	• No weekly LGF Skill Cards	Weekly LGF Skill Cards
	• No specific FMS and Team	• Specific FMS and Team Challenge
	Challenge stations	stations
	• No participation in dance	• Participation in an innovative LGF and
	activities	FMS dance
Coach	One Provisional training	One initial 2-hour Coach Education
	workshop prior to programme	Workshop followed by 8-weekly 1-
	initiation, not theoretical in	hour sessions, theoretical in design
	design	Provision of CPD
	• No provision of CPD	• Utilisation of SAAFE and SHARP
	• No on-demand 'WhatsApp'	research-informed coaching principles
	instant messaging service	(see Figure 6.1)
	G4G Coaching Manual	• On-demand 'WhatsApp' instant
	(hardcopy) with no research-	messaging service
	informed principles	• G4G Coaching Manual (hardcopy and
		electronic resources)
Parent/	• Parents/guardian evening (not	• Parents/guardian evening (theoretical
Guardian	theoretical in nature) on week 4	in nature) on week 4
	No technological resources	• Technological resources – 'WhatsApp'
	available for parents	instant messaging and 'Facebook'
	• No parental involvement in	• Parental involvement in session
	sessions	activities with daughters (week 8)

Table 1.1: Differences between the existing	G4G programme and revised programme	
---	-------------------------------------	--

G4G: Gaelic4Girls; LGF: Ladies Gaelic Football; FMS: fundamental movement skills; TGfU: Teaching Games for Understanding; CPD: continuous professional development
#### 1.1.3 Summary and significance of G4G study

In Ireland, this PhD research study is a unique first of its kind exploratory intervention trial describing the design, development, implementation, and evaluation of the community sports-based revised G4G programme to increase PA participation, FMS proficiency and psychological wellbeing amongst 8 to 12-yearoldIrish girls. Specifically, the combined integration of the player, coach, parent/guardian, community and efficacy of intervention resources are rich in originality and embedded within previously successful intervention evidence. This feasible and cost-efficient intervention programme, which will be discussed more specifically in a later stage of the thesis, is the first of its kind in an Irish 8 to 12year-old LGF population. The revised programme is warranted, innovative and indeed novel and simultaneously targets the increase of PA and FMS proficiency, alongside the enhancement of psychological wellbeing through the deliverable mechanism of weekly LGF sessions, by club coaches, assisted through researchinformed coach educational workshops. The G4G study was initiated at University College Cork in 2016, in accordance with the Medical Research Council (MRC) (2000) framework for trials of complex interventions. Specifically, this framework (as shown in Figure 2.2 in Chapter 2) served as a template for the phases of investigation in the evaluation of the revised G4G intervention programme. This PhD will address the Pre-clinical, Phase 1 (Modelling) and Phase 2 (Exploratory Trial) of this MRC framework.

# **1.2 AIMS AND OBJECTIVES**

The overall aim and objectives of this thesis are presented below. Information relating to the rationale aims, and objectives of specific studies can be found in their respective chapters (Chapter 4, Chapter 5, Chapter 6, and Chapter 7).

# <u>Aim of Research</u>

To design, develop, implement, and evaluate a targeted community sports-based intervention, designed to promote the quantity and quality of participation in PA within and outside of school among 8- to 12- year old Irish girls.

#### Primary Objectives

1. To collect baseline data (2016) from a socio-ecological perspective on PA levels, FMS proficiency, and psychological correlates of PA of 8- to 12- year-old Irish girls (chapters 4 and 5).

2. To use information gathered (from baseline studies) to clarify the relative importance of different types of correlates (psychological, social and environmental variables) and design a theory-based, multi-component, community sports-based Gaelic4Girls'(G4G) intervention, to suit the specific needs for increasing 8- to 12-year-old Irish girl's PA levels, FMS proficiency, and psychological well-being (chapter 6).

3. To evaluate if involvement in the 10-week research-informed G4G intervention leads to an overall increase in PA participation, FMS proficiency, and psychological wellbeing of the experimental group receiving the intervention, in comparison to a second intervention group (receiving the existing LGFA programme) and control group (chapter 7).

4. To establish a research-informed evidence base for the G4G programme and identify potential improvements to the G4G intervention structure through focus group discussion with G4G participants (chapter 7).

# Secondary Objectives

 To examine if an overall relationship between actual FMS proficiency, perceived motor confidence and competence, and PA existed among 8- to 12year old girls (baseline data, chapter 4).

# **1.3 RESEARCH QUESTIONS**

1. What are the PA levels, FMS proficiency levels of Irish girls aged 8- to 12- years (at baseline, 2016)?

2. What is the relationship between actual FMS proficiency, perceived motor skill confidence and competence, and PA levels among 8- to 12- year-old girls (at baseline, 2016)?

3. What are the psychological correlates of PA (enablers and inhibitors) amongst a sample of rural and urban Irish 8- to 12- year-old girls (again at baseline, 2016)?

4. What are the essential components required for developing, designing, and implementing an Irish LGF revised programme G4G, as guided by the literature and baseline data measurements)?

5. Is it possible to increase levels of PA participation, FMS proficiency and enhance psychological wellbeing over time (pre-post-test) in 8- to 12-year-old girls through the G4G intervention, utilising coach pedagogies and parental support structures?

6. What are the potential improvements to the G4G intervention structure among G4G participants post-intervention through focus group discussion with G4G participants?

### **1.4 STRUCTURE OF THE THESIS**

This thesis contains eight chapters along with additional references and appendices. This chapter provides a brief introduction and rationale for the thesis, with further details being provided in Chapter 2. Chapter 2 critically reviews the currently available evidence-based literature on youth PA and sport participation. Chapter 3 provides an overview of the various methods used in the studies, described in Chapter 4 to Chapter 7. Chapter 4 is a cross-sectional baseline study informing the design and development of the research-informed G4G intervention study, and is considered a secondary outcome objective of this thesis. Chapters 5 to 7 are the primary objectives of this thesis consisting of studies relating to the design, development, implementation, and evaluation of the G4G intervention amongst girls aged 8 to 12 years old in Ireland.

14

Each study includes a brief rationale of the purpose of the chapter and its contribution to the field, the author's contribution to the paper, abstract, keywords, introduction, methods, results, discussion, and a chapter conclusion. Any relevant supplementary material (questionnaires, G4G intervention material etc.) are available in the appendices. The studies presented in Chapters 4 and 5 are peer-reviewed published articles. Chapters 6 and 7 are currently under review for publication in Journals.

The following is a breakdown of the specific chapter contents:

# Chapter 1: Introduction

A brief introduction to the phenomenon of physical inactivity among 8- to 12- yearold girls and details of intervention gaps within the literature, conducted to date. The background to the G4G programme within an Irish OYS context is outlined. The overarching aim and objectives of the thesis are also presented at the end of the chapter.

# <u>Chapter 2:</u> Critical Review of Literature

Chapter two presents a critical review of the available literature and is presented in four parts. Part A; details the current high physical inactivity levels of 8- to 12- yearold girls and how the field of sport is a potential avenue to increase PA participation, Part B; highlights the personal and social correlates of PA, Part C; details the importance of FMS proficiency and highlights the current low levels of motor skill proficiency among young girls, and Part D; examines the interventions which have been conducted to-date with this sample, examining both school and community-based interventions. Potential gaps in the evidence are highlighted, and recommendations from other research are considered, justifying the research undertaken in this thesis.

#### Chapter 3: Research Methodology

Chapter 3 describes in further detail some of the methods used across the studies described in Chapters 4 to 7, including details on the study population, ethical considerations, setting, data collection methods, and analysis procedures.

<u>Chapter 4:</u> The relationship between actual Fundamental Motor Skill Proficiency, Perceived Motor Skill Confidence and Competence, and Physical Activity in preadolescent girls

This is a cross-sectional study conducted across three all-female primary schools examining the relationship between actual FMS proficiency, perceived motor confidence and competence, and PA in a sample (n=160) of 8- to 12- year-old pre-adolescent girls. One of the main pillars for the G4G intervention was FMS integration; for this reason, the levels of FMS proficiency and the association of perceived competence and confidence needed to be assessed amongst an Irish pre-adolescent cohort to inform the development of the 'G4G' intervention.

# <u>Chapter 5:</u> Enhancing the Evidence Base for Irish female youth participation in physical activity the development of the Gaelic4Girls programme.

This chapter describes a mixed-method study conducted across three all-female primary schools examining the baseline PA, FMS, and various psychological correlates of PA data of 8- to 12- year-old Irish pre-adolescent youth. Focus group interviews were then used to explore PA barriers and motivators of the cohort. Based on the data analysis, chapter 5 introduces the reader to the development of the 'G4G' intervention, specifically tailored to the needs of this age group. The unique PA, FMS, and psychological component analysis of this chapter has guided the research team and helped inform the development of the G4G intervention, specifically the delivery of the pedagogical coaching approaches utilised as part of the G4G intervention study. Collectively, chapters 4 and 5 combined provided meaningful information for the overall design and development of the G4G intervention.

# <u>Chapter 6:</u> An overview of the G4G intervention – the theory-based design, study components and content.

Chapter 6 introduces the reader to the rationale for the G4G intervention and most importantly, gives an accurate and descriptive account of the intervention components. In chapter 6, the reader will be introduced to the guiding principles of the community sports-based G4G intervention, specifically the participant, parental/guardian and coach education focus.

# Chapter 7: Evidence for the effectiveness of the G4G intervention

Chapter 7 assesses the effectiveness (pre-post) of the multi-component 10-week G4G intervention on PA levels, FMS proficiency and the psychological wellbeing of 8- to 12- year old girls, when compared to a second intervention group (the traditionally delivered, national G4G programme, as run by the LGFA), and a third control condition (received no G4G intervention). G4G programme acceptability was also assessed via focus group interviews.

# Chapter 8: Discussion and Thesis Conclusion

In this final chapter, the findings from the individual studies conducted as part of this thesis are considered and common themes across studies are discussed. The contribution to the current evidence base is highlighted, along with areas for future research and recommendations for policy and practice.

# **1.6 DEFINITION OF TERMS**

*Correlates of Physical Activity:* The determinants/factors which influence physical activity behaviour and participation. They are classified into demographic, personal and biological factors (e.g. age), behavioural factors (e.g. activity history), psychological/cognitive/emotional factors (e.g. attitudes, self-efficacy), social environment factors (support from family and friends) and physical environment factors (e.g. access to facilities) (Martins et al., 2017; Van Der Horst et al., 2007).

*Fundamental Movement Skills:* Fundamental movement skills are the basic observable patterns of locomotor, manipulative, or stability movements underlying more complex sport-specific skills (Gallahue & Ozmun, 2012). Examples, exhibited during PE and PA, include running, hopping, skipping (locomotor), balancing, twisting, dodging (stability), throwing, catching and kicking (object control) (Stodden et al., 2008).

*Ladies Gaelic Football Association:* The Ladies' Gaelic Football Association (Irish: Cumann Peil Gael na mBan) is the main governing body for Ladies' Gaelic football. The Ladies Gaelic Football Association (LGFA), founded in 1974, oversees club registration and organises the National league and championship competitions in Ireland.

*Locomotor Subtest:* In the context of FMS, the locomotor subtest measures the gross motor skills that require fluid coordinated movements of the body as the child moves in one direction or the other (Ulrich, 2000).

*Mastery/Near Mastery:* In the context of FMS, 'mastery' is defined as the correct performance of all components of a skill (Van Beurden et al., 2002). 'Near mastery' is defined as the correct performance of all components but one (Van Beurden et al., 2002).

*Moderate to vigorous-intensity physical activity (MVPA):* MVPA is any activity with a metabolic equivalent (MET) value of between 3 and 5.9 and vigorous-intensity physical activity as  $\geq$ 6 MET. Energy expenditure or METs are the measures of the amount of oxygen consumed while at rest (Ainsworth et al., 2011).

*Object-Control Subtest:* In the context of FMS, the object control subtest measures gross motor skills that demonstrate efficient throwing, striking, and catching movements (Ulrich, 2000).

*Organised Youth Sport:* Organised youth sport is defined as PA that is directed by adult or youth leaders/coaches and involves rules and formal practice and competition (Logan et al., 2019). Community club sports are included in this definition.

*Physical Activity:* Physical activity is described as any bodily movement produced by the skeletal muscles resulting in a substantial increase over resting energy expenditure (Caspersen et al., 1985). Examples of physical activity include play, lifestyle activities such as walking and cycling (active transport), sport and recreational activities, domestic tasks such as household chores and gardening (WHO, 2018).

*Physical Inactivity:* Physical inactivity is defined as engaging in insufficient levels of PA and not meeting the current PA recommendations (Tremblay et al., 2017).

*Preadolescence:* Preadolescence is defined as a stage of human development following early childhood and preceding adolescence. It commonly ends with the beginning of puberty. For example, the age range is commonly designated as 9- to– 13- years old (McNeely & Blanchard, 2009).

*Psychological Correlates of Physical Activity:* Factors influencing girls' PA levels and participation such as perceived competence, self-efficacy motivation, attitude and enjoyment of PA, as well as external factors such as competing priorities, friendship group changes, 'sporty' stereotypes, and family and peer support (Telford et al., 2016; Gillison et al., 2012). In the literature, the terms 'psychological' (Uijtdewilligen et al., 2011) and 'psychosocial' are often used interchangeably when examining the PA correlates amongst youth.

*Reliability:* Reliability refers to the degree to which an instrument yields consistent result (Haradhan, 2017).

*Static Balance:* In the context of FMS, the static balance on one foot is an important non-locomotor skill that is used in gymnastics, dance, diving and many team sports and is defined as being able to maintain a stationary position throughout the movement (NSW Department of Education and Training, 2000).

*Test of Gross Motor Development 2:* A criterion- and norm-referenced FMS instrument designed to measure the process of how children coordinate their trunk and limbs performing FMS (a movement task) rather than assessing the end product result (Ulrich, 2000).

*Validity:* Validity refers to the degree to which an instrument accurately measures what it intends to measure (Liu, 2010).

# **1.7 G4G SCHEMATIC OVERVIEW**

In summary, Figure 1.1 provides a visual map of the thesis highlighting the eight chapters and four associated studies (articles).



Figure 1.1 G4G schematic overview

# CHAPTER 2 LITERATURE REVIEW



# **2.1 INTRODUCTION**

This chapter aims to examine pertinent literature in addressing the aims of this study. This literature review will provide an overview of the importance and prevalence of physical activity (PA) and sport participation in children, specifically among young girls. The chapter comprises four main sections. First, the benefits of being physically active, partaking in organised youth sport (OYS), and the associated health risks if not undertaken will also be considered. The second section reviews the correlates of PA and sport participation among youth, with a particular focus on psychological correlates. The third section reviews the importance of fundamental movement skills (FMS) and its association with PA findings, motor development in children will also be discussed. The fourth and final section summarises the effectiveness of school-based and community sports-based intervention strategies that target the promotion of PA among children, specifically girls. Finally, the theoretical frameworks utilised within this study, and an outline of the theoretical approach and evaluation framework for the intervention will be provided.

# **2.2 PHYSICAL ACTIVITY AND HEALTH**

#### **2.2.1** The importance of physical activity

The World Health Organisation (WHO) (2014) defines physical activity (PA) as any bodily movement produced by the skeletal muscles that substantially increases energy expenditure. PA can be undertaken in many different ways: walking, cycling, sports, active forms of recreation (such as dance, yoga), work (lifting, carrying or other active tasks), and as part of domestic tasks around the home (cleaning, care duties). Indeed regular participation in PA is imperative for good health (Eime et al., 2013), and is associated with an array of physical, psychological and social health benefits (Biddle et al., 2019; Janssen & Leblanc, 2010) for children and young people aged 5 to 18 years old.

It is well established that PA plays a key preventive role in the physical and mental health of children and adolescents (Biddle et al., 2019) including the management, prevention and treatment of the leading non-communicable diseases (NCDs), namely heart disease, stroke, diabetes and breast and colon cancer (WHO, 2010) and chronic conditions such as obesity and depression (Beaulac et al., 2011). PA is also linked to lower rates of antisocial behaviours, increased participation in prosocial activities, improved educational progress (Budd, 2016; Mavilidi et al., 2019) and aspirations (Gray et al., 2014), improved mental health (Schuch et al., 2016), and improved quality of life and well-being (Das & Horton, 2012).

# 2.2.2 Physical activity (PA) guidelines

Whilst PA of any type will deliver health benefits and consequently should be encouraged, it is preferable to achieve a certain level of activity to deliver sufficient benefits (Healthy Ireland Survey, 2015). The WHO (2010) endorse the PA guidelines, stipulating that in order to enhance health, youth (5-17 years) should accumulate at least 60 minutes of moderate-to-vigorous PA (MVPA) daily. The current Irish PA guidelines for health specify that all children and young people (2-18 years) should be active, at a moderate to vigorous level, for at least 60 minutes every day incorporating muscle strengthening, flexibility and bone-strengthening exercises up to 3 times a week (Department of Health, 2016). Other European countries align with the WHO guidelines including France, Norway, Sweden, and the United Kingdom (UK) (Davies et al., 2019; Kahlmeier et al., 2015).

Compared to the WHO (2010) recommendation, the minimum recommended duration of PA for children and young people (5-18 years) is lower in Malta (30-60 minutes) (Micaleff, 2010) and higher in Finland (1-2 hours) (Gråstén et al., 2014). Switzerland (Martin et al., 2009) and the United Kingdom (UK) (Davies et al., 2019) also recommended more PA for young children, which are, however, not included in the WHO recommendations. In addition to this, the most recent Australian 24-Hour Movement Guidelines for Children (5-12 years) and Young People (13-17 years) endorse an accumulation of 60 minutes of MVPA daily (Australian Department of Health, 2019). Likewise, the most recent Canadian 24-Hour Movement Guidelines for Children sfor children and youth (5-11 years), recommending at least 60 min/day of MVPA for disease prevention and health promotion (Lau et al., 2016). Similarly, the most recent and widely endorsed PA guideline for children and young people in the USA stipulates that to enhance health and fitness benefits, youth should accumulate at least 60 minutes of MVPA daily (Department of Health & Human Services, 2018).

To summarise, many of the recent PA guidelines worldwide for children and youth have seen the commonality of a minimum of 60 minutes PA per day, with some countries endorsing recommendations of more and less PA, and other countries advocating the inclusion of activities to strengthen muscle and bone.

#### 2.2.3 Global prevalence of PA in children and youth

Globally, according to recent data obtained from 146 countries, 81·0% of children (77·6% of boys and 84·7% of girls) aged 11 to 17 years fail to meet the recommended public health guidelines of 60 minutes of MVPA per day (Guthold et al., 2019). According to WHO (2018) worldwide, 1 in 4 adults, and 3 in 4 adolescents (aged 11– 17 years), do not currently meet the global recommendations for PA. Results and analysis from 49 countries in the most recent Global Matrix 3.0 of Report Card Grades on PA for Children and Youth (Aubert et al., 2018) reported that an average low grade of "C–," D+," and "C–" was obtained for the low and medium Human Development Index (HDI) countries (e.g. India and South Africa), high HDI countries (e.g. China, Mexico, Thailand), and very high HDI countries (Australia, England, USA), respectively. The C- and D+ grades reported specifically indicate that between 34-46% of children and youth from the participating 49 countries met the recommended minimum PA guidelines for health (Aubert et al., 2018). In line with this, according to Tremblay et al. (2016), overall grades for children's PA in a worldwide report card were low/poor.

In an Irish context, the most recent 2016 Report Card on PA (a national document, containing all data on indicators related to children's PA levels from Northern and Southern Ireland), awarded children and youth an overall grade of 'D' for low PA participation (Harrington et al., 2016). These findings are accumulative national evidence, which specifically indicates that only 25% of 9-year-olds and a further 19% of primary school children in Ireland are meeting the recommended minimum PA guidelines for health (Harrington et al., 2014, 2016). Most recent nationally representative Irish data (n=6,651; mean age=13.8  $\pm$  2 years) from the 'Children's Sport Participation and Physical Activity Study' (CSPPA) (Woods et al., 2018) found that only 17% of children (n=6,651) aged 9-11 and 10% of adolescents

aged 12 to 17 years old self-reported meeting the recommended 60-minute MVPA daily. Other findings from the most recent Health Behaviour in School-Aged Children (HBSC) study (Költő et al., 2020) found that in Ireland, only 23% of children from 5th class to 5th year who were aged 10 to 17 years (N = 12,002) report being physically active. Growing Up in Ireland national longitudinal study of children found that only one-quarter (25%) of 9-year-olds met the recommended level of PA – 60+ minutes every day. The percentage meeting this recommended level of PA was even lower for girls (22% compared to 28% of boys).

Similarly, a systematic review of PA levels in Canada and the US observed between 26.5% (self-report) to 45.7% (pedometry/accelerometry) of youth (<18 years) meeting the recommended guidelines (Foulds et al., 2013). The newly released 2018 United States Report Card on PA for Children and Youth findings align with Irish studies reporting approximately 24% of children 6 to 17 years of age participating in the recommended 60 minutes of PA every day, earning a D- in overall PA (Katzmarzyk et al., 2018). Similarly, according to Australian data from the Active Healthy Kids Report (2014), less than one-fifth of Australian youth aged 5 to 17 years old are meeting the recommended PA guidelines. The prevalence of PA amongst Irish adolescents is also very low when compared in a European context with 35.9% of adolescents (14.45  $\pm$  0.67 years) in France and Spain meet the 60-minute guidelines (Aibar et al., 2013) when measured by ActiGraph GT3X accelerometry. The most recent 2016 Report Card on PA in Belgium awarded children and youth a lower overall grade of 'F+' for overall PA levels (Wijtzes & Verloigne, 2016), indicating that less than 20% of children and youth are sufficiently active and are not meeting the recommended PA guidelines.

Regarding the reported levels of youth PA across European countries, a recent systematic literature review (Van Hecke et al., 2016), inclusive of 30 articles of cross-European studies, reported that there is substantial variability between countries in overall levels of PA and in the prevalence of compliance to recommended PA levels in youth (Van Hecke et al., 2016). Specifically, results revealed large differences between countries in the prevalence of compliance to PA recommendations (i.e. 60 min of daily MVPA) measured subjectively (5 to 47%) and objectively (ranged from 0 to 60%, with accelerometer measured minutes of daily MVPA ranging from 23 to 200 min) among

children and adolescents. Overall, boys and children were more active than girls and adolescents (Van Hecke et al., 2016).

# 2.2.4 Gender differentials in PA

Further to these low levels of youth PA participation, global (Inchley et al., 2020; Marques et al., 2016; Pearce et al., 2012; Silva et al., 2013; Telford et al., 2016; Van Hecke et al., 2016) and Irish (Költő et al., 2020; Woods et al., 2018) specific evidence has shown a consistent gender-based disparity, whereby girls are significantly less active than boys (Marques et al., 2016; Pearce et al., 2012; Silva et al., 2013; Telford et al., 2013; Telford et al., 2016; Van Hecke et al., 2016). These differences are meaningful across all age groups and in nearly all countries and regions (Demetrio et al., 2019). Interestingly, the difference in PA is greatest for vigorous PA, less marked for moderate PA and does not exist for light PA (Bucksch et al., 2014). A clear decrease in PA participation during adolescence has been further observed (Cooper et al., 2015; Jago et al., 2017), with a higher risk of decline amongst girls (Owen et al., 2017; Ortega et al., 2013).

# 2.2.4.1 Declining levels of PA from childhood to adolescence

Many international studies acknowledge age-related differences in PA levels, with a noticeable decline occurring during adolescence and into adulthood (Kemp et al., 2019; O'Donovan et al., 2010), specifically for girls (Van Hecke et al., 2016). For example, the Health Behaviour in School-Aged Children (HBSC) study (Gavin et al., 2015) found that 31% of females aged 11 years old and 41% of males the same age (N=13,611; mean age: 13.5) self-reported accumulating at least 60 minutes of MVPA daily. By age 14, a substantially lower 19% of females and 32% of males report meeting this PA guideline (Gavin et al., 2015). In Ireland's most recent HBSC study (Költő et al., 2020), a higher percentage of boys (28%) aged 10 to 17 years are more likely than girls (18%) of the same age to report being physically active on 7 days in the last week (n=12,002). Further to this, self-report data from the 2018 Active Healthy Kids Australia (AHKA) PA Report Card, show 18% of 12-17-year-olds; 6-22% of 15-17-year-olds; and 15-41% of 5-17-year-olds accumulate 60 mins of MVPA every day (or on average) in the past week (Schranz et al., 2018).

Nevertheless, while levels of MVPA among adolescents are typically much lower than recommended (Hallal et al., 2012) there is very little evidence that MVPA declines with the onset of adolescence, or that adolescent declines in MVPA are more marked in girls than boys (Farooq et al., 2018). Dumith et al.'s (2011) systematic review (with 26 eligible longitudinal studies) of changes in PA during adolescence aligns with this statement. Findings indicated that although the decline among girls was higher in younger ages (9–12 years), it was higher in older ages (13–16 years) among boys. Of these eligible studies the majority were assessed by questionnaire; only 2/26 included objective measures of the intensity of PA (accelerometry), most (16/26) were from the USA, and the representativeness of the samples in the eligible studies was usually unclear (Dumith et al., 2011). In addition to this, the National Centre for Health Statistics (2014) reported that 27% of boys engaged in MVPA for at least 60 minutes a day, compared with 22.5% of girls, further highlighting the gender disparity in PA.

#### 2.2.4.2 Gender division during primary and post-primary school years

In the Irish CSPPA study (Woods et al., 2018), only 13% of CSPPA participants reported meeting the national PA guidelines of at least 60 minutes of MVPA daily. This 13% was made up of 17% of primary pupils and 10% of post-primary pupils, with fewer girls meeting the PA guidelines, compared to boys (9% vs. 17%). This gender difference was evident at primary school (13% vs. 23%) and post-primary school (7% vs. 14%). Telford et al, (2016) investigated gender differentials in PA across 29 schools in a recent Australian youth study (N=276 boys, 279 girls; aged 8 to 12 years old) and reported that males were significantly (p<0.001) more active than female youth. In addition to this, the National Centre for Health Statistics (2014) reported that 27% of boys engaged in MVPA for at least 60 minutes a day, compared with 22.5% of girls, further highlighting the gender disparity in physical activity. Similarly, a recent Chinese adolescent study (N=250, mean age = 15.55 ± 0.72 years), reported that 80% of participants were insufficiently active (Chen et al, 2014) with males significantly more active than females with regard to meeting the PA guidelines (25.1% male, when compared to 14.6% females) (Chen et al., 2014).

Key findings from the Girls Active Survey (2017) from 138 secondary schools in England and Northern Ireland also reported low levels of PA, revealing that only 8% of girls, compared to 16% of boys (aged 11-18 years) meet the Chief Medical Officer's recommendation that young people aged 5 -18 should do 60 minutes of PA every day (n= 26,000) (Youth Sport Trust, 2017). Consistently, research indicates the low levels of PA, the associated age-related decline in PA, and the apparent gender differences of girls being significantly less active than boys worldwide (Ishii et al., 2015; Pearce et al, 2012; Sallis et al., 2016). Based on the consistently reported research informed literature, girls have been identified as a high priority population for childhood PA promotion (Biddle, et al, 2014).

#### 2.2.5 Physical inactivity

Despite the widely-known health benefits of PA and recommended guidelines, studies continue to show that physical inactivity, defined as engaging in insufficient levels of PA and not meeting the current PA recommendations (Tremblay et al., 2017), has become a global health concern among children and adolescents (Guthold et al., 2019). Physical inactivity (PI) has been well established as one of the leading risk factors for non-communicable (not contagious and is non-transmissible) disease (Andersen et al., 2016) and has been strongly associated with many major diseases worldwide, such as obesity, diabetes, cardiovascular disease and some cancers) (Lee et al, 2012; Haileamlak, 2019).

PI among school-aged children and youth has found to be associated with a myriad of unfavourable and adverse physical, mental, social, and cognitive health outcomes (Janssen & LeBlanc, 2010; McMahon et al., 2017; Poitras et al., 2016), lower physical fitness, and lower PA levels in later life (Telama et al., 2005). PI is the fourth leading cause of death worldwide (Kohl et al., 2000), and is estimated to be associated with one million deaths per year in the European Region and five million deaths annually worldwide (Guthold et al., 2018). Globally, PI is estimated to cost INT\$ 54 billion in direct health care, in 2013, of which 57% is incurred by the public sector and an additional INT\$ 14 billion is attributable to lost productivity (Ding et al., 2016). A recent systemic review (de Moraes et al., 2013) showed that the worldwide prevalence of insufficient physical activity (IPA) varied from 18.7% to 90.6%, with a median of 79.7%.

#### 2.2.5.1 Sedentary Behaviour

As previously mentioned, recent evidence also highlights that the amount of time spent being sedentary is an important risk factor for several aspects of ill health, including being overweight, obesity and associated metabolic diseases (Lavie et al., 2019). Sedentary Behaviour (SB) has been defined as any waking behaviour characterised by a low energy expenditure  $\leq 1.5$  metabolic equivalents of task (METs), while in a seated, reclined or lying posture (Tremblay et al., 2017) (e.g., prolonged sitting, seated screen time, motorised transportation). Interestingly, as levels of MVPA decrease during childhood (Nadar, 2008), the time children engage in sedentary leisure-time pursuits increases (Pate et al., 2011).

Several guidelines are recommending less than two hours of recreational screen time per day in youth (Arundell et al., 2016; Australia's Physical Activity and Sedentary Behaviour Guidelines and the Australian 24-Hour Movement Guidelines, 2019; Tremblay et al., 2011). Likewise, the UK guidelines recommend minimising the time spent sedentary for extended periods (UK Chief Medical Officers' Physical Activity Guidelines, 2019). Most recently, the Australian 24-Hour Movement Guidelines for Children (5-12 years) and Young People (13-17 years) advocate that children and young people (aged 5–17 years) should limit sedentary recreational screen time to no more than 2 hours per day for optimal health benefits (Australian Government Department of Health, 2019). Despite these recommendations, many studies now provide evidence that a high proportion of children and adolescent youth engage in high amounts of daily television viewing, video games, with screen time recreation (defined as two or more hours of television, computer and/or video game usage) being the most common (Barnett et al., 2018; Houghton et al., 2015; Robinson et al., 2017; Suchert et al., 2016).

Findings from the American Heart Association Scientific Statement found that school-aged children are sedentary for 8 of their daily waking hours on average; most are engaging in excessive screen time (Barnett et al., 2018). In Ireland, the CSPPA 2018 study (Woods et al., 2018) reported that the average sedentary leisure time was 5.1 hours/day for primary school children (n = 1549) and 6.6 hours/day for post-primary children (n = 5102) with 63% percent of primary and 42% of post-primary pupils met the sedentary screen time guideline of no more than 120 minutes (Woods

et al., 2018). Similar findings were reported in Ireland's 2014 and 2016 Report Cards (Harrington et al., 2014; Harrington et al., 2016), with SB awarded a grade C-, highlighting that 46% of children watching < 2hrs TV/day and SB awarded a grade C-. The literature suggests that there is a clear need to increase the current PA levels and decrease sedentary behaviours among children and adolescents worldwide (Guthold et al., 2019; Kumar et al., 2015; Woods et al., 2018).

It is important to note that screen time increases substantially with age, most notably during preadolescence and according to a scientific statement from the American Heart Association, adolescents are the most sedentary of paediatric populations and are engaging in the most total recreational screen-based media (Barnett et al., 2019). According to recent projections by the WHO Regional Office for Europe, collaborating with the UK Health Forum, levels of obesity are forecast to increase globally, and Ireland could have one of the highest rates of obesity in Europe by 2030 (Bel-Serrat et al., 2017). Evidence in Ireland and worldwide, highlight that there is a need to develop specified SB guidelines for children and youth (Bull et al., 2020; Woods et al., 2010). This is particularly important in Ireland and for girls, as no current recommendations on the maximum amount of sedentary time exist.

# 2.2.6 Participation in Organised Youth Sport (OYS) – an avenue to increase youth PA

Sport is a popular type of leisure-time PA, especially for children (Basterfield et al., 2015; Eime et al., 2016) which is often organised, usually competitive and played as part of a team or as an individual (Eime et al., 2013). Organised youth sport (OYS) participation is one such strategy with the existing potential to increase overall PA levels in young people (Howie et al., 2018; Wiium & Säfvenbom, 2019), and is positively associated with an increased likelihood of complying with national PA and SB guidelines (Dalene et al., 2019;Vella et al., 2013). Specifically, participation in OYS has been recommended as an opportunity to increase young peoples' MVPA levels (Guagliano, 2014; Lee et al., 2018), and was identified by the International Society for PA and Health as "an investment that works" to promote PA (Trost et al., 2014).

Community sport participation during leisure time promotes PA for children and adolescents; hence findings associating it with improved physical health, and positive psychological and social health outcomes (Eime, 2013). Sports club membership predicts higher levels of leisure-time PA engagement, specifically among girls (Marques et al., 2016), and OYS participation increases the likelihood of meeting daily PA recommendations for improved health (Kokko et al., 2018). In a recent study (Marques et al., 2016), specifically investigating associations between OYS participation and objectively measured PA (N=973, mean age =  $14.1\pm2.4$ ), results reported that those who were engaged in OYS were more likely to achieve PA guidelines (p<0.01), spend more time in moderate PA (MPA) (p<0.01), vigorous PA (VPA) (p<0.001) and MVPA (p<0.001), than those who did not participate in OYS. Further to this, more boys (51.3%) than girls (28.3%) were reported to be involved in OYS participation (p<0.001).

Organised sports have a prominent place in the Irish identity (Federation of Irish Sport, 2013). Encouragingly most recent Irish representative data revealed that 75% of primary school children (aged 10-12 years) and 67% of post-primary children (aged 12-18 years) reported being members of at least one community sports club (Woods et al 2018). A further 80% of primary and 58% of post-primary school pupils reported participating in community sport at least once a week (Woods et al., 2018). Organised sport participation was graded at a C- in Ireland's 2014 and 2016 Report Cards (Harrington et al., 2014; Harrington et al., 2016), indicating between 33 – 64% of children participating in OYS. In a European context, Kokko et al., (2018) study, comparing sports club participation in children and adolescents across six European countries, namely Belgium (Flanders), Czech Republic, Finland, France, Ireland and Sweden indicated that approximately two-thirds of children and adolescents take part in sports club activities in the given countries.

# 2.2.6.1 Community club sport and PA

It is also argued that early participation in specific types of OYS in childhood is associated with higher levels of PA during adolescence (Basterfield et al., 2015; Gallant et al., 2017) and into adulthood (Bélanger et al., 2015; Hirvensalo & Lintunen,

2011). Although it does not necessarily prevent the decline in PA during adolescence (Belanger et al., 2009), youth participation in organised PA such as hockey, soccer, volleyball, martial arts, track and field, dancing and gymnastics is associated with higher PA levels in adulthood (Telama et al., 2005; Kjønniksen et al., 2008). For example, Kjønniksen and colleagues (2008) conclude that if young people are members of sports clubs from an early age, they are more likely to be physically active in early adulthood, especially if they also are members during adolescence (Maillane-Vanegas et al., 2017).

Similarly, results from a 13-year study (Belanger et al., 2015) on 12-13-yearold Canadian adolescents (N = 673), revealed a positive linear relationship between the number of years participating in sports and running in adolescence and PA level at age 24 years ( $\beta$  (95 % confidence interval) = 0.09 (0.04-0.15); 0.08 (0.01-0.15), respectively). Authors of this study suggest that efforts to establish the habit of participation in sports and running in adolescence may promote higher PA levels in adulthood (Belanger et al., 2015). Engström (2008), in a 38-year follow-up study, found different results, concluding that neither membership nor amount of time spent on sports at the age of 15 has any significant association with exercise habits in middle age. Engström's study instead indicates that a broad and varied experience of sports during childhood and adolescence has an impact on exercise habits later on in life.

# 2.2.6.2 OYS participatory benefits

Participation in OYS during childhood and adolescence has important benefits for physical, psychological, and social health (Eime et al, 2013; Mills et al., 2019). Such benefits include, higher rates of physical fitness (Hoffman et al., 2005), greater involvement in PA over time (Nelson et al., 2011), lower rates of sedentary behaviour (Vella et al., 2013), higher rates of self-esteem, improved skill development, heightened levels of muscular strength and endurance (Felfe et al., 2016), more positive social interaction, decreased levels of depressive symptoms (Eime et al, 2013) and lower rates of obesity (Drake et al, 2012). In addition to this, OYS can be an avenue for children to develop increased positive social skills like cooperation, responsibility, empathy and self-control, as well as promoting good citizenship,

positive peer relations, leadership skills, and a sense of initiative (Bailey et al., 2009; Coté et al., 2006; Fraser-Thomas et al., 2005).

A number of studies have recently been published examining OYS clubs as a setting to promote health (Geidne et al, 2013; Kokko et al., 2019; Moeijes et al., 2019). Many studies suggest that childhood participation in community-based sport or PA may lead to enhanced psychosocial development and physical health outcomes (Cairney & Veldhuizen, 2017; Moeijes et al., 2018). One longitudinal study of club sport participation over a three-year period during adolescence in Germany, as well as identifying physical benefits, showed that sport club activities had a positive influence on the development of self-esteem, with girls discovering sports as a source of self-esteem earlier than boys (Brettschneider, 2001). Similarly, in a qualitative parental study of young people participating in sport, a range of personal, social, and physical benefits factors, as well as life skills and self-concept, were stated as benefits of participation (Neely & Holt, 2014).

# 2.2.6.3 Declining participation rates among youth

Despite the associated benefits, there is extensive research showing that participation in OYS and PA dramatically decreases during adolescence (Beltran-Valls et al., 2019; Eime et al., 2016; Farooq et al., 2018). There is evidence that participation in sport peaks at around 11 to 13 years of age, before declining through adolescence (Zimmermann-Sloutskis et al., 2010). Similarly, Eime et al. (2013) propose there is evidence that sports participation is a young persons' activity, with reports that participation levels peak at ages 12–13 years (Olds et al., 2009). In Ireland, the majority of children participate in organised sport, and most begin before 10 years of age (Woods et al., 2018). However, research indicates that dropping out from OYS increases during adolescence in Ireland, as in other countries globally (Bélanger et al., 2009; Eime et al., 2019; Lagestad & Sorensen, 2018).

In an Australian study of a total of 465,403 sport participants, it was reported that the participation rate per resident population in Victoria, Australia was 40% for ages 10–14 and 23% for 15–19 years (Eime et al., 2016). In the Eime et al., (2013)

study, older adolescent (16–18 years) females (n = 732) shifted their participation away from organised, competitive modes and settings towards non-organised and noncompetitive modes and settings and were more likely to then participate in individual types of PA. Another study by Eime et al., (2016), reported that from a peak through ages 10–14 (28% of all participants), participation declined dramatically during ages 15-19 years (15% of all participants) (Eime et al., 2016). Similar findings were revealed in a study in Europe focusing on children aged 9 to 15 years showed sports participation decreased across all ages in all countries (Riddoch et al., 2004). A recent longitudinal study (Eime et al., 2020) including registered women and girls of all ages (4-96 years at baseline) in an almost exclusively female sport, found that most participation (almost one third) occurs between the ages 4-14 years. This proportion diminished through ages at commencement from 10 to 19, reaching a low point of 4% for ages 15–19, then rebounded slightly, reaching 7% for ages 30+. In general, the above research provides evidence that as age increases participation in sport decreases. However, these studies are often limited to self-report sample surveys and/or to specific age ranges (Eime et al., 2016).

In Ireland, even though a high proportion of children participate in organised sport, a high dropout rate exists with increasing age (Harrington et al., 2016; Woods et al., 2018). For instance, new data from Woods et al., (2018) reveal that similar to active participation in community sport, club membership levels also decline, in overall terms and the numbers of clubs, as children progress through their school years. While less than one quarter (24%) of pupils in first year (aged 12-13) of post-primary are not members of a sports club this has increased to 47% by the time they get to sixth year (ages 17-19).

# 2.2.6.4 Gender differences in OYS among youth

Gender differences favouring boys are especially prevalent with regard to structured PA and sports or vigorous PA (Pizarro et al., 2017; Schmidt et al., 2019). Furthermore, boys participate in sport more frequently than girls and are more physically active from childhood into adolescence (Owen et al., 2017; Trang et al., 2012), and girls have been shown to participate less in organised sport (Vella et al., 2014). Data from HBSC

showed that 75% of 10 to 14-year-old boys report playing with a club compared to 59% of girls (Gavin et al., 2015). In an Irish context, at post-primary level, 38% of pupils reported never participating in community sport, with non-participation among girls now standing at 45% (Woods et al., 2018). Results from the "Keeping Them in The Game" study (Lunn et al., 2013) in Ireland report that 1 in 10 girls drop out of playing regular sports during the primary to secondary school transition (12–13 years old). This constitutes a tremendous dropout rate and is especially pronounced among girls and adolescents (Eime et al., 2020). To prevent dropouts, identifying and understanding the precise factors that contribute to dropping out among 8- to 12- year-old girls is critical.

# 2.2.7 Factors influencing 'drop-out' in OYS

The extant literature has found that dropouts may be attributed to a lack of enjoyment, lack of the opportunity to play, lower perceptions of competence, poor performance, social pressure, slower maturation, injuries, lack of fulfilment of basic psychological needs, and poor relationships with teammates and/or coaches (Crane & Temple, 2014; Lagestad & Sorensen, 2018). Basterfield et al. (2016) longitudinal study of children and adolescents perceived barriers to sports participation in the Gateshead Millennium Study (n>500) found that at 12 years, perceived barriers were predominantly classed as intrapersonal ('they're boring') or social environmental ('my friends don't go'). Similarly, in a recent systematic review (Somerset & Hoare, 2018) on barriers to voluntary participation in sport for children (up to 18 years of age), frequently reported barriers across both quantitative and qualitative studies (n=22) included 'time, 'cost, 'opportunity/accessibility' and 'friends', 'not being good at sport' (n = 6) and 'fear of being judged/embarrassed'. Results from a systematic review of dropout from organised sport among children and youth (n = 1125, 001) revealed five major areas emerged: 1) lack of enjoyment, 2) perceptions of competence, 3) social pressures, 4) competing priorities and 5) physical factors (maturation and injuries) (Crane & Temple, 2014).

#### 2.2.7.1 The 'problem' with girls' – motivators and barriers to sport participation

Underlying motivators and barriers to whether or not an individual engages in PA is important to investigate (Hoare et al., 2017), particularly for girls as it is widely reported that girls are less physically active than boys throughout childhood (Sallis et al., 2016), and the age-related decline in PA participation, particularly from early adolescence onwards, is steeper for girls than for boys (Owen et al., 2017). Current research on female youth reveals that having fun (Pawlowski et al., 2014; Visek et al., 2015), keeping fit, and being with friends (Carlin et al., 2015) are key motivators for female youth PA participation. Girls' perception of benefits of PA (e.g. the benefit of being healthy and body image) has also shown to be positively associated with PA (Cheng et al., 2003; Haerens et al., 2007).

Other studies have shown that internal factors such as confidence, pressure, embarrassment, or negative self-belief, and external factors such as lack of female role models, and other hobbies or commitments that result in a lack of time or preference for sport are the main barriers for female youth (Wetton et al., 2013). Additionally, the weather (Pawlowski et al., 2014), access to facilities, and lack of enjoyment have also been shown to be key barriers among female youth. Similarly, self-report and focus group findings from numerous studies (Perry et al., 2011; Verloigne et al., 2016) indicate that social pressures, namely parents, coaches, and peers, are reported as participatory barriers. Finally, a higher perception of barriers to be active is associated with less PA among girls (Zaragoza et al., 2011). Moreover, resources facilitating (positive encouragement/praise) or impeding engagement (negative coach relationship) among 8–12-year-old female youth are deemed important findings in the context of participatory barriers in OYS (Battaglia et al., 2017; Fenton et al., 2017; Laird et al., 2018).

In order to develop and implement targeted strategies to increase team sport participation and, consequently, PA levels among young girls, there is a need to understand trends in, and influences on, 8- to 12- year-old girls' participation in team sport (Allison et al., 2017). Examining girls' PA perceptions, motivations, barriers, and psychological correlates (for example, self-efficacy and attitudes) through in-depth qualitative and quantitative research allows for greater understanding of the rationale behind their PA-related choices (Whitehead & Biddle, 2008).

In the introductory section to this literature review, PA for youth, with a specific focus on female youth, was contextualised and rationalised under specific subheadings. The findings strongly indicate that participation in regular PA by those aged 5-18- years of age is associated with an array of physical, psychological and social health benefits (Biddle & Assare, 2011; Das & Horton, 2012; Owen et al., 2017). In terms of the recommended PA guidelines for health, it has become widely accepted in the literature and across many countries that all children and young people should engage in moderate to vigorous intensity PA for at least 60 minutes every day (Australian Department of Health, 2019; Kahlmeier et al., 2015). The review progressed on to assessing the current global levels of PA among youth and gender differentials and evidence suggests that levels of PA among youth remain low (Woods et al., 2018; Guthold et al., 2019). Most notably, levels of PA decline dramatically during adolescence (Kemp et al., 2019), specifically for girls (Van Hecke et al., 2016; Woods et al., 2018). Most recent evidence highlights that the decline in MVPA and increased sedentary time observed from childhood to adolescence are of concern and might increase the risk of developing obesity and other chronic diseases later in life (Ortega et al., 2013).

OYS as an avenue to promote PA among youth was also discussed, specifically outlining the participatory benefits, with many studies suggesting that childhood participation in community-based sport or PA may lead to enhanced psychosocial development and physical health outcomes (Felfe et al., 2016); Moeijes et al., 2018). In Ireland, even though a high proportion of children participate in organised sport, a high dropout rate exists with increasing age (Harrington et al., 2016; Woods et al., 2018), with a higher rate of decline among girls. Finally, underlying motivators and barriers to whether or not girls engage in PA were investigated in section 2.2.8. Furthermore, to develop and implement targeted strategies to increase team sport participation and, consequently, PA levels among girls, there is a need to understand trends in, and influences on, pre-adolescent girls' participation in a team sport (Allison et al., 2017), specifically the correlates of PA, in particular, the psychological (psychosocial), social support and FMS influences on PA participation.

# **2.3 CORRELATES OF PA (PERSONAL AND SOCIAL FACTORS)**

#### 2.3.1 An introduction to PA correlates

To design more effective intervention programmes and strategies to promote physically active lifestyles, especially in the community club sport environment, it is important to know the main 'determinants' (correlates) associated with PA behaviour in youth (Lounsbery & McKenzie, 2015). Although there are several reviews (Van Der Horst et al., 2007) and reviews of reviews (Bauman et al., 2012; Sterdt et al., 2014) regarding the main correlates of PA, few studies summarise the findings connecting them to community sport and lack a clear identification of correlates for girls. The correlates of PA identified in the reviews amongst youth have been classified into the following factors:

- Demographic, personal and biological (age, sex, ethnicity, genetic influences, overweight/obesity).
- Behavioural (activity history, dietary habits, processes of change, coping skills).
- 3) Psychological/cognitive/emotional (attitudes, perceived barriers, enjoyment, intention to be active, self-efficacy, perceived competence, self-motivation)
- 4) Social environment (support from family friends, parent PA).
- 5) Physical environment (access to facilities, availability of conducive environments) (Liangruenrom et al., 2019; Van Der Horst et al., 2007).

Many variables affect youth PA participation, with substantial evidence documenting how children and young people are influenced by their parents, peers, and siblings (Laird et al., 2018; Reimers et al., 2019; Salvy et al., 2012), as well as their environment (Pouliou et al., 2015; Van Der Horst et al., 2007). Determinants of PA are increasingly being understood using socio-ecological models, whereby intrapersonal/individual (e.g., biological, psychological, and behavioural aspects), interpersonal (e.g., relationships with parents, relatives, peers, and socio-cultural networks), environmental (e.g., access/availability of tools/services, and proximal/distal built/natural surroundings), and policy (e.g., organisational and governmental aspects) variables are identified as influences on participation (Martins et al., 2017; Niemistö et al., 2019; Vancampfort et al., 2019).

However, heterogeneity in the design of correlates research has hampered understanding of those factors that are most consistently predictive of PA in young people, with studies differing according to sample characteristics, measurement instrumentation and the context of PA under examination (Atkin et al., 2016). For example, correlates of PA have been shown to vary according to gender (Lawman et al., 2011; Sterdt et al., 2014), age (Van Sluijs et al., 2007), and socioeconomic status (SES) of study samples (Stanley et al., 2012). Correlates are also shown to differ according to the approach to PA assessment, such as objective measures and selfreport (Wenthe et al., 2009), and PA contexts, such as free play and organised sport (Stanley et al., 2014). Despite this, analysing the correlates of PA from a socioecological perspective may help to identify factors that promote versus discourage PA (Wilkie et al., 2018). It is important to note that pre-adolescence and adolescence is a crucial period of life during which important shifts in psychosocial behaviour occur and many future health behaviours are established (Mohamadian & Ghannaee Arani, 2014).

# 2.3.2 Psychological correlates of PA - factors influencing girl's PA engagement and experiences

Factors influencing girls' PA levels and participation include psychological correlates, such as perceived competence, self-efficacy motivation, attitude and enjoyment of PA, as well as external factors such as competing priorities, friendship group changes, 'sporty' stereotypes, and family and peer support (Gillison et al., 2011; Reimers et al., 2019; Telford et al., 2016; Slater & Tiggemann, 2010). In the literature, the terms 'psychological' (Uijtdewilligen et al., 2011) and 'psychosocial' are often used interchangeably when examining the PA correlates amongst youth, however for this literature review and thesis, the specific terms as used in each of the individual research papers will be referenced. Systematic reviews have synthesised evidence on correlates associated with PA in adolescent girls (Biddle et al., 2011; Laird et al., 2016). These reviews have identified personal and demographic, psychological, environmental and social correlates to be consistently associated with PA in girls aged 8

to 12-years specifically may inform more effective intervention design for this vulnerable population (Sallis et al., 2000).

Nonetheless, this area of research is complex given the number of settings likely to influence children's PA behaviour (e.g. home, school and neighbourhood environments), as illustrated by socioecological model approaches (Sallis et al., 2006). It is therefore important to consider a wide range of correlates from multiple domains, given that focusing on a single domain may give rise to inaccurate conclusions (Brodersen et al., 2005). Despite this, there is a lack of studies simultaneously exploring a range of potential correlates from multiple domains (van Sluijs et al., 2013). One personal factor that may be particularly important for young girls is self-efficacy (Roesch et al., 2013).

Over the past decade social-cognitive research has specifically identified selfefficacy (SE) for PA as a strong correlate of PA among children, youth and adults (Dishman et al., 2005; Liangruenrom et al., 2019; Vitali et al., 2019). Perceived SE is defined as "*a judgment of one's capability to accomplish a certain level of performance*" (Bandura, 1986, p. 391). SE with respect to PA stands for a child's belief concerning his or her skills for successfully completing the task associated with PA (Ray & Henry, 2011). Increasing SE through PA is considered to be an important part of health improvement efforts against physical inactivity (Bauman & Craig, 2005). In other words, improved SE may lead to higher levels of PA at some later point in time (Pekmezi et al., 2009).

In the context of OYS, SE has been extensively examined (Li et al., 2018; Morano et al., 2019; Plotnikoff et al., 2014; Schroeder et al., 2020). Results from a systematic review of reviews (Sterdt et al., 2014) reported that three reviews (Biddle et al., 2005; Lubans et al., 2008; Van der Horst et al., 2007) found a consistently positive association between SE and participation in sport by adolescents. In Wang et al. (2016) recent study of 8–13-year-old Chinese children (n = 449; 252 males, 197 females), SE was positively related to PAQ-C and objective MVPA (r: 0.22–0.63). Additionally, Bauman et al. (2012) also considered SE a significant determinant in PA of children and adolescents. Results from a recent Canadian longitudinal study (Li et al., 2018), examining the relationship between generalised SE and PA in school-aged children (n= 2278; 1120 girls and 1158 boys), showed that children with high perceived SE had higher free play and organised activity participation relative to other children over time.

De Bourdeaudhuij and Sallis (2002) demonstrated that psychosocial variables, including SE, vary in importance according to sex and age groups. SE beliefs, in general, have been shown to influence girls' PA (Duncan et al., 2005; Motl et al., 2002; Morano et al., 2019; Neissaar & Raudsepp, 2011), and has indeed been identified as one of the most important correlates of adolescent girls' PA (Graham et al., 2014): a higher feeling of SE, or beliefs in one's capability to be physically active, was related to more PA among girls. Findings from the Lubans et al (2012) study on 1518 adolescent girls, observed that SE was the most strongly associated correlate with PA self-reported PA behaviour. Furthermore, SE appears to be particularly salient for adolescent girls (Chen et al., 2017; Young et al., 2014). Dishman and colleagues (2005) reported that the positive effect of a PA intervention targeting adolescent girls was partially mediated by increased efficacy beliefs about PA participation. According to Bandura (1997), SE not only influences whether a person will engage in a particular behaviour but also whether he or she will persist in their efforts in the face of barriers. Thus, SE may be an important mediator for girls' enjoyment and engagement in PA and warrants further investigation (Spence, 2010).

A less-studied PA correlate is enjoyment (Silva et al., 2012), although it is one of the main reasons why people perform PA (Granero-Gallegos et al., 2017). Enjoyment is often cited as an important correlate or predictor of PA participation and it is included in many health promotion models and behaviour motivation theories (Dacey, 2009). Enjoyment is an intrinsic and affective component of motivation (Sallis et al., 2000) that leads directly to behaviour changes (Vallerand et al., 1987). In the PA and sport domains, enjoyment is conceptualised as a positive affective response resulting from participation that reflects generalised feelings typically described as pleasure, liking, and fun (Kimiecik & Harris, 1998).

Within the PA context, enjoyment represents a positive attitude toward PA practice (Wankel, 1993) and constitutes one of the most important correlates for PA participation (Gavin et al., 2014), particularly for girls. An important source of enjoyment is the degree to which PA participation results in the satisfaction of individuals' basic psychological needs of competence, relatedness, and autonomy

(Teques et al., 2017). Previous studies have acknowledged the experience of enjoyment as a critical factor in motivating young girls to continue participating in activity environments (Brazendale et al., 2015; Cairney et al., 2012). Enjoyment of PA is positively correlated with PA participation levels for children and adolescents (Woods et al., 2012; Sallis et al., 2000) more specifically, among girls during early adolescence (Budd et al., 2018; Post & Palacios, 2019) and sustained PA throughout adulthood (Budd et al., 2016). For example, in Robbins et al. (2019) 'Girl on the Move' 17-week intervention study, enjoyment had a positive direct effect (B = 24.48, p < .001) on young girl's (M age: 12.05 years (SD = 1.01; N = 1519), MVPA from baseline to post-intervention. Contrastingly, Budd et al. (2018) study, examining the role of PA enjoyment as a mediator of social and physical environments to MVPA of early adolescent girls, found that there was no relationship between PA enjoyment among 1721 sixth grade girls.

Enjoyment in youth sport is considered the greatest predictor of commitment and lack of enjoyment is the most frequently cited predictor of dropout among young girls (Crane & Temple, 2014; Visek et al, 2015). Enjoyment is consistently associated with continued sport participation (Gardner et al., 2017; MacDonald et al., 2011; Wetton et al., 2013). A recent Australian qualitative study (Litchfield & Elliott, 2020) exploring how Netball clubs and organisations utilise enjoyment to attract and retain girls' participation. Interestingly, findings indicate that clubs and organisations appreciated the importance of enjoyment, however, their current practices did not consistently reflect this understanding. The experience of enjoyment during physical education (PE) classes is associated with enhanced intrinsic motivation, increased PA participation, and the adoption of active and healthy lifestyles (Dishman et al., 2005; Jaakkola et al., 2017; Morano et al., 2019; Vitali et al., 2019). Thus, understanding enjoyment motives and other variables are known to influence PA levels, successful motor experiences, and improvement of physical fitness can help researchers and practitioners design more effective intervention strategies to promote healthy lifestyles among school-aged children (Morano et al., 2019). Dzewaltowski (2002) suggests that to promote the enjoyment and SE (as previously discussed), environments should be structured to encourage: connection, autonomy, skill-building, and healthy norms (CASH). Specifically, girls should be engaged in environments where the group norm is healthy behaviour and where they have a sense of belonging and control while allowing them to develop their skills (Spence et al., 2010).

# 2.3.3 Social influences on PA

Social networks, or specific sets of linkages between people (Mitchell, 1969), can influence PA through various mechanisms (Laird et al., 2018). Social support has been defined as resources provided from interactions with significant others (e.g., parents, friends) within a social network that can influence behaviour (Langford et al., 1997). Social support for PA has been identified as an important social correlate of PA in young girls (Laird et al., 2016; Liangruenrom et al., 2019; Shen et al., 2018), and even considered by some a key process promoting and facilitating PA among adolescents (Ren et al., 2020). Social support can be emotional (e.g., encouragement, praise), instrumental (e.g., logistic support), informational (e.g., instruction), or co-participation (e.g., engaging in PA with child) (Yao & Rhodes, 2015). Additionally, social support can be perceived (e.g., the support a person thinks that they get) or received (e.g., the support a provider thinks that they give) (Laird et al., 2018). Within the PA literature, modelling (e.g. associations between activity levels of provider and child) and co-participation (e.g. performing physical activities together) have also been considered forms of social support (Reimers et al., 2019; Yao & Rhodes, 2015).

Some recent evidence would suggest that social support is a potentially important mediator of increased PA in young girls (Casey et al., 2009; Coen et al., 2018; Ren et al., 2020; Van Lippevelde et al., 2012), and has been targeted by several intervention programmes (Cowley et al., 2021; Eather et al., 2013; Story et al., 2017). Numerous studies have focused on girl's social support for PA in children and adolescents and these have also been systematically reviewed (Beets et al., 2010; Maturo & Cunningham, 2013; Mendonça et al., 2014; Yao & Rhodes, 2015). Most of these reviews have focused on parental influences with results suggesting positive significant associations between parent support and child and adolescent PA (Beets et al., 2010; Liszewska et al., 2018; Petersen et al., 2020; Yao & Rhodes, 2015). Regarding the influence of environmental factors, it has been shown that particularly

the social environment, and more specifically peers and parents, can have an important impact on girls' PA levels (Verloigne et al., 2016).

# 2.3.3.1 Parental PA influences

Research has documented parents' many and important roles in OYS (Fyfe-Johnson et al., 2019; Petersen et al., 2020). Studies have found parents to be initiators of sport participation and motivators for further activity (MacPhail & Kirk, 2006; Wheeler, 2012). They contribute as providers of resources and transport, as role models and interpreters (Smoll et al., 2011). The influence of parents on adolescent girls' PA level can occur through modelling and co-participation (Timperio et al., 2013), but especially through supporting them, such as encouraging them or by providing logistic support (Beets et al., 2010; Sebire et al., 2013). Furthermore, parental modelling and social support are considered to enable or to foster PA participation in children and youth (Reimers, 2019). Parental modelling of activity is positively related to children's participation in PA like outdoor play, sports or walking for transport (Rodrigues et al., 2017; Schoeppe et al., 2017; Mutz & Albrecht, 2017). For example, the provision of instrumental support like driving a child to places where they can play sports or buy equipment are necessary assistances to engage in some sporting activities (Reimers, 2019). Interestingly, in a study of fifth-grade students from Germany, Schoeppe et al. (2016) confirmed the same-sex imitation hypotheses by finding relationships between girls' leisure-time PA and maternal sport participation and between boys' activity and paternal sport participation. Similar results were found by Lijuan et al. (2016) in Chinese children, by Kirby et al. (2013) in Scottish adolescents, and by Cheng et al. (2014) in Brazilian adolescents.

Edwardson and Gorely's (2010) review (including 96 studies by intensities and types of PA), examining parental support of PA for young people, found that for children, parental involvement (overall PA and leisure-time PA), and overall support (organised PA) were associated with types of activity, whereas for adolescents, the main associations were for parental support (MVPA, vigorous PA, overall PA), attitudes/beliefs (MVPA, overall PA), transport (MVPA, organised PA), and encouragement (PA frequency). This review investigated cross-sectional and

longitudinal studies and their data was analysed separately by children and adolescents. Results from a recent study in the Western Cape in South Africa, investigating factors influencing participation in PA among 11-13-year-old school children (n= 348), reported parental influence as the strongest predictor (r = 0.236, p < 0.01). Similarly, Pugliese and Tinsley (2007) found in their meta-analysis a small significant positive association for parental behaviour (p < .001). The results showed that encouragement (p < .01), instrumental behaviour (e.g. paying membership dues; p < .05) and modelling (p < .001) were significantly positively related to children's PA (Pugliese & Tinsley, 2007).

Parental support and direct help from parents have previously been correlated with PA in female youth (Carlin et al., 2015; Eddolls et al., 2016; Kirby et al., 2013). For example, in a recent study of 226 girls (mean age  $16.0 \pm 1.0$  years), higher parental modelling, co-participation, and encouragement related to a higher PA level (selfreported) among adolescent girls (p<0.05) (Verloigne et al., 2016). However, results from one meta-analysis suggest that social support is not a strong predictor of PA in adolescent girls (10 to 19 years), although parents and friends may have a role in enhancing PA (Laird et al., 2016). Findings from Laird et al.'s (2016) systematic review and meta-analysis both support and contest the findings from another recent meta-analysis by Yao and Rhodes (2015), who identified positive associations between parent support and PA in children and adolescents (r=.38). Laird's systematic review and meta-analysis identified more modest effect sizes than Yao and Rhodes (2015) for parent support on adolescent girls' PA (r = .19), which aligns more closely with a metaanalysis by Pugliese and Tinsley (2007) (r = .17). These differences could be attributed to the fact that Yao and Rhodes (2015) and Pugliese and Tinsley (2007) considered all children and adolescents, whereas Laird et al (2016) only synthesised associations reported for adolescent girls.

Most recently, Morgan et al. (2019) novel Dads And Daughters Exercising and Empowered (DADEE) programme in Australia provides evidence that efforts to increase PA behaviour in young girls (n=153; 4 to 12 years) would benefit from a meaningful engagement of fathers (n=115; 29-23 years). Significant effects (p < .05) were identified for daughters' FMS competence (objective: d = 1.1-1.2; perceived: d = 0.4-0.6), a range of fathers' PA parenting practices (d = 0.3-0.8), and screen-time

for daughters (d = 0.5-0.8) and fathers (d = 0.4-0.6) at post-intervention and followup. As previously highlighted, parents already play an important role in influencing girls' health behaviour at a younger age (Edwardson & Gorely, 2010), but the influence of peers may increase when girls move toward adolescence (Kirby et al., 2011) (roughly the period between ages 10 and 19 years), they spend increased time with peers enhancing the potential for the norms and behaviours of peers to influence their PA levels (Duncan et al., 2007).

# 2.3.3.2 Peer PA influence

Children live in peer-rich worlds and spend significant amounts of time with peers and friends at school, on playgrounds, and while participating in organised activities such as sport (Stearns et al., 2019). It has even been suggested that peers or friends are more important than parents in influencing adolescent girls' PA levels (Lytle et al., 2009). The nature of the relationship between children and their parents transforms significantly during pre-adolescence to adolescence (Smetana et al., 2006). As adolescent girls spend less time with their parents and more time with their friends (Larson et al., 2002), friends might be better positioned to influence adolescent girls' PA than parents (Laird et al., 2016).

The influence of friends and the importance of having someone to participate is a key emergent theme among children, and these are key factors in young female's participation in PA (Carlin et al., 2015). Children like doing PA with friends rather than alone (Bristow & Saunders, 2014), and describe their friends as influencing their PA in various ways such as modelling, co-participation, and encouragement (Jago et al., 2009). Some studies also indicated the role of social support and social modelling from peers predicting PA in children and adolescents (Reimers, 2019). The presence of peers and the presence of peers being physically active were associated with an increase in PA in a wide range of children and adolescents from the age of 3–5 to 15 years old (Lau et al., 2016; Ward et al., 2017).

Friends' PA is consistently related to youth PA (Stearns et al., 2019; Beets et al.; Voorhees et al., 2005). Maturo & Cunningham (2013) systematic review
(including 106 studies) on the influence of friends found that children's PA was positively associated with encouragement from friends (43 of 55 studies indicating a positive relationship), friends' own PA (30 of the 35 identified studies), and engagement with friends in PA (9 of the 10 identified studies). A study by Jago et al (2009) examined whether the number of friends in an adolescent's social circle was associated with PA in a sample of 10 to 11-year-old children, and found that an increase in the number of friends was associated with an increase in adolescent PA. Besides support from parents, children who have supportive friends and peers have also been found to be more physically active (Fitzgerald et al., 2012), whereas those who experience negative peer interactions may become less physically active (Salvy et al., 2012).

Peer-influence, which increases as girls transition from childhood to adolescence (Davison & Jago, 2009), plays a pivotal role in girls' PA (Fitzgerald et al., 2012), and can comprise social support, presence of peers during PA, peer norms, friendship quality, changes to friendship groups, activity preferences of peers, affiliation to peer groups, and peer victimisation (Fitzgerald et al., 2012; Cavallo et al., 2014). Despite being a key part of the complexity of girls' PA, until recently, peers have been relatively overlooked in many PA interventions (Ginis et al., 2013). A recent longitudinal study (Basterfield et al., 2016) on perceived barriers to sports participation were collected from a birth cohort, the Gateshead Millennium Study (n>500) at ages 9 and 12 years. The 12-year-olds' concerns relating to their social environment, emphasising the importance of friendship groups at this age: 'I don't take part in any outside school clubs, because none of my friends do' (ID15, girl, 12) years). Some of the findings of this study echo those of Stanley et al. (2012) with children aged 10-13 years in Australia highlighting the importance of friends and parental support concerning PA (not specifically sport participation). This emerges during early adolescence, as they become more aware of what their friends think of them, and the need to feel accepted and similar (Erdvik et al., 2019).

Peer acceptance and friendship quality are two important dimensions of peer influence that have been linked with an increased commitment to sports, greater enjoyment, and improved psychosocial wellbeing among adolescents (Fitzgerald et al., 2012; Stearns et al., 2019). There is evidence that suggests friends may have a

direct impact on adolescent girls' attitudes and beliefs toward PA, and that adolescents are more likely to either choose friends who have similar PA behaviour or alter their PA behaviour in response to their friends (Salvy et al., 2012; Marks et al., 2015). Findings from the recent Corr (2019) systematic review of qualitative studies on adolescent girls' perceptions of PA found that for some girls, if their friends were involved in PA, then it became a motivating factor for them to be active too as well as increasing enjoyment levels and providing positive activity experiences: '*My best friend dances with me and it's way more motivating cause then you know you won't be alone when you need to go somewhere, and you tell yourself it's going to be real fun'* (Bélanger et al., 2011). In Corr et al.'s (2019) systematic review of qualitative studies on adolescent girls' perceptions of PA, girls discussed how they failed to participate in PA because 1) their friends didn't, 2) if they participated, it meant they missed opportunities to spend time with friends, 3) fear of ridicule from friends, and 4) friends teasing them and excluding them from conversations.

Peers are a potential key source of support for adolescent PA and for PA efficacy beliefs (Duncan et al., 2005; Duncan et al., 2007). Peer support may serve several different functions (e.g., social integration in PA, emotional support, informational support, tangible aid, a reassurance of worth), which may help bolster a youth's PA efficacy beliefs and assist in overcoming perceived barriers to PA participation (Duncan et al., 2015). Evidence also suggests that adolescent girls socialise in groups with similar PA levels and, over time, their PA behaviours reflect those of their peers (Macdonald-Wallis et al., 2011). Weiss et al. (1996) found some evidence to suggest that friends influence perceptions of self-esteem, enjoyment and performance in sports. A focus group study by Jago et al. (2009) found that support from friends encouraged young people to try new physical activities, and coparticipation with friends was linked with enjoyment of PA. Whilst these findings give some indication of the possible ways in which social support influences PA, it is important to note that the majority of these studies considered boys and girls together. Considering potential gender differences, a specific focus on how social support influences adolescent girls' PA may be important to inform targeted PA interventions for adolescent girls (Laird et al., 2018). Peer-based interventions could be an effective means of helping adolescent girls become more physically active (Fitzgerald et al., 2012; Willis et al., 2019).

Studies have examined both friend and family support for PA, and results have shown that some studies have found a positive association (Beets et al., 2010; Reimers et al., 2019), while others have not (Cohen et al., 2015). Seabra et al.'s (2012) study examining correlates of PA in school-children aged 8-10 years, found perceived acceptance by peers in games and sports and parental encouragement were positively and significantly related to PA in girls (n= 683). Additionally, a study by Jago et al. (2009) found that children reported friends influencing their activity behaviour through 3 main mechanisms: 1) modelling of PA, 2) verbal encouragement of PA, and (3) co-participation in PA with participants. Peer social support had a direct effect on MVPA ( $\beta$  = .33, p < .05), also significantly influenced levels of enjoyment ( $\beta$  = .47, p < .01) and self-efficacy ( $\beta$  = .67, p < .01).

Although several reviews have considered the role of social networks, in particular social support, on PA in children and adolescents (e.g., Beets, et al., 2010; Edwardson & Gorely, 2010; Yao & Rhodes, 2015), only one focused specifically on adolescent girls (Laird et al., 2016). This review identified significant positive associations between social support from friends and family and adolescent girls' PA. The review also highlighted the complexity of the relationship between social support and PA, with numerous types and providers of support identified, and varying effect sizes for these associations with PA identified between studies. Further, according to socio-ecological models (Kremers et al., 2006; Verloigne et al., 2016), these associations between peer and parental factors and adolescent girls' PA could be mediated by personal factors, which means that peer and parental factors may influence girls' PA indirectly through personal factors. This implies that these personal factors are also associated with adolescent girls' PA, specifically self-efficacy (Morano et al., 2019; Vitali et al., 2019), as noted previously.

Strategic advice by Woods et al. (2018) on increasing the PA participation levels of Irish children and youth indicates that family and peer support are crucial influences for developing PA opportunities. Given the continued influence parents and peers play as children move into adolescence (Eddolls et al., 2016; Van Der Horst et al., 2007), it would be prudent for female OYS interventions to consider active peer and parental involvement, as these reinforcing factors may positively influence girls' PA participation (both directly and indirectly) through the internalisation of several personal and psychological variables (Verloigne et al., 2016). Other social agents, such as youth sport coaches and coordinators, can structure the PA environment to maximise feelings of competence and personal mastery, with children more likely to enjoy the activity, maintain interest in involvement, and commitment to the activity (Granero-Gallegos et al., 2017). Studies have reported that youth who feel socially connected to their coach, or other children in a sports team (Carlin et al., 2015; Farmer et al., 2018) tend to have greater satisfaction and positive emotions in the context of youth sport.

#### 2.3.3.3 Coach influences

Social agents, such as OYS coaches can be viewed as important role models that could influence young players' PA developmental experiences (Santos et al., 2019), empowering or enabling participants to take part in PA within an OYS context (Griffiths & Armour, 2014). Coaches can offer support and guidance to athletes that ultimately facilitate the formation of strong bonds (Rhind & Jowett, 2010). This support has been shown to impact enjoyment, motivation and the development of key competences (MacDoanld et al., 2011). A systematic review on social support in youth sport (Sheridan et al., 2014) of 73 studies revealed that coaches were identified as the most prevalent provider of social support through offering participants unique forms of tangible, informational, emotional and esteem support. Furthermore, authors of this review concluded that coach, (along with parent and peer) support plays a significant role in shaping youth sport experiences both from a positive (athlete motivation levels, elite sport participation) and negative (drop-out) perspective (Sheridan et al., 2014).

It is of particular importance that every coach (volunteer, part-time and full time) should engage with coach education and receive adequate training (Côté & Gilbert, 2009), reinforcing the coaches' influence on children's involvement and development in sport (Trudel et al., 2010). Based on Côté and Gilbert's (2009) understanding of effective coaching, it is necessary to systematically integrate coaches' knowledge to increase athletes' confidence, competence, connections and character given the particular characteristics of a sports context (i.e. aims, age group, and coaching domain) (Côté & Gilbert, 2009; Flett et al., 2013).

Previous studies (Sullivan et al., 2012) show that coaches who participated in coach education programmes positively affected coaches' efficacy beliefs, knowledge, and behaviours (Trudel et al., 2010). A review on the effectiveness of coach education in OYS indicated that non-formal learning methods within coach education (such as continuous professional development (CPD), clinics, workshops or seminars (Maclean & Lorimer, 2016) addressed specific topics that are more authentic, meaningful, and contextualised to a coach (Mallett et al., 2009). Empirical studies that found OYS coaches who participated in non-formal coach training programmes enhanced their ability to improve their team's environment, build positive coach-athlete relationships and foster positive outcomes in their athletes (Koh et al., 2014; Walker et al., 2018), including enhanced participation rate and performance (Galatti et al., 2016).

Furthermore, an important element of OYS participation is the coach education component (Owen et al., 2018; Robbins et al., 2013, 2019). The literature widely acknowledges that youth PA interventions through the provision of non-formal coach education, such as Continuous Professional Development (CPD) opportunities (training, resource manual, and opportunities to work with coaches and instructors) can positively support programme implementation, and childhood engagement in PA (Casey et al., 2014; Koh et al., 2014). Despite research on the effectiveness of nonformal coach training programmes, little is known about how knowledge is incorporated and transmitted in these programmes (Trudel et al., 2010). As a result, it has been argued that there is a need for innovative coach education approaches that can better equip coaches with the professional competencies needed to deal with the dynamic nature of coaching within an OYS setting (Morgan et al., 2013). Hence, quality of interactions and relationships with parents, coaches, and teammates - whom young girls look up to as models, rely upon for support and seek out for performance feedback, social reinforcement, and skill instruction - hold the key to whether sport's potential to optimise positive outcomes is realised (Weiss et al., 2012).

From the outlined studies and research, there appears to be robust evidence that parental, peer and coach support, can positively contribute towards youth PA participation. Similar to the domain of OYS, with parental, peer and coach influences, fundamental movement skills (FMS) have been shown to have a positive association with PA (Robinson et al., 2015), indicating that FMS may be a potentially viable correlate of youth PA participation.

#### 2.3.4 Fundamental Movement Skill (FMS) influences on PA

Research suggests that actual movement competency (e.g., fundamental movement skill (FMS) proficiency) in early childhood may be an important prerequisite for engagement in PA later in life (Loprinzi et al., 2015). A fundamental movement pattern is an observable performance of basic locomotor, manipulative, or stability movements, that involves combining movement patterns of two or more body segments, and the integration of developmentally appropriate arm, trunk, and leg actions (Gallahue & Ozmun, 2012). They can be broken down into two main categories: locomotor skills (LS) and object control skills (OS). LS require the fluid coordination of the body as it moves in one direction or another and includes skills such as running, galloping, jumping, hopping, leaping and skipping. OS require the efficient control of another object such as a bat or ball and includes skills such as throwing, catching, kicking, rolling, striking and dribbling (Department of Education Victoria, 1996; Gallahue and Ozmun, 2006; Stodden et al., 2008). Balance (stability) is also identified as a measure of FMS (Ulrich, 2013). These basic movement skills are essentially the groundwork and 'building blocks' that underpin all movement (Babic et al., 2014; Barnett et al., 2016).

FMS proficiency leads to specialised movement sequences required for adequate participation in many organised and non-organised physical activities for children, adolescents and adults (Goodway et al., 2020; Hardy et al., 2010). Studies suggest that in both pre-schoolers (Cliff et al., 2009) and school-aged children (Luban et al., 2010) there is a positive relationship between FMS competence and increased levels of PA. A systematic review by Logan et al. (2015) found relationships between FMS and PA in early through to late childhood in cross-sectional studies. Additionally, a recent systematic review and meta-analysis, exploring the relationship between FMS interventions and PA levels in children, found significant improvement in FMS with teacher-led interventions of three or more sessions per week (Engel et al., 2018). Furthermore, developing FMS during childhood may be a significant step towards establishing a lifelong commitment to PA (Nilsen et al., 2020). One correlate which

requires additional examination in the literature is the levels of FMS proficiency amongst young people, specifically young girls, and this association with PA (see section 2.4 for a complete discussion of FMS).

In section 2.3 the correlates of PA amongst youth were examined, with a specific focus on female youth. The identification of SE, parental, peer and coach influence and perceived skill competence (FMS) are considered to be the potentially important mechanisms promoting PA. SE was the only psychological correlate which was consistently associated with PA and sport participation in many of the most recent reviews for children and youth over the last decade (Sterdt et al., 2014; Uijtdewilligen et al., 2011; Van Der Horst et al., 2007; Wang et al., 2016). A plethora of research has shown that parental support and direct help from parents have previously been correlated with PA in female youth (Carlin et al., 2015; Eddolls et al., 2016; Kirby et al., 2013; Petersen et al., 2020). Peer-influence, which increases as girls' transition from childhood to adolescence (Davison & Jago, 2009), plays a pivotal role in girls' PA (Fitzgerald et al., 2012; Stearns et al., 2019). Furthermore, peer-based interventions could be an effective means of helping adolescents become more physically active (Willis et al., 2019).

Other social agents, such as OYS coaches who can also be viewed as important role models that could influence young players' PA developmental experiences (Santos et al., 2019), empowering or enabling participants to take part in PA within an OYS context (Griffiths & Armour, 2014). To enhance coach influence, the literature widely acknowledges that youth PA interventions through the provision of non-formal coach education, such as CPD opportunities (training, resource manual, and opportunities to work with coaches and instructors) can positively support programme implementation, and childhood engagement in PA (Casey et al., 2014; Engel et al., 2014; Koh et al., 014). Understanding how social support influences PA behaviour is imperative to inform the development of PA intervention strategies aimed at engaging inactive girls in PA (Laird et al., 2018). Finally, FMS as an important and viable correlate for youth PA engagement with positive associations found in the literature (Lubans et al., 2010). There is evidence emerging that developing FMS during childhood may be a significant step towards establishing a lifelong commitment to PA (Burns et al., 2017; Logan et al., 2015).

#### **2.4 FUNDAMENTAL MOVEMENT SKILLS (FMS)**

The purpose of section 2.4 is to review the relevant literature regarding the development of FMS proficiency and assessment in children and young people. The first theme explored is an overall introduction and description of FMS.

#### 2.4.1 Introduction and description of FMS

As explained previously in section 2.3.5, FMS have been described as the building blocks for movement and form the foundation for many of the specialised movement skills needed to participate successfully in sport and PA (Gallahue and Ozmun, 2006). Zuvela et al. (2011) described FMS as basic motor skills underlying more complex sport-specific skills. These skills are also regarded as the foundation of body movements (Barnett et al., 2013). In essence, FMS are basic abilities and skills of a child to perform an organised series of basic movements (locomotor, manipulative, or stability movements), that involve various body parts and provide the basis of achieving a high level of motor competence to develop normally, maintain health, and gain athletic excellence (Gallahue & Ozmun, 2012; Lloyd et al., 2014).

A more contemporary view is that the term 'fundamental' is too narrow and does not sum up the full range of skills that may promote continued PA participation across the lifespan (Hulteen et al., 2018). Skills involved in activities such as swimming, or cycling, require competency in specific coordinative movement patterns (e.g., swimming strokes) that do not easily fit into the traditional classifications of FMS (i.e., locomotor, object control, stability skills) and therefore have not been widely addressed in the literature base (Hulteen et al., 2015). Hulteen and colleagues (2018) suggest that the term 'foundational movement skills' better reflects the wide variety of skills of which an individual should develop competency. Foundational movement skills are defined as goal-directed movement patterns that directly and indirectly impact an individual's capability to be physically active and can be developed to enhance PA participation and promote health across the lifespan (Hulteen et al., 2018). Proficiency in a range of FMS is considered to be the foundation for an active lifestyle (Cattuzzo et al. 2016), and the primary school window (5 to 12 years old) represent the "golden years" of motor skill development (Clark & Metcalfe, 2002).

A helpful heuristic device describing the sequential progression of motor development across the lifespan can be highlighted in Gallahue & Ozmun's (2012) hourglass model; this model, conceptualises, describes, and provides explanations of the rate of movement skill acquisition from infancy throughout life (see Figure 2.1 below).



Figure 2. 1: The phases and stages of motor development (taken from Gallahue and Ozmun, 2012).

The development of FMS is subdivided into three stages of development (Pienaar et al., 2016). The initial phase describes the execution of these skills in an uncoordinated manner. Children are typically in this phase of development between the ages of 2 and 3 years. Children are generally in the elementary phase, the second phase of development, between the ages of 4 and 5 years, typifying a child displaying improved coordination and rhythm, exerting finer control over movement during the execution on these skills, albeit not yet fluent (Gallahue & Ozmun, 2006). A child can execute a fundamental movement skill in a well-coordinated, mechanically correct and fluent manner (Gallahue & Ozmun, 2006) in the mature phase of development, usually occurring between the ages of 6 and 7 years old.

Motor development refers to the continuous, age-related changes in movement that occur across one's lifespan and the interacting individual, task, and environmental factors that underlie these changes (Clark & Whithall, 1989). Pope et al. (2011) defined the fundamental movement phase as the period when motor skills are obtained, by a progressive improvement of coordination. Gallahue and Ozmun's (2006) hourglass model for motor skill development suggests that both boys and girls have the potential to demonstrate mastery of FMS (i.e., perform the skill(s) correctly) between the ages of five to seven years. To achieve such mastery, these movement skills must be learned and practiced and reinforced (Robinson and Goodway, 2009); they are not acquired naturally (Pang & Fong, 2009). Developing proficiency in any FMS can take a long time – up to 10 hours of quality teaching (Hands, 2012). Therefore, the early years ( $\pm$  3–7 years) are a critical period in the development of these skills (Gallahue et al., 2012). Stodden et al. (2008) report that FMS are important skills to master during the childhood years, as they will be used later in their lives for sport-specific application and lifelong utilisation. Oliver et al. (2011) and Jukic et al. (2019) further propose the development of FMS is necessary to become competent in more advanced PA contexts such as sports. Not only this, but there is also a growing body of work that explores the connections between early movement skill development and an array of health-related outcomes including social, cognitive, and emotional development (Hendrix et al., 2014; Iivonen & Sääkslahti, 2014).

#### 2.4.2 Benefits of FMS

FMS contribute to the general development and wellbeing of young children (Pienaar et al., 2013), which is a prerequisite for daily life skills (Stodden et al., 2008) and therefore impact children's physical, social and cognitive development (Iivonen & Sääkslahti, 2014). FMS proficiency is positively associated with many physiological, psychological, and behavioural benefits (Lubans et al., 2010), including fitness (Burns et al., 2017) and PA levels (Burns et al., 2017; Logan et al., 2015). Previous research has demonstrated positive associations between FMS and areas of health including PA, fitness, healthy body composition, self-concept, and executive functioning (Lubans et al., 2010; Hendrix et al., 2014). Childhood is a critical time for FMS development as recent reviews have found FMS proficiency to be positively

associated with a range of benefits, including health, fitness, academic outcomes, participation in organised sports and sustained engagement in PA (Hardy et al., 2012; Jaakkola et al., 2015; Lubans et al., 2010; Nilsen et al., 2020; Stodden et al., 2014). In a systematic review of the literature, Lubans et al. (2010), demonstrated the numerous health benefits of FMS proficiency including increased PA and fitness, a lower body mass index (BMI), and improved mental health outcomes. However, the majority of the articles included in the review were cross-sectional; thus, the causal relationship between these skills and health-related variables is not yet fully understood (Lubans et al., 2010).

#### 2.4.3 The relationship between FMS and PA

Children develop their FMS through engagement in PA (Figueroa & An, 2016), as increased PA provides more opportunities to promote neuromotor development, which in turn promotes FMS development (Fisher et al., 2005). At the same time, learning to move is a necessary skill underlying PA (Stodden et al., 2008). Based on the conceptual model introduced by Stodden et al. in 2008 (Stodden et al., 2008), the relationship between FMS and PA is likely to be bi-directional. In addition, the relationship may differ at different stages of a child's development. While Stodden et al. (2008) hypothesised engagement in PA to be important for the development of FMS during the early years, FMS levels were hypothesised to become more important for PA participation as the child gets older (and becomes more motor competent) (Stodden et al., 2008). So far, only a few studies are examining the causal relationship between the quality of FMS and PA (Barnett et al., 2011; Jaakkola & Washington, 2013; Xin et al., 2020), i.e. that having high FMS level may increase options for participation in PA, as well as increased participation leading to further development of motor skills. Some studies suggest a reciprocal relationship between PA and FMS (Barnett et al., 2011; Kambas et al., 2012; Stodden et al., 2008).

Lloyd and colleagues (2014) conducted a 20-year follow-up study of the longterm association between motor skills at age 6 and PA at age 26. They found that in females, overall motor skill proficiency at 6 years of age was positively related to leisure time PA at 26 years of age (Lloyd et al., 2014). In a review article, Logan et al. (2015) reported a small-to-moderate positive relationship between FMS and PA among 6- to 12-year-olds. It should be noted however, that only one of the 13 studies included was conducted in Europe and self-report PA questionnaires were used in the majority of the studies. Numerous studies have examined the cross-sectional relationship between FMS and PA in children, supporting a low to moderate, positive association (r<0.50) between FMS and levels of total PA (TPA), light PA (LPA), and MVPA (Robinson et al., 2015; Lubans et al., 2010). Cross-sectional studies among pre-schoolers (Barnett et al., 2013; Roscoe et al., 2019), children (Barnett et al., 2011; Bolger et al., 2018; Jaakkola & Washington, 2013), and adolescents (Okely et al, 2002) have identified FMS proficiency as an important correlate of PA participation (Lubans et al, 2010). Notably, longitudinal evidence indicates that childhood motor proficiency is also predictive of child (Nilsen et al., 2020) and adolescent (Barnett et al., 2015) PA participation.

In Ireland, findings from a recent study by Bolger et al. (2019) among 7-10 year old (n=296) primary school Irish children (Bolger et al., 2019) revealed small positive correlations between FMS and moderate PA (r(71) = .259, p < .05), and between FMS and total PA (r(71) = .355, p < .05). Engel et al. (2018) recent systematic review and meta-analysis, investigated the relationship between FMS and PA in preschoolers (3-5 years) and primary-school-aged children (5-12 years) following an FMS intervention and found a significant improvement in overall FMS, PA, MVPA, and SB among 18 studies. Holfelder & Schott (2014) systematic review, investigating the relationship between FMS and PA in children and adolescents found strong evidence from cross-sectional studies for a positive relationship between FMS and organised physical activities. This may foster lifelong PA participation amongst children (Field & Temple, 2017), providing greater opportunity for children to engage in a variety of physical activities, games, and sport. Nevertheless, evidence is emerging from the literature that FMS and physical fitness are declining in children internationally (Behan et al., 2019; Bolger et al., 2019; Foulkes et al., 2015; Pienaar et al., 2015).

#### 2.4.4 FMS proficiency levels among youth

Ample literature consistently illustrates that children and adolescents do not obtain proficiency in FMS development (Behan et al., 2019; Bolger et al., 2019; Farmer et

al., 2018; McGrane et al., 2016; Hardy et al., 2010; Mitchell et al., 2013; O' Brien et al., 2016a) and are falling below the expected FMS proficiency levels for their age group (Duncan et al., 2017; Eather et al., 2018; Hardy et al., 2013). As a result, this may affect their sports-specific skill development, and their PA participation (Gallahue et al., 2012; Robinson et al., 2015). Current FMS proficiency levels of children worldwide have been reported to be low (Bardid et al., 2016; Belton et al., 2014; Bryant et al., 2014; Lester et al., 2017; Mitchell et al., 2013; Roscoe et al., 2019; Spessato et al., 2013). Recent studies and trends indicate less FMS proficiency among children when compared to a previous generation (Bardid et al., 2016; Spessato et al., 2013).

On a national level, findings from a recent Irish study (Behan et al., 2019), examining FMS proficiency across the full range of Irish primary school children (n = 2098, 47% girls, age range 5–12 years) found that FMS mastery or near mastery of skills ranged from 16% for the overhand throw, to 75.3% for run. Similarly, Kelly et al. (2019) examined FMS proficiency among male (N = 216) and female (N = 198) Irish primary school pupils from Year 2 to Year 7 (9.0  $\pm$  1.7 years) and reported that the percentage mastery ranged between 1.4% (gallop) and 35.7% (slide), indicating that Irish children's FMS proficiency is consistently low. Worryingly, another Irish study has highlighted that children are entering adolescence lacking in basic FMS (O'Brien et al., 2016a), which is expected to have a subsequent effect on their sportspecific skill development and consequently their PA participation (Field & Temple, 2017; Gallahue et al., 2012).

In other countries, similar trends have been observed, for example, there is evidence to suggest that proficiency levels of FMS among children are generally quite low in an Australian study carried out by Okely and Booth (2004) reporting that the prevalence of mastery and near mastery across a range of FMS was generally low. The proportion of students who displayed mastery of skill did not exceed 35% for any of the FMS (Okely & Booth, 2004). This finding is supported by more recent studies amongst Australian primary school children (Hardy et al., 2013; Hardy et al., 2016) which indicate that children's FMS proficiency is consistently low. In a recent national PA report card, where an 'A' represented the highest score (81–100% mastery), Australian children received a D (21–40% mastery) for 'movement skills' (Schranz et al., 2018). Similar representative studies of children from Brazil and the United States

recorded a low prevalence of mastery for each FMS assessed (Goodway et al. 2010; Spessato et al., 2013).

In the UK, a large proportion of 10-11-year old children rated as non-proficient in overall FMS competency levels (Foweather, 2010), and in Singapore the majority of children aged 6–9 years old score 'below average' or 'poor' on both locomotor and object control skills (Mukherjee et al., 2017). Similarly, research carried out by Mitchell et al. (2013), in a study on 5-13-year-olds in the Waikato region of New Zealand, found that less than half of the participants exhibited proficiency in striking (40%), throwing (31%) and kicking (21%). Likewise, results from O'Brien et al.'s (2016a) study in Ireland found that 11% of early adolescents assessed achieving mastery or near mastery on the nine FMS assessed. The FMS proficiency of Hong Kong children, however, is considerably higher than the aforementioned studies (Goodway et al. 2010; Okely and Booth, 2004), with 24% of participants achieving mastery in a range of 12 FMS (Pang & Fong, 2000).

It is reasonable to state that a growing body of literature illustrates that children and adolescent youth do not obtain proficiency in FMS development (Duncan et al., 2017; Farmer et al., 2017; Hardy et al., 2010; Mitchell et al., 2013; Mukherjee et al., 2017; Okely and Booth, 2004; Roscoe et al., 2019). The fact that FMS proficiency is consistently low amongst primary school children indicates the importance of early intervention programmes in primary school (Hardy et al., 2010), particularly for atrisk vulnerable populations including girls. Further to this, research reports gender differences in the execution of FMS (Barnett et al., 2016; Bolger et al., 2018; Eather et al., 2018), with girls performing lower in object control skills, involving catching, kicking, and throwing, in comparison to boys (Barnett et al. 2016; Butterfield et al., 2012).

#### 2.4.5 Gender differences in youth FMS

Both age and sex have been found to influence FMS proficiency among children (Bardid et al., 2016; Bryant et al., 2014; Freitas et al., 2015; Spessato et al., 2013). Older children typically exhibit superior FMS proficiency than their younger counterparts (Bardid et al., 2016; Freitas et al., 2015; Mitchell et al., 2013; Spessato et

al., 2013) as a result of natural maturation and additional instruction, practice, and feedback (Lander et al., 2015). Gender differences in FMS children are well established (Eather et al., 2018; Barnett et al., 2013; Bolger et al., 2018; O' Brien et al., 2018). In terms of overall FMS proficiency levels, research indicates that boys have commonly been found to demonstrate higher levels of overall FMS compared to girls (Behan et al., 2019; Cohen et al., 2015), although some studies have reported no sex-related differences within overall FMS performance (Hardy et al., 2010; Kordi et al., 2012).

More specifically, research reports gender differences in the execution of FMS (Barnett et al., 2016; Hardy et al., 2010), with girls performing less well in object control skills, involving catching, kicking and throwing, in comparison to boys (Barnett et al., 2016; Butterfield et al., 2012; Eather et al., 2018). In an Irish context, research conducted on 8-12 -year old primary school girls (n = 331; mean age 10.92±1.22) reported 98.1% did not achieve the FMS proficiency expected for their age (Farmer et al., 2018). Additionally, low, moderately, and highly active participants differed significantly in terms of overall FMS (p = .03), and locomotor control scores (p = .03) (Farmer et al., 2018). Similarly, research that was conducted on primary school children (between ages of 5-14 years) both in New South Wales and America showed that girls performed less well in object control skills, involving catching, kicking and throwing, in comparison to boys (Barnett et al., 2010; Butterfield et al., 2012). Recent Australian research shows girls consistently demonstrate poorer FMS proficiency than boys, particularly for object-control skills. For example, only 14% of Australian girls have mastered the kick and over-arm throw upon entering secondary school, compared to 52% and 53% of boys (respectively) (Hardy et al., 2017).

#### 2.4.5.1 Gender differences in object control (OC) skills

More specifically, further research was carried out to examine gender differences within locomotor and object control subtests (Barnett et al. 2013; Hardy et al, 2010). Boys have been found to perform better in object control skills – catch, throw, strike, kick, bounce and roll (Bardid et al., 2016; Behan et al., 2019; Hardy et al., 2010; Lin & Yang, 2015; O'Brien et al., 2018; Spessato et al., 2013). In a recent Australian study,

an investigation of OC skill execution in 4 to 12-year-old primary-school-aged girls on 153 girls (n= 153; M age =7.7 years old, SD=1.8) reported <5% (of the total group, 4 to 8-year-olds or 9 to 12-year-olds) demonstrated mastery or advanced skill level in the strike, stationary dribble, overhand throw or kick (Eather et al., 2018). These findings are in agreement with previous research among primary school children of all ages including 3- to 10-year-old Brazilian children (Spessato et al., 2013), 7- to 10year-old Portuguese children (Freitas et al., 2015), and among Australian primary school children (van Beurden et al., 2002). Findings from Eather et al.'s (2018) study demonstrate that the vast majority of girls failed to master any of the OC skills irrespective of age-group, and although mastery levels for the catch were highest, only 34% of girls mastered or displayed near mastery skill level in this skill. These results add to the growing body of research highlighting low FMS levels in girls – especially OC skills (Foweather, 2010; Hardy et al., 2017; Mitchell et al., 2013; Mukherjee et al., 2017; O'Brien et al., 2016; Pang and Fong, 2009). Foulkes et al. (2015) reported better kicking and overhand throwing skills in boys aged 3-5 years in North-west England, while girls were more competent in the run, hop, and gallop. Foulkes and colleagues argue that the gender differences may be due to 1) the influence of sociocultural or environmental factors, and 2) boys and girls are likely to participate in differing games and physical activities that may contribute to observed sex differences in competence (Foulkes et al., 2015). Evidence indicates that similar patterns exist among older children (Barnett et al., 2010; LeGear et al., 2012; Okely & Booth, 2004; van Beurden et al., 2002) and adolescents (McGrane et al., 2016; O' Brien et al., 2016b; O'Brien et al., 2018), indicating that sex differences in OC skills are established in early childhood and may track into later childhood and adolescence.

#### 2.4.5.2 Gender differences in locomotor (LOM) skills

On the contrary, several studies suggest that girls are likely to display greater proficiency levels in LOM skills (Bolger et al., 2018; Field & Temple, 2017; Roscoe et al., 2019). However, findings regarding LOM skill proficiency are not consistent with some studies reporting no sex-related differences (Bakhtiar, 2014; Kordi et al., 2012). A study conducted by Robinson (2011) found that boys are better than girls in locomotor skills, and in other studies that girls are better than boys (Cliff et al., 2009;

Hardy et al., 2010). The greater LOM proficiency among girls has been previously accounted for by the types of activities that girls are more likely to participate in such as dance (Smyth, 2016), a trend that is also evident in an Irish context (Woods et al., 2018).

#### 2.4.5.3 Causes for gender differences in FMS proficiency

Several studies have reported that an increase in FMS gender differences are observed during puberty and adolescence (Barnett et al., 2010; Bolger et al., 2018; Spessato et al., 2013). The varying levels of FMS proficiency between boys and girls are attributed to a variety of factors including body size, strength, growth, and maturation as well as social and environmental factors (Beunen et al., 2006; Kohl & Cook, 2013). Prior to puberty, however, boys and girls are very similar physically with little differences in biological characteristics including body type, body composition, strength, and limb length (Hardy et al., 2010). Sex differences in FMS proficiency have been explained by the type of activities that children undertake, with boys and girls possessing very similar biological characteristics such as genotype, body composition, strength, and limb length before puberty (Hoffman, 2009). However, differences in motor skill performance at pre-pubertal stages are still widely reported to be present, which would suggest that these differences are due to social and environmental factors such as the influence of family, peers, teachers, and the physical environment rather than biological (Hardy et al, 2010; Junaid & Fellowes, 2006).

Researchers have attempted to explain the extensive differences between sexes, especially in OC related activities such, with some researchers suggesting that environmental and socio-cultural factors explain why boys generally outperform girls at OC skills (as boys generally spend more time participating in different ball games and gross motor activities) (Deaner et al., 2012; Niemistö et al., 2019). Research has shown that boys participate more in ball sports (OC related activities) while girls participate more in dance and gymnastics (LOM related activities) (Bardid et al., 2016; Hardy et al., 2010; Lunn, 2010; Woods et al., 2018). Hardy et al. (2012) reported that the observed gender difference in skills may be accounted to gender norms that preference children's activities, or that girls are not exposed or provided with reinforcing opportunities to develop OC skills. Indeed, a culturally relevant Irish national study investigating the childhood PA participation levels found that boys are much more likely to participate in team sports/invasion games, such as soccer, GAA, and rugby (Wood et al. 2018) which could be a credible reason for their increased OC FMS proficiency. Additionally, girls were reported to have higher participation levels in dance, swimming, and gymnastics (Wood et al. 2018), therefore explaining their higher competency in LOM skills.

Evidence from previous studies indicates that gender differences in FMS development across the age spectrum do exist (Barnett et al. 2013; Behan et al., 2019; Bolger et al., 2018; Eather et al., 2018; Gromeier et al., 2017). To cater to the needs of both boys and girls, researchers also need to understand these gender differences. Additionally, researchers need to examine whether there is consistency in these gender differences in order to tailor FMS intervention programmes that are unique to the child's developmental requirements (Goodway et al., 2010). Understanding patterns and consistency of FMS acquisition across developmental time for boys and girls is important, as it can help determine how to intervene to improve motor skill proficiency (Barnett et al. 2013). To maximise learning experiences for girls, it is important that researchers not only identify FMS component mastery levels, but that evidence-based and age-appropriate FMS programmes are developed that specifically target areas of need (Hardy et al., 2012).

The fact that FMS proficiency is consistently low amongst primary school children indicates the importance of early FMS intervention programmes in primary school (Hardy et al., 2010), particularly for at-risk vulnerable populations including girls. These low levels of FMS proficiency may translate into a lack of confidence in performing specific skills (Feltz, 1988). Research suggests that without FMS proficiency and a positive perception of such, children may be less likely to engage in PA (Lubans et al., 2010). To better understand the acquisition of FMS alongside levels of PA, it is crucial to consider mediators, such as confidence, that may account for the motor development of children (Lubans et al., 2010).

#### 2.4.6 Perceived motor competence and confidence

Within the realm of motor development, various terms such as "self-confidence", "self-efficacy", "perceived ability", and "perceived competence" have been used to describe one's perceived capability to accomplish a certain level of performance (Feltz, 1988). As previously mentioned, motor skill competence (MSC) is a mastery of physical skills and movement patterns that enable enjoyable participation in physical activities (Castelli & Valley, 2007). Numerous studies have confirmed the proposed positive relationship between children's motor competence and their PA levels (e.g., Holfelder & Schott, 2014; Logan et al., 2015; Lubans et al., 2010). Perceived skill competence (PSC) refers to an individual's perception of their actual movement capabilities (Harter, 1999). In other words, an individual's awareness and belief of their capability to perform both gross and fine motor tasks (Rudisil et al., 1993). Stodden et al. (2008) postulates that when children reach middle-to-late childhood (7-10 years old) their cognitive ability has developed to a stage where they can compare themselves to their peers (Harter, 1982). Furthermore, children who obtain a higher PMC due to a higher actual motor competence (AMC) will perceive tasks as being easier and are more likely to engage in them. This engagement in PA will allow a child to practice and develop their AMC, sustain their FMS proficiency and PA level (Stodden & Goodway et al., 2008).

Similarly, self-confidence (SC) refers to the perceived ability to accomplish a certain level of performance (Feltz, 1988). Research carried out in this field suggests that as children gain confidence in performing fine and gross motor skills, and build a sufficiently diverse movement repertoire, they acquire a high level of movement proficiency that is positively associated with the quality of their psychomotor and cognitive health (Chaddock-heyman et al., 2013). Various studies have been conducted assessing the relationship between perceived movement and actual movement competence (Babic et al., 2014; De Meester et al., 2016; Robinson, 2010) among children, while far fewer refer to perceived confidence (perception of ability/SE) (Robinson, 2010; Barnett et al., 2008), and specifically so within the female childhood population. A recent Irish study investigating the relationship between FMS proficiency and perceived physical self-confidence (PSC) levels among adolescents found a significant correlation (r = 0.305) between both variables for

females (McGrane et al. 2016). The McGrane et al., (2016) findings suggest that if a female has low actual FMS proficiency, she may in turn have low PSC levels or vice versa. Another recent Irish study (Farmer et al., 2017), investigating the relationship between PA, FMS proficiency and perceived PSC among pre-adolescent girls (n =160; M age =  $10.69 \pm 1.40$  years) found that while there were high levels of perceived PSC reported, there was no significant correlation between actual FMS proficiency and perceived PSC among the cohort. However, results reported that perceived PSC is a significant predictor (beta = 0.183) of the pre-adolescent participants' overall PA level.

Vedul-Kjelsås et al. (2012) also found that FMS and self- perception was most strongly correlated among girls (r = 0.312, mean age 11.46 years). Viholainen et al.'s study (2014) on self-concept, FMS, and psychosocial well-being also found that FMS proficiency is connected to self-concept and psychosocial well-being among adolescent girls. In contrast to this, McIntyre et al. (2010) compared the influence of both PMC and MSC on children's PA and found that MSC but not PMC, influenced children's PA behaviour significantly and explained 9% variance of 9-year-olds girl's PA level. It is important to note that it is considered 'normative' for young children to overestimate their abilities, because of cognitive limitations in finding it hard to distinguish between their 'ideal' in terms of competence, and their reality (Harter, 1984). Nonetheless, research suggests that inflated PSC can drive the acquisition of movement skills because children will continue to engage in mastery attempts in activities for which they believe they are skilful (Stodden et al., 2008). Furthermore, this early period has been termed a 'window of opportunity', as children (even if low skilled) can still be keen participators in PA (LeGear et al., 2012). It is, therefore, imperative to gather quantitative information about the actual and perceived movement skill level of the child (Hands, 2008) to allow for the implementation of interventions during this early period. Furthermore, the assessment of FMS (actual MSC) and PSC is particularly important during the preschool and school years as it monitors developmental changes and identifies developmental delays in children (Gallahue & Ozmun, 2006).

#### 2.4.7 FMS assessment in youth

Instruments which assess a child's FMS proficiency would provide opportunities for professionals working with young children in the context of sport and PE to 1) have a more objective understanding of children's skills, 2) better meet children's developmental demands, and 3) analyse the effectiveness of their interventions (Platvoet et al., 2018). Nevertheless, although monitoring children's FMS is considered to have numerous advantages, currently the implementation of FMS measurement in settings like sports training and PE is rather limited and the number of large-scale longitudinal studies on this topic are scarce (Lloyd et al., 2014). Platvoet and colleagues (2018) postulate that most FMS assessment instruments focus on identifying children with FMS development disorders in a clinical setting (one-on-one setting) and are rather time-consuming. Consequently, most existing instruments do not cover the whole performance spectrum (from low to high performance), making it difficult to differentiate between performance levels (Platvoet et al., 2018).

There is no gold standard of motor assessment for children and the available tests vary in their ease of use and interpretability in clinical and research settings (Piek et al, 2008). Griffiths et al., (2018) systematic review of seven movement skill assessment tools, defines these measurement instruments as either norm- or criterion-referenced (Logan et al., 2011). Criterion-referenced tests are designed to be scored as items or criteria are demonstrated; meaning that the score is a reflection of a child's competence on the test items (Griffiths et al., 2018). Most available assessments, however, are norm-referenced, meaning that a child's results are reported concerning a specific population (Spittle et al., 2013). Currently, there are several instruments, which can measure children's FMS performance level, e.g., the Movement Assessment Battery for Children 2 (Henderson et al., 2008), the Bruininks-Oseretsky Test for Motor Proficiency 2 (Bruininks & Bruininks, 2005), and the Test of Gross Motor Development (Ulrich, 1985, 2000). For the purpose of this literature review, the TGMD-2 and Get Skilled Get Active FMS assessment instruments will be discussed.

The Test for Gross Motor Development-2 (TGMD-2) has been extensively used as a process assessment tool of children's motor skill performance (3–10 years old) (Issartel et al., 2017; Barnett et al., 2014). This process-orientated tool consists of six locomotor skill subtests (run, gallop, jump, slide, hop, and leap) and six object control subtests (striking a stationary ball, stationary dribble, catch, kick, overhand throw, and underhand roll) (Ulrich, 2000). The TGMD-2 is the only tool that assesses gross motor skills in isolation and that focusses on quality of performance (Griffiths et al., 2018). For each of the skills, there are between three and five performance criteria (Ulrich, 2000). Participants perform each of the 12 FMS with each gross motor skill including several behavioural components presented as performance criteria (Ulrich, 2000). The TGMD-2 is notable for the inclusion of credit towards incomplete skills in the scoring system, in which points are awarded for the quality of the action performed, instead of satisfactory completion of the task only. These actions include preparatory movements prior to running and jumping, or arm position during movements (Griffiths et al., 2018). The TGMD-2 is designed to assess the FMS in children and has been used in many studies (Farmer et al., 2018; Mitchell et al., 2013; Rudd et al., 2016; Slykerman et al., 2016), due to its empirically determined reliability and established validity (Barnett et al., 2014; Houwen et al., 2010; Issartel et al., 2017; Lopes et al., 2016).

Lopes et al.'s (2016) study investigated the reliability and construct validity study of the test of TGMD-2 in 5-10-year-old Portuguese children and found that the test-retest reliability analysis was good, with an agreement ratio of .96 (.09) for 12 skills. Cronbach's alpha values showed acceptable internal consistency (.69 for 12 items, .46 for locomotor skills, and .64 for object control skills). Results from this study provide strong support for the construct validity of the TGMD with correlations between the object control and locomotor factors yielding a high positive value (0.82). Kim et al. (2014) study investigated the reliability and validity of the TGMD-2 for South Korean children. The composite reliability of locomotor (.91) and object control (.85) factors highlights that the TGMD-2 is a usable process-orientated gross motor assessment. Finally, Valentini (2012) further evaluated the reliability and validity of the TGMD-2 in 3,124 Brazilian children; all factor loadings were again statistically significant with construct validity evident.

In recent years, the TGMD has been revised to its third edition, the TGMD-3, to accommodate potential changes in the normative population, as well as to incorporate recommendations from experts in the field of motor development and practitioners who frequently use the assessment (Ulrich, 2013). Noticeable changes

included dropping two skills from the TGMD-2, the leap, and underhand roll, and replacing these with the skip, underhand throw, and one-hand strike in the TGMD-3. Validity and reliability of the TGMD-3 have been recently reported among children providing confidence for the usage and collection of new norms (Magistro et al., 2020; Valentini et al., 2017; Webster & Ulrich, 2017).

An Australian FMS measurement tool 'Get Skilled: Get Active' has been developed to help support teachers with the increase of primary school students' levels of gross motor skill proficiency (NSW Department of Education and Training, 2000). This process-orientated FMS instrument consists of 12 FMS; a unique strength of the 'Get Skilled: Get Active' resource is that all categories of FMS are included within this assessment protocol (locomotor, object- control and stability) (Gallahue and Ozmun, 2006; Haywood and Getchell, 2009). These 12 skills (run, balance, vertical jump, catch, hop, side gallop, skip, overarm throw, leap, kick, two-handed strike and dodge) were selected because collectively, they represent a platform for the development of 65 specialised skills, enabling students to participate in a wide range of physical activities (NSW Department of Education and Training, 2000).

All 12 FMS are composed of observable, behavioural components that together constitute a mature performance of the skill (Okely and Booth, 2004). Each FMS has been broken down into individual components that are developed at different stages of learning; checklists for teachers also exist to help identify student performance (NSW Department of Education and Training, 2000). A substantial amount of evidence in Australia highlights that the 'Get Skilled: Get Active' FMS assessment protocol is an appropriate, reliable, culturally acceptable and valid instrument for measuring levels of gross motor skill proficiency amongst children and adolescents (Barnett et al., 2010; Hardy et al., 2010; Okely & Booth, 2004).

In summary, the TGMD-2 and Get Skilled Get Active are designed to assess the gross motor functioning in children 3 through 10 years of age and as mentioned previously, has empirically determined reliability and validity for this age cohort (Barnett et al., 2014; Issartel et al., 2017; Lopes et al., 2016). Monitoring children's FMS is considered to have several advantages, however currently the implementation of FMS measurement in settings like sports training and PE is rather limited and the number of large-scale longitudinal studies on this topic are scarce (Lloyd et al., 2014). Nevertheless, the assessment and teaching of FMS in young children is an important core component of many early childhood PA and sport interventions (Hands & McIntyre, 2015). The assessment of movement skills is a critical component of many disciplines and professions (Basman, 2019). Although the specific contexts and applications may vary, the basic concepts and strategies are similar to plan efficient and effective movement programmes or to support the involvement of a child with special needs in the wider community (Hands, 2012).

# 2.4.8 Advancing towards specialised movement skills in a youth sporting context – the relationship between sports participation and the level of motor coordination in childhood.

A positive relationship between participation in organised sport and motor performance in children exists (Elhakeem et al., 2018; Hardy et al., 2014; Kokstejn et al., 2019; Larsen et al., 2015). Motor performance is a combination of FMS and fitness-related components (Biddle, 2007), and is likely an important factor in developing good sports-specific skills and physical activities (Jiménez-Díaz et al., 2019; Okely et al., 2011). Recent evidence shows that motor performance in children predicts PA levels three years later (Larsen et al., 2015) and that early life motor performance predicts engagement with sport and exercise throughout life to older adulthood (Elhakeem et al., 2018). For example, Hardy et al. (2014), using a cross-sectional design, found that participation in organised PA was positively associated with fundamental motor skill performance in 10–16-year-olds. In a prospective longitudinal study, children with high motor competence had greater levels of participation in organised sport at a two-year follow-up than children with low motor competence (Fransen et al., 2014).

Additionally, in a longitudinal study investigating the relationship between sports participation and the level of motor coordination among 6-9-year-old children (n=371), with two-time points, children who reported participating in organised club sport had better motor coordination than children who partially participated or did not participate in organised club sport (Vandrope et al., 2012). Vandrope and colleagues reported that the basic level of motor coordination and the amount of club sports

participation significantly predicted sports participation two years later (Vandorpe et al., 2012). Recent findings from the Vallence et al., (2019) CHAMPS Study-DK complement the above findings, showing that participation in organised sport from 1067 Danish school-aged children (6 to 12 years) is positively associated with motor performance across the coordination fitness spectrum. In particular, participation in organised handball and gymnastics was positively associated with coordination-related and fitness-related motor performance. Despite the limitation of the lack of process-oriented outcomes in this study, the findings from Vallence et al. (2019) suggest that participation in organised handball and gymnastics, or any sport requiring coordination, agility, and fitness, could be a promising method for improving motor performance in children.

Findings from previous community sports-based youth interventions (Camiré & Trudel, 2014; Chan et al., 2019; Koh et al., 2014; Lubans et al., 2017; Santos et al., 2019) provide evidence that FMS coach support strategies may be an effective way to improve FMS competencies and mastery. The acquisition of FMS is not only achieved through natural development and maturation, but also through continuous interaction with a stimulating and supportive social and physical environment including attractive and sufficient space, a stimulating social attitude, as well as a professional instructional approach (Chambers & Gregg, 2016; Wick et al., 2017). As such, a variety of evidenced-based approaches have been used to teach FMS utilising a variety of pedagogical approaches. Thus, a broad range of both 'instructional models' (Gurvitch & Metzler, 2013), as well as teaching strategies (Lee et al., 2020; Rogers et al., 2020), can be implemented when teaching FMS. FMS can be taught and practiced within a game-like environment, where gameplay, either structured or non-structured is integrated into the curriculum or practice environment (Barnett et al., 2016).

For example, Launder and Piltz (2013) in their Play Practice Model suggest expertise in skills can be taught within the game context. Additionally, sport-specific skills, through a Teaching Games for Understanding (TGFU) instructional modelapproach (Sheppard, 2014) has been shown to empower young children, deepen their knowledge of game tactics, help improve skills, and contribute to enjoyment levels (Butler & Griffen, 2010). The important point to note when motor skills are taught together/within game components is that FMS contributes to the development and provide a framework for instruction within integrated models of instruction (especially for non-specialist PE teachers in primary schools) (Barnett et al., 2016). A less studied phenomenon is that FMS in children can also be developed and improved through the medium of dance Lykesas et al., 2014; Robertson-Wilson et al., 2016). Creative dance is currently an exceptional contemporary pedagogical and teaching approach that assists and supports significantly the harmonious kinetic development of individuals (Lykesas et al., 2014). Among youth, dance programmes have facilitated changes in physical outcomes, such as fitness (e.g., Connolly et al., 2011; Urmston et al., 2012), as well as in psychological outcomes (e.g., Connolly et al., 2011; Duberg et al., 2013; Tsompanaki et al., 2019; Urmston et al., 2012).

A variety of intervention models have been trialled with school children to enhance children's FMS (e.g., (Bryant et al., 2016; Duncan et al., 2017; Fagenbaum et al., 2013; Robinson et al., 2015). These aforementioned interventions have had success but largely focus on the practice of FMS skills in isolation and without the context of sports performance (Duncan et al., 2020). However, motor coordination is not the sole determent and the initiation and continuation of sport participation in childhood (Vandorpe et al., 2012). Weiss (1993) postulates that 1) developing and demonstrating competence (skill, fitness etc.), 2) gaining social acceptance and support (parents, peers, coach etc.), and 3) enjoyment derived from participation are the major motives essential to initiating and maintaining involvement in OYS. Evidence indicates that planned instructional programmes emphasising practice on skill acquisition in early childhood and the transition into middle childhood can enhance the development of basic movement patterns in children 4 and 5 years of age and more complex skills in older children (Malina, 2012). Guided instruction by specialists, trained parents or qualified coaches, appropriate motor task sequences, adequate time for practice, and constructive feedback and guidance are essential components of successful instructional programmes at young ages (Kohl et al., 2013; Pearson et al., 2015).

The importance of FMS acquisition among youth in the PA and sporting contexts was rationalised in this section. The literature and research evidence strongly suggests that FMS proficiency have a significantly positive role to play in the development of sport-specific skills (Hardy et al., 2012; Jaakkola et al., 2015; Stodden et al., 2008), and increasing habitual PA level amongst children and adolescents (Lee

et al., 2018; Logan et al., 2015; Lubans et al., 2010; Robinson et al., 2015). A recent systematic review of the health benefits associated with FMS competency found strong evidence for a positive association between FMS competency and PA in children (Engel et al, 2018). It is evident in the recent literature that the causal relationship between FMS and PA is a subject that continues to be addressed by current research, and further studies would strengthen the evidence base (Bremer & Cairney, 2018).

Despite the numerous associated benefits, it has become evident that a poor trend of skill execution and low proficiency levels exists amongst children and young people worldwide (Bardid et al., 2016; Bryant et al., 2014; Eather et al., 2018; Lester et al., 2017; Mitchell et al., 2013). Gender differences in FMS among children also exist (Eather et al., 2018; Barnett et al., 2013; Bolger et al., 2018), and research indicates that boys have commonly been found to demonstrate higher levels of overall FMS compared to girls (Behan et al., 2018; Barnett et al., 2016). It is, therefore, essential for researchers to examine whether there is consistency in these gender differences to tailor FMS intervention programmes that are unique to the child's developmental requirements (Goodway et al., 2013). Research suggests that without FMS proficiency and a positive perception of such, children may be less likely to engage in PA (Lubans et al., 2010), and this is of particular importance for vulnerable at-risk girls. When children perceive themselves as confident in performing FMS, and build a sufficiently diverse movement repertoire, they acquire a high level of movement proficiency that is positively associated with PA (Chaddock-Heyman et al., 2013).

Assessing a child's FMS proficiency would provide opportunities for professionals working with young children in the context of sport to (1) have a more objective understanding of children's skills, (2) better meet children's developmental demands, and (3) analyse the effectiveness of their interventions (Platvoet et al., 2018), as was discussed. Findings from previous community sports-based youth interventions (Lubans et al., 2017; Rogers et al., 2020) provide evidence that FMS coach support strategies may be an effective way to improve FMS competencies and mastery. A variety of evidenced-based approaches have been used to teach FMS utilising a variety of pedagogical approaches including 'instructional models' (Gurvitch & Metzler, 2013) as well as teaching strategies (Rogers et al., 2020), namely TGfU (Sheppard, 2014), the Play Practice Model (Launder & Piltz, 2013), and also

through the medium of dance (Connolly et al., 2011; Robertson-Wilson, 2016). School and community-based programmes that include developmentally appropriate FMS learning experiences delivered by PE specialists or highly trained classroom teachers significantly improve FMS proficiency in youth (Morgan et al., 2013).

## 2.5 BRIDGING THE GAP: PROMOTING GIRLS' PA AND SPORT

### PARTICIPATION

The evidence in this literature review has clearly outlined and emphasised that PA is important to young people for health, specifically on children's physical, psychosocial, and intellectual development (Eime et al., 2013; Sallis et al., 2016; Woods et al., 2018), and for this reason intervention programmes have been designed, developed, implemented and evaluated with the purpose of increasing PA and/or FMS (Costello & Warne, 2020; Chan et al., 2016; Cohen et al., 2015; Mostafavi et al., 2013; Rudd et al., 2016; Sutherland et al., 2013).

# 2.5.1 Call for multi-component community sports-based PA interventions for girls

Pre-adolescent girls are an important target for PA behavioural change strategies, as this age cohort may enhance tracking into the crucial period of adolescence (Eime et al., 2020). The quantification of PA intervention effectiveness for this age group of girls has not been adequately reported (Biddle et al, 2014). A comprehensive review was reported by Van Sluijs et al (2007), and PA interventions conducted with pre-adolescent children showed no or inconclusive effectiveness when analysed across a variety of settings. Further to this, no distinction was made in the results by gender, thus while we know that PA levels of boys and girls differ, we do not yet know whether targeting girls alone is more effective than a mixed-sex intervention.

Previous systematic reviews (Camacho-Minano et al., 2011; Voskuil et al., 2017; Salmon et al., 2007) and a meta-analysis (Pearson et al., 2015) have assessed interventions to promote PA in adolescent girls across the school and community settings. Voskuil et al. (2017) reported highly variable effect sizes, inferring that PA interventions only had a small effect on objectively measured PA in girls aged 6–18 years (Voskuil et al., 2017). Camacho-Minano et al. (2011) found overall mixed results regarding the effectiveness of PA interventions for adolescent girls aged 5–18 years but, suggested that multicomponent school-based interventions, which included PE that addressed the unique needs of girls were the most effective. Pearson et al. (2015) reported small but significant effects (g= 0.35, p < 0.001) for the effectiveness of PA interventions on girls aged 12 to 18 years. Biddle et al, (2014) meta-analysis study, conducted to test whether an intervention to increase PA in young girls (aged 5-11)

years), reported a significant small positive treatment effect (p<.001) for experimental groups participating in PA interventions (N=1641). The differential between treatment and control groups indicated that there was an equivalent of 12.17% more PA for girls participating in the treatment conditions.

To develop and implement targeted strategies to increase team sport participation and, consequently, PA levels among pre-adolescent and adolescent girls, there is a need to understand trends in, and influences on, girls' participation in team sport (Eime et al., 2020), based on the above mixed results regarding the effectiveness of PA interventions for girls. Although previous systematic reviews have examined the effectiveness of interventions designed to increase PA in general (Biddle et al., 2014; Cavill et al., 2012; Metcalf et al., 2012; Yuksel et al., 2020), to date there is no known systematic review investigating the effectiveness of PA interventions that aim to promote girl's participation in team sport in a community setting (Allison et al., 2017). Furthermore, efforts specifically focusing on PA promotion are increasing and improving the PA levels of youth has been recognised as an important public health challenge (Sallis et al., 2016).

#### 2.5.2 PA and sport interventions - evidence for successful implementation

#### 2.5.2.1 School-based PA interventions

Schools are an ideal environment to promote PA in children and help them learn how to maintain activity (Trost et al, 2014). Indeed, a recent multi-level, worldwide review highlighted local school contexts as important correlates to PA in children (Gomes et al., 2017). In accord with the WHO (Langford et al., 2015), Booth and Okely (2005) highlighted the compulsory nature of attendance, teachers as credible change agents, and access to facilities as the primary strengths of a school as an intervention setting. Schools have been a popular setting for such interventions, as they offer continuous, intensive contact with children, and the school infrastructure and physical environment, policies, curricula, and staff have the potential to positively influence children's health (Jurak et al., 2013). Research suggests that children exposed to school-based PA interventions are approximately three times more likely to engage in MVPA during the school day than those not exposed (Dobbins et al., 2013). Evidence from systematic reviews (Kreimler et al., 2011; Van Sluijs et al., 2007) shows that the

most effective school-based interventions among adolescents used whole-of-school approaches that link curricula activities with the broader school environment and the local community. School-based interventions that are multi-component, simultaneously targeting curricular, school environment and policy, and community links, are a promising approach for promoting PA (Okely et al, 2011).

A 2015 meta-analysis of PA interventions with adolescent girls showed that public health-focused interventions can be effective, but the intervention effect was enhanced if the interventions included only girls, was school-based and employed a theory of behaviour change (Pearson et al., 2015). Most recently, Owen et al.'s (2017) systematic review and meta-analysis on the effectiveness of school-based PA interventions for adolescent girls found intervention effects were significant but very small (k = 17, g= 0.37, p < 0.05), which indicates that changing PA behaviours in adolescent girls through school-based interventions is challenging. Hence, tailoring interventions to the interests and needs of girls are likely to enhance effectiveness (Jago et al., 2015).

The Biddle et al., (2014) meta-analysis study on interventions seeking to increase PA in young girls (aged 5-11 years), reported a significant small positive treatment effect (p<.001) for experimental groups participating in PA interventions (N=1641). In addition to this, Girls on the Run (GOTR) programme (N=877), found statistical significance in participation self-esteem (p=0.001), body satisfaction and PA frequency (p=0.010) among elementary school girls (Gabriel et al., 2011). In line with the aforementioned studies, TAAG (Trial of Activity for Adolescent Girls) found positive PA outcomes for adolescent girls (n=3504). Results indicated that intervention schools (n=36) were more physically active than girls in control schools (mean difference 10.9 MET-weighted minutes of MVPA, 95% CI=0.52–21.2) (Webber et al., 2008).

In the UK, Harrington et al.'s (2018) recent cluster randomised controlled trial (N=1211), examining the effectiveness of the 'Girls Active' school-based PA programme, found no difference in change in accelerometry measured MVPA (mins/day; 95% confidence intervals) between Girls Active control and intervention schools at 14 months (1.7; -0.8 to 4.3), however, there was at seven months (2.4; 0.1 to 4.7). Dewar et al (2013) Nutrition and Enjoyable Activity for Tween Girls (NEAT) programme in New South Wales, Australia, evaluated the impact of a 12-month

school-based multi-component programme on adolescent girls' (n=357,  $13.2 \pm 0.5$  years) PA from 12 secondary schools. In contrast to the aforementioned studies, results found the intervention did not impact upon any of the PA outcomes or hypothesised mediators of PA behaviour change, and no significant results were found for within or beyond school day PA.

Moreover, other interventions targeting pre-adolescent and adolescent girls have experienced similar challenges in their attempts to improve PA behaviours. For example, Girls in Sport (GIS) study (Okely et al., 2017), a group-randomised trial aiming to prevent a decline in MVPA among adolescent girls (N=1518 girls, mean age  $13.6\pm0.02$ ) found no statistically significant differences between the intervention and control groups on any of the variables. Similar to Okely et al.'s (2017) GIS study, a more recent school-based PA intervention, specifically identified as 'The Bristol Girls Dance Project' (BGDP) found no significant difference between the intervention and control groups (Jago et al., 2015) among 11 and 12-year-old girls (N=508). Inevitably, there is a need to seek new ways to assist pre-adolescent and adolescent girls to be physically active by identifying ways to support and encourage sustained engagement in PA over the life course (Jago et al., 2015).

The effectiveness of multi-component school-based PA interventions, particularly those that include links to family and community has consistently been demonstrated in reviews (Cleland et al., 2012; Harrington et al., 2018; Kriemler et al., 2011; Okely et al., 2017; Van Sluijs et al., 2007). These interventions have had some success, with short-term improvements in school-based PA, but limited evidence of positive effects on PA outside of school or during leisure time (for example, sport) (Van Sluijs et al., 2007). Multi-component interventions and interventions underpinned by theory may be the most effective approaches to positively change adolescent girls' PA (Owen et al., 2017).

#### 2.5.2.2 Community-based PA interventions for girls

As well as the school environment, the family and community environment are key for young people's behaviour and for promoting PA (Van Sluijs et al., 2011). Community-based programmes have the capacity to involve parents and other adult role models from the community (e.g. relatives, coaches, older players) who can influence the PA behaviour of children (Legg et al., 2015). For any PA intervention to be sustainable, the primary responsibility for its maintenance and management must come from the community (Draper et al., 2009). Sims et al. (2015), in a recent systematic review and meta-analysis of controlled studies, examined the effectiveness of 22 interventions on sustained childhood PA. Not only did results reveal that males reported higher levels of MVPA than females (2.67mins/day: 95% CL 2.05, 3.28) but community settings were more effective than school settings (1.70 mins/day: 95% CL 4.84, 8.25) for sustained childhood PA interventions, furthermore suggesting that PA community-based PA and sports interventions may open avenues for PA sustainability among vulnerable at-risk girls.

For example, VERBTM Summer Scorecard (VSS), is a community-wide intervention that sought to increase early adolescent PA participation (9-13 years) (R.D DeBate et al., 2009). Results found that the intervention group (n=1,253) was significantly related to meeting recommendations for vigorous PA (VPA) (OR = 2.08, P = 0.0259), being physically active on weekends (OR = 1.84, P = 0.0017), and reporting more days of trying a new game or sport (OR = 1.49, P = 0.046), unlike the comparison group (n=866). Contrary to DeBate et al. (2009) findings, Robbins et al. (2019) evaluated the effect of a 17-week Girls on the Move (GOTM) intervention on increasing MVPA among fifth to eighth-grade girls in 24 schools in Midwestern USA. Findings reported no between-group differences occurred for weighted mean minutes of MVPA per week at post-intervention (B = -0.08, p = .207) or 9-month follow-up (B = -0.09, p = .118) while controlling for baseline MVPA. The Childhood Health Activity and Motor Performance School Study Denmark (CHAMPS-DK), focused on 1124 primary school-aged participants, found children playing soccer at any frequency demonstrated 3 to 15 times increased odds of achieving at least 60 minutes of MVPA per day as compared to children not participating in organised sport (Hebert et al., 2015).

Community sports-based settings play an imperative role in young girls' participation and retention in OYS (Eime et al., 2020; Mäkelä et al., 2017). Club based participation in sport has been associated with higher rates of meeting the PA guidelines compared to those not participating in sport (Hebert et al., 2015; Marques et al., 2016; Vella et al., 2013). Despite this, there is a lack of effective intervention strategies to promote PA in female youth (Anokye et al., 2018; Baker et al., 2011),

and particularly within a sporting context (Whitley et al., 2019). Although previous systematic reviews have examined the effectiveness of interventions designed to increase PA in general (Cavill et al., 2012; Metcalf et al., 2012), to date, there is no known systematic review investigating the effectiveness of PA interventions that aim to promote participation in team sport specifically (Allison et al., 2017). Additionally, evidence on the use of complex interventions delivered by sports organisations to improve sports participation is lacking (Mansfield et al., 2015). Cavill et al.'s (2012) literature review corroborated this finding and particularly emphasised the low quality of studies. This is particularly so for community-based sports interventions (Cavill et al., 2012) which were characterised by sparse evidence (n=6), methodological weakness (e.g. only one data point pre-intervention and post-intervention and/or very small sample sizes), with mixed results or with some evidence relevant to only very specific populations (e.g. men who play football) (Cavill et al., 2012).

Despite the recommendations for multi-level community-wide interventions for increasing PA, interventions focused on the individual level persist, due, in part, to the challenges that community-based programmes present (Elder et al. 2006; Koorts & Gillison, 2015). For example, community-based programmes may require various levels of activities, such as tailoring to specific audiences, developing partnerships, organising community-based events, promoting them through the mass media, and having social support systems that require coordination of planning, implementation, and evaluation (Elder et al. 2006; Reger-Nash et al. 2006). In terms of the opportunities for organised sports participation outside of school hours, Eime et al. (2015) have proposed young children can participate in modified sports programmes as a pathway to club sport competition (Eime et al., 2015).

#### 2.5.2.3 Modified sports programmes – an avenue to increase PA participation

Modified sports programmes are offered to engage children in play activities designed, among other things, to develop FMS and sport-specific skills for future participation (Cote et al., 2009). These programmes are modified to match the developmental capacity of children and are aimed at the development of FMS and sport-specific skills, rather than competition (Eime et al., 2018). Essentially, the sport is modified to match the developmental capabilities of children by adapting games and activities

through changes to the rules, equipment, and/or physical space to encourage inclusion and maximise participation (Eime et al., 2018). Modified sports are entry-level programmes; sports organisations (clubs and governing bodies) should have an interest in retaining players, which requires these young people to transition into clubbased competition.

In the Eime et al. (2015) longitudinal study, female players (n= 13,760; aged 4-10 years) of the modified sports programme were followed over 4 years to determine their pattern of transition to club sport competition. Results showed that the majority (59%) transitioned and the peak sport entry age with the highest rates of transition was 7-9 years (Eime et al., 2015). Furthermore, modified sports programmes should be encouraged in an Irish youth sporting context, as recent findings reveal a high proportion of Irish children participate in organised sport, however, a high dropout rate exists with increasing age (Harrington et al., 2016; Woods et al., 2018). Most recent literature has found that less than one quarter (24%) of pupils in 1st year (aged 12-13) of post-primary are not members of a sports club this has increased to 47% by the time they get to 6th year (ages 17-19) (Woods et al., 2018). Therefore, from a health perspective implementation of modified sports programmes continued participation in sport and PA is important (Eime et al., 2018). Consequently, multifaceted interventions that target multiple settings such as schools and communities, and include modified sports programmes are considered important for making positive changes to female youth PA levels (Eime et al., 2015; Murillo Pardo et al., 2013; Van Sluijs et al., 2007).

#### 2.5.3 Effective PA intervention components

A recent systematic review and meta-analysis of PA community sports-based interventions among girls (Pearson et al., 2015) found that greater intervention effectiveness appears to result from multi-component interventions, theory-based, (Sluijs et al., 2007) focused on girls only (Camacho-Minano et al., 2011), target both PA and SB (Pearson et al., 2015), and include a non-formal coach education element (Koh et al., 2014). Based on the available evidence, Ling et al. (2015) concluded that multi-component interventions are more successful than single-component interventions. Multi-component interventions including a number of methods have

proven successful in targeting and improving youth PA participation (Messing et al., 2019; Pearson et al., 2015), namely:

1) combinations of support components (i.e. coaches, family, friends, etc.)

2) individual components (i.e., specifically tailored programmes for individuals/groups)

3) choice components (different options to facilitate behaviour change)

4) educational (i.e., coach education) and environmental components

Other important themes identified as central influences on young people's meaningful experiences and engagement in PA and sport in previous studies include; the social interaction, fun challenges, motor skill focus, non-competitive nature, and personally relevant PA opportunities (Beni et al., 2017; Springer, 2013). James et al. (2018) propose that policy-makers and those involved in PA delivery (e.g., schools, local council and local activity providers) should include young people in designing interventions and facilities to ensure they are meeting the needs of this age group and providing the right opportunities for teenagers to be active. That is unstructured, local, low cost, fun, sociable opportunities and the right facilities to be active (James et al., 2018). Providing girls with such ownership over the intervention design and ensuring sessions are fun and enjoyable, is consistent with multiple principles of the Self-Determination Theory (SDT), including intrinsic motivation, relatedness needs, autonomy support and autonomous motivation (Ryan & Deci, 2000).

Most recently, James and colleagues qualitative study on teenage recommendations to improve PA for their age group (James et al., 2018), identified key recommendations which included 1) make activities more specific to teenagers, 2) give teenagers a choice of activities/increase variety of activity and 3) provide activities that teenage girls enjoy (e.g., fun, sociable and not competitive sport). Furthermore, recommendations made by teenagers to improve activity provision, uptake and sustainability of PA engagement for both themselves and their peers (James et al., 2018). Through doing so, a further understanding can be gained of the current barriers, facilitators and motivation (Patay et al., 2015; Quested et al., 2013; Taylor et al., 2012).
Interventions that had a positive effect on PA behaviour in children integrated structured PA into the formal curriculum (Goh et al., 2016; Martin & Murtagh, 2017; Ward et al., 2010) and one of these components should be a structured PA programme (Innerd et al., 2019). Ward et al. (2010) stated that there is a need for teacher training and that structured PA programmes should be implemented by staff who are specialised and well-trained. Having well-qualified staff could also support skill development and competence in children (Messing et al., 2019). As previously discussed, sports participation can be instrumental in the physical, social and emotional development of children and adolescents (Eime et al., 2013). Yet, the quality of instruction from sports coaches is highly variable, and not all youngsters have positive experiences with sport (Langan et al., 2013; Langan et al., 2015; Merkel, 2013). Indeed, attrition rates for sports participation are substantial (Eime et al., 2019), particularly during the teenage years, and 'lack of enjoyment' and 'problems with the coach' are commonly cited reasons for drop-out (Farmer et al., 2018; Somerset & Hoare, 2018; Wetton et al., 2013). As such, there is scope to improve the quality of instruction across organised PA settings (Fenton et al., 2016).

A systematic review of interventions to promote PA within young and adolescent girls reported that the strongest evidence of success was associated with multi-component school-based interventions that are combined with community opportunities for PA that address multiple levels of influence on behaviour (i.e. as outlined in socio-ecological frameworks) and include enjoyable PE experiences as the main component (Camacho-Miñano et al., 2011). Effective interventions are also reported to consist of both a PA and cognitive component (Demetriou et al., 2019). Peer support strategies are also considered promising; whereas family support has been reported by some as ineffective (Camacho-Miñano et al., 2011). Strategies considered most appropriate for public health agencies and partners to promote PA to adolescents include community-wide campaigns, increased access to places in which to be active, educational information or outreach programmes, and enhanced PE programmes integrated within a socio-ecological perspective (Casey et al., 2014; Mehtälä et al., 2014).

In addition to this, recommendations for school- and community-based PA programmes from various countries have FMS development as an integral aspect of physical education, school and community sport (Morgan et al., 2013; O' Brien et al., 2016b). Recent evidence supports a positive association between competence in FMS

and PA in young people (Hulteen et al., 2018). For example, the effectiveness of a community-based FMS intervention, "Multimove for Kids" project, reported that the intervention group demonstrated a higher gain in both LOM ( $\beta$  = 3.78, SE = 1.08, p < 0.001) and OC ( $\beta$  = 4.46, SE = 1.06, p < 0.001) skills than the control group in children aged 3–8 years (Bardid et al., 2017). Additionally, girls demonstrated a lower gain in OC skills ( $\beta$  = -3.50, SE = 0.49, p < 0.001) and higher gain in LOM skills ( $\beta$  = 1.01, SE = 0.44, p = 0.022) than boys, regardless of group (Bardid et al., 2017). A variety of intervention models have been trialled with school children with a view to enhancing children's FMS (e.g., (Bryant et al., 2016; Duncan et al., 2017; Robinson et al., 2015). These aforementioned interventions have had success but largely focus on the practice of FMS skills in isolation and without the context of sports performance (Duncan et al., 2020). Findings from previous community sports-based youth interventions (Lubans et al., 2017) provide evidence that FMS coach support strategies may be an effective way to improve FMS competencies and mastery.

With a specific focus on increasing PA amongst girls, research has shown the importance of providing a wide range of appealing activities (Morgan et al., 2019), providing girl-only sessions (Mitchell et al., 2015), and underpinning an intervention framework with theory (Owens et al., 2017). Recognising the importance of motivation for the long-term adoption of a behaviour (Deci & Ryan, 2000), many interventions focus on attempts to promote action by converting motivation into action (Hardcastle et al., 2015). Such motivation can come from observing positive results in others and it has been suggested that role models can inspire for young people to become involved in, or maintain involvement in, sport and PA (Lawler et al., 2020). Meier (2015) specifically recommends the use of female role models within local communities to increase PA among females.

These recommendations also highlight the importance of ensuring sustainable engagement in PA through the development of school and community links (Morgan et al., 2019). Interventions linking the school and wider community have shown to be effective in sustaining changes in PA levels among adolescents (Casey et al., 2014; Van Sluijs et al., 2007), which ultimately requires effective partnership working within communities. That said, there is a need, therefore, to design and implement preventative strategies to help evade the age-related decline in PA that is characteristic of adolescent girls (Morgan et al., 2019).

## 2.5.4 Research considerations and frameworks for designing and evaluating PA and sport intervention studies

Various frameworks for the development of health promotion interventions have been suggested including intervention mapping, the behavioural epidemiology framework and the MRC (Medical Research Council) framework for developing and evaluating complex interventions (Craig et al., 2008; Fernandez et al., 2019; Sallis et al., 2000). These frameworks generally suggest using behaviour change theory, existing evidence and conducting formative research with the target group (Craig et al., 2008; Fernandez et al., 2019; Sallis et al., 2000).

The original UK Medical Research Council (MRC) framework for evaluating complex interventions recommended sequential phases of development, feasibility testing and evaluation, culminating in the estimation of an effect size via a randomised controlled trial (RCT), prior to wider implementation (Campbell et al., 2000). The 2008 update of the MRC guidance for complex intervention development and evaluation provides a four-phase, cyclical framework advising health researchers to answer a range of sequential questions regarding complex intervention theory, feasibility and acceptability, effectiveness and cost-effectiveness, and sustainability (Craig et al., 2008), as shown in Figure 2.2.



Figure 2.2: Medical Research Council (2000) Framework- Phases of investigation in the evaluation of a complex intervention.

The revised 'Gaelic4Girls' intervention includes the pre-clinical Phase 0, Phase 1 and Phase II, as highlighted in Figure 2,2.

- 1. *The first phase (intervention development):* involves the development of an intervention's theoretical rationale, often depicted in a 'logic model' describing inputs that the intervention involves, the processes that these initiates, and the mechanisms via which these are intended to realise positive outcomes. This phase should identify underpinning 'active ingredients' and how intervention components are expected to synergistically interact with one another, and with the context of delivery, to generate outcomes (both intended and unintended) (Bonell et al., 2014).
- The feasibility and piloting phase: include testing the feasibility and acceptability of the proposed intervention and its evaluation methods. Although the exact distinction between feasibility and pilot studies is contested (Lancaster, 2015), pilot studies may simply be a smaller version

of the main trial, aiming to implement the intervention and its trial on a smaller scale (for example, with smaller samples, in fewer sites and/or for shorter follow-up periods), while feasibility studies may focus only on select intervention or trial elements about which there is particular uncertainty. Further refinements may be made to the intervention theory after this phase to optimise the intervention design, logic model and the proposed evaluation design prior to testing effectiveness and cost-effectiveness (O'Cathain et al., 2019).

3. *RCTs and 'implementation studies':* Once a well-theorised intervention has been developed and feasibility questions addressed, RCTs are recommended to examine their effectiveness (and cost-effectiveness) whenever randomisation is practicable (Craig et al., 2008). Finally, 'implementation studies' are also needed to address the scale-up of interventions into routine practice (Craig et al., 2008). The cumulative effect of these processes should be the generation of a strong theoretical and evidence base for public health intervention which provides greater confidence that outcomes observed during trials can be replicated in real-world settings, and which supports the ongoing cycle of developing and evaluating complex interventions (Fletcher et al., 2016).

Additionally, according to the MRC (2008), there are some fundamentally important considerations for researchers to take into account developing an intervention including:

- Being clear about what you are trying to do: what outcome you are aiming for?
- How you will bring about change?
- Does your intervention have a coherent theoretical basis? Have you used this theory systematically to develop the intervention?
- Can you describe the intervention fully, so that it can be implemented properly for your evaluation, and replicated by others?
- Does the existing evidence ideally collated in a systematic review suggest that it is likely to be effective or cost-effective?

Can it be implemented in a research setting, and is it likely to be widely implementable if the results are favourable? (Craig et al., 2008)

Sport-based youth development interventions founded on principles of inclusion and participation (Borbee et al., 2019) expand their focus beyond physical health, promoting holistic development in physical, cognitive, affective, social, and lifestyle domains (Hermens et al., 2017). Sport-based youth development interventions underpinned by theoretical frameworks which involve youth in physical activities that intentionally foster developmental assets (e.g., values, commitment to learning, social competencies, positive identity) and surround youth with protective factors (e.g., support, relationships, experiences, resources, opportunities) are important facilitators of health for at-risk youth (Hermens et al., 2017; Holt et al., 2016; Whitley et al., 2019).

#### 2.5.5 Theoretical considerations for successful PA and sportinterventions

Even though quite a few studies refer to a theoretical model (e.g., social cognitive theory, the theory of planned behaviour, transtheoretical model, positive youth development, self-determination theory) to explain PA behaviour and sport participation enhancement (Holt et al., 2016; Michie et al., 2011), very few of them explicitly design and evaluate their intervention based on the respective theory (Demetriou & Höner, 2012; Owen et al., 2017). The current G4G intervention is underpinned by the established SDT (Deci & Ryan, 2002), and elements of the social-ecological model (SEM) (Stokols, 1996), because it's theoretical foundations are concerned with how the psychological and socio-environmental conditions (e.g., created by a coach) can support an individual's motivation (Fortier et al., 2012).

#### 2.5.5.1 Self-Determination Theory (SDT)

SDT (Deci & Ryan, 2000) is a motivational theory that has received significant research attention and support in predicting PA as well as in the development of PA and OYS interventions (Fenton et al., 2016; Fortier et al., 2012). The SDT has been successfully applied in order to understand the psychological processes likely to impact upon PA engagement across many contexts, including youth sport (Fenton et

al., 2014; Fortier et al., 2012; Lawler et al., 2020). SDT is uniquely placed among theories of human motivation to examine the differential effects of qualitatively different types of motivation that can underlie behaviour (Deci & Ryan, 2000). Originating from a humanistic perspective, hence fundamentally centered on the fulfilment of needs, self-actualisation, and the realisation of human potential, SDT is a comprehensive and evolving macro-theory of human personality and motivated behaviour (Deci & Ryan, 2000). Specifically, three psychological needs are hypothesised which are considered to be psychological nutrients for autonomous motivation and psychological well-being (Sebire et al, 2013). The needs are autonomy (i.e., to be choiceful and the origin of one's action), competence (i.e., to feel effective and confident in one's abilities and actions) and relatedness (i.e., to feel a sense of meaningful and mutual connectedness with others) (Deci & Ryan, 2000; Fortier et al., 2012).

According to this theory, the level of satisfaction of the basic psychological needs of competence, relatedness and autonomy decides whether a person is intrinsically motivated to engage in a certain behaviour or not (Demetriou & Bachner, 2019). As Player (2010) posits, one advance in the process of creating the right environment in OYS is supporting a child's intrinsic motivation. Among different theories used to understand PA behaviours, Deci & Ryan's (1985) SDT proposes that meeting participant's three psychological needs will help them remain intrinsically motivated in an activity (Deci & Ryan., 2008; Fortier et al., 2012; Guzman & Kingston, 2012). According to SDT theorists, the importance of the individual fulfilling the needs of competence, relatedness, and autonomy determines the degree to which an individual will be self-motivated (Baric et al., 2014). Furthermore, Deci and Ryan asserted that fulfilment of basic needs leads to greater intrinsic motivation or the pleasure and enjoyment experienced by engaging in an activity rather than the goal associated with the behaviour (Deci & Ryan, 2000).

Most cross-sectional studies have confirmed the positive influence of intrinsic motivation on PA (Sebire et al., 2016; Ntoumanis et al., 2020; Demetriou & Bachner, 2019). Results from Sebire et al. (2013) cross-sectional school-based intervention study, based on SDT to promote girls' PA, found the girl's (n=600) motivation that is based on enjoyment and inherent satisfaction of PA is associated with their objectively-assessed PA. Additionally, such motivation was positively associated with perceptions of psychological need satisfaction (Sebire et al, 2013). It is less clear though if positive

changes in intrinsic motivation – and more precisely in the three underlying basic psychological needs – can serve as significant mediators in interventions trying to promote PA of children and adolescents (Owen et al., 2014). However, a few longitudinal studies are pointing to the promotion of the basic psychological needs as a promising means for increasing PA in youth (Demetriou & Bachner, 2019; Laroche et al., 2019; Lawler, 2016). For example, in a study by McDavid et al. (2014) examining early adolescents over a period of three years, students' satisfaction of autonomy and relatedness as well as their intrinsic motivation in the context of PE significantly predicted the development of leisure-time PA. SDT is particularly appropriate for understanding pre-adolescent girls' PA participation viewpoints (Player, 2010) because the theoretical framework focusses on understanding the quality of an individual's motivation (e.g. how self-determined their reasons for PA participation are). Previous research suggests that more autonomous PA motivation is positively associated with child and adolescent PA participation (Fenton et al., 2016; Dishman et al., 2016), and positive psychological outcomes (such as quality of life and physical self-concept etc.) (Standage et al., 2012; White et al., 2018). This hypothesis is supported by empirical research among children (Demetriou & Bachner, 2019; Sebire et al., 2013) and adolescents (Owen et al., 2014). Within SDT, people's psychological needs can be supported or undermined by the motivational climate that an authority figure (e.g., teacher/coach) creates through their motivating or teaching style (Deci & Ryan, 2008; Su & Reeve, 2011). Need supportive styles are underpinned by the provision of autonomy support, structure and involvement which is reflected in how teachers' (or sports coaches) conduct their classes and interact with pupils/players (Keegan et al., 2010; Occhino et al., 2014).

SDT advocates finding ways in which young girls can develop a sense of choice and ownership over their own PA (autonomy), feel capable engaging in PA (competence), and are supported within a broader social context (relatedness) that will foster more self-determined forms of motivation (e.g. participating for fun or personally valued benefits), which are in turn positively associated with heightened PA levels (Sebire et al., 2016). SDT, therefore, suggests that PA interventions, such as the current G4G programme, which target enjoyable and valued activities, and foster perceptions of ownership, competence and belonging, are more likely to result in a sustained behaviour change (Demetriou & Bachner, 2019; Schneider et al., 2018; Sebire et al., 2013).

#### 2.5.5.2 Social Ecological Model (SEM)

The revised G4G intervention was also designed using elements of the Social Ecological Model (SEM) (Stokols, 1996), grounded in three key elements (organisational, interpersonal and individual). An ecological model used to examine levels of PA in youth may reasonably examine intrapersonal factors (for example, attitudes and opinions youth hold about being active), their own behavioural experiences (for example, the extent to which being physically active is part of their behavioural routine), their social environment (for example the types of role models and encouragement that they receive from others to be active) and factors in the physical environments of their neighbourhoods (such as how safe it is to play outside or the availability of parks and recreational areas (Hearst et al., 2012).



Figure 2. 3: Social-ecological model of children's participation in physical activity programmes - adapted from Sallis, Owen, & Fisher (2008).

The SEM provides an overarching framework for understanding the impediments and enablers to PA behaviour as it not only focuses on individual characteristics but also considers the social and physical environment context which can include family, friends, neighbourhood associates, formal and informal organisations, design of urban environment and facilities which promote or prevent PA (Fleury & Lee, 2006; Gilescorti & Donovan, 2002). As shown in Figure 2.3, the individual and his/her knowledge, skills, and attitudes are at the centre of the model. Layers of the model emanate from the individual to levels of the environment. Interpersonal factors including family, friends, and social networks, organisations and social institutions, and community relationships between organisations form the levels of the model. The outermost layer is society, encompassing public policy reflected in national, state, and local laws and regulations (Solmon, 2015). There is a reciprocal nature to the relationships between the layers, and to design environments that promote PA and other health behaviours, it is necessary to consider how multiple levels of the model interact (Solmon, 2015).

In order to structure relevant determinants, the conceptual framework for this research adopted Bronfenbrenner's ecological model of child development and wellbeing (Bronfenbrenner, 1994, 1977; Spence, 2003). This model proposes that a child's development is affected by multiple levels of influencers including the micro-system which includes direct influencers such as family, school and neighbourhood (Bronfenbrenner, 1994). Bronfenbrenner's model advocates the need to address factors at multiple levels to understand and change PA behaviours (Cadogan et al., 2014). Multi-level approaches derived from such ecological models have been recommended to examine PA determinants (Sallis et al., 2011). The SEM has been applied in a variety of settings to identify key elements for success and to guide efforts to promote PA including community-based PA interventions (Haggis et al., 2013), and PE (Elder et al., 2006; Langille & Rodgers, 2010; Martins et al., 2017; Belton et al., 2019). For example, the Sigue la Huella PA intervention (Murillo Pardo, 2014) on 12-15-year-old adolescents (n=368) was underpinned by the SEM, findings reported that MVPA increased in the experimental group ( $\beta = 7.02, 95\%$  confidence interval (CI) = 1.27 to 12.78, p = 0.017). A recent study by Zeng et al. (2019) on social-ecological correlates of FMS among preschool-aged children in Colorado found positive associations on a child, family, and environmental level. Child perceived cognitive competence, parent education, and the physical environment were positively associated with locomotor skills (p=0.02) and explained 5.5% of the variance at the environmental level (Zeng et al., 2019).

Ultimately, a social ecological perspective provides an innovative theoretical framework to guide the investigation of multiple factors influencing PA behaviour change (Stokols, 1996; Sallis et al., 2008). Given the fact that PA must take place in

specific physical settings that are likely to influence an individual's choice to be physically active, a social ecological perspective is particularly suited for studying PA (Zhang et al., 2012)

#### 2.5.5.3 Theory of Planned Behaviour (TPB)

The Theory of Planned Behaviour (TPB) (Ajzen, 1985), an extension of the Theory of Reasoned Action (TRA; Fishbein & Ajzen, 1975), is another prominent psychological model that has been proven to be effective, and versatile for examining the antecedents of PA behaviour among children and adolescents (Hagger et al., 2002; Cousson-Gélie et al., 2019; Darabi et al., 2017; Wang & Wang, 2015). According to TPB, behavioural intention is influenced by attitude, subjective norms, and perceived behavioural control (PBC) (Ajzen, 1985; Azjen, 1991). Moreover, human behaviour is guided by different subjective probabilities (e.g. Fishbein/Ajzen, 1975), that means beliefs about the consequences of the behaviour, beliefs about the normative expectations of other people and beliefs about the presence of factors which may facilitate or impede the performance of the behaviour (Sommer, 2011). Beliefs are based on a wide range of background factors. In their aggregates, behavioural beliefs produce attitude towards behaviour, normative beliefs result in subjective norms and control beliefs generate perceived behaviour control. The combination of all these elements leads to the formation of a behavioural intention (Ajzen 2002, p. 107). The behavioural intention could be described as "... instructions that people give to themselves to behave in a certain way" (Triandis 1980, p. 203). In other words, intention represents the motivation of an individual's conscious plan to exert effort to perform the behaviour. The intention could be understood as an immediate antecedent to behaviour (Ajzen 2002, p. 107). Figure 2.4 presents an overview of the interplay between the so-called background factors, beliefs, aggregates, intention and behaviour (Ajzen 2005, p. 135).

The relation between intention and behaviour could be described as follows (Sheeran 2002, p. 1): "*People do what they intend to do and do not do what they do not intend*". In the present G4G thesis, attitude represents the positive or negative evaluation of engagement in MVPA, whereas subjective norms reflect perceived social pressures to perform MVPA. PBC refers to resources and obstacles that facilitate or impede engagement in MVPA behaviour (Ajzen, 1985; Azjen, 1991). Wang & Wang (2015) postulate that children who have high motivation to engage in MVPA are more

likely to have high MVPA levels. Second, children who express strong feelings of control over their PA are likely to be engaged in more MVPA (Wang & Wang, 2015).



Figure 2. 4: Theory of Planned Behaviour. Source: Ajzen (2005) p.135

Consistent with the TPB, intentions are powerful predictors of behaviour in PA contexts (Hagger et al., 2002) and thus are used as indicators of future sports participation behaviour (Atkins et al., 2013; Chuan et al., 2014; Gardner et al., 2017). Gardner et al. (2017) explored whether enjoyment and behavioural intentions to continue in OYS predicted participation and dropout behaviour among 327 adolescent children (mean age = 13.01 y at baseline). At one-year follow-up, findings revealed that 247 children (75.5%) continued participating in their main sport and 26 individuals (8%) dropped out, with both enjoyment and intentions inversely associated with dropout. However, further research is needed to determine whether these findings generalize across a range of sports to support the use of behavioural intentions as a valid and reliable indicator within youth sport research (Sommer, 2011).

TPB has been applied to examine the PA of children and adolescents from numerous countries (Duncan et al., 2012; Gourlan et al., 2015; Cousson-Gélie et al., 2019; Kuen & Mok, 2013). Kuen and Mok (2013) case study on the application of TPB predicting the PA intention and behaviour of secondary school students (n=486; 11-18-year-olds) in Hong Kong, found the variables of TPB explained 53.1% of PA intention with significant factors of subjective norm and perceived behavioural control (PBC), increasing to 56.2%. When predicting PA behaviour, the TPB accounted for 26.6% of the variance (Kuen & Mok, 2013). Most recently, Lee et al.'s (2019) longitudinal study revealed that the TPB significantly predicted Taiwanese children's exercise intentions and behaviours (both p < .05) during a 6-month interval. A systematic review of 237 independent prospective tests found that the TPB accounted for 19.3% of the variability in health behaviour with intention being the strongest predictor (McEachan et al., 2011). It was also found that the TPB was considerably less predictive of behaviour when studies used a longitudinal rather than a 'shortitudinal' design, and when outcome measures were taken objectively rather than as a self-report.

The TPB is one of the most widely tested models of the factors influencing health-related behaviours (Glanz & Bishop, 2010) and has been shown to be particularly suited to the prediction of PA (McEachan et al., 2011). Findings support the use of behavioural intentions as indicators of sport participation/dropout behaviour and may aid the development of interventions aimed at preventing future dropout (Gardner et al., 2017). As guided by the aforementioned studies, it is encouraged that clinical practitioners, teacher and coaches working with children can help youth engage in regular exercise by enhancing their intentions and perceived behavioural control, and cultivating positive attitudes and subjective norms when planning exercise intervention programmes (Lee et al, 2019).

Understanding the psychological, social, and environmental factors that influence the behaviour of children and adolescents is a crucial component in designing any PA intervention (Eime et al., 2013; Van Der Horst et al., 2007). In this literature review, the author has predominantly examined the significance of PA, sports participation and PA correlates, including FMS amongst young people, as potentially successful contributors for community-based sports programmes (Camacho-Minano, 2011; Pearson et al., 2015; Robinson et al., 2015; Van Sluijs et al., 2011; Woods et al., 2018). Due to the pronounced gender differences in PA during adolescent years and the great disparity for girls between the reality and the targets of the recommended PA level, it is appropriate that pre-adolescent girls are identified as a high priority group for PA promotion (Camino-Macho, 2011).

In this final stage of the literature review, specifically, section 2.5, the evidence highlighted the call for multi-component community sports-based interventions to increase youth PA (Pearson et al., 2015; Voskuil et al., 2017; Jones et al., 2020) among young girls as an important public health challenge (Sallis et al., 2019; Woods et al., 2018). Recent findings from many comprehensive review articles have found that the community sports-based setting provides an ideal opportunity to increase PA among youth; sufficient evidence is now available to recommend global implementation of multi-component community sports-based programmes (Jones et al., 2020; Salmon et al., 2007; Voskuil et al., 2017;). Community-based programmes may require various levels of activities, such as tailoring to specific audiences (namely at-risk of drop-out adolescent girls), and Eime et al. (2015) have proposed young children can participate in modified sports programmes as a pathway to club sport competition. Specifically, a recent systematic review and meta-analysis of PA community sports-based interventions among girls (Pearson et al., 2015) found that greater intervention effectiveness appears to result from multi-component interventions, theory-based, (Sluijs et al., 2007) focused on girls only (Camacho-Minano et al., 2011), target both PA and sedentary behaviour (Pearson et al., 2015), provide activities that teenage girls enjoy (e.g., fun, sociable and not competitive sport), include a FMS development and a non-formal coach education element (Trudel et al., 2010).

Despite this, there is a lack of effective intervention strategies to promote PA in female youth (Anokye et al., 2018), particularly within a sporting context (Whitley et al., 2019), with no known systematic review investigating the effectiveness of PA interventions that aim to promote participation in team sport specifically (Allison et al., 2017). From this review of literature, it is evident that the scientific knowledge of what works is still evolving, and this author agrees it is critical that:

'The conduct and publication of well-designed evaluations of well-defined interventions is crucial to advance the field of youth physical activity promotion and make us better understand which intervention strategies may or may not work, why, and for whom'. (Van Sluijs & Kriemler, 2016, p.1).

## CHAPTER 3

### METHODOLOGY



#### **3.1 OVERVIEW OF CHAPTER**

This chapter presents the description of the research process, providing information concerning the methodological approach adopted as well as a justification of the various methods used in the studies conducted as part of this thesis. The chapter also describes the various stages of the research, which include ethical considerations, the selection of participants, the data collection process, and the process of data analysis. The chapter ends with a discussion of validity and reliability in both quantitative and qualitative research and discusses how these two requirements were met in the current study along with the role of the researcher in qualitative research concerning reflexivity.

As part of this research investigating PA, FMS, and psychological correlates among 8 to 12-year-old girls, four distinct, but connected, studies were undertaken. These included;

- A cross-sectional baseline study (quantitative only) in a school-based setting (Chapter 4), specifically focusing on actual and perceived FMS and the relationship with PA among 8 to 12-year-old pre-adolescent girls;
- A second cross-sectional baseline study investigating PA levels, FMS proficiency and psychological correlates of 8 to 12-year-old pre-adolescent girls in a school-based setting (Chapter 5);
- A mixed-methods intervention protocol study conducted in a communitysports based setting (Chapter 6);
- A non-randomised controlled three-arm exploratory before-and-after study in three community sports-based settings (Chapter 7). Further details of the individual studies can be seen in Table 3.1.

	Study 1	Study 2	Study 3	Study 4
Chapter	4	5	6	7
Study Type	Cross-sectional	Cross-sectional	Quasi- experimental intervention protocol study	Non-randomised controlled, three- armed exploratory, before-and-after
Study Design	Quantitative	Mixed methods	Mixed methods	Mixed methods
Sample Size	N= 160	N= 331	N=117	N=120
Data Collection Dates	March to May 2016	March to May 2016	March to June 2017	March to June 2018
Focus Groups	N/A	6 x FG with child participants (n=37)	N/A	1 x FG G4G participants (group 1) n = 6
Primary Outcome Measures	PA using PAQ-C PSC using PSC scale PPSC using a modified version of the Athletic Competence sub- scale	PA using PAQ-C and PACE FMS using TGMD, TGMD-2, and GSGA 9 x Psychological correlates (explained in 3.3)	PA using PAQ-C and PACE FMS using TGMD, TGMD-2, and GSGA Psychological correlates (see Table 3.2)	
Other outcomes	N/A	BMI -Mass using Seca weighing scales -Height using a portable stadiometer	B -Mass using Seca weighir -Height using a po	MI 761 dual-platform ng scales ortable stadiometer

### Table 3. 1: Characteristics of studies conducted as part of the thesis

PA: Physical Activity; PAQ-C: Physical Activity Questionnaire for Older Children, PACE: Physician-based Assessment and Counseling for Exercise, FMS: Fundamental Movement Skills; TGMD: Test of Gross Motor Development, GSGA: Get Skilled Get Active, BMI: Body Mass Index

#### **3.2 Research philosophy and paradigm**

The term research philosophy refers to a system of beliefs and assumptions about the development of knowledge (Saunders et al., 2009). It is important when undertaking research for the researcher to make clear their philosophical position regarding their understanding of what it means to be in the world (ontology) and how they know what they know (epistemology). Prior to choosing the research methodology, the lead researcher underwent the process of exploring and understanding their research philosophy honing the skill of reflexivity, that is, to question your thinking and actions, and learn to examine your own beliefs with the same scrutiny as you would apply to the beliefs of others (Gouldner & House, 1971).

As a researcher, it is required to develop your reflexivity, to become aware of and actively shape the relationship between your philosophical position and how you undertake your research (Alvesson & Sköldberg, 2000). Subsequently, the lead researcher utilised a reflexive tool called Heightening your Awareness of your Research Philosophy (HARP), designed by Bristow & Saunders (2014) to help with the exploration of research philosophy. HARP consists of six sections, each comprising of five statements (a total of 30 statements). Each section considers one aspect within the philosophical beliefs (ontology, epistemology, the purpose of research, the meaningfulness of data and structure), with each statement epitomising a particular research philosophy's position. The lead researcher indicated her agreement or disagreement with each statement, and thus discovered similarities and differences with different aspects of each research philosophy. More specifically, the HARP tool was used within this thesis as a starting point and enabled the lead researcher to ask oneself more refined questions about how they see research and help think about their values and beliefs concerning research. Ultimately, this helped the lead researcher make their values and assumptions more explicit while considering the potential fit between their own beliefs and those of the four major philosophies used in research (the positivist, the interpretivist, the pragmatist, and the realistic research philosophies).

As guided by the HARP research reflexive tool, the current thesis uses a pragmatic paradigm approach following both positivism and interpretivism to seek the answers to the research questions. One philosophical position is 'positivism', which implies that the goal of the research is to produce objective knowledge

through direct observation (Madill et al., 2000). Interpretivist research, on the other hand, implies that the goal of the research is to create new, richer understandings and interpretations of social worlds and contexts (Saunders, 2009). As a research paradigm, pragmatism is based on the proposition that researchers should use the philosophical and/or methodological approach that works best for the particular research problem that is being investigated (Teddlie & Tashakkori, 2009). It is often associated with mixed-methods or multiple-methods (Creswell, 2013), where the focus is on the consequences of research and the research questions rather than on the methods.

Pragmatism as a research paradigm accepts that there can be single or multiple realities that are open to empirical inquiry (Creswell & Clark, 2011). Pragmatist scholars have offered their opinion that there is an objective reality that exists apart from human experience. However, this reality is grounded in the environment and can only be encountered through human experience (Goles & Hirschheim, 2000; Morgan, 2014; Teddlie & Tashakkori, 2009).

#### 3.2.1 Justification of the pragmatic approach

A major underpinning of pragmatist epistemology is that knowledge is always based on experience, and reality is based on beliefs and habits that are socially constructed (Yefimov, 2004). One's perceptions of the world are influenced by our social experiences. Each person's knowledge is unique as it is created by her/his unique experiences. Nevertheless, much of this knowledge is socially shared, as it is created from socially shared experiences (Kaushik & Walsh, 2019). Therefore, all knowledge is social knowledge (Morgan, 2014). The nature of this thesis explores social knowledge within a community sports-based setting among young girls and coaches within a broader team dynamic system. The interpretivism research philosophy aligns with the design, development, implementation, and evaluation of the G4G intervention programme, with the creation of new, richer understandings and interpretations of social worlds and contexts (Saunders et al., 2009).

#### **3.3 RESEARCH DESIGN**

#### 3.3.1 Mixed methods

The studies described in chapters 5 to 7 of this thesis adopt a mixed-methods approach. A mixed methodology has been defined by Tashakkori and Creswell (2007, p.4) as, 'research in which the investigator collects and analyses data, integrates findings and draws inferences using both qualitative and quantitative approaches or methods in a single study or programme of inquiry'. As has been previously established in Chapter 2 of this thesis, community sports-based PA interventions for 8 to 12-year-old girls to date are limited and have only yielded small to moderate changes in PA behaviours. Qualitative data was collected to understand human behaviour, emotions, attitudes, and experiences. This was combined with quantitative data to assess other outcome variables including PA levels, FMS proficiency, and several psychological variables. Collecting data from different sources avoids problems associated with evidentiary inadequacy (Erickson, 1986) and allows triangulation between data sources to support findings and assertions (Glesne, 1999).

#### **3.3.2 Justification of a mixed-methods approach**

In Chapter 5, the combined quantitative and qualitative data aided in seeking a current representative baseline picture of pre-adolescent self-reported PA levels, FMS proficiency, and investigated correlates of PA among the target population. Additionally, qualitative data aided in the exploration of PA enablers and barriers, which was necessary to ensure appropriate intervention functions and pedagogical coaching techniques were being selected for the target group when designing the G4G intervention. In Chapter 7 qualitative data was collected post-intervention as a method to gain further insight into participant perceptions of the programme and provide vital opinions that otherwise may not have been detected through quantitative data alone.

Van Maanen et al. (2007) recommends the use of qualitative design when developing an understanding of the lived experiences of others. Because there are multiple perspectives describing experiences and perceptions of the G4G programme

implementation, this approach to the study was most appropriate. This combination of multiple perspectives is a strategy to add rigor, breadth, richness, and depth to inquiry (Flick, 2007). Choy (2014) advocates that a complementary approach between qualitative and quantitative approaches for a research topic may provide better results than using only one isolated methodology. Engaging the intended user groups (i.e., children) within the design, eliciting their perspectives on PA and content is central to a phased approach to complex intervention design, and is deemed essential to their success (Davison et al., 2013). Whilst quantitative approaches report PA prevalence and identify associations (Cooper et al., 2015; Robbins et al., 2019), they provide limited contextual understanding or explanation as to why some children are more active than others and offer little insight into intervention design (Noonan et al., 2016). So whilst we get the broad picture of the impact of a sports-based programme for 8 to 12-year-old girls through the quantitative data, the qualitative aspect provides us with more detailed information on a subset of the group who have successfully changed their PA behaviour and how we can engage other vulnerable girls at-risk of drop out with PA and sport participation.

#### 3.3.3 Cross-sectional design

Chapter 4 of this thesis is cross-sectional in design, investigating the relationship between actual FMS proficiency, perceived self-confidence, and PA of 8 to 12-yearold girls in a school setting. Chapter 5 (same dataset as Chapter 4) also includes a cross-sectional design, investigating self-reported PA levels, FMS proficiency levels, and correlates of pre-adolescent girls in a school setting. For both chapters, a crosssectional design was used as this method can highlight associations that may exist between certain variables, and therefore can provide justification for the development of future research and interventions (Levin, 2006). Conducting a crosssectional study is useful as correlations between variables can be identified, however, the design is limited as they do not permit the distinction between cause and effect (Mann, 2003). Despite this limitation, a cross-sectional design is useful in estimating the prevalence of behaviour within a certain population (Sedgwick, 2014), and therefore was chosen as the method as a baseline representation to assess the primary PA, FMS and psychological PA correlates of 8 to 12-year-old girls.

#### **3.3.4 Intervention protocol study**

In Chapter 6, the G4G intervention protocol is reported; a 10 week specifically designed multi-component PA and FMS community sports-based coaching intervention, informed by contextually relevant data (Chapters 4 and 5), targeting 8- to 12- year olds, from a community sports club setting.. The lead researcher was guided by the transparent reporting of evaluations with nonrandomised designs (TREND) statement for reporting intervention effects (Reeves & Gaus, 2004).

# 3.3.5 Quasi-experimental non-randomised controlled trial, three-armed exploratory pre-post design

In Chapter 7, the G4G intervention using a quasi-experimental, non-randomised controlled trial (NCRT) design, assessed the effectiveness of a 10 week specifically tailored multi-component PA and FMS community sports-based coaching intervention, informed by contextually relevant data (Chapter 4 and 5) on improving i) overall PA levels, ii) FMS proficiency, and iii) psychological wellbeing when compared with a second intervention group – the existing G4G national programme (intervention group 2) and a control group 3 receiving no programme. Furthermore, the intervention group 1, intervention group 2, and control group 3 were identified using convenience sampling by the lead researcher, although it was requested that they did not select groups based upon judgements of who might benefit most from being involved in the intervention. Using a quasi-experimental design, the programme is viewed as an 'intervention' in which a treatment – comprising the elements of the programme/policy being evaluated – is tested for how well it achieves its objectives, as measured by a prespecified set of indicators (White & Sabarwal, 2014). Quasi-experimental studies are less expensive and less time consuming than Randomised Controlled Trials (RCTs) (Schweizer et al., 2016), and meet some requirements for causality including temporality, the strength of association and the dose-response (Shadish, 2011). The study followed the

transparent reporting of evaluations with non-randomised designs (TREND) statement for reporting intervention effects (Reeves & Gaus, 2004).

#### **3.4 THEORETICAL FRAMEWORK**

The G4G intervention is underpinned by self-determination theory (SDT) (Deci & Ryan, 2008), and elements of the social-ecological model (SEM) (Stokols et al., 1996), and the Theory of Planned Behaviour (TPB) (Ajzen & Madden, 1986), because its theoretical foundations are concerned with how the psychological (behavioural) and socio-environmental conditions (e.g., created by a coach) can support an individual's motivation (Fortier et al., 2012). A detailed description of the theoretical frameworks is presented in the final section of the Literature Review and Chapter 6.

The qualitative aspect within this thesis utilises a theoretical research approach that is phenomenological in nature, whereby the lead researcher describes the lived experiences of individuals about a phenomenon as described by participants (Creswell, 2014). For example, increasing our knowledge about the experiences of 8 to 12-year-old girls, parents, and coaches' psychological correlates of PA and sport and the influence of these experiences on their sport participation and coaching practice. Thematic analysis, a method for identifying, analysing, and reporting patterns (themes) within data (Braun & Clarke, 2006) underpinned the qualitative framework within this thesis. Thematic analysis was carried out in which data were analysed inductively using a constant comparison method (Glaser & Strauss, 1967). Thematic analysis is widely used in the field of PA research (Koh et al., 2014; S. Whitehead & Biddle, 2008) providing a purely qualitative, detailed, and nuanced account of data (Braun & Clarke, 2006). A more detailed description of the thematic analysis of this thesis is presented in the Data Analysis section (3.10).

#### **3.5 ETHICAL CONSIDERATIONS**

In accordance with the University's regulations for non-clinical research involving human participants, a research proposal form was completed and submitted to the UCC Social Research Ethics Committee (SREC) on 1<sup>st</sup> December 2015 for the

current research, 'The design, development, implementation, and evaluation of the Gaelic4Girls intervention'. This document described the purpose, design, and methodology of the research project. It also gave details of the participants' involvement in the project and outlined any ethical considerations. Due to the nature of the project, the main ethical issue to be considered was confidentiality. The SREC initially reviewed the submitted proposal, and a report was returned in early January detailing suggested changes and revisions. These revisions were predominantly referring to clarity of terms and procedures. A revised submission was then submitted to SREC on 21<sup>st</sup> January and approval was granted on the 16th of March (Reference no. log2015-005) (see Appendix A).

#### 3.5.1 Informed consent

Within this thesis, informed consent was required for all students; with students under the age of sixteen requiring parental/guardian consent. When designing the Informed Consent form (Appendix), the researcher consulted the Federal Policy for the Protection of Human Subjects, which lists the policy requirements for research involving human subjects (OHRP, 2001) including;

- A statement that the study involves research, an explanation of the purposes of the research and the expected duration of the subject's participation, a description of the procedures to be followed, and identification of any experimental procedures.
- A statement describing the extent, if any, to which confidentiality of record identifying the subject will be maintained.
- An explanation of whom to contact for answers to pertinent questions about the research and research subjects' rights, and whom to contact in the event of a research-related injury to the subject.
- A description of any reasonably foreseeable risks or discomfort to the subject.
- A statement that participation is voluntary, refusal to participate will involve no penalty or loss of benefits to which the subject is otherwise entitled, and the subjects may discontinue participation at any time without penalty or loss

of benefits to which the subject is otherwise entitled (Drowatzky, 1996) (Office for Human Research Protections, 2001).

#### 3.5.2 Inclusion criteria:

- Students (female-only) between the ages of 8-12 who are in selected class groups in the selected schools at the time of testing (Chapter 4 and 5).
   Participants who expressed an intention to attend the G4G programme for its full duration (Chapter 6 and 7).
- Informed parental consent and child assent were the requirements for eligible participation in all studies. Students who give their consent (or if under 16 are given parental consent) to participate in the research study.
- Any student unable to partake in the FMS testing (due to illness/injury) may partake in the other parts of the study.
- For coaches to be eligible to partake in the programme, recognised LGF coaching credentials from the National Ladies Gaelic Football Association (LGFA), and previous experience of coaching girls within the existing community sports-based setting were required.

#### 3.5.3 Exclusion criteria:

- Anyone under the age of 16 without parental consent.
- For the FMS measurement, volunteers were excluded from the research if they:
- Carried an injury or illness that affected their involvement in FMS testing.
- Had a history of clinical conditions that may preclude them from exercise.

#### 3.5.4 Confidentiality and anonymity

Participants were reassured that all information provided would be treated in the strictest of confidence. The anonymity of the participants was respected. All participants were given an ID number, which corresponded to the data collected from that specific point. The data gathered from the study was entered into, and will be stored in, the University College Cork (UCC) Network File Storage (NAS) for the

duration of 7 years following the completion of the study, in line with University regulations for examinations. The data will then be destroyed by the principal investigator. All personal details will be stored in a secure place, which can only be accessed by the named researchers. The ID number on the questionnaire was only used by the researcher to identify a subject, meanwhile maintaining confidentiality from others.

#### 3.5.5 Other ethical considerations

To ensure a duty of care and safeguarding of participants, all members of the research team were trained in handling and responding to sensitive issues that may arise during the data collection process, particularly during interviews/focus groups. All the assistant researchers are pre-service physical education teachers (Sports Studies Physical Education Course at University College Cork) and are Garda-Vetted and trained in field-based research. There was no coercion or undue influence of research participants to take part in the research in the recruitment of participants. All participants were free to withdraw from the research at any stage without penalty and without providing a reason. Participants could also require that their data be withdrawn from the study.

#### **3.6 PARTICIPANTS AND SETTING**

In this research, girls approximately 8-12 years old were chosen as the target group. This age group was chosen as girls within this category are attending primary school, giving the researcher easy access to many eligible participants. Given that a systematic review and meta-analysis by Farooq et al. (2018) found that on average there is a significant annual decline in MVPA across all age groups from around age 6 years onwards in girls (-10.2%) and from age 9 among boys (-7.8%), roughly equivalent to a 6 minute per day decline in MVPA per year justifies the need to intervene with this population and promote healthy lifestyle behaviours. Club LGFA coaches (male and female) within the clubs were also chosen as a target group to coach the girls in the G4G programme. The coaches had recognised coaching credentials from the national LGFA, and previous experience of coaching girls within the existing community sports-based setting was required.

#### 3.6.1 Sampling

A convenience sample (based on the researcher's proximity to the schools and clubs) of cross-sectional (n=331 and n=160), quasi-experimental (n=117), and NRCT (n=118) data were collected from Irish female youth as part of all four study protocols within this thesis. In Chapters 4 and 5, eligible-primary school pupils, enrolled between second and sixth class (8–12 years) from three, all-female schools (one rural, and two rural/urban) were invited to participate. A representative socioeconomic status demographic was selected, with the inclusion of one school identified as DEIS (Delivering Equality of opportunity In Schools). For the studies described in Chapter 4 and Chapter 5, a school setting was used as the location for cross-sectional baseline data collection. For both studies' convenience sampling was used to recruit from three female-only primary schools. Research has suggested that single-sex groupings are more beneficial to females when it comes to PA (Bender & Litfin, 2015). The G4G programme specifically targets females only and for this reason, female-only educational primary schools were chosen to seek a current representative picture of current PA levels, FMS proficiency, and psychological perceptions of PA and sport participation without the influence of boys in a coeducational setting.

In Chapter 6 and Chapter 7, the G4G intervention protocol study and G4G NRCT intervention were both conducted in a community sports setting in close proximity to the local designated LGF club and the research team. As outlined in Chapter 2, there is currently a paucity of research focusing on multi-component sport interventions for girls aged 8 to 12-year-old in community settings, with the large majority being conducted in school (i.e. before, after or during school time), therefore, the scientific reporting of a multi-component community sports-based intervention is investigated within this thesis. In Chapter 6, convenience sampling was used to recruit from the five girls' only primary schools (rural and rural/urban) in the County Cork region. All schools were in close proximity to the local designated Ladies Gaelic Football (LGF) club, and the research team. For chapter 7, convenience sampling was used in the NRCT to recruit participants from the eighteen girls' only primary schools (rural and rural/urban), from the three surrounding local LGF clubs in the Cork and Kerry regions (Munster Province,

Ireland). Similar to Chapter 6, all schools were in close proximity to the local designated LGF clubs and the research team.

To be considered eligible for all four studies, participants (girls only) were formally enrolled in between the years of second to sixth class of primary school (aged 8- to 12- years old), and had expressed an interest (via assent and parental consent) to take part in the data collection procedures, and had expressed intention to attend the programme for its full duration (Chapter 6 and 7). As described in the Ethical Consideration section, full written assent and parental/guardian consent prior to participation were required for all four studies. If a participant or their parent/guardian did not consent, they were not permitted to participate in the data collection.

#### **3.7 DATA COLLECTION PROCEDURE**

#### 3.7.1 Recruitment

Prior to the commencement of the baseline studies (Chapter 4 and 5), the leading researcher emailed the school Principal from all three selected primary schools, calling for expressions of interest for the lead researcher to visit the school, and specifically to inform the girls about the G4G programme. Subsequent to the granted approval from school Principals, the lead researcher visited the school to meet with the Principal and all classes from second to sixth class, where a full outline of the G4G intervention and the associated data collection measurements were provided. Information leaflets (plain language statements) and informed consent forms (Appendix B) were distributed to students who expressed an interest in taking part in the research data collection process. One class group was randomly selected from each school representing each year group - the rationale for the selection of one class from a year group, as opposed to sampling across the full year group was to minimise interruptions to the schools and the teachers.

The same recruitment process was used for the G4G Intervention studies (Chapter 6 and 7). Clubs were initially contacted by telephone and a meeting was arranged with the club Chairperson/Secretary to outline the project. After the meetings, selected clubs that agreed to take part were provided with an information sheet. For these studies, the lead researcher, along with the two G4G coordinators (head club coaches from the selected LGF intervention club) visited the primary schools, subsequent to the granted approval from school Principals. Information leaflets and informed consent forms (Appendix B) were handed out to students who expressed an interest in taking part in the research data collection process. For coaches to be eligible to partake in the programme, recognised coaching credentials from the national LGFA, and previous experience of coaching girls within the existing community sports-based setting was required.

#### 3.7.2 Field training

Prior to data collection, field staff underwent a rigorous and robust field researcher training workshop in the measurement protocol associated with FMS, self-report questionnaires, FG discussions, and body composition. To enhance the quality of the data across all collection sites, the research assistants (all pre-service physical education teachers) were formally trained by the lead researcher in the standardised measurement procedures and protocols. In total, from 2016-2018, there were 17 field staff, including the two lead investigators, involved in the study's data collection process. Each research assistant (N=15) attended 2-hour robust field training workshop two weeks prior to data collection and was given detailed manuals, checklists, and scripts to read when informing the participants about the measures. To ensure consistency and accuracy, this protocol manual, which included specific instructions for conducting all assessments, was also developed and used by research assistants to standardise procedures and for quality assurance (See Appendix E). In terms of the research rigour associated with school and club-based measurements, the leading investigators (N=2) for the G4G study are qualified post-primary specialist physical education (PE) teachers, as recognised by the Teaching Council of Ireland.

#### 3.7.3 Procedure

All baseline data (Chapters 4 and 5) were collected on participants within their class groups (max n = 30) during specific school visits over an 8-week period from March

to May 2016. The intervention protocol study (Chapter 6) took place in both a community sports club setting and school setting (pre-test in March 2017 and post-test in May 2017). Objective measurements, such as FMS, and body composition were carried out during a 3-hour timetabled block in the community sports club setting, while subjective self-report measurements and FG interviews were taken during a specific school visit, in a supervised classroom, and computer lab, respectively. Due to the numerous schools (n=18) targeted in Chapter 7, all objective and subjective measurements took place in the community sports club settings for the three intervention groups (pre-test in March 2018 and post-test in May 2018).

During FMS data collection, one trained field staff member provided approximately every eight participants with an accurate demonstration of the FMS to be performed. Procedures outlined in the respective examiners' manuals were closely adhered to. Participants performed the skill on three occasions, including one familiarisation practice and two performance trials, as reported in previous Irish adolescent movement skill data collection (O' Brien et al., 2016a, 2016b), and were recorded using a digital video camera (3 × Canon type Legria FS21 cameras; Canon Inc., Tokyo, Japan) to allow for greater measurement scrutiny and accuracy of measurement precision (Okely et al., 2004) during analysis. To ensure consistency and accuracy in the data collection procedure the same researchers were allocated to the same measure/skill in each school visit.

- Researcher A, station 1 (run and skip)
- Researcher B, station 2 (kick and catch)
- Researcher C, station 3 (vertical jump, bounce, and balance)
- Researcher D, station 4 (weight and height measurements)
- Researcher E, researcher D, and lead researcher, focus group interviews
- Researcher A, B, C, D and lead researcher, questionnaire administration

Self-report data were collected on participants (n = 275) in their class groups, with a ratio of 1 researcher to 10 students, for questionnaire completion. The study was briefly explained, and instructions were provided on how to complete the questionnaire. All questions were completed using the online tool SurveyMonkey (San Mateo, CA), with an ID number assigned. For the final study (Chapter 7), all questionnaires were administered and completed in hard copy format, as it was not

feasible timewise for the research team to visit all 18 included schools from the three clubs. All participant questionnaire data was inputted into the online SurveyMonkey tool manually by the research team and was double-checked by the lead researcher prior to data analysis.

Following the completion of the self-report, anthropometric, and FMS data collection, a sub-sample of participants were randomly selected to participate in FGs for Chapter 7. All FGs were conducted in a school classroom and/or community sports club, under supervision, and lasted 45–50 min each, on average. Prior to FG commencement, all child participants and their parents/guardians received and signed a consent form and a plain language statement providing details of the research. Girls from second to sixth class were invited to participate in the FG study by way of a letter, which explained the purpose of the study. Interested girls were given a letter of introduction and a consent form for their parents. The girls were also asked to sign a consent form on the day of the FG allowing for the sessions to be audiotaped.

#### **3.8 Measures/Instruments**

In section 2.2.2 in Chapter 2, the worldwide recommended PA guidelines for health among children and youth were outlined. In order to monitor trends in PA, examine associations between PA and health outcomes, and evaluate the effectiveness of PA enhancing interventions, valid, reliable, responsive, and feasible measures of PA are needed (Benitez-Porres et al., 2016; Hidding et al., 2018; Jekauc et al., 2013). PA can be assessed in two main ways: by self-report (or parental proxy reports for children aged <10 years) or objectively through using equipment that measures movement and/or physiological responses to physical exertion (Mindell et al., 2014). Ideally, as Trost (2007) posits, a PA measurement tool should provide valid and reliable assessments of all four dimensions of activity behaviour (frequency, duration, intensity and type) in all of the PA domains (leisure-time PA, occupational PA, transportation activity and household-tasks). Moreover, the selection of a method to determine PA must be based on careful consideration of the advantages and disadvantages of the instrument, as well as indications for application and evidence to

support that the instrument is reliable and valid (Haskell et al., 2009; Treuth et al., 2005).

#### **3.8.1 Self-reported PA**

Subjective measurement tools for PA consist of self-report questionnaires, selfadministered recalls, interviewer-administered recalls, diaries, and proxy reports completed by parents and teachers (Innerd et al., 2015), and are preferred in large epidemiological studies (Bervoets et al., 2014). The Physical Activity Questionnaire for Older Children (PAQ-C), developed in Canada to assess the moderate-vigorous physical activity (MVPA) levels of children of an average age of 8–14 in grades 4–8, is a 7-day recall instrument based on self-reporting (Kowalski et al., 2004). A review by Biddle and colleagues (2011) identified the PAQ-C as one of the most promising self-report tools available in the field (Biddle et al., 2011). The PAQ-C has been identified as one of the very few self-report instruments that has acceptable validity, reliability and practicality for use with children and adolescents (Biddle et al., 2011; Benitez-Porres et al., 2016; Chinapaw et al., 2010).

Self-report measures of PA for 8 to 12-year-old girls was measured using the Physical Activity Questionnaire for Older Children (PAQ-C) in all four studies in this thesis (Kowalski et al., 2004).. It is a 7- day recall questionnaire that provides a summary score of PA derived from 8 items, each scored on a 5-point Likert scale (Kowalski et al. 2004). The PAQ-C for all four studies included 15 physical activities, 10 leisure/free-time activities, activities in school (physical education), transport activities (walking to and from school), and other activities. The participants were told to recall what activities they had engaged in the previous 7 days and how many times and the number of minutes they participated in each of these activities.

Habitual PA was also assessed using two questions from the Physician-based Assessment and Counseling for Exercise (PACE) questionnaire: How many days in the last week (PACE 1) and a usual week (PACE 2) does the subject do at least 60 min of physical activity? The PACE questionnaire presents test-retest reliability assessed by the intraclass correlation coefficient (ICC) of 0.77 (López Sánchez et al., 2016), and, due to its simplicity and ease of understanding, the PACE questionnaire was suitable for the 8–12-year-old cohort (Appendix C). Asking about the habitual

activity (Corder et al., 2008) risks responses reporting 'most active weeks' instead of 'usual' activity or activity in the specified time period, most commonly for children the previous 7 days. For the baseline studies (Chapter 4 and 5), the compound result was obtained from both questions ((PACE 1 + PACE 2)/2) and students were categorised as low active (meeting guidelines on 0, 1, 2, or 3 days a week), moderately active (meeting on 4 or 5 days), or highly active (meeting on 6 or 7 days) (Prochaska et al., 2001).

Studies that have assessed the validity and reliability of the PAQ-C have been varied in study designs and populations, but have overall been convincing (Aggio et al., 2016; Gobbi et al., 2016; Saint-Maurice et al., 2014). For example, Erdim et al. (2019) recent Turkish study among 9-14-year-old children (N = 784) provided evidence for acceptable reliability and validity of the PAQ-C, with results reporting the Cronbach's alpha was 0.77 and intra-class correlation coefficient (ICC) was 0.91. Additionally, Wang et al. (2016) validation study of the PAQ-C among 8-13-year-old Chinese children (n = 742) reported the Cronbach alpha coefficient ( $\alpha$ =0.79), composite reliability value ( $\rho$ =0.81), and the ICC ( $\alpha$ =0.82) indicating the satisfactory reliability of the PAQ-C score. This study also found to have a moderate correlation with accelerometry (r=0.33, p<0.05), suggesting that this instrument is valid at ranking children's MVPA.

Nevertheless, Hidding et al., (2018), most recent systematic review of PA questionnaires for youth, found no questionnaires were identified with conclusive evidence for both acceptable validity and reliability, partly due to the low methodological quality of the 162 studies included. Authors of this systematic review (Hidding et al., 2018) further recommend high-quality studies on measurement properties of PA questionnaires as an urgent priority. Despite the wellestablished psychometric properties and desirable measurement characteristics of the PAQ-C, compared to other self-report measures for youth (Chinapaw et al., 2010), more emphasis on the content validity of PA questionnaires is needed to confirm that questionnaires measure what they intend to measure to heighten the strength of PA accuracy.

Furthermore, structured questionnaires can provide useful insight into PA types and domains, which may be particularly relevant to assess behavioural changes

over time and/or in response to PA interventions (Voss et al., 2013). Self-report tools are commonly used in the research field of PA due to their low cost and convenience in large-scale population surveys (Helmerhorst et al., 2012). However, questionnaires have their limitations as well, such as the potential for social desirability and recall bias, as are memory errors; in children, however, a predominant concern is that their developing cognitive abilities may limit their ability to think abstractly and perform detailed recall, in particular when asked to report time (Welk et al., 2000). Asking about the habitual activity (Corder et al., 2008) risks responses reporting 'most active weeks' instead of 'usual' activity or activity in the specified time period, most commonly for children the previous 7 days (Craig et al., 2009). Shorter time periods give greater recall accuracy but determine irregular activities less accurately. Additional concerns exist over the accuracy of questionnaire data from children under 10 years due to their cognitive underdevelopment (Sallis, 1991) and may translate to poor validity coefficients (Helmerhorst et al., 2012), where a tendency exists for questionnaires to over-report PA levels compared with directly measured PA (Adamo et al., 2009). Nonetheless, PAQ's provide an instrument that can independently collect information in addition to be a supplementary tool to further enrich the information gathered from objective measurement (Kohl et al., 2000; Sallis, 1991). Thus, for measuring PA a combination of the more objective measures such as accelerometers and self-report questionnaires seems most promising.

#### 3.8.2 Objective measures of PA

Objective methods include measures that directly assess one or more dimensions of PA (e.g. Frequency, intensity, time, type), and can capture a variety of metrics such as the number of steps, minutes of activity, the intensity of activity, and bouts of activity (Strath et al., 2013). Although it has been argued that there is no "gold standard" for objective PA measurement, (Kelly et al., 2016; Ridgers & Fairclough, 2011) commonly used tools include wearable monitors (e.g. accelerometers, pedometers, and heart rate monitors) as well as indirect calorimetry (doubly labelled water) and direct observation (Sylvia et al., 2014). PA is a multifaceted and complex behaviour, and research has shown that these objective measures are more precise compared to subjective measures (Freedson et al., 2012), better capture the intricacies of PA dimensions, (Kelly et al., 2016; Prince et al., 2008), and provide

more continuous evaluation of free-living activity (Yang & Hsu, 2010). Of the aforementioned objective techniques for PA measurement, accelerometry will be discussed only for the purpose of this literature review.

Accelerometers are becoming the most widely used measure of PA in public health research (Hildebrand et al., 2016; Trost, 2007; Vale et al., 2015) as they provide an objective assessment of energy expenditure and time spent in different intensities of PA (Crouter et al., 2018). Over the past decade, there has also been increasing use of accelerometry to estimate PA in children (Crouter et al., 2018; Rowlands et al., 2014). Accelerometers are considered an effective PA measurement instrument for children and adolescents because of their small size, robust design and relatively modest cost (Fairclough et al., 2019; Pate et al., 2006; Yildirim et al., 2011), and are considered more reliable and valid in children than alternative methods such as self-report (Rowlands, 2007). Despite this, they are relatively expensive to use, compared to other methods of PA assessment, require more complex data handling and processing techniques to estimate PA and their accuracy in tasks requiring greater use of the upper body, cycling, or non-linear movement is not fully established (Rowlands, 2007). Moreover, acceleration is only an indicator of PA, and accelerometers register more movement (counts) during some activities (e.g. walking) than others (e.g. cycling) (Robertson et al., 2011; Smith et al., 2018).

Accelerometers provide quantitative information relating to the vertical accelerations of the trunk and body segments at a user-specified time (Welk, 2002; Trost et al., 2005), and measure complex movements of children's free play MVPA, including step and non-step movements. Current accelerometer technology has introduced algorithms to measure steps, along with counts (Howe et al., 2018). Raw data produced by accelerometers are transformed using various algorithms into PA summaries, which have different labels (e.g. steps, calories, activity counts) and can be aggregated at different temporal resolutions (e.g. minutes, hours, or days) (Karas et al., 2018). Currently, numerous accelerometer-based (counts/min) prediction equations and cut-points are available for estimating children's PA energy expenditure (PAEE) and PA intensity (Trost et al., 2011; Evenson et al., 2008). To accurately estimate PA in paediatric populations (Duncan et al., 2016; Roscoe et al., 2017) accelerometry derived cut-points are useful in determining the extent to which children meet current PA guidelines for health (Duncan et al., 2019). Determination

of cut-points that are specific to age group (e.g., children), model of accelerometer and wear location are critical in ensuring the accuracy of PA assessment (Duncan et al., 2019).

The ActiGraph<sup>™</sup> line is the most commonly used and extensively validated accelerometer in PA research for the measurement of PA levels, particularly in studies involving children and adolescents (Addy et al, 2014; Bingham et al., 2016; Cain et al., 2013; Trost et al., 2005; Trost et al., 2011). These small devices are usually worn at the waist (Bassett & John, 2010; Montove et al., 2016) and should be as close as possible to the body's centre of mass (Trost et al., 2005). One of the most recent versions of the ActiGraph model is the GT3X (John et al., 2010) which measures movement in three planes. Previous studies hypothesised that the triaxial accelerometer (De Vries et al., 2011). Most recently, Lynch et al.'s (2019) systematic review, which evaluated the accuracy of tri-axial and omnidirectional accelerometers for measuring PA and sedentary behaviour in children, showed that triaxial and omnidirectional accelerometers accurately categorize sedentary behaviour and PA in children when using energy expenditure, as measured by indirect calorimetry (Lynch et al., 2019).

An important recommendation is for researchers to state their methodological 'decision rules' (i.e. choice of epoch length for recording data, criteria for valid days and number of wearing days, etc.) (Cain et al., 2013). However, even with the use of accelerometers, uncertainties remain with regards to the interpretation of the data generated. With advances in technology, variability in data collection settings has increased, presenting some challenges for comparisons between studies and with historical data (Orme et al., 2014). Accelerometers were historically most often set-up to record body movement at 60-s epochs (Ainsworth et al., 2000). Some studies have investigated the effect of epoch duration on measured time spent in different PA intensities in children (Nilsson et al., 2002) and the consensus is that shorter epochs should be used to obtain a more accurate representation of young people's PA levels (Bailey et al., 1995). The epoch is the amount of time over which activity counts are integrated and recorded, and it affects estimates of MVPA in children (Cook et al., 2012; Nicaise et al., 2011). Because of the limited memory capacities of earlier accelerometer models, a 1-minute epoch has been the most commonly used,
but shorter epochs appear to more accurately record the intermittent, short bursts of PA common in young people (Corder & Van Sluijs, 2010). Regarding the minimum number of valid wear days, recommendations typically range from 3 to 10 days (Cain et al., 2013; Grydeland et al., 2014) and reliability estimates differ by age and sex (Fairclough et al., 2007; Trost 2005). Ward and colleagues (2005) suggested that 7 days might be a reasonable standard for all ages. It is recommended that at least one weekend day is required to obtain reliable estimates (Rowlands, 2007; Trost et al., 2005).

Given the increase in popularity of accelerometry as an assessment tool, there have also been considerable efforts made to calibrate accelerometer cut-off points, which is needed to more accurately estimate PA in paediatric populations (Duncan et al., 2016; Phillips et al., 2014; Roscoe et al., 2017). The precise interpretation of accelerometer-based cadence (steps/min) cut-points is necessary for accurately measuring and tracking children's MVPA (Howe et al., 2018). One major concern is the validation of wear and non-wear time when the accelerometer is worn at the hip (Vanhelst et al., 2019). Additionally, the choice of wear time algorithms may introduce significant errors in PA levels assessment (Aadland et al., 2018; Banda et al., 2016; Chinapaw et al., 2014). Algorithms have been proposed to consider the treatment of consecutive zeros recorded by the accelerometer as an indicator of non-wear times when participant wear at the hip (Vanhelst et al., 2019).

Several cut-offs are used as an indicator of a non-wearing period, such as 10, 15, 20, 30, or 60min of continuous zeros (Cain et al., 2013; Ekelund et al., 2004; Ruiz et al., 2011; Troiano et al., 2008). In a review, similarly, Esliger et al. suggested to use 20min of consecutive zero counts as a criterion in children (Esliger et al., 2005). This conclusion concurs with another study performed in 369,517 children aged 8–13years (Janssen et al., 2015). Authors compared three non-wear time algorithms (10, 20 and 60min of consecutive zero counts) with a data reduction log (Vanhelst et al, 2019), and found that 20min is the more appropriate. Different decision rules for defining non-wearing time further reduce comparability across studies, affecting the number of valid days, sample sizes, and particularly, estimates of sedentary time (Choi et al., 2011).

Over the past decade, there has also been increasing the use of accelerometry to estimate PA in children (Rowlands et al., 2014; Crouter et al., 2018). Several reviews concluded that accelerometers provide an accurate, reliable, and practical objective measure of PA in children and adolescents (Corder et al., 2008; Rowlands, 2007). Such methods provide objective data which is more reliable and valid in children than alternative methods such as self-report (Rowlands, 2007). However, this method has some limitations for data collection and processing criteria decisions.

# 3.8.3 Psychological correlates of PA

A variety of psychosocial outcomes were assessed using existing questionnaires that have demonstrated reliability and validity for use with this age group. The psychological correlates assessed within this thesis are as follows; self-efficacy (SE), physical self-worth (PSW), physical performance self-concept (PPSC), attitudes, subjective norms, behavioural control, intentions, enjoyment, perceived selfconfidence (PSC), perceived motivators and barriers to PA participation and perceived peer and parental social support. All psychological correlates were included as part of the overall G4G questionnaire, which was piloted before administration with a small convenient sample of 8-12-year old girls. Information regarding all psychological correlates for the PA instruments used in the four studies within this thesis is presented in Table 3.2. Any modifications made to questionnaires, along with the internal consistency (scale reliability measure) values and relevant references are also reported in Table 3.2. These internal consistency values were greater than .70, representing acceptable internal consistency values (Nunnally & Bernstein, 1994). Specific detail regarding psychological instruments is highlighted in the instrument section of the methodology in all four chapters. See Appendix C to view questionnaire.

Variable	Instrument	Scale	Max score	Reference (taken from)	А	Study
SE	Modified version of the Children's Physical Activity Self-Efficacy Survey (PASES)	11-items (4-likert)	44	Saunders et al., 1997, adapted by Sherwood et al.,2004	.73	Study 2 Study 4
Perceived Self Confidence	Physical self-confidence scale	15-items	150	McGrane et al., 2017	.93	Study 1 Study 4
PA Enjoyment	Modified version of Choices questionnaire	10-items 5-likert)	50	Motl et al., 2001	.80	Study 4
PA Attitudes	The Fife Active Survey Scale	4-item (4-likert)	16	Rowe & Murtagh, 2012	.82	Study 2 Study 4
Subjective Norms	Taken from Rhodes & Smith (2006) Questionnaire	6-item (4-likert)	24	Rhodes & Smith 2006	.81	Study 2
Behavioural Control		3-item (4-likert)	12		.68	
Intentions		1-item (4-likert)	4		.66	
PA Motivators	Modified version of the Physical Activity Outcome Expectancies Scale	11-item (4-likert)	44	Taylor, Chan, & Cummings, 2004	.74	Study 2
PA Barriers	PASS	10-item (4-likert)	40	Inchley, Kirby, & Currie, 2008	.84	

# Table 3. 2: G4G psychological correlates of PA assessment instruments and scales, with alpha Cronbach Coefficients

Physical	Athletic Competence subscale taken from the	5-item	20	Sherwood et al, 2004	.73	Study 1
Performance	Self-Perception Profile for Children	(4-likert)				G: 1 0
Self-Concept						Study 2
Physical Self-	The Child and Youth Physical Self-	5-item	20	Inchley, Kirby, & Currie, 2008	.66	Study 2
Worth	Perception Profile (taken from PASS)	(4-likert)				
Peer/Social	Modified version of Choices Questionnaire	5-item	25			
Support		(5-likert)		Birnbaum et al., 2005	.74	
				Source of all 1007		Cto la 1
		E · .	25	Saunders et al., 1997	70	Study 4
Family Support		5-item	25		./6	
		(5-likert)				

Abbreviations: PA = physical activity;  $\alpha =$  Cronbach's Alpha Coefficient; SE = self-efficacy; PSC = perceived self-confidence; PPSC: physical performance self-concept; PSW: physical self-worth; PASS: Physical Activity in Scottish School age children

#### 3.8.4 FMS assessment

The FMS proficiency of participants across seven-movement skills was assessed in conjunction with the behavioural components from three established instruments, namely the Test of Gross Motor Development (TGMD) (Ulrich, 1985), Test of Gross Motor Development-2 (TGMD-2) (Ulrich, 2000), and the Get Skilled Get Active (GSGA) resource (Ryde, 2000). Four skills (run, stationary dribble, kick, and catch) were assessed using the TGMD- 2. The remaining three skills included the skip and the vertical jump, which were both assessed using the TGMD (Ulrich, 1985), and balance, which was assessed using the GSGA Manual (NSW Department of Education and Training, 2000). Each of these instruments and their associated protocols has established validity and reliability in children and are designed to give an objective measurement of gross motor skill proficiency (Farrokhi et al., 2014; Issartel et al., 2017). The seven-item test included culturally relevant skills to LGF: three locomotor skills (run, skip, and vertical jump), one stability (balance) skill, and three object skills (stationary dribble, catch, kick), all of which are culturally relevant to ladies football and deemed important to Irish sporting culture (Woods et al., 2018).

Each skill comprises three to six performance criteria across the seven selected FMS, which are scored as either present (1) or absent (0) and scored over two test trials (Appendix D). Scores for each child were calculated by totalling the correctly performed criteria for two trials for each skill (i.e., if a skill comprises three performance criteria, the score range is 0–6). The maximum subtest scores for locomotor and object control skills were 30 and 22, respectively, with a minimum of 0. The sum of both subtests yielded the total gross motor skill score (total FMS). Raw scores were used in the data analysis as this is recommended for research purposes (Ulrich, 2000).

#### **3.8.5 Focus groups**

FGs were used (Chapter 5 and 7) within the studies described in this thesis for two reasons; to involve participant voice during intervention development stages and to gain further insight into participant perceptions post-intervention. Research to increase PA in children and inform intervention design has, to date, largely

underrepresented children's voices (Bentley et al., 2012; Noonan et al., 2016), justifying the use of child participant FGs within this thesis. The FGs within this thesis used a novel combination of qualitative techniques to explore children, views, experiences, and perceptions of PA, sports participation, and the G4G intervention, as well as offering a formative opinion about future intervention design. FGs were conducted by female researchers who had experience facilitating discussions. In addition, field notes were taken to maintain contextual details and non-verbal expressions for data analysis and interpretation (Tannehill et al., 2013). A semistructured guide was employed and topic guides and questions (Appendix E, F and G) were used to facilitate discussion.

For all child participant FGs, a write, draw, show, and tell (WDST) qualitative technique was incorporated as an ice-breaker activity to further stimulate children's thinking and facilitate discussion around PA and sport participation (Pearce et al., 2009) (Appendix E, F and G). The inclusion of the WDST technique as an ice breaker task was used at the beginning of each child FG to provide children the opportunity to experience talking aloud to the group and to establish an environment in which sharing and listening was valued (Gibson, 2012).

- Firstly, to allow children to express their perceptions of PA visually, we invited children to independently (i.e., not completed in conjunction with peers) draw (and colour) themselves with their '5 favourite physical activities surrounding them'.
- The drawing took the focus away from direct questioning and consensus, to that of a more child-centred approach that better allowed for the lived experience to be shared (Horstman et al., 2008).
- Children were provided with a blank paper, a pencil, and colours and were further encouraged to write down some key 'buzz words to best describe their favourite physical activities next to the drawings'. Children subsequently provided a verbal explanation of the meaning behind their drawn and written responses. Throughout the write and draw activity, the first author separately engaged children in informal conversations for them to articulate what they were drawing and why.

In-depth and semi-structured interviews explore the experiences of participants and the meanings they attribute to them. Researchers encourage participants to talk about issues pertinent to the research question by asking open-ended questions, usually in one-to-one interviews. In some instances, the interviewer might have reworded, reordered, or clarified the questions to further investigate topics introduced by the respondent (Tong et al., 2007). The semi-structured FG interview style was based on the following qualitative research interview method described by Neuman (2000):

- Questions were pre-planned but are tailored to the participant's situation.
- The interviewer showed interest in responses and encouraged elaboration.
- An interview is like a friendly conversational exchange but with slightly more direction from the interviewer.
- An interview is interspersed with jokes, aside from stories and diversion which are also recorded.
- Open-ended questions are used, and probes are frequent.
- The interviewer and participant jointly control the pace and direction of the interview.
- The interviewer adjusts to the participant's norms and language usage.

All FGs were recorded and transcribed verbatim using pseudonyms to protect participant identities. Further details on the focus groups can be found in the relevant chapters (Chapter 5 and 7).

# 3.8.6 Body mass index (BMI)

Anthropometric measurements were also assessed as secondary outcome measures in all four studies. Height was measured using a portable stadiometer (Leicester portable height scale; Marsden, Rotherham, UK). Participants were measured with their footwear and any obstructive hair accessories removed. The participant was asked to stand straight, feet together, eyes looking forward and to feel the weight evenly on both feet. At maximum inhalation, the horizontal bar was lowered enough to touch the top of the head (International Society for the Advancement of Kinanthropometry, 2001). Measurement was recorded to the nearest 0.1cm. The process was repeated after repositioning the horizontal bar. As per protocol, if the first two measures differed by more than 0.5cm, a third measurement was taken. The mean of the two closest values was then used.

Mass was measured using Seca 761 dual-platform (Chino, CA) weighing scales. Participants were measured without footwear and heaving outer clothing was removed. The participant was asked to stand on the scales, look straight ahead, arms by their side, and measured to the nearest 0.1kg. Two measurements were recorded, if the two measurements differed by 0.5kg, a third measurement was taken and the mean of the two closest values was then used. Body mass index (BMI) was calculated using the following equation: weight (kg)/height (m)<sup>2</sup>. In Chapter 4 and Chapter 5, children were categorised into PA groupings and classified as normal weight, overweight, and obese according to gender and age-specific BMI cut-off values proposed by the (Cole & Lobstein, 2012) classification. Subsequently, underweight/normal weight, overweight, and obese subjects were grouped into three weight cohorts.

# **3.9 INTERVENTION DESIGN AND COMPONENTS**

Chapter 6 outlines a detailed description of the G4G Intervention methodological approach including intervention components, materials, and resources, and the theoretical framework underpinnings of the intervention. Intervention Material (manuals/skill cards etc.) are provided in Appendix I, J, and K.

## **3.10 DATA ANALYSIS**

Various methods of analysis were used concerning the quantitative data collected depending on the aim of the study in question. All PA and FMS data were scored according to their relevant scoring protocol and were managed using Excel (Microsoft Excel, Office 365) and analysed using the statistical software programme SPSS version 20.0 for Windows (IBM, Armonk, NY). Descriptive statistics and frequencies for all FMS, PA, and psychological variables were calculated. Before investigating potential correlations, means, standard deviations and frequencies were assessed to ensure the data was normally distributed. All statistical assumptions were tested for normality, outliers, and homogeneity of variance. In instances where

significant main effects were found, post hoc comparisons were carried out using the Tukey honest significance difference (HSD) to determine where the differences occurred. To calculate effect sizes, Cohen d values (1988) were applied, with d=0.2 representing a small effect size, 0.5 representing a medium effect size, and 0.8 representing a large effect size. Statistical significance was set at P < 0.05. A Bonferroni adjustment to the alpha level was made to reduce the risk of Type I error.

In Chapter 4, a one-way between-groups analysis of variance (ANOVA) was used to explore the impact of PA grouping (low, moderate, or high active) on all psychological correlates. In Chapter 5, which investigates the relationship between actual FMS proficiency, perceived motor skill confidence and competence, and PA in 8–12-year-old girls, a two-way between-groups ANOVA was used to identify any significant correlations and explore the impact of the two variables (perceived PSC, PPSC) and PA grouping (low, moderately, or highly active) on total FMS, LOC, and OC scores. Additionally, a Spearman's correlation coefficient test was used to investigate the relationship between actual FMS proficiency levels and perceived PSC levels. A multiple linear regression was conducted to explore the predictive ability of actual FMS, perceived PSC, and PPSC on PA measurement.

For Chapter 7, assessing the effectiveness of the 10-week multi-component community-sports based PA G4G intervention, a series of mixed-model analyses of variance (i.e. between-within subjects) (MANOVAs) were conducted to assess the impact of the intervention group 1 against the intervention group 2 and control group 3 across the PA, FMS, and psychological variables. A summary table of the statistical data analysis used for each study/chapter (2016, 2017, 2019) is presented in Table 3.3.

	Study	Objective	Statistical Analysis		
Study 1	•	Analyse the relationship between perceived PSC, PPSC, actual FMS,	One-way between-groups analysis of variance		
		and PA in Irish female children.	Two-way between-groups analysis of variance		
			Follow-up Bonferroni-corrected post hoc tests		
			Spearman's correlation coefficient		
			Multiple linear regression		
Study 2	•	Investigate the PA patterns, FMS proficiency, and psychological factors influencing PA in 8–12-year-old Irish girls	Descriptive statistics and frequencies		
			One-way between-groups analysis of variance		
	psychological correlates by physica activity patterns		Follow-up Bonferroni-corrected post hoc tests		
Study 3	•	Reports on the theory-based design protocol of the 'Gaelic4Girls'(G4G) study, a novel, multi-component, community sports-based coaching intervention for increasing 8 to 12- year-old Irish girl's physical activity (PA) levels, fundamental movement skill (FMS) proficiency, and psychosocial well-being.	Statistical analysis not undertaken or warranted for this study.		
Study 4	•	Assess whether 8 to 12-year-old girls who attended a multi-component research-informed tailored 10-week 'Gaelic4Girls' (G4G) community sports-based intervention, could: i) increase their overall PA levels, ii) FMS proficiency, and iii) psychosocial wellbeing when compared with a second intervention group and a control group	Mixed-model analyses of variance (i.e. between-within subjects) (MANOVAs) Follow-up Bonferroni-corrected post hoc tests		

Table 3. 3: Summary	of study	objectives	and statistical	analysis acros	s all studies
1 abic 5. 5. Summary	or study	objectives	and statistical	analy 515 act 05	s all studies

As mentioned previously, the FG audio recordings were transcribed verbatim. The leading author double-checked each transcript against the original recording for robustness and accuracy prior to analysis. Transcripts were analysed using thematic analysis procedures (Braun & Clarke, 2006) involving coding, categorisation, and summarizing (Elo & Kyngäs, 2008). A thematic analysis (Joffe & Yardley, 2003) was carried out in which the data were analysed inductively.

- a) Familiarisation with the data: first the transcripts were read and reread.
   Using an iterative process, meaningful quotes were then identified.
- b) Generation of comments and descriptive codes for each transcript: phrases from the transcripts were highlighted and grouped, from which themes and subthemes emerged (Neergaard et al., 2009; Pawlowski et al., 2014).
- c) Defining and naming codes: quotes considered to represent the same meaning were clustered together and assigned a code, while quotes considered to represent a different concept were given a new code. All of the data were searched until all meaningful data had been identified, clustered, and coded.
- d) Identification of a pattern across all data to derive overarching themes: the resulting labels were then scrutinised and organised into themes, each theme comprising labels considered to be conceptually similar. A constant comparison method (Glaser and Strauss, 1967) was implemented in which quotes were continually compared with one another.
- e) Reviewing themes throughout the dataset: following the recommendation of Tesch (1990), the whole process remained flexible, meaning that labels and themes could be modified and refined until the most 'reasonable' reconstruction of the data had been developed (Lincoln & Guba, 1985). The categories that were developed were largely based on the research questions.
- f) The last step in the analysis consisted of writing the report and some initiatives were taken to ensure the trustworthiness of the interpretations provided (King & Horrocks, 2010), as outlined below in section 3.10.1.

#### **3.11 RESEARCH RIGOUR, VALIDITY, AND RELIABILITY OF DATA**

#### 3.11.1 Cronbach's alpha co-efficient

Within this thesis, all scales utilised in the questionnaires for all four studies were tested for internal consistency. One of the most commonly used indicators of internal consistency is Cronbach's alpha coefficient, that is, how closely related a set of items in a scale are as a group and was assessed as a measure of scale reliability in this thesis. Ideally, the Cronbach alpha coefficient of a scale should be above .7 (DeVillis, 2012). Cronbach alpha values are, however, quite sensitive to the number of items in a scale. Negatively worded items in some of the scales were 'reversed' before checking reliability. Table 3.2 reports the Cronbach alpha coefficient values for all PA, FMS, and psychological variables assessed as part of the four studies.

#### 3.11.2 FMS inter-reliability scoring

Once data collection was completed and prior to data analysis, a minimum of 95% inter-rater, intra-rater reliability, and interobserver agreement was achieved by trained field researchers for all seven FMS. Skills were then analysed as per assessment tool guidelines scoring a "1" if the component of the skills is present and a "0" if it is absent (Ulrich, 2000). For each FMS, the two test trials were added together to get the total score for each skill. Scores were then totalled to give the overall locomotor, overall object. Skill mastery was determined when the child correctly demonstrated each performance criterion of the skill in both trials. If children displayed correct performance on all or all but one skill component, they were classified as having achieved mastery or near mastery, respectively, for that skill (Okely & Booth, 2004). Near mastery was defined as the correct performance of all components but one on both trials (van Beurden et al., 2002).

#### 3.11.3 Trustworthiness of data

The trustworthiness of FG data was established through data triangulation, peer debriefing, and dependability. Data triangulation was implemented by examining data from different respondents but collected using the same method. Data were triangulated between the prospective PE teachers/sports coaches and the lead

researcher to examine if concordances and discrepancies were arising. The research team looked for patterns and contradictions investigating the same problem, which brought about different perceptions of inquiry, helping to reduce bias while strengthening the integrity of the FG findings. The research team sought support (peer debriefing) from other academic professionals (staff within the department) to provide scholarly guidance. Such investigators looked at the study background information, data collection methods and process, data management, transcripts, data analysis procedure, and research findings (Pitney & Parker, 2009). Dependability was implemented through the code recode strategy (code agreement), whereby the researcher coded the same data twice, giving 1- or 2-weeks' period between each coding. The results from the two coding processes were compared to see if the results were the same or different (Ary et al., 2010). Disagreements were discussed until an agreement was reached (Braun & Clarke, 2006; King & Horrocks, 2010). The coding results of this current study were in agreement and therefore enhanced the dependability of the qualitative inquiry. The lead researcher also double-checked each transcript against the original recording for robustness and accuracy prior to analysis.

# 3.11.4 Reporting of data

For Chapter 6 and 7, the transparent reporting of evaluations with nonrandomised designs (TREND) statement for reporting was used (Reeves & Gaus, 2004). Consolidated criteria for reporting qualitative research (COREQ): a 32-item checklist (Tong et al., 2007) developed to promote explicit and comprehensive reporting of qualitative studies (interviews and focus groups) was used by the lead researcher within this thesis. The criteria in the 32-item checklist helped the lead researcher to report important aspects of the research team, study methods, context of the study, findings, analysis, and interpretations. Furthermore, the checklist aims to promote complete and transparent reporting among researchers and indirectly improve the rigour, comprehensiveness, and credibility of the interview and focus-group studies (Tong et al., 2007).

#### 3.11.5 Peer-review process

Chapters 4 and 5 in this thesis are published peer-reviewed articles and therefore underwent the peer-review process with experts in the field of youth PA and sport participation, further establishing the validity of the presented research. Chapters 6 and 7 have been extensively reviewed by both Ph.D. Supervisors and have been submitted to Journals for publication.

## **3.12 ROLE OF RESEARCHER**

Gee (1999) states that the researcher as an interviewer in qualitative research plays an important role in how the interviewees construct their reality. The researcher's outlook on life, his/her life experiences, and observations have a high likelihood of influencing the process of collecting, analysing, and interpreting data. The lead researcher acknowledges that personal characteristics including her sport playing profile and professional occupation may have caused some bias in the selection, data collection, and intervention phases within this thesis. The lead researcher is a senior intercounty LGF player in Cork and is known within the GAA community. In addition to this, the lead researcher played a semi-emic role in the research process for the intervention studies (Chapter 6 and 7), as a partial intervention participant in the provision of coach education weekly workshops.

The lead researcher is also a qualified post-primary Physical Education teacher and has numerous years of teaching and sports coaching experience in the GAA. As a result of the lead researchers known athletic profile and sports education background, it was easier for the researcher to establish rapport with the participants and create a safe environment where the participants could construct the meaning of their experiences without feeling that they were being judged.

Furthermore, the lead researcher was transparent and reflexive about the processes by which data have been collected, analysed, and presented. Specifically, the lead researcher was critically self-reflective about her preconceptions, relationship dynamics, and analytic focus; (Polit & Beck, 2013). Patnaik (2013) postulates that an understanding of one's attitudes, values, and biases is a useful tool in not only gaining deeper insight into the research but also in ensuring that the focus

remains on the research and its participants. Within this thesis, the lead researcher kept a research journal explicating personal reactions and reflections, insights into self and past, in a separate journal. After every interview, the lead researcher spent substantial time noting immediate observations, thoughts, and interpretations before the data was subjected to structured analysis. This enabled the lead researcher to capture in its raw state her attitudes and responses to the participant she had met. The lead researcher utilised introspective reflexivity and acknowledged that the researcher's experiences, attitudes, and emotions affect engagement with the participants and subsequent analysis of data. This attempted to maintain research focus by bracketing biases and attitudes of the researcher to minimise, if not prevent, her influence on the research process (Patnaik, 2013).

# **3.13 CHAPTER CONCLUSION**

This chapter has provided an overview of the methods used in the studies described in this thesis, along with a brief justification for doing so. The next four chapters will provide more detail on the individual studies and the methods used. These upcoming chapters are presented as research articles, two of which are published (Chapter 4 and 5), and the remaining two (Chapters 6 and 7) have been prepared for publication in peer-reviewed journals.

# **CHAPTER 4**

THE RELATIONSHIP BETWEEN ACTUAL FUNDAMENTAL MOTOR SKILL PROFICIENCY, PERCEIVED MOTOR SKILL CONFIDENCE AND COMPETENCE, AND PHYSICAL ACTIVITY IN 8–12-YEAR-OLD IRISH FEMALE YOUTH



# Chapter 4: The Relationship between Actual Fundamental Motor Skill Proficiency, Perceived Motor Skill Confidence and Competence, and Physical Activity in 8–12-Year-Old Irish Female Youth

## Preamble

The following article reports one of the 'Gaelic4Girls' (G4G) baseline studies on pre-adolescent girls' actual and perceived motor skill confidence and competence, and relationship with PA. This is a published manuscript which has been peer-reviewed and published in the Sports Journal online [September 27<sup>th</sup>, 2017]. The following is the citation for this article:

Farmer, O., Belton, S., & O'Brien, W. (2017). The Relationship between Actual Fundamental Motor Skill Proficiency, Perceived Motor Skill Confidence and Competence, and Physical Activity in 8–12-Year-Old Irish Female Youth. *Sports*, *5*(74). <u>https://doi.org/10.3390/sports5040074</u>

# Statement of authorship;

I hereby declare that I, Orlagh Farmer, am the principal author of this article. The following statements outline my contributions to the work:

- I was the principal investigator responsible for all the data collected (2016) in this chapter. I organised all stages of data collection consisting of timetabling undergraduate students, contacting all primary schools (principals, teachers and students), equipment preparation, data storage, data cleaning and data inputting.
- Organised the rigorous training and protocol workshop for field staff assistants, which was an important component of this baseline study.
- Made substantial contributions to the conception and design of the work; the acquisition, analysis and interpretation of data for the work.
- Wrote the first draft of the full article and contributed significantly to the subsequent redrafts until submission.
- Final approval of the version to be published, and agreement to be accountable for all aspects of the work in ensuring that questions related to the accuracy or integrity of any part of the work are appropriately investigated and resolved.

#### **4.1 PURPOSE OF CHAPTER**

#### 4.1.1 Rationale

To date, various studies have been conducted assessing the relationship between perceived movement and actual movement competence (Britton et al., 2020; DeMeester et al., 2016) among children, while far fewer refer to perceived confidence (perception of ability/self-efficacy) (Robinson, 2010; Barnett et al., 2008), and specifically so within the female childhood population. As this area is relatively unexplored, this baseline study aimed to examine the relationship between actual FMS proficiency, perceived motor confidence and competence, and PA in a sample (n=160) of 8- to 12-year-old pre-adolescent girls. One of the main pillars for the G4G intervention was FMS integration; for this reason, the levels of FMS proficiency and the association of perceived competence and confidence needed to be assessed amongst an Irish pre-adolescent cohort to inform the development of the 'G4G' intervention.

#### 4.1.2 Contribution to the field

This study makes an important contribution to the field, demonstrating that despite the low levels of actual FMS proficiency, Irish girls may be inaccurately overestimating their perceived confidence levels for movement. It also adds to the evidence base implying that PA groupings (low-moderate and highly active) may impact perceived PSC levels. From this study, it appears that pre-adolescent females need additional hours of instructional practice towards the acquisition of actual FMS proficiency. This has important implications for researchers and those working with pre-adolescent girls, highlighting the importance of the promotion of autonomous and competencebased FMS activities among PA groupings, particularly targeting low active pre-adolescent girls).

# 4.2 ABSTRACT

**Background:** This study examines the relationship between actual fundamental motor skill (FMS) proficiency, perceived motor confidence and competence, and physical activity (PA) among female children (n= 160; mean age =  $10.69 \pm 1.40$  years).

**Methods:** The Test of Gross Motor Development-2nd Edition (TGMD-2) was used to assess seven FMSs (locomotor, object-control, and stability). Motor confidence and competence were assessed using a valid skill-specific scale, and a modified version of the Self-Perception Profile for Children. PA levels were assessed using self-report (PA Questionnaire for Older Children (PAQ-C)) and classified as low, moderate, and high active. One-way and two-way ANOVAs (post-hoc honest significant difference (HSD)) and correlation coefficients were used to analyse the data.

**Results:** Findings indicate that the majority of youth (71.8%) were not meeting the minimum 60 min of daily PA recommended for health, and that 98.1% did not achieve the FMS proficiency expected for their age. While there were high levels of perceived physical self-confidence (PSC) reported within FMS skill-specific tasks, there was no significant correlation observed between actual FMS proficiency and perceived PSC among the cohort. Results show that low, moderately, and highly active female participants differ significantly in terms of their overall FMS (p = 0.03) and locomotor (LOC) control scores (p = 0.03). Results from a two-way between-groups analysis of variance also revealed no statistically significant interaction effect between PA grouping and physical performance self-concept (PPSC) on overall FMS proficiency levels. Results of a multiple linear regression indicate that perceived PSC is a significant predictor (beta = 0.183) of participants' overall PA levels.

**Conclusion:** Data show a need for targeting low levels of PA, and low FMS proficiency in female youth, and for developing interventions aiming to enhance perceived PSC levels.

**Keywords:** fundamental movement skills; perceived physical self-confidence; physical performance self-concept; physical activity; female youth.

## **4.3 INTRODUCTION**

Regular participation in physical activity (PA) and sport is positively associated with an array of physical, psychological, and social health benefits (Eime et al., 2013). Despite the widely known positive health benefits of PA, children and adolescents are not meeting the daily recommendations of at least 60 min of moderate to vigorous physical activity (MVPA) (Hallal et al., 2012).

Further to these low levels of PA participation, research consistently indicates a gender-based disparity in PA amongst youth, with girls significantly less active than boys (Telford et al., 2016). Research also demonstrates that PA decreases significantly during the transition from childhood to adolescence, with girls in particular showing sharper declines in participation (Bradley et al., 2011). There is a need to pay attention to girls' PA patterns and influences (E. Budd, 2016). Numerous factors have been shown to influence child PA participation (Loprinzi et al., 2015), and research suggests that actual movement competency (e.g., fundamental movement skill (FMS) proficiency) in early childhood may be an important prerequisite for engagement in PA later in life (Loprinzi et al., 2015).

Actual movement competency, which also has been noted in previous literature as motor coordination, motor skill proficiency, FMS, or motor ability (Robinson et al., 2015), can be defined as the basic observable building blocks (Cohen et al., 2014) for movement. These provide the foundation for the specialised, and sport-specific movement skills (Babic et al., 2014) required for participation in a variety of physical activities including games, sports, and recreational activities. FMS can be categorised as locomotor (e.g., run, skip, jump), object-control (e.g., throw, catch, kick), and stability (e.g., static balance) skills (Gallahue et al., 2012). Children move and engage in PA through the execution of FMS (Cohen et al., 2015). Previous research has demonstrated positive associations between FMS and areas of health including PA, organised youth sport, and self-concept (Lubans et al., 2010). Despite FMS contributing to the general development and wellbeing of young children, literature consistently illustrates that proficiency in FMS among children and adolescents is low (O' Brien et al., 2016a), with only 50% of children demonstrating competency in a broad range of skills (Gallahue et al., 2012; Hardy et al., 2013). By the time children reach 10 years of age, they have developmental capabilities to master FMS performances, (Gallahue et al., 2012) however, this is

often not the case. These lower levels of FMS proficiency may translate into a lack of confidence in performing specific skills (O' Keefe & Smyth, 1999). Research suggests that without FMS proficiency and a positive perception of such, children may be less likely to engage in PA (Lubans et al., 2010). To better understand the acquisition of FMS alongside levels of PA, it is crucial to consider mediators, such as confidence, that may account for the motor development of children (Lubans et al., 2010).

Within the realm of motor development, various terms such as "selfconfidence", "self-efficacy", "perceived ability", and "perceived competence" have been used to describe one's perceived capability to accomplish a certain level of performance (Feltz, 1988). Perceived competence refers to an individual's perception of their actual movement capabilities (Harter, 1999). Stodden and Goodway (2008) postulate that higher perceived competencies are related with FMS proficiency, and increased levels of PA. Similarly, self-confidence refers to the perceived ability to accomplish a certain level of performance (Feltz, 1988). Research carried out in this field suggests that as children gain confidence in performing fine and gross motor skills, and build a sufficiently diverse movement repertoire, they acquire a high level of movement proficiency that is positively associated with the quality of their psychomotor and cognitive health (Chaddock-Heyman et al., 2013).

Like self-confidence, research has shown that PA and self-concept are connected in different ways (Planinšec et al., 2004). Physical self-perceptions, including physical performance self-concept (PPSC), are significant correlates of PA in children (Crocker et al., 2000). Indeed, increasing perceptions of competence and levels of FMS proficiency are potential strategies to promote PA.

Various studies have been conducted assessing the relationship between perceived movement and actual movement competence (Babic et al., 2014; DeMeester et al., 2016; Robinson, 2010) among children, while far fewer refer to perceived confidence (perception of ability/self-efficacy) (Barnett et al., 2008; Robinson, 2010), and specifically so within the female childhood population. A recent Irish study investigating the relationship between FMS proficiency and perceived physical self-confidence (PSC) levels among adolescents found a significant correlation between both variables for females (McGrane et al., 2016).

Little is known regarding the influences of FMS and perceived movement self-confidence on PA among Irish pre-adolescent girls. At present, there are no studies examining the relationship between perceived PSC, PPSC, and actual FMS involving Irish female children. Consequently, the aim of the present study was to analyse the relationship between perceived PSC, PPSC, actual FMS, and PA in Irish female children.

# 4.4 METHODS

# 4.4.1 Participants and study design

Two hundred and twenty-one participants from three primary schools were originally invited to participate in the study, with full consent received from 166 participants (75% of total sample). In total, 160 female participants (with a mean age of 10.69 years (SD =  $\pm 1.40$ ) had full data available for all measurements, including FMS assessment, the PA questionnaire (PAQ), and the PSC and PPSC scales, as presented in Figure 4.1.

This cross-sectional, mixed methods study was granted full ethical approval by the Institutional Social Research Ethics Committee at University College Cork (Social Research Ethics Committee, UCC) in 2016. Informed assent for participation was granted by all participants and consent from their parent(s)/guardian(s) and the school principals prior to participation; all participants were free to withdraw from the research at any stage.



Figure 4.1: Study design and flow of data collection (n = 160).

# 4.4.2. Recruitment and data collection

Convenience sampling was used to recruit three all-female primary schools (rural and urban) in the Cork region, Ireland. A representative social economic status demographic was selected, with the inclusion of one school identified as 'DEIS' (delivering equality of opportunity in schools). The lead researcher emailed the principals of all three schools, calling for expressions of interest, and written consent to participate in the data collection process. To be eligible for the study, female participants needed to be formally enrolled in between the years of second to sixth class and were required to provide written assent and parental/guardian consent prior to participation. If a participant or their parent/guardian did not consent, they were not permitted to participate in the data collection. Prior to data collection, ten field staff underwent a rigorous and robust 8-hour training workshop (across two days) in the measurement protocol associated with FMS and self-report questionnaires. This involved an objective criteria-informed process to ensure field staff were consistent in the data collection measurement protocol.

The data were collected on participants within their class groups (max n = 30) during specific school visits (n = 3). Objective measurements, such as FMS, were carried out during a timetabled block, with a ratio of one researcher to five students. Subjective self-report measurements took place in a supervised classroom, or computer lab, and the ratio of participant to researcher was 10:1. The study was briefly explained, and instructions provided on how to complete the questionnaire. Participants were encouraged to take their time, reflect on their answers, and to be as honest as possible. All questionnaires were completed online through the tool 'Survey Monkey'. In cases where computer networks failed, participants completed hardcopies of the questionnaire. Throughout the duration of this baseline data collection, participants were assigned identification numbers for anonymity purposes.

# 4.4.3 PA self-report assessment

Moderate to vigorous PA (MVPA) was assessed using a modified version of the Physical Activity Questionnaire for Older Children (PAQ-C) (Kowalski et al., 2004). Studies have established the reliability and validity of the 7-day recall on children (Moore et al., 2007). The PAQ-C for this study included 15 physical activities, 10 leisure/free-time activities, and activities in school (Physical Education), transport activities (walking to and from school) and other activities. The participants were told to recall what activities they had engaged in the previous seven days and how many times and number of minutes they participated in each of these activities. Habitual PA was also assessed using two questions from the Physician-based Assessment and Counseling for Exercise (PACE) questionnaire: how many days in the last week (PACE 1) and in a usual week (PACE 2) does the subject do at least 60 min of physical activity. The PACE questionnaire presents a test–retest reliability assessed by the Intra-class Correlation Coefficient (ICC) of 0.77 (López Sánchez et al., 2016), and due to its simplicity and ease of understanding, the PACE questionnaire was suitable for the 8–12-year-old cohort. In the current study, the Cronbach alpha coefficient was 0.63, suggesting that the scale has good internal consistency. Data were collected on participants in their class groups with a ratio of one trained field staff to ten students, for questionnaire completion. The study was briefly explained, and instructions were provided on how to complete the questionnaire. The compound result was obtained from both questions ((PACE 1 + PACE 2)/2) and students were categorised as low active (meeting guidelines on 0, 1, 2, or 3 days a week), moderately active (meeting on 4 or 5 days), or highly active (meeting on 6 or 7 days) (Rowe & Murtagh, 2012).

#### 4.4.4 Measures: fundamental movement skills (FMS)

The FMS proficiency of seven movement skills were assessed (n = 160) in conjunction with the behavioural components from three established instruments, namely the Test of Gross Motor Development (TGMD) (Ulrich, 1985), Test of Gross Motor Development-2 (TGMD-2) (Ulrich, 2000), and the Get Skilled Get Active resource (New South Wales Department of Education and Training, 2000). Each of these instruments and their associated protocols have established validity and reliability in children and are designed to give an objective measurement of gross motor skill proficiency. The test consists of two subscales, locomotor control (LOC) skills and object-control (OC) skills and was designed to measure criterion elements of FMS performance in children aged 8-12 years. The seven-item test included: three LOC skills (run, skip, and vertical jump), one stability (balance) skill, and three OC skills (stationary dribble, catch, kick), which combine to give an overall maximum raw score of 60. A total score for all seven skills was calculated for each participant, along with an OC score, and LOC score. Overall mean OC and LOC scores were also calculated. The raw skill scores were then added to obtain a raw LOC subtest score (ranging from 0–36 points) and a raw OC subtest score (ranging from 0–24 points).

During data collection, the seven skills were assessed during a three-hour timetabled block and one trained field staff member provided every participant (in groups of 8) with an accurate demonstration of the FMS to be performed.

Participants performed the skill on three occasions, including one familiarisation practice, and two performance trials, as reported in previous Irish adolescent movement skill data collection studies (O' Brien et al., 2016a, 2016b). Participant performance, along with execution of the required skill, were recorded using digital video cameras (3× Canon type Legria FS21 cameras; Canon Inc., Tokyo, Japan) to allow for greater measurement scrutiny, and accuracy of measurement precision during analysis. The FMS scoring process was completed at a later date by the principal investigators. The number of FMS performance criteria varied from six to twelve across the range of selected FMS. There was a total of 29 performance criteria for all seven fundamental movement skills. Once data collection was completed, the principal investigators were required to reach a minimum of 95% inter-observer agreement for scoring all seven FMS. If children displayed correct performance on all or all but one skill component, they were classified as having achieved 'mastery' or 'near mastery', respectively, for that skill. 'Near mastery' was defined as correct performance of all components but one on both trials (van Beurden et al., 2002).

# 4.4.5 Perceived physical self-confidence

Participants' perceived PSC levels were assessed using the PSC scale (McGrane et al., 2016), which has shown excellent test–retest reliability, and internal consistency, with a Cronbach alpha coefficient of 0.94 (McGrane et al., 2016). In the current study, the Cronbach alpha was 0.88, suggesting very good internal consistency reliability for the perceived PSC scale for the seven skills with this sample. The PSC scale consists of 15 questions in which participants rate their confidence at performing 15 separate FMSs. The identified skills included within this instrument are considered central to the Irish youth sporting culture (Cohen et al., 2014; van Beurden et al., 2002) Participants rated their confidence at performing each skill on a Likert scale of 1–10, with "1" being not confident at all, and "10" being very confident. This present study assessed 7 of the 15 actual movement skills, consistent with the seven actual FMSs assessed, therefore, the maximum PSC score which could be achieved was 70 if participants scored their confidence at 10/10 for performing all seven skills. Similar to previous research (McGrane et al., 2016),

participants were divided into three tertiles based on perceived PSC using visual binning in SPSS (≤28.7 was the low PSC group, 28.8–49.3 was the medium PSC group, and 49.4+ was the high PSC group).

#### 4.4.6 Physical performance self-concept

Participants' PPSC were assessed using a modified version of the Athletic Competence sub-scale, taken from the Self-Perception Profile for Children (Harter, 1982) (Sherwood et al., 2004). The scale for this study consists of eleven statements (e.g., 'I am good enough at sports'). Students were asked to indicate how true each statement was for them ('very true', to 'not at all true', range 1-4), and the participant selected the statement that best described them. The more positive the statement in the questionnaire, the higher the value. For example, 1 = not at all true and 4 = very true for nine of the 'positive' statements (e.g., 'I feel positive about myself physically' and 'I am good enough at sports', etc. 1 = not at all true and 4 =very true). This was reversed and recoded for two of the 'negative' PPSC statements (i.e., 'I wish I could feel better about myself physically' and 'I like to watch sports rather than play'), whereby in this case 4 = not at all true and 1 = very true. Reliabilities for the subscales of the PPSC range from 0.73 to 0.84 [35], and the testretest reliability coefficient (r = 0.51) was moderate (Sherwood et al., 2004). In the current study, the Cronbach alpha coefficient was 0.70, thus suggesting that the scale has acceptable internal consistency.

#### 4.4.7 Data analysis

The FMS, PSC, PPSC, and PA data set were analysed using IBM, USA for SPSS software (version 20.0 for Windows (SPSS Inc., IBM Corp., Armonk, NY, USA). Statistical significance was set at p < 0.05. Where participants had incomplete data for a given variable, participants were excluded from analysis of this variable specifically. The number of days participants self-reported meeting the 60 min PA guidelines were analysed descriptively using means, standard deviations, and proportions. Descriptive statistics and frequencies for all variables were calculated.

One-way between-groups analysis of variance were conducted to explore the impact of PA groupings (low, moderately, and highly active) on overall FMS, LOC and OC proficiency levels, perceived PSC and PPSC levels. As the assumptions for ANOVA were met, two-way between-groups ANOVAs were used to explore the impact of the two variables (perceived PSC, PPSC) and PA grouping (low, moderately, or highly active) on total FMS, LOC, and OC scores. In instances where significant main effects were found, post-hoc comparisons were carried out using the Tukey (HSD) test to determine where the differences occurred. Considering the relatively large numbers of means (FMS, LOC, and OC), the use of Tukey HSD comparisons is a powerful, and accepted form of statistical post-hoc assessment.

Spearman's correlation coefficient test was used to investigate the relationship between actual FMS proficiency levels and perceived PSC levels. A multiple linear regression was conducted to explore the predictive ability of actual FMS, perceived PSC, and PPSC on PA measurement.

# 4.5 RESULTS

## 4.5.1. Physical activity

Self-report PA data showed that 10.2% of participants were meeting the 60 min guideline on 0–3 days a week (low active), with 40.95% meeting the guideline on 4 or 5 days a week (moderately active), and the remaining 48.85% of participants meeting the guidelines on 6 or 7 days a week (highly active). The percentage of participants meeting the 60 min PA guideline on all 7 days was 28.2%.

#### 4.5.2 Fundamental movement skills

Only three participants (1.9%) possessed complete mastery level across all seven OC and LOC skills. Overall, the vertical jump and the skip were the poorest performed skills across the cohort, where 38.6% and 36.9%, respectively, achieved mastery. The best performance was for the kick (OC), where 68.4% and 24.7%, respectively, achieved mastery and near mastery. The overall mean FMS score of participants was 48.75 (SD =  $\pm$ 5.83), out of a possible score of 62. Mean overall FMS scores for the

three PA groups were as follows: low = 46.25 (SD =  $\pm$ 7.60), medium = 49.92 (SD =  $\pm$ 3.80), and high = 48.74 (SD =  $\pm$ 6.04). Mean skill scores by PA grouping (low, moderately, and highly active) for actual FMS are given in Table 4.1. A one-way between-groups analysis of variance was conducted to explore the impact of PA groupings (low, moderately, and highly active) on overall FMS proficiency levels. Participants were divided into three groups according to their PA level (Group 1: low active; Group 2: moderately active; Group 3: highly active). There was no statistically significant difference at the p < 0.05 level in overall FMS proficiency for the three PA groupings.

Table 4.1: Descriptive data for each variable (frequency, means, standard deviation, and physical activity (PA) grouping (low, moderately, and highly active).

Frequencies of PA Groupings							
Variable	Ν	М	SD	LA	MA	НА	
РА	158	1.57	0.67	16	36	106	
FMS	158	48.75	5.84	16	36	106	
PSC	152	57.55	14.34	15	35	102	
PPSC	150	18.89	3.33	14	35	101	
LOC	158	29.71	4.20	16	36	106	
OC	158	19.04	2.98	16	36	106	
Age	160	10.69	1.40	16	36	106	

N = number; M: Mean; SD: standard deviation; LA: low active; MA: moderately active; HA: highly active; LOC: locomotor control; OC: object-control.

	FMS ( <i>n</i> = 158)				PSC ( <i>n</i> = 152)			
Skill	Low Act	Mod Act	High Act	OMS	Low Act	Mod Act	High Act	OMS
	( <i>n</i> = 16)	( <i>n</i> = 36)	( <i>n</i> = 106)		( <i>n</i> = 15)	( <i>n</i> = 35)	( <i>n</i> = 102)	
Run	6.79 SD = ±1.42	6.67 SD = ±1.51	6.59 SD = ±1.41	6.63 SD = ±1.43	6.53 SD = ±3.50*	7.06 SD = ±3.12	8.27 SD = ±2.66*	7.81 SD = ±2.92
Skip	4.60 SD = ±1.18	5.00 SD = ±1.33	4.63 SD = ±1.12	4.72 SD = ±1.18	6.93 SD = ±3.22	6.86 SD = ±3.27	7.90 SD = ±2.71	7.56 SD = ±2.92
VJ	9.67 SD = ±2.38	9.94 SD = ±2.01	9.97 SD = ±2.24	9.94 SD = ±2.19	7.13 SD = ±3.25	7.26 SD = ±3.23	8.19 SD = ±2.72	7.87 SD = ±2.92
Balance	8.63 SD = ±1.31	9.17 SD = ±1.16	8.73 SD = ±1.34	8.82 SD = ±1.30	8.27 SD = ±2.34	8.15 SD = ±2.55	8.99 SD = ±2.07	8.73 SD = ±2.23
Kick	7.13 SD = ±1.50	7.58 SD = ±0.77	7.37 SD = ±1.04	7.39 SD = ±1.05	8.33 SD = ±1.92	8.09 SD = ±2.78	8.88 SD = ±1.88	8.65 SD = ±2.13
Bounce	6.06 SD = ±2.21	6.60 SD = ±1.99	6.77 SD = ±1.68	6.66 SD = ±1.81	9.27 SD = ±1.44	8.15 SD = ±2.85	9.08 SD = ±2.10	8.89 SD = ±2.26
Catch	5.13 SD = ±1.20	5.11 SD = ±1.01	5.43 SD = ±0.92	5.33 SD = ±0.98	7.07 SD = ±2.99*	8.62 SD = ±2.26	8.72 SD = ±2.34*	8.53 SD = ±2.43
LOC	27.94 SD = ±6.04	30.81 SD = ±2.71	29.60 SD = ±4.22	29.71 SD = ±4.20	7.22 SD = ±3.08	7.33 SD = ±3.04	8.34 SD = ±2.54	7.63 SD = ±2.89
OC	18.31 SD = ±3.44	19.11 SD = ±2.46	19.13 SD = ±2.98	19.04 SD = ±2.91	8.22 SD = ±2.12	8.29 SD = ±2.63	8.89 SD = ±2.11	8.47 SD = ±2.29
OMS	46.25 SD = ±7.60	49.92 SD = ±3.80	48.74 SD = ±6.04	48.75 SD = ±5.84	53.53 SD = ±14.48	53.23 SD = ±17.46	59.62 SD =±12.76	57.64 SD =±14.23*
PPSC	16.64 SD = ±3.63*	17.89 SD = ±3.80*	19.54 SD = ±2.89*	18.89 SD = ±3.33*				

Table 4. 2: Mean scores for fundamental movement skills (FMSs), perceived physical self-confidence (PSC), and physical performance self-concept (PPSC) by physical activity (PA) grouping (low, moderately, and highly active).

\*: Statistical significance  $p \le 0.05$ . OMS: Overall mean score; Mod: Moderately; VJ: Vertical jump; LOC: Mean locomotor score; OC: Mean

object-control score.

#### 4.5.3 Perceived physical self-confidence

Interestingly, 40 participants (26% of the cohort) rated themselves as the top score on the perceived PSC scale for all seven FMSs, while only three participants (1.9% of the cohort) possessed complete mastery level across all seven FMSs in the actual FMS assessment. The overall mean perceived PSC score of participants was 57.64 (SD =  $\pm$ 14.23), out of a possible score of 70. Mean skill scores by PA grouping (low, moderately, and highly active) for perceived PSC are given in Table 4.1. There was also a statistically significant difference at the *p* < 0.05 level in overall perceived PSC scores for the three PA groups (F (2, 143) = 3.34, *p* = 0.038). There were significant PA grouping differences observed in perceived PSC, with highly active participants scoring significantly higher than low active participants in two of the seven individual skills, as highlighted in Table 4.1. This included one LOC skill (run; *p* = 0.020) and one object control skill (catch; *p* = 0.046).

#### 4.5.4. Physical performance self-concept

The overall mean PPSC score of participants was  $18.89 (SD = \pm 3.31)$ , out of a possible score of 24. Mean skill scores by PA grouping (low, moderately, and highly active) for PPSC are given in Table 4.1. A one-way between-groups analysis of variance was conducted to explore the impact of PA grouping on overall PPSC levels. There was a statistically significant difference at the p < 0.05 level in PPSC scores for the three PA groups (F (2, 147) = 7.32, p = 0.001). The actual difference in mean scores between the groups was moderate. The effect size, calculated using eta squared, was 0.09, which in Cohen (1988) terms would be considered a medium effect size. Post-hoc comparisons using the Tukey HSD test indicated that the mean score for highly active (M = 19.54, SD = 2.89) participants was significantly different (p = 0.05) from the low active participants (M = 16.64, SD = 3.63). Likewise, the mean score for the moderately active group (M = 17.89, SD = 3.80) was significantly different than the highly active group (p = 0.02).

#### 4.5.5 Interrealtionship between variables

A two-way between-groups analysis of variance was conducted to explore the impact of PA groupings and perceived PSC on overall FMS proficiency levels. Participants were divided into three groups according to their PA level (Group 1: low active; Group 2: moderately active; Group 3: highly active). The interaction effect between PA grouping and perceived PSC was statistically significant (F (4, 143) = 2.76, p = 0.03). There was also a statistically significant main effect for PA category (F (4, 143) = 4.76, p = 0.01).

Additionally, a two-way between-groups analysis of variance was conducted to explore the impact of PA grouping and perceived PSC on overall FMS LOC proficiency levels, as shown in table 4. 2. The interaction effect between the two variables was statistically significant (F (4, 143 = 2.78, p = 0.029). There was also a statistically significant main effect for PA category (F (2, 143) = 6.54, p = 0.002), and PSC grouping (F (2, 143) = 3.77, p = 0.025). This means that low, moderately, and highly active participants differ in terms of their overall locomotor scores, and there is also a difference in scores for participants with low, modertely, and highly perceived PSC scores. Results from a two-way between-groups analysis of variance also revealed no statistically significant interaction effect between PA grouping and PPSC on overall FMS proficiency levels.

The relationship between actual FMS proficiency levels and perceived PSC levels was investigated using Spearman's correlation coefficient. Preliminary analyses were performed to ensure no violation of the assumption of normality, linearity, and homoscedasticity. There was no correlation found between the two variables (rho = 0.11, n = 154, p = 0.191).

A multiple linear regression was calculated to predict overall PA levels based on participants' perceived PSC, PPSC, and actual FMS proficiency. A significant regression equation was found (F (3, 145) = 3.127, p = 0.028), with an R<sup>2</sup> of 0.061. The model explains only 6.1% of the variance in overall PA levels. Of the three variables, perceived PSC makes the largest, unique, significant contribution (beta = 0.183) while PPSC and actual FMS proficiency made no statistical contribution (beta = 0.119 and 0.011, respectively). This indicates that PSC is a predicting variable in female participants' overall PA levels.

# **4.6 DISCUSSION**

Results of this study show that the percentage of participants meeting the 60 min MVPA guidelines on all 7 days was 28.2%. These low levels of PA participation are

relatively consistent with other studies on a national Irish level (Layte & Mccrory, 2011; Woods, et al., 2010). Although higher than the 16% reported figure for a similar age group in the Irish Health Behaviour in School-aged Children (HBSC) study (Gavin et al., 2015), these low levels are in line with this study's findings.

The majority of female adolescents in this study (98.1%) failed to reach a level of mastery across key FMSs, indicating that basic movement skill proficiency amongst this selected pre-adolescent Irish female youth cohort is low. These observed low findings of FMS proficiency are consistent with recent Irish adolescent research (O' Brien et al., 2016a). Results from the present cohort indicate that only three participants were fundamentally competent across all seven object-related, stability, and locomotor skills, despite children having the developmental capacity to master these skills by ten years of age (Gallahue et al., 2012). Previous international research examining the FMS proficiency of female children support these low levels of FMS competency (Duncan et al., 2017; Hardy et al., 2012).

In the present study, 26% of participants (n = 40) rated themselves as being fully proficient at performing the seven FMSs on the perceived PSC scale. However, actual FMS assessment results reveal only three participants (1.9%) possessed complete proficiency across all seven FMSs. On the perceived PSC scale (0–10), participant mean values were generally in the upper thresholds (mean values of  $\geq 6.5$  within Table 4.1), indicating higher levels of perceived PSC amongst this selected female cohort. High reported PSC in the current study aligns with recent research on an Irish cohort, which highlighted that female adolescents in particular consistently scored a mean of 6.8 or above (out of 10) in confidence, regardless of their actual ability (McGrane et al., 2016). Similarly, in another recent Irish adolescent study (O'Brien et al., 2017), the results purport that participants had considerably higher perceived PSC levels when compared to their actual skill proficiency in FMS.

Futhermore, the results of this study suggest that despite the low levels of actual FMS proficiency, Irish girls may be inaccurately overestimating their perceived confidence levels for movement. Harter & Pike (1984) support this view in postulating that it is considered 'normative' for young children to overestimate their abilities, because of cognitive limitations in finding it hard to distinguish between their 'ideal' in terms of competence, and their own reality (Harter & Pike, 1984). Research suggests that inflated perceived competence can drive the acquisition of movement skills, because children will continue to engage in mastery attempts in activities for

which they believe they are skilful (Stodden et al., 2008). Furthermore, this early period has been termed a 'window of opportunity', as children (even if low skilled) can still be keen participators in activity (LeGear et al., 2012). It is, therefore, imperative to have an understanding of children's perceived and actual movement skill proficiency to allow for the implementation of interventions during this early period.

In this study, no correlation was found between perceived PSC and actual FMS proficiency levels. This indicates that children's perceptions of their LOM and OC skill proficiencies are not associated with the 'reality' in terms of their actual FMS proficiencies. This lack of association in the current study can perhaps be explained by the fact that the participants' mean age was 10.69 years (SD =  $\pm 1.40$ ), and according to Barnett et al. (2015), young children may not have the capacity to differentiate perceived competence in terms of the different skill types of LOM and OC. The findings in the current study are not in line with McGrane et al. (McGrane et al., 2016), where perceived PSC and actual FMS proficiency levels were moderately correlated among Irish females (r = 0.305). These findings (McGrane et al., 2016) suggest that if a female has low actual FMS proficiency, she may in turn have low PSC levels, or vice versa. Likewise, in a recent study by O'Brien et al. (2017), the results revealed that the perception of adolescent females in relation to their movement confidence did not equate to their actual movement skill proficiencies. Similar results were also found in the Vedul-Kjelsås et al. (2015) study, where actual FMS proficiency and self-perception were strongly correlated with girls (r = 0.312, mean age 11.46 years). Nevertheless, the results of the current study align with Stodden et al.'s (2008) developmental theory, which suggess that the transition into middle childhood (9-11 years) marks an important period when perceived physical competence should be lower. Stodden et al. (2008) postulate that the shift from early childhood to middle childhood marks the beginning of a period of vulnerability during which children who have lower actual movement skill competence will demonstrate lower perceived movement skill competence and are less physically active.

Consistent with other studies (Stellino & Sinclair, 2013; Ulrich, 1987), the analysis of psychological variables reveals an association with female children's PA levels. Although there were no associations found in the current study, there were significant differences found among the PA groupings (low, moderately, and highly active) for perceived PSC and LOM scores and overall FMS levels, which exhibited small effects. The interaction effect between PA grouping and perceived PSC was statistically significant. There was also a statistically significant main effect for PA category. This indicates that there is a significant difference in the effect of perceived PSC on overall FMS scores for the different PA groupings (low active, moderately active, and highly active). Furthermore, this implies that PA groupings may impact perceived PSC levels. Research suggests that those who are not confident about their ability (in this case those in the low perceived PSC group) will not want to put themselves in a situation where they may display low ability levels, which in turn may affect their performance (Harter & Pike, 1984). Furthermore, a comprehensive understanding of the issues surrounding perceived PSC and motivation for PA by developmental stages in female adolescents are vital for improving interventions, and enhancing psychological well-being (Porter, 2016).

To the authors' knowledge, this is the first study of its kind in Ireland that seeks to critically examine late childhood, and early adolescent female perceived PSC, PPSC, actual FMS proficiency, and PA participatory levels. Nevertheless, one limitation of this current study is the use of a convenience sample, which can lead to the under-representation or over-representation of particular groups within the sample. Another limitation of the present study is its cross-sectional design. The results do not provide causal evidence regarding relationships among actual FMS proficiency, perceived PSC, PPSC, and PA. To gain more insight in the direction of these relationships and to understand how associations among these variables may change over time, longitudinal or experimental studies should be conducted.

## 4.6.1 Future practical implications

Pre-adolescent girls are an important target for PA behavioural change strategies, as this age cohort may enhance tracking into the crucial period of adolescence (Biddle et al., 2014). The quantification of intervention effectiveness for this age group of girls has not been adequately reported (Biddle et al., 2014) specifically examining perceived PSC and actual FMS proficiency. Indeed, as investigated in the present cross-sectional study, components that foster the development of both actual and perceived confidence levels may have the potential to significantly improve the long-term impact of childhood and adolescent movement.

Considering the observed low levels of actual FMS proficiency amongst female pre-adolescents, developing a specifically designed movement-oriented intervention would be a strategic step towards improvement. Furthermore, the promotion of autonomous and competence-based activities among PA groupings (particularly targeting low active girls) may exist within future interventions. From this study, it appears that females need additional hours of instructional practice towards the acquisition of actual FMS proficiency.

# **4.7 CONCLUSIONS**

In terms of perceived PSC, participants generally displayed higher levels of confidence, however, these results do not appear to be associated with their actual movement-based tasks. Results from this study suggest that future interventions may need to specifically address the low levels of actual movement skill proficiency with developmentally appropriate strategies for understanding perceived confidence at the FMS level. Likewise, increasing perceptions of PSC and levels of actual FMS proficiency are potential strategies to promote physical activity (Foweather, 2010). Furthermore, physical education professionals and youth sports programmes should target both actual and perceived motor confidence and competence in order to promote lifelong PA in children.

Acknowledgments: The authors wish to acknowledge the help of undergraduate students from the Sports Studies and Physical Education degree programme involved in data collection. Sincere thanks also to the participants, parents, teachers, and principals from the three primary schools involved in this phase of the study. The authors would also like to express gratitude to the National Ladies Gaelic Football Association (LGFA) in Ireland for their continuous support and assistance.

## **Declaration of interest statement**

The authors declare that there are no conflicts of interest within this manuscript publication.
### 4.8 LINK TO CHAPTER 5

As discussed, there is a noticeable gap in the literature relating to young girls actual and perceived movement skill proficiency; therefore, chapter 4 assessed the actual and perceived competence and confidence of FMS and the relationship with PA amongst 8- to 12-year-old pre-adolescent girls. The findings from chapter 4 constituted one part of baseline data and assisted the development, design and originality of the G4G intervention, specifically the FMS component. The reader will be able to identify the findings from the baseline data collection (chapter 4), incorporated within the intervention; furthermore, the reader will recognise specific PA patterns, FMS proficiency levels, and psychological correlates of PA (chapter 5), as integrated within the G4G intervention. The overall objective of chapter 5 is to contextualise the development of a targeted community sports-based intervention (G4G) for Irish 8 to 12-year-old girls.

# **CHAPTER 5**

# ENHANCING THE EVIDENCE BASE FOR IRISH FEMALE YOUTH PARTICIPATION IN PHYSICAL ACTIVITY—THE DEVELOPMENT OF THE GAELIC4GIRLS PROGRAMME



### Chapter 5: Enhancing the Evidence Base for Irish Female Youth Participation in Physical Activity—The Development of the Gaelic4Girls Programme

### Preamble

The following article reports the second 'Gaelic4Girls' (G4G) baseline study, which further informs the development of the G4G intervention. Cross-sectional data on PA levels, psychological correlates of PA, anthropometric characteristics, and the FMS proficiency of 8- to 12-year-old girls were collected. Focus group interviews were used to explore perceptions of PA, barriers and motivators to PA in the cohort. This is a published manuscript which has been peer-reviewed and published in the Women in Sport and Physical Activity Journal (WSPAJ) online [March 12<sup>th</sup>, 2018].

The following is the citation for this article:

Farmer, O., Duffy, D., Cahill, K., Lester, D., Belton, S., & O'Brien, W. (2018).
Enhancing the evidence base for Irish female youth participation in physical activity
the development of the Gaelic4Girls program. *Women in Sport and Physical Activity Journal*, 26(2), 111–123. <u>https://doi.org/10.1123/wspaj.2017-0046</u>

### Statement of authorship;

I hereby declare that I, Orlagh Farmer, am the principal author of this article. As the data presented in chapter 5 is the same dataset as chapter 4, the statements outlining my leading contributions to this work are presented in chapter 4.

### **5.1 PURPOSE OF CHAPTER**

### 5.1.1 Rationale

At present, there is a specific gap in the literature, as no other study has assessed PA levels and psychological factors of PA and FMS, both quantitatively and qualitatively, as part of an existing Irish childhood female sports-based development programme. Very few studies have aimed to involve youth voice in the design of PA interventions, resulting in the development of interventions which may not meet the needs and interests of those involved. This study investigated the PA patterns, FMS proficiency, and psychological factors influencing PA in 8- to 12- year-old Irish girls. Focus group interviews were used to explore perceptions of PA, barriers and motivators to PA and sport participation in the cohort. The overall objective of this chapter was to contextualise the development of a targeted community sports-based intervention (G4G) for Irish 8 to 12-year-old girls.

### 5.1.2 Contribution to the field

This study strengthens the evidence base of the sparsely available literature on PA, FMS and psychological wellbeing for 8 to 12-year-old girls, justifying the development of a multi-component PA programme targeting low levels of PA, FMS proficiency in female youth sport interventions, through addressing self-efficacy levels, inclusive of fun, and socially-stimulating PA environments.

### **5.2 ABSTRACT**

**Background:** The purpose of the current research was to gather baseline data on female youth to inform the development of a targeted physical activity (PA) and sports-based intervention, specifically identified as "Gaelic4Girls".

**Methods:** Cross-sectional data on PA levels, psychological correlates of PA, anthropometric characteristics, and the fundamental movement skill (FMS) proficiency of female youth (n = 331; M age  $10.92\pm1.22$ ) were collected. A subsample (n = 37) participated in focus group (FG) interviews exploring perceptions of health/sport and identifying barriers/motivators to participation. PA levels were assessed using self-report (PA Questionnaire for Older Children) and classified as low, moderate, and high active.

**Results:** One- and two-way ANOVAs (post hoc Tukey honest significant difference [HSD]) were used to analyse the data. The FGs were transcribed verbatim, coded, and thematically analysed. Findings indicated that the majority of youth (71.8%) were not meeting the minimum daily PA recommendations for health, and that 98.1% did not achieve the FMS proficiency expected for their age. Low, moderately, and highly active participants differ significantly in terms of overall FMS (p = .03), and locomotor control scores (p = .03). FG findings report fun and friendship as key PA motivators, too much competitiveness as barriers, and positive outside encouragement from family/friends/coaches as facilitators encouraging PA engagement.

**Conclusion:** Findings highlight the need for targeting low levels of PA, FMS proficiency in female youth sport interventions, through addressing self-efficacy levels, inclusive of fun, and socially stimulating PA environments.

**Keywords:** fundamental movement skill, physical activity motivators and barriers, psychosocial physical activity correlates, sport participation

### **5.3 INTRODUCTION**

It is well established that regular participation in physical activity (PA) is imperative for good health (Eime et al., 2013). While PA of any type will deliver an array of physical, psychological, and social health benefits (Biddle & Asare, 2011), research continues to show that lack of PA participation among children and adolescents is a global concern (Guthold et al., 2010). According to previous data obtained from 105 countries, 80% of children aged 13 to 15 years fail to meet the recommended public health guidelines of 60 min of moderate-to-vigorous physical activity (MVPA) per day (Hallal et al., 2012). In an Irish context, the most recent 2016 Report Card on PA (a national document, containing all data on indicators related to children's PA levels from Northern and Southern Ireland) awarded children and adolescents an overall grade of "D" for low PA participation (Harrington et al., 2016).

Further to these low levels of PA participation, a consistent gender-based disparity in PA among adolescents exists, whereby females are significantly less active than males (Marques et al., 2016; Telford et al., 2016). A notable decrease in PA participation during adolescence has been observed, with a higher risk of decline among girls (Bradley et al., 2011; Woods et al., 2010). On a national level, the Growing Up in Ireland National Longitudinal Study (Layte & McCrory, 2011) found a significant gender differential visible among 9-year-olds, with boys (29%) more likely than girls (21%) to meet the PA guidelines. Based on the consistently reported research-informed literature, girls have been identified as a high-priority population for childhood PA promotion (Biddle et al., 2014). Nevertheless, organised youth sports (OYS) have been shown to have a positive association with PA participation. These OYS provide an excellent opportunity to be physically active and have been recommended as a heightened opportunity to increase young peoples' MVPA participation (Guagliano et al., 2012; Mandic et al., 2012). In a recent study (Marques et al., 2016) investigating associations between organised sport (OYS) participation and objectively-measured PA (N= 973, M age =  $14.1\pm2.4$ ), results reported that those who were engaged in OYS were significantly more likely to achieve PA guidelines, spend more time in moderate physical activity (MPA), vigorous physical activity (VPA), and MVPA than those who did not participate in OYS. Further to this, more boys (51.3%) than girls (28.3%) reported to be involved in OYS participation (Marques et al., 2016).

Like PA, participation in OYS during childhood and adolescence has important benefits for physical, psychological, and social health (Eime et al., 2013), and has been shown to promote greater involvement in PA over time (Nelson et al., 2011). Previous research suggests participation in OYS can be an avenue for developing social skills such as cooperation, responsibility, empathy, and selfcontrol, as well as promoting good citizenship, social success, positive peer relations, leadership skills, and a sense of initiative (Côté & Hancock, 2016). It is also important to note that OYS participants cumulatively acquire transferable movement skills for enhanced competence and confidence (Rosewater, 2009), all of which are imperative benefits for female youth PA participation.

Similar to OYS, fundamental movement skills (FMS) have been shown to have a positive association with PA (Robinson et al., 2015). FMS are the basic observable building blocks for movement and provide the foundation for specialised and sport-specific movement skills required for participation in a variety of physical activities (Babic et al., 2014), including games, sports, and recreational activities (Hands, 2012). Studies have shown that higher levels of FMS proficiency will provide greater opportunity for children to engage in a variety of physical activities, games, and sport (Wick et al., 2017). This is pivotal to note within the female preadolescent context, as girls are less active than boys.

Several studies indicate that OYS is positively associated with increased PA, FMS, and areas of health, including fitness, healthy body composition, and selfconcept (Hendrix et al., 2014). By the time children reach 10 years of age, they should have developmental capabilities to master FMS performances (Gallahue et al., 2012). Nevertheless, literature consistently illustrates that children and adolescents do not obtain proficiency in FMS development (Farmer et al., 2017; Mitchell et al., 2013; O' Brien et al., 2016a, 2016b; O'Brien et al., 2017).

Further to this, research reports gender differences in the execution of FMS (Barnett et al., 2016; Eather et al., 2018), with girls performing lower in object control skills, involving catching, kicking, and throwing, in comparison to boys (Barnett, et al. 2016; Butterfield et al., 2012). This emerging evidence suggests that children are making the transition to adolescence without acquiring basic movement skill proficiency (O'Brien et al., 2016a, 2016b; Belton et al., 2014). The fact that

FMS proficiency is consistently low among Irish adolescents (O'Brien et al., 2016a, 2016b; O'Brien et al., 2017) indicates the importance of early intervention programmes in Irish primary schools, particularly for the female youth population.

Examining girls' PA perceptions, motivations, barriers, and psychological correlates (for example, self-efficacy and attitudes) through in-depth qualitative research allows for greater understanding of the rationale behind their PA-related choices (Whitehead & Biddle, 2008). Underlying motivators and barriers to whether an individual engages in PA is important to investigate (Hoare et al., 2017). Current research on female youth reveals that having fun (Pawlowski et al., 2014; Visek et al., 2015), keeping fit, and being with friends (Carlin et al., 2015) are key motivators for female youth PA participation. Other studies have shown that internal factors such as confidence, pressure, embarrassment, or negative self-belief, lack of female role models, and other hobbies or commitments that result in a lack of time or preference for sport are the main barriers for female youth (Wetton et al., 2013). Similarly, the weather (Pawlowski et al., 2014), access to facilities, and lack of enjoyment have also been shown to be key barriers among female youth.

A variety of theories have been employed to explain and predict the adoption of an active lifestyle and PA participation in children and adolescents (Inchley et al., 2011). Among the well-researched theories include Icek Ajzen's Theory of Planned Behaviour (TPB; Ajzen, 1988), stating that behavioural intention to be physically active is influenced by attitude, subjective norms, and perceived behavioural control (PBC). In the present study, self-efficacy (SE) and attitude represents the positive or negative evaluation of engagement in MVPA, whereas subjective norms reflect perceived social pressures to perform MVPA. PBC refers to resources and obstacles (barriers) that facilitate or impede engagement in MVPA behaviour. Research has shown that PA and self-concept are connected in different ways (Planinšec et al., 2004). Furthermore, to engage young girls more effectively, the factors which influence PA behaviour must be better understood (Inchley et al., 2011).

Although previous studies have demonstrated that highly skilled children are significantly more active than children with lower levels of FMS proficiency in school and community-based PA interventions (Morgan et al, 2013), little is known regarding the influence of FMS on female children's PA in an Irish OYS context.

Such a context includes the Gaelic4Girls programme: an emerging intervention strategy. Furthermore, it is not yet fully understood how OYS in female youth could be optimised to facilitate continued participation and increased PA, specifically in a female sport context such as Ladies Gaelic Football (a team field sport that is popular in Ireland and organised through the Ladies Gaelic Football Association). The Gaelic4Girls (G4G) research project is part of an existing national Ladies Gaelic Football (LGF) programme in Ireland for 8–12-year-old female adolescents. Based on results of previous literature, the potential impact of developing FMS through OYS may be important for promoting long-term health-enhancing PA across childhood.

The purpose of this study was to investigate the PA patterns, FMS proficiency, and psychological factors influencing PA in 8–12-year-old Irish girls. The second purpose was to investigate differences in psychological correlates by physical activity pat- terns. Results of this current study will aim to inform the development of a multi-component, community-sports-based Irish intervention, specifically identified as G4G. At present, there is a specific gap in the literature, as no other study has assessed PA levels and psychological factors of PA and FMS, both quantitatively and qualitatively, as part of an existing Irish childhood female sports-based development programme.

### **5.4 METHODS**

### 5.4.1 Overview of the study

Cross-sectional empirical baseline data were collected and analysed to inform the design and development of a larger multi- component, community-sports-based physical activity intervention (G4G) aimed at increasing PA participation for Irish female youth for ladies' football clubs in Ireland. Baseline data for the study (initiated in October 2015) was gathered over an 8-week period from March to May 2016, which specifically included PA (accelerometry and self-report), FMS, focus group (FG) discussions, and anthropometric characteristics (height and mass). Full ethical approval for this cross-sectional, mixed-methods study was granted by the Institutional Social Research Ethics Committee at University College Cork in 2016.

Parents and children were informed about the full study prior to the physical and psychological data collection. Informed assent for participation was granted by all participants and written consent was obtained from their parent(s)/ guardian(s) and the school principals prior to participation; all participants were free to withdraw from the research at any stage.

### 5.4.2 Participants and setting

A convenience sample (based on the researcher's proximity to the schools) of crosssectional data were collected from Irish female youth (n = 331) as part of the baseline study protocol. Eligible primary school pupils enrolled between second and sixth class (8– 12 years) from three, all-female, rural and rural/urban schools were invited to participate, as presented in Figure 5.1. A representative social economic status demographic was selected, with the inclusion of one school identified as DEIS (delivering equality of opportunity in schools). A total of 443 participants from the three schools were originally invited to participate in the study, with full consent received from 358 participants (81% of total sample). Of the participants, 20 children were in second class (6%), 57 were in third class (16%), 100 were in fourth class (28%), 97 were in fifth class (27%), and 84 were in sixth class (23%). In total, 275 female participants (with a M age of 10.92±1.22 years) had full data available for the PA questionnaire (PAQ), 228 had full data for the anthropometric and FMS assessment, and 37 participants had full data for the FG discussions, as presented in Figure 5.1.



Figure 5.1: Study design and flow of baseline data collection (N=344)

### 5.4.3 Measures/Instruments

### 5.4.3.1 PA self-report assessment

MVPA was assessed using a modified version of the Physical Activity Questionnaire for Older Children (PAQ-C) (Kowalski, Crocker, & Donen, 2004). Studies have established the reliability and validity of the 7-day recall on children (Moore et al., 2007). The PAQ-C for this study included 15 physical activities, 10 leisure/free-time activities, activities in school (physical education), transport activities (walking to and from school), and other activities. The participants were told to recall what activities they had engaged in the previous 7 days and how many times and number of minutes they participated in each of these activities.

Habitual PA was also assessed using two questions from the Physician-based Assessment and Counselling for Exercise (PACE) questionnaire: How many days in the last week (PACE 1) and in a usual week (PACE 2) does the subject do at least 60 min of physical activity? The PACE questionnaire presents a test-retest reliability assessed by the intraclass correlation coefficient (ICC) of 0.77 (López Sánchez et al., 2016), and, due to its simplicity and ease of understanding, the PACE questionnaire was suitable for the 8–12-year-old cohort.

Questions on psychological correlates of PA (self-efficacy, physical selfworth, PA attitudes, perceived benefits and barriers to PA participation, subjective norms, behavioural control, and intentions) were all taken, with scales intact, from the FifeActive survey (Rowe & Murtagh, 2012). Participants' SE was assessed using a modified version of the Children's Physical Activity Self-Efficacy Survey, an 8item scale modified by Sherwood et al. (2004). Physical self-worth was assessed using a global physical self-worth subscale from the Child and Youth Physical Self-Perception Profile (taken from Physical Activity in Scottish School age children [PASS]) (Inchley et al., 2008). Participants' physical performance self-concept was assessed using the Athletic Competence subscale taken from the Self-Perception Profile for Children (Sherwood et al, 2004). Participants' perceived PA motivators and barriers were assessed using a modified version of the Physical Activity Outcome Expectancies Scale (Taylor et al., 2004) and PASS, respectively (Inchley et al., 2008). PA attitudes, subjective norms, behavioural control, and intentions were assessed using the scale from Rhodes and Smith's (2006) questionnaire. In the

current study, the Cronbach alpha coefficient for all measures were calculated, and all scales have good internal consistency (see Table 5.2).

#### 5.4.3.2 Anthropometric measurements

As reported in O'Brien et al. (2016b), mass was measured to the nearest 0.1 kg using Seca 761 dual-platform (Chino, CA) weighing scales. Height was measured to the nearest 0.1 cm using a portable stadiometer (Leicester portable height scale; Marsden, Rotherham, UK). Body mass index (BMI) was calculated (n = 228) using the following equation: weight (kg)/height (m)<sup>2</sup>. Children were classified as normal weight, overweight, and obese according to gender and age-specific BMI cut-off values proposed by the Cole and Lobstein (2012) classification. Subsequently, underweight/normal weight, overweight, and obese subjects were grouped into three weight cohorts.

### 5.4.3.3 Fundamental movement skills (FMS) assessment

The FMS proficiency of seven motor skills were assessed (n = 228) in conjunction with the behavioural components from three established instruments, namely the Test of Gross Motor Development (TGMD; Ulrich, 1985), Test of Gross Motor Development-2 (TGMD-2; Ulrich, 2000), and the Get Skilled Get Active (NSW Department of Education and Training, 2000). Four skills (run, stationary dribble, kick, and catch) were assessed using the TGMD- 2. The remaining three skills included the skip and the vertical jump, which were both assessed using the TGMD (Ulrich, 1985), and balance, which was assessed using the Get Skilled Get Active Manual (NSW Department of Education and Training, 2000). Each of these instruments and their associated protocols have established validity and reliability in children and are designed to give an objective measurement of gross motor skill proficiency. The four locomotor skills (run, skip, vertical jump, and balance) and three object-control skills (stationary bounce, kick, and catch), all of which are culturally relevant to ladies football and deemed important to Irish sporting culture (Woods et al., 2010), were assessed during a 3-hr timetabled block.

Participants performed the skill on three occasions, including one familiarisation practice and two performance trials, as reported in previous Irish

adolescent movement skill data collection (O'Brien et al., 2016a, 2016b), and were recorded using a digital video camera (3 × Canon type Legria FS21 cameras; Canon Inc., Tokyo, Japan) to allow for greater measurement scrutiny and accuracy of measurement precision (Okely et al., 2004) during analysis. The FMS scoring process was completed at a later date by the principal investigators. Each skill comprises three to six performance criteria across the seven selected FMS, which are scored as either present (1) or absent (0) and scored over two test trials. Scores for each child were calculated by totalling the correctly performed criteria for two trials for each skill (i.e., if a skill comprises three performance criteria, the score range is 0–6). The maximum subtest scores for locomotor and object control skills were 30 and 22, respectively, with a minimum of 0. The sum of both subtests yielded the total gross motor skill score (total FMS). Raw scores were used in the data analysis as this is recommended for research purposes (Ulrich, 2000).

Skill mastery was determined when the child correctly demonstrated each performance criteria of the skill on both trials. Once data collection was completed, the principal investigators were required to reach a minimum of 95% interobserver agreement for scoring all seven FMS. If children displayed correct performance on all or all but one skill component, they were classified as having achieved mastery or near mastery, respectively, for that skill (Okely & Booth, 2003). Near mastery was defined as correct performance of all components but one on both trials (Van Beurden et al., 2002).

### 5.4.3.4 Focus groups

FG interviews were used to explore students' perceptions of what it means to be healthy and physically active, and to identify their motivators and barriers to PA and ladies football participation. A semi-structured interview guide, using questions designed by the research team, was developed (see Table 5.1). Six FG interviews (two FG interviews in each school with six participants in each FG) were conducted.

### **5.4.4 Procedures (field training and data collection)**

Prior to data collection, six field staff underwent a rigorous and robust field researcher training workshop in the measurement protocol associated with FMS, self-report questionnaires, FG discussions, accelerometery, and body composition. This involved an objective criteria-informed process to ensure field staff were consistent in the gross motor skill and movement tasks. In an effort to ensure consistency and accuracy, a protocol manual, which included specific instructions for conducting all assessments, was also developed and used by research assistants to standardise procedures and for quality assurance. In terms of the research rigor associated with school-based measurements, the leading investigators for the G4G study are qualified post-primary specialist physical education (PE) teachers, as recognised by the Teaching Council of Ireland.

The baseline data were collected on participants within their class groups (max n = 30) during specific school visits. Objective measurements, such as FMS, accelerometry, and body composition were carried out during a 3-hr timetabled block, while subjective self-report measurements and FG interviews were taken during a specific school visit, in a supervised classroom, and computer lab, respectively. During FMS data collection, one trained field staff member provided approximately every eight participants with an accurate demonstration of the FMS to be performed. Procedures outlined in the respective examiners' manuals were closely adhered to. Self-report data were collected on participants (n = 275) in their class groups, with a ratio of one researcher to 10 students, for questionnaire completion. The study was briefly explained, and instructions were provided on how to complete the questionnaire. All questions were completed using the online tool SurveyMonkey (San Mateo, CA), with an ID number assigned.

Thirty-seven participants were randomly selected in the three all-female schools, following the completion of the self-report, anthropometric, and FMS data collection. FGs occurred in a school classroom, under supervision, and lasted 45–50 min each, on average. Prior to FG commencement, all participants and their parents/guardians received and signed a consent form and a plain language statement providing details of the research. Girls from second to sixth class were invited to participate in the FG study by way of a letter, which explained the purpose of the

study. Interested girls were given a letter of introduction and a consent form for their parents. Those girls who returned a completed parental consent form participated in the FG at a later time. The girls were also asked to sign a consent form on the day of the FG allowing for the sessions to be audio taped. All FGs were conducted by the leading investigator, who has training and experience facilitating FGs. The FGs were conducted at school in a meeting room, with chairs arranged in a circular manner around a boardroom style table, as suggested by (Kitzinger, 1995). Each session was also attended by an assistant who took comprehensive notes in order to aid the subsequent transcription process, as suggested by Krueger & Casey (2000). Patton (2002) technique of probing through follow-up questions (i.e., detail probes, elaboration probes, clarification probes, and contrasting probes) was employed to gain a mutual understanding between the participants and the lead researcher during interviews. Participants were reminded that they could withdraw from the interview at any stage and that all recordings would remain confidential. The FG interviews were recorded by dictaphone and were transcribed verbatim.

Each of the six FG interviews were conducted by two researchers: a facilitator and a note taker. The facilitator's role was to guide the FG, stimulate interaction among students toward the theme, oversee group discussion, and encourage all students to respond. The note taker kept a record of the discussion as it evolved to add details for instances where the recording was not audible (Tannehill et al., 2015). The FG commenced with a brief introduction by the moderator, which included a short discussion on how the FG would work. The moderator then asked an opening question, namely, "What does being healthy mean to you?" The opening question was designed to be easy to answer, with the purpose of encouraging the participants to talk and feel comfortable in the group setting (Krueger & Casey, 2000). The first key question (see Table 5.2) was then asked, with girls being allowed to talk until no more views were expressed. The moderator then followed up the points the girls had made, using follow-up (e.g., "Can you tell me more about that?") or clarifying questions (e.g., "Was that in an all-girls team or a mixed team?") to explore issues at a deeper level. Throughout the duration of this baseline data collection, participants were assigned ID numbers for anonymity purposes. Trustworthiness of data was established through data triangulation, peer debriefing,

and dependability. Data triangulation was implemented by examining data from different respondents but collected using the same method.

The research team looked for patterns and contradictions investigating the same problem, which brought about different perceptions of inquiry, helping to reduce bias while strengthening the integrity of the FG findings. The research team sought support (peer debriefing) from other academic professionals (staff within the department) to provide scholarly guidance. Such investigators looked at the study back- ground information, data collection methods and process, data management, transcripts, data analysis procedure, and research findings (Pitney & Parker, 2009). Dependability was implemented through the code recode strategy (code agreement), whereby the researcher coded the same data twice, giving 1- or 2-weeks' gestation period between each coding. The results from the two codings were compared to see if the results were the same or different (Ary et al., 2010). The coding results of this current study were in agreement and therefore enhanced the dependability of the qualitative inquiry.

Phase	Focus Group Activity/Questions	Purpose							
Opening – Ice-Breaker	Participants were asked to draw a picture of what being healthy means to them ('Draw a picture	Informal conversation ice-							
Activity	of yourself at the weekends surrounded by all of your favourite physical activities and objects on	breaking activity used to create a							
	this sheet'). Participants also coloured the sheet.	relaxing environment							
1.									
Health-related PA	• What does being healthy mean to you?	Discussion conducted to reveal							
	• What do you think makes you healthy?	participant's understanding of							
	• What do you think PA is?	PA/Health.							
	• Do you know how much PA per day you should be doing? Prompt: How many minutes								
	• How do you think PA affects your health?								
	PA definition/explanation & note-taker summarised key words/themes/comments from								
	this section. Facilitator will ask do the group have further comments they would like to								
	make.								
2.									
PA motivators/	• What are your favourite types of PAs/sports?	Discussion conducted to identify							
hannians and proformed	• What PAs/sports do you take part in?	preferred PAs, motivators and							
activities	• What kind of things do you do when you play outside?	barriers that influence children's							
	• Where/when/with whom do you take part in these activities?	PA/sport participation.							
	• Why do you take part in PA/sport?								
	• Who or what helps you to take part in PA/sport?								

### Table 5.1: Procedure & questions used for the G4G baseline focus groups (n=6)

• Why don't you like taking part in PA/sport?

٠	Is there anything stopping you from becoming more active than you already are?
	Reinforced PA explanation and note-taker summarised this section. Facilitator asked the
	group if they had further comments to add.

3.	• Is there anything that would help you to take part in more PA/sports?	A discussion on what								
PA opportunities &	• Is there anything that would help you to enjoy PA/sport more?	participants deemed important in								
possible PA enhancers	• Do you think some people have more opportunities to play sports than other people?	influencing their participation								
	Why might that be the case?	in, and barriers to, PA and sport								
	• Do some people give up sports? Why, in your opinion does this happen?									
	<ul><li>What do you think we could do to get more people playing sports?</li><li>If you could play any sports in the world what would they be and why?</li></ul>									
	Reinforced PA explanation and note-taker summarised this section. Facilitator asked the									
	group if they had further comments to add.									
4.	• What do you know about LGF?	Discussion based on								
LGF/OYS	• Do you play LGF? (club/school)	participant's knowledge of LGF,								
environment	• Do you like playing it? What's your favourite part?	and possible strategies to promote LGF/sport was then conducted								
	• What don't you like about LGF?									
	• What would encourage you to play LGF?									
	• What would stop you from playing LGF?									
	• What do you think we could do to get more girls playing LGF?									

Abbreviations: LGF = Ladies Gaelic Football (popular Irish female sport); OYS = organised youth sport; PA = physical activity

### 5.4.5 Data analysis

The FMS and PA data set were analysed using SPSS version 20.0 for Windows (IBM, Armonk, NY). Participants with incomplete data for a given variable were excluded from analysis of this variable specifically. The number of days that were self-reported to meet the 60-min PA guideline were analysed descriptively using means, standard deviations, and proportions. Students were categorised as low active (meeting guidelines on 0, 1, 2, or 3 days a week), moderate active (meeting on 4 or 5 days), or high active (meeting on 6 or 7 days) (Rowe & Murtagh, 2012). Descriptive statistics and frequencies for all FMS, PA, and psychological variables were calculated. All statistical assumptions were tested for normality, outliers, and homogeneity of variance. A one-way between-groups ANOVA was used to explore the impact of PA grouping (low, moderate, or high active) on all psychological correlates. To reduce the risk of Type I error, a Bonferroni adjustment to the alpha level was made by dividing p < .05 by 8 (number of variable comparisons); thus, the alpha level was set to p < .006. In instances where significant main effects were found, post hoc comparisons were carried out using the Tukey honest significance difference (HSD) to determine where the differences occurred. To calculate effect sizes, Cohen d values (1988) were applied, with d=0.2 representing a small effect size, 0.5 representing a medium effect size, and 0.8 representing a large effect size.

FG audio recordings were transcribed verbatim. The leading author doublechecked each transcript against the original recording for robustness and accuracy prior to analysis. Transcripts were analysed using thematic analysis procedures (Braun & Clarke, 2006) involving coding, categorisation, and summarizing (Elo & Kyngäs, 2008). The FGs were coded and, under each heading, the coded comments were clustered into categories based on similar content. Then, a thematic analysis was conducted to produce an in-depth description and understanding of the transcripts; phrases from the transcripts were highlighted and grouped, from which themes and subthemes emerged (Neergaard et al., 2009; Pawlowski et al., 2014).

### **5.5 RESULTS**

Of the 331 participants involved in this study, the mean age was 10.92 years (SD = 1.22). Participants had an average BMI of  $19.03 \pm 3.41$  kg/m2 (n = 228), with 73.2%

categorised as underweight/ normal weight, 20.2% overweight, and 6.6% obese, using the Cole and Lobstein (2012) classification for girls.

### 5.5.1 Physical activity

Self-reported PA data showed that 11.6% of participants are meeting the 60-min guideline for 0–3 days a week (low active), with 41.4% meeting the guideline 4 or 5 days a week (moderately active), and the remaining 47% of participants meeting the guidelines 6 or 7 days a week (high active). The percentage of participants meeting the 60-min PA guideline on all 7 days was 28.2%. Self-reported PA participation data found that the largest motivators were "being healthy" (91.0%), "having fun" (93.4%), and "making new friends" (89.8%). Conversely, the largest participation barrier to PA was "not being as good as others at it" (37.4%). Other participation barriers to PA include "difficulty in accessing facilities to do PA" (26.6%), "fear of injury" (25.4%), and "not having enough time" (22.6%).

### 5.5.2 Fundamental movement skills

Only six participants (2.63%) possessed complete mastery level across all sevenobject control and locomotor skills. Overall, the vertical jump and the skip were the poorest performed skills across the cohort, where 38.1% and 39.1%, respectively, achieved mastery (see Figure 5.2). The best performance was for the kick (object control), where 63.7% and 25.7%, respectively, achieved mastery and near mastery. Mastery levels for all seven skills are reported in Figure 5.2. The overall mean FMS score of participants was 48.86 (SD = 5.61), out of a possible score of 62.



Figure 5.2: Percentage of advanced skill proficiency of 8-12-year-old participants (*N*= 228).

### 5.5.3 Psychosocial correlates of PA

A one-way between-groups analysis of variance was conducted to explore the impact of physical activity groupings (high, moderate, and low active participants) on levels of self-efficacy, as measured by the PASES scale. Participants were divided into three groups according to their PA level (Group 1: low active; Group2: moderately active; Group3: highly active). There was a statistically significant difference at the p < .004 level in self- efficacy scores for the three PA groupings: F(2, 255) = 17.21, p = .00. The actual difference in mean scores between the groups had a medium to large effect size, using the Cohen d value of 0.74. Post hoc comparisons using the Tukey HSD test indicated that the mean score for high active participants (M=38.26, SD = 3.89) was significantly different from low active (M=35.53, SD = 5.33) and moderately active (M= 33.84, SD =4.53) participants. Of the remaining psychological variables assessed (see Table 5.2), PA selfconfidence, perceived self-worth, physical performance self-concept, PA motivators, PA barriers, and PA attitudes demonstrated significant main effects for PA grouping, with small to medium effect sizes in each case using Cohen d. Post hoc comparisons, using the Tukey HSD, showed that in each of the aforementioned variables, bar PA motivators, the mean scores for the low active group were significantly reduced, when compared to the high active group. Significant differences in mean scores for PA groupings are shown in Table 5.2.

Variable	Ν	Low	Mod	High	OMS	Main Effect	d	α	Post hoc M Differences	
Self-efficacy	258	33.84*	35.53*	38.26*	37.84	F(2, 555) = 17.21, p = .004	0.74	.73	Low & high	p < .047
		(25)	(60)	(173)					Mod & high	p < .047
Self-confidence	257	46.60*	51.54*	61.28*	58.00	F(2, 554) = 5.78, p = .002	0.41	.85	Low & high	p < .047
		(25)	(59)	(173)					Mod & high	p < .015
Barriers	258	32.72*	33.47*	36.25*	35.15	F(2, 555) = 6.63, p = .002	0.46	.84	Low & high	p < .025
		(25)	(60)	(173)					Mod & high	p < .01
PA motivators	253	32.12	32.03*	33.66*	33.11	F(2, 550) = 5.73, p = .004	0.41	.74	Mod & high	p < .007
		(25)	(59)	(169)						
Attitudes	255	13.88*	14.15*	14.92*	14.61	F(2, 552) = 6.16, p = .002	0.46	.82	Low & high	p < .024
		(25)	(59)	(171)					Mod & high	p < .017
Perceived self-worth	258	17.72*	18.95*	20.23*	19.59	F(2, 555) = 5.80, p = .003	0.41	.66	Low & high	p < .009
		(25)	(60)	(173)						

 Table 5.2 Mean (SD) Across physical activity grouping with significant main effects and effect size of one-way ANOVAs

Physical performance	257	16.58*	17.35*	19.42*	18.68	F(2, 554) = 17.3, p = .000	0.47	.73	Low & high	p < .000
self-concept		(24)	(60)	(173)					Mod & high	p < .000
Subjective	255	16.24*	17.44	18.27*	17.84	F(2, 552) = 3.92, p = .021	0.35	.81	Low & high	p < .027
norms/social support		(25)	(59)	(171)						

Abbreviations: Mod = moderate; OMS= overall mean score; PA = physical activity. \*p < .05.

\*p < .05.

### 5.5.4 Focus groups

Three key themes emerged from the FG data that were pertinent to students' perceptions of health, and what participants deemed important in influencing their participation in, and barriers to, PA and sport. These themes were: (a) motivators: PA/sport is fun; (b) barriers: lack of time, distance, weather, and friend-related factors; and (c) factors influencing PA/sport engagement and disengagement (dropouts).

### 5.5.4.1 PA motivators: fun, enjoyment, and friends

The key emergent themes in relation to motivators for PA and sport were fun, enjoyment, and a sense of social satisfaction. Examples of this include a participant stating: "I love doing it (physical activity/ sport) because it just makes me feel really happy; it's just very enjoyable". Similarly, other students commented on the rationale behind partaking in PA and sport: "so I will be stronger and independent" and "you can learn new things about the sport". Interestingly, the influence of friends was a common recurring theme, with the social aspect "to make friends" attributed by participants as a PA motivator, with participants stating: "I love to socialize with everybody on the team", and "I like seeing my friends there".

5.5.4.2 PA barriers: lack of time, fear of injury, weather, and friend- related factors In relation to the barriers to PA participation, insufficient time to participate (particularly on parent's side, coupled with too many other commitments) was identified as the main barrier among participants. One student stated: "I have a lot of things going on during the week, and I just don't have enough time". Similarly, another participant commented: "Like, my mom and dad won't have time to bring you there because they have to bring like my brothers and sisters around too". Several participants voiced concerns in relation to fear of injury as a barrier to PA and sport participation, with one participant stating: "Some people get a bit too competitive and that kind of like makes you feel not that confident, like, some people just get a bit too into it and yeah know pushing and shoving.... and there might be a couple of injuries and competitiveness". Other participants commented on the weather as a barrier; responses included: "I don't really like being wet" and "like, if it was too hot, you would be tired and mightn't want to do it". Interestingly, friends acting as barriers negatively impacted self- esteem, which was another emergent response among participants. One student indicated: "like, you'd see other people doing gooder (at sport), and then you would feel quite bad about yourself, because it's just not your thing I suppose".

5.5.4.3 Factors influencing PA/sport engagement and disengagement (dropouts) Findings illustrate that positive outside encouragement/support from family, friends, and coaches act as models of motivation, encouraging PA engagement among the female participants. Supporting statements included: "she's very encouraging", "she [teacher] thinks it's important to, like, get active" and 'they'd (sports role models) give you tips on how to become better and you'd take them". Contrastingly, participants cited that negative relationships with coaches promote drop-out from the PA/sport, or participants are less likely to stay engaged. This is depicted with one student stating, "if there's a really mean coach I'd probably stop the sport because they'd give out to you", with another participant's reason for disengaging as, "too much competitiveness amongst girls". In addition to this was a lack of enjoyment ("not fulfilled, bored, I just thought it was way too strict, like, it wasn't fun") along with having a dislike for the teacher/coach ("the teacher was mean") and a lack of support/encouragement from coach/teacher ("you feel that you're not able because they are giving out you so much".

### **5.6 DISCUSSION**

Results of this study indicate that a large number of Irish female youth are insufficiently active to benefit their current and future health. The percentage of participants meeting the 60-min MVPA guidelines on all 7 days was 28.2%. These low levels of PA participation are relatively consistent with other studies on a national Irish level (Layte et al., 2011; Harrington et al., 2014). Although higher than the 16% reported for a similar age group in the Irish Health Behaviour in Schoolaged Children (HBSC) study (Gavin et al., 2015), comparable findings were reported for other European countries (Inchley et al., 2017; Verloigne et al., 2012) and in New Zealand (Maddison et al., 2016), which are in line with this study's findings. Given the consistently reported age-related decline in PA participation for girls (Bradley et al., 2011), the need for intervention to address these low levels reported for young girls aged 8–12- years-old is palpable.

The majority of female youth in this study (97.74%) failed to reach a level of mastery across key FMS, indicating that basic movement skill proficiency among pre-adolescent Irish youth is low. These observed low findings of FMS proficiency are consistent with recent Irish adolescent research (O'Brien et al., 2016a). Results from the present cohort indicate that only six participants were fundamentally competent across all seven object-related and locomotor skills, despite children having the developmental capacity to master these skills by 6 years of age (Gallahue & Ozmun, 2006). Previous international research examining the FMS proficiency of female youth supports these low levels of FMS mastery (Bryant et al., 2016; Cliff et al., 2012; Foweather, 2010). The poorest performed skill across the cohort was the vertical jump, where only 38.1% achieved mastery. This finding for the low prevalence of vertical jump proficiency among female youth participants is in line with the Physical Education, Physical Activity and School Sport (PEPASS) year 2 study by Foweather (2010), specifically across 10–11-year-old children, where only 16% of female participants obtained skill proficiency in the vertical jump. It is critical that children develop competence in locomotor skills, as they appear to be a key determinant of children's PA, physical fitness, and body fatness (Foweather, 2010). It would appear that Irish female youth may be engaging in sport-specific skills without learning the prerequisite criteria for basic jumping movement patterns. This highlights a further implication, indicating that these FMS are not developed

during PE or in the sporting environment (O'Brien et al., 2016a). Furthermore, this points to the targeting of improving FMS proficiency as a strategic supplement of future school-based and community interventions in the promotion of PA in female youth. A strategic step for the future G4G intervention in Ireland may include the provision of opportunities for young girls to execute FMS through fun games and developmentally-appropriate activities within school and community settings, as these environments have been empirically shown to significantly improve FMS proficiency in youth (Morgan & Barnett, 2013).

Consistent with other recent studies (Stellino & Sinclair, 2013; Verloigne et al., 2016), the analysis of psychological variables demonstrated differences and significant main effects with female children's PA levels. This was evident in the current study for self-confidence, perceived barriers, attitudes, and perceived performance self-concept (PPSC), all of which exhibited small to medium effect sizes, as shown in Table 5.1. Self-efficacy, interestingly, was the only variable from the analysis which exhibited medium to large effect sizes from the PA groupings. A significant difference in the mean scores were found in the attitudes of participants toward PA, self-efficacy, self- confidence, perceived barriers, and PPSC between low, moderately, and high active participants, with moderately active participants scoring significantly lower than their high active counterparts in each case and low active participants scoring significantly lower than their high active participants in each case. This is relatively consistent with other studies (Belton et al., 2014; Bauman et al., 2012), further implying that levels of perceived self-efficacy, selfconfidence, attitudes, PPSC, and barriers differ among low, moderately, and high active female youth, which may impact future PA intention and participation.

These variables (self-efficacy, self-confidence, attitudes, PPSC, and barriers), each of which demonstrated differences and significant main effects with female children's PA levels in the current study are categorised as important factors for the TPB. The TPB posits that behavioural intention to be physically active is influenced by attitude, subjective norms, and PBC (Ajzen & Madden, 1986). Current research suggests that children who express strong feelings of control and have high motivation to engage in PA are likely to have high MVPA levels (Wang & Wang, 2015). In the current study, self-report and FG findings indicate that social pressures, namely parents, coaches, and peers, were reported as participatory barriers, and these

findings are in line with previous research on female youth (Verloigne et al., 2016; Perry et al., 2011). Regarding PBC, resources facilitating (positive encouragement/praise) or impeding engagement (negative coach relationship) among 8–12-year-old female youth are deemed important findings in the context of this G4G study, and are consistent with recent studies (Battaglia et al., 2017; Fenton et al., 2017; Laird et al., 2018).

The FGs in this current study identified several issues which will be important to consider in the design of the G4G intervention. Friendship and the influence of peers were consistent themes both for influences on current activity and suggestions on how to promote participation ("I love to socialize with everybody on the team", and "I like seeing my friends there".) This theme emerged irrespective of current PA levels and underlines the importance of peers at this stage of the life cycle (Fitzgerald et al., 2012). The influence of friends and the importance of having someone to participate with have been shown to be key factors in young female participation in PA (Carlin et al., 2015; Fitzgerald et al., 2012; Sawka et al., 2013).

Additionally, a systematic review of 106 studies on PA influences identified that children's PA was positively associated with encouragement from friends, friends' own PA participation, and engagement with friends in PA (Maturo & Cunningham, 2013). Feeling connected to others, such as parents, teachers, and peers, can help cultivate the development of intrinsic motivation (Crespo et al., 2013), particularly for female youths. Similarly, Dudley et al. (2015) posits that one of the primary reason's youth participate in PA is for the sheer enjoyment they experience while moving in social situations and interacting with their peers. This suggests, particularly for this current G4G study, that it is important to emphasize the social aspect within PA interventions for young girls, specifically by focusing on doing physical activities and skills with friends and promoting the message to encourage each other to participate in these activities (Voorhees et al., 2005).

In addition to this, participants cited the important role of parents within their PA and sport involvement. Parental support and direct help from parents have previously been correlated with PA in female youth (Carlin et al., 2015; Eddolls et al., 2016). Given the continued influence parents play as children move into adolescence (Van Der Horst et al., 2007), it would be prudent for the G4G

intervention to consider active peer and parental involvement, as these reinforcing factors may positively influence girls' PA participation (both directly and indirectly) through the internalisation of several personal psychological variables, such as self-efficacy to be active and the perceived benefit of having fun (Verloigne et al., 2016).

The community-based Gaelic Athletic Association (GAA) club environment in Ireland is also a key opportunity for the promotion and provision of PA opportunities through sport, specifically for LGF participation. Previous crosssectional studies indicate that youth sports club participants tend to have higher levels of PA, increased self-esteem, and positive peer relationships (Basterfield et al., 2015; Geidne et al., 2013; Silva et al., 2013). Considering that LGF is a favourite form of sport among female youths and adolescents and is currently recognised as one of the fastest growing female sports in Europe (LGFA, 2016), it is indeed a desirable environment in which girls can engage.

The G4G programme, through a sports-based community environment, is a key opportunity to intervene, and may overcome many barriers to female youth participation. It is group-based and offers social interaction by promoting movement and skill competence in a non-competitive context. This programme provides an appropriate medium through which to increase PA participation and basic movement skill proficiency among female Irish youths. Results from the "Keeping Them in The Game" study (Lunn et al., 2013) in Ireland report that 1 in 10 girls drop out of playing regular sports during the primary to secondary school transition (12–13 years old). This is of concern, and it is proposed that the next phase of the G4G intervention will be implemented through the provision of LGF clubs and the community environment.

Preadolescent girls are an important target for PA behavioural change strategies, as this age cohort may enhance tracking into the crucial period of adolescence (Biddle et al., 2014). The quantification of intervention effectiveness for this age group of girls has not been adequately reported (Biddle et al., 2014). This suggests that the school and community centre are key organisational structures to implement this proposed intervention, supported by professionals (Murillo Pardo et al., 2014). Therefore, school and com- munity-based interventions that are multicomponent in nature, simultaneously targeting curricular, school environment,

policy, and community links, are a promising approach for promoting PA (Biddle et al., 2014). A critical challenge facing governments, schools, sports groups, and communities is to develop forms of PA that are sensitive to girls' needs and interests.

There were some limitations of the present study. First, the data relied on self-reports of PA and psychological variables. Although previous studies showed that the PA and the psychological measures had good reliability and acceptance validity, more objective measures of activity, such as accelerometer data, should have an additional value. Secondly, the use of a convenience sample is a limitation, which can lead to the under-representation or over-representation of particular groups within the sample. Another limitation of the present study is its crosssectional design and small sample size. The results do not provide causal evidence regarding relationships among actual FMS psychological correlates of PA and PA. To gain more insight in the direction of these relationships and to understand how associations among these variables may change overtime, longitudinal or experimental studies should be conducted with a bigger sample size.

Acknowledgements: The authors wish to acknowledge the help of undergraduate students from the Sports Studies and Physical Education degree programme involved in data collection. Sincere thanks also to the participants, parents, teachers, and principals from the three primary schools involved in this phase of the study. The authors would also like to express gratitude to the National Ladies Gaelic Football Association (LGFA) in Ireland for their continuous support and assistance.

### **Declaration of interest statement**

The authors declare that there are no conflicts of interest within this manuscript publication.

### 5.8 LINK TO CHAPTER 6

The unique element of this chapter was the assessment of PA patterns, FMS proficiency and psychological correlates of 8 to 12-year-old girls, as part of the baseline data. This process allowed the G4G research team to identify weaknesses in skill performance and pre-adolescent needs-driven perceptions to inform the development of the G4G intervention. Chapter 5 is directly associated and logically connected with the following overview of the G4G intervention (chapter 6). In this next chapter, the components of the G4G intervention will be outlined, described and discussed, as underpinned by Deci & Ryan (2002) Self-Determination Theory (SDT) and Stokols (1996) Social-Ecological Model (SEM). The reader will be able to identify the findings from baseline data collection (chapters 4 and 5), integrated within the research-informed G4G intervention (chapter 6).

## **CHAPTER 6**

RATIONALE AND STUDY PROTOCOL FOR THE 'GAELIC4GIRLS' PROJECT: A NOVEL MULTI-COMPONENT COMMUNITY SPORTS-BASED COACHING INTERVENTION FOR INCREASED PHYSICAL ACTIVITY PARTICIPATION OF PRE-ADOLESCENT GIRLS



Chapter 6: Rationale and Study Protocol for the 'Gaelic4Girls' Project: A Novel Multi-Component Community Sports-Based Coaching Intervention for Increased Physical Activity Participation of Pre-Adolescent Girls

### Preamble

The following article reports the 'Gaelic4Girls' (G4G) intervention protocol study; a 10 week specifically designed multi-component PA and FMS community sportsbased coaching intervention, informed by contextually relevant data (chapters 4 and 5), targeting 8- to 12- year olds, from a community sports club setting. The design of this study was guided by Deci & Ryan's (2002) Self-Determination Theory (SDT) and Stokols (1996) Social-Ecological Model (SEM). This manuscript is currently under review in the Sports Coaching Review Journal.

Manuscript submitted as follows:

Farmer, O., Cahill, K., & O'Brien, W (2020) "Rationale and Study Protocol for the 'Gaelic4Girls' Project: A Novel Multi-Component Community Sports-Based Coaching Intervention for Increased Physical Activity Participation of Pre-Adolescent Girls. *Sports Coaching Review Journal* (under review).

### Statement of authorship;

I hereby declare that I, Orlagh Farmer, am the principal author of this article. The following statements outline my contributions to the work:

- I was the principal investigator responsible for all the data collected (2017) in this chapter. I organised all stages of data collection consisting of timetabling undergraduate students, contacting all primary schools and the LGFA club involved (principals, teachers, coaches, and students), equipment preparation, data storage, data cleaning and data inputting.
- Organised the rigorous training and protocol workshop for field staff assistants which was an important component of this baseline study.
- Made substantial contributions to the conception and design of the work; the acquisition, analysis and interpretation of data for the work.

- Wrote the first draft of the full article and contributed significantly to the subsequent redrafts until submission.
- Final approval of the version to be published, and agreement to be accountable for all aspects of the work in ensuring that questions related to the accuracy or integrity of any part of the work are appropriately investigated and resolved.

### **6.1 PURPOSE OF CHAPTER**

### 6.1.1 Rationale

There are a lack of effective intervention strategies to promote PA in female youth (Anokye et al., 2018), and particularly within a sporting context (Whitley et al., 2019). In this chapter, the reader is introduced to the theorybased design protocol of the 'Gaelic4Girls'(G4G) study intervention, specifically the participant, coach (pedagogical focus), and parental (support) components, designed to increase 8 to 12-year-old pre-adolescent girls PA, FMS proficiency and psychological wellbeing. By actively engaging the G4G participant, coach, parent, guardians and community sports club in the intervention process, the G4G intervention is adhering to previously successful intervention evidence in

a youth sporting context, as underpinned by the SDT and SEM.

### 6.1.2 Contribution to the field

In terms of originality, the theoretical components of the intervention address PA, FMS, and psychological content in the promotion of skill competency, attitudes, self-efficacy, enjoyment and social support structures towards the importance of regular PA participation in OYS. The theoretical framework underpinning this study may be used to guide future theory-based community-sports based coaching interventions, targeting both PA promotion for8 to 12-year-old girls and the enhancement of coach education within a youth sport context.
## **6.2 ABSTRACT**

**Background:** This paper reports on the theory-based design protocol of the revised 'Gaelic4Girls'(G4G) study, a novel, multi-component, community sports-based coaching intervention for increasing pre-adolescent Irish girl's physical activity (PA) levels, fundamental movement skill (FMS) proficiency, and psychosocial well-being.

**Methods:** A target sample of 241 female-only participants, aged 8 to 12 years old received the revised G4G intervention. A sub-sample (n = 117; mean age =  $10.44 \pm 1.40$  years) were assessed at pre- and post-intervention phases (March-June 2017), using empirically established self-report and fundamental movement skill proficiency. The revised G4G intervention (child, coach, and parental/guardian component) was delivered once a week, for 60 minutes, over 10 weeks in a rural community sports club setting, as guided by the Social-Ecological, and Self-Determination theoretical frameworks.

**Conclusion:** The theoretical framework underpinning this study may be used to guide future theory-based community-sports based coaching interventions, targeting both PA promotion for pre-adolescent girls and the enhancement of coach education within a youth sport context.

**Keywords:** organised youth sport intervention; fundamental movement skills; psychosocial correlates of physical activity; coach education

## **6.3 INTRODUCTION**

Regular physical activity (PA) is associated with numerous physical, and psychological health benefits for children and young people aged 5 to 18 years old (Owen et al., 2017). A large body of evidence, however, consistently shows that a large proportion of children and adolescents fail to meet the current recommendation of 60 minutes of moderate-to-vigorous PA (MVPA) daily (Hallal et al., 2012). It is widely reported that girls are less physically active than boys throughout childhood (Sallis et al., 2016), and the age-related decline in PA participation, particularly from early adolescence onwards, is steeper for girls than for boys (Owen et al., 2017).

Recent systematic reviews examining the effectiveness of interventions to increase PA in children have reported only small treatment effects, specifically in terms of increasing girls' MVPA (Owen et al., 2017; Voskuil et al., 2017). Previously reported evidence would suggest that pre-adolescent girls are an important target population for PA behaviour change and are identified as a high priority group for PA promotion (Camacho-Minano et al., 2011). Evidence would support the importance of intervening during the pre-adolescent period for PA promotion, specifically as activity levels have been previously shown to track into adolescence (Biddle, 2014).

Organised youth sport (OYS) participation is one such strategy with the existing potential to increase overall PA levels in young people (Hebert et al., 2015), and for these reasons, community-sports based settings are recognised as key health settings in promoting knowledge of PA, and healthy lifestyle behaviours (Kokko, 2014). Sports club membership predicts higher levels of leisure-time PA engagement among girls (Marques et al., 2016), and OYS participation increases the likelihood of meeting daily PA recommendations for improved health (Hebert et al., 2015). Evidence suggests that an array of physical and psychosocial health benefits are associated with children and adolescents' participation in team OYS settings, such examples include improved skill development, heightened levels of muscular strength and endurance, increased self-esteem and positive peer relationships (Felfe et al., 2016). Despite these widely endorsed benefits for OYS, there is a lack of effective intervention strategies to promote PA in female youth (Anokye et al., 2018), and particularly within a sporting context (Whitley et al., 2019).

Aligned with these intervention strategies, coaches in OYS settings are prominent stakeholders who have the potential to impact the health and well-being of children and adolescents (Guagliano et al., 2014). Specifically, these coaches can be viewed as important role models who can influence players' PA developmental experiences (Santos et al., 2019), empowering and enabling participants to engage in PA (Griffiths & Armour, 2014) within an OYS context. It is, therefore, of particular importance that coaches at the grass-roots level (volunteer, part-time and full time) engage in regular coach education programmes (Côté & Gilbert, 2009) reinforcing their unique position towards children's involvement in sport (Bäckmand et al., 2010). Based on Côté and Gilbert's (2009) understanding of effective coaching, it is necessary to systematically integrate coaches' knowledge to increase athletes' confidence, competence, connections, and character, given the particular characteristics of a sport context (i.e. aims, age group, and coaching domain) (Côté & Gilbert, 2009; Flett et al., 2013).

Previous studies (Sullivan et al., 2012) have highlighted that those who participate in coach education programmes can positively improve their coaching efficacy, beliefs, knowledge, and behaviours (Trudel et al., 2010). A recent review on the effectiveness of coach education in OYS indicated that non-formal learning methods within coach education (such as continuous professional development (CPD), clinics, workshops or seminars) (Maclean & Lorimer, 2016) are more authentic, meaningful, and contextualised to a coach (Mallett et al., 2009). Despite some of the research on the effectiveness of non-formal coach education training programmes, little is known about how the acquisition of knowledge is transmitted through these types of programmes (Trudel et al., 2010). Research in practice would now suggest that there is a need for innovative coach education approaches, that can better equip coaches with the professional competencies needed to deal with the dynamic nature of coaching within OYS settings (Morgan et al., 2013).

Similar to the domains of both OYS and coach education, fundamental movement skills (FMS) have also been shown to have a positive association with youth PA participation (Robinson et al., 2015). Proficiency in a range of FMS is considered to be the foundation for an active lifestyle (Cattuzzo et al., 2016), and the primary school window (5 to 12 years old) represent the "golden years" of motor skill development (Clark & Metcalfe, 2002). In order to achieve the overall recommendation of increasing PA levels of Irish youth, previously reported national

surveillance data on childhood PA recommended that the development and promotion of FMS should be of priority (Woods et al., 2010). Existing evidence would suggest that children with better skills develop enhanced physical selfperceptions, and this enables further PA engagement, leading to greater motor competence (Robinson et al., 2015; Stodden et al., 2008). Children who fail to acquire a proficient level of FMS often lack the confidence and motivation to engage in PA (Whitehead & Biddle, 2008) and are more likely to drop out of OYS during adolescence (Woods et al., 2010). Nationally in Ireland (Farmer et al., 2018) and internationally (Eather et al., 2018), the proportion of school-aged children achieving FMS proficiency is low. Efforts to promote PA in youth, particularly among girls, alongside the professional development of coaches, may lead to a greater understanding of evidence-based strategies to improve FMS proficiency (Morgan et al., 2013).

In the last decade, there has been a gradual increase in the number of randomised controlled trials (RCT's) that have evaluated the impact of PA-based interventions on young girls (Jago et al., 2015; Sutherland et al., 2016), however, minimal evaluations of pre-adolescent girls within a community sports-based setting exist. Of those RCT's which have been undertaken, results have found small effects when PA was measured objectively (Okely et al., 2011). The need to promote higher habitual MVPA engagement amongst children (particularly girls), in parallel to the provision of coach education in the youth sport context, has been underlined as prudent strategies (Fenton et al., 2017).

In light of the previous considerations, this paper reports the study protocol and methodological procedures used in the design, development, and implementation of the 10-week G4G community sports-based coaching intervention. The novel coaching strategies and pedagogies utilised within G4G aim to increase overall PA levels, FMS proficiency, and psycho-social well-being among pre-adolescent girls. The Gaelic4Girls (G4G) research project is part of an existing national Ladies Gaelic Football (LGF) coaching programme in Ireland, designed for 8 to 12-year-old girls. LGF is a favourite form of PA engagement among Irish youth (Woods et al., 2010), and as a national 'Gaelic' game, the sport is a desirable activity in which preadolescent girls can participate.

In this study, the G4G coaching intervention seeks to highlight the novel coaching pedagogies from an intervention design and dosage perspective for

heightening young girl's participation in LGF in a rural community sports club setting in the County Cork region of Ireland. The G4G project specifically targets pre-adolescent girls and OYS coaches, intending to create a community club sporting environment for LGF, which encourages and supports the full involvement of girls in PA. This proposed research objective will be targeted through a specifically designed multi-component community-sports based coaching intervention, informed by contextually relevant data (Chapters 4 and 5). The purpose of this paper was to provide a detailed insight into the theory-based design protocol of the G4G intervention study, namely the G4G participant, coach, parent/guardian, and intervention components.

## **6.4 METHODS**

#### 6.4.1 Study design

The G4G intervention, a 10-week multi-component PA and FMS community sportsbased coaching intervention targeted 8- to 12- year olds, from five girls' only primary schools, all of which were located in a proximal location to the surrounding rural LGF club. Assessment measures were conducted at pre-intervention [March 2017] and were repeated at post-intervention [June 2017] phases. Ethical approval for this study was obtained from the Social Research Ethics Committee of the researchers' institution (University College Cork) in March 2016.

## 6.4.2 Recruitment, setting, and participants

Convenience sampling was used to recruit from the five girls' only primary schools (rural and rural/urban) in the County Cork region. All schools were in close proximity to the local designated LGF club and the research team. To be considered eligible for this study, participants (girls only) were formally enrolled between the years of second to sixth class of primary school (aged 8- to 12- years old) and had expressed an intention to attend the programme for its full duration. Informed parental consent and child assent were the requirements for eligible participation in this study. Information sheets and consent forms were administered to students who expressed an interest in taking part in G4G. Informed assent for participation was

granted by all participants, and written consent was obtained from their parent(s)/ guardian(s) prior to the physical and psychological data collection measurements. For coaches to be eligible to partake in the programme, recognised LGF coaching credentials from the National Ladies Gaelic Football Association (LGFA), and previous experience coaching girls within the existing community sports-based setting were required; all participants were free to withdraw from the research at any stage.

Prior to the commencement of the G4G intervention study, the leading researcher emailed the school Principal from all five primary schools, calling for expressions of interest for the lead researcher to visit the school, and specifically to inform the girls about the G4G programme. Subsequent to the granted approval from school Principals, the lead researcher, along with the two G4G coordinators (head club coaches from the selected LGF intervention club) visited the five primary schools, where a full outline of the G4G intervention and the associated data collection measurements were provided.

Three hundred and seven participants from the five primary schools were invited to participate in this study. Two hundred and forty-one participants took part in the 10-week G4G intervention, with 188 participants providing fully available date at both pre and post data collection time points (78 % of the total sample).

## **6.5 GAELIC4GIRLS INTERVENTION**

## 6.5.1 Theoretical framework

The revised G4G intervention is underpinned by self-determination theory (SDT) (Deci & Ryan, 2008), and elements of the social-ecological model (SEM) (Stokols, 1996), because its theoretical foundations are concerned with how the psychological and socio-environmental conditions (e.g., created by a coach) can support an individual's motivation (Fortier et al., 2012). SDT is particularly appropriate for understanding pre-adolescent girls' PA participation viewpoints (Player, 2010b) because the theoretical framework focusses on understanding the quality of an individual's motivation (e.g. how self-determined their reasons for PA participation are). Previous research suggests that more autonomous PA motivation is positively associated with child and adolescent PA participation (Dishman et al., 2016), and

positive psychological outcomes (such as quality of life and physical self-concept, etc.) (Standage et al., 2012). SDT within G4G contends that finding ways in which pre-adolescent girls can develop a sense of choice and ownership over their own PA (autonomy), feel capable engaging in PA (competence), and are supported within a broader social context (relatedness) will foster more self-determined forms of motivation (e.g. participating for fun or personally valued benefits), which are in turn positively associated with heightened PA levels (Sebire et al., 2016). SDT, therefore, suggests that PA interventions, such as the named G4G programme, which target enjoyable and valued activities, and foster perceptions of ownership, competence, and belonging, are more likely to result in a sustained behaviour change (Sebire et al., 2013).

The revised G4G intervention was also designed using elements of the Social Ecological Model (SEM) (Stokols, 1996), grounded in three key elements (organisational, interpersonal, and individual). At the organisational level, initial support from the G4G lead researcher allowed for the development of a multi-component community OYS coaching intervention (see Figure 6.1). At the interpersonal level, ongoing support was provided for the G4G coaches from both the lead researcher and the G4G coordinators. The individual-level involved developing coaches', parents, and guardian's understanding of the benefits surrounding children's PA, FMS, and effective coaching practice.

## → SHARP and SAAFE Principles: ◄

Stretching whilst moving; High Repetitions of motor skills; Accessibility through differentiation; Reducing sitting & standing; and Promoting on-field physical activity

Supportive in coaching; maximise participant's opportunities to be physically Active; create an Autonomous learning environment; design and deliver session experiences that are Fair; provide an Enjoyable experience by focusing on fun and variety.



Figure 6.1. Theoretical underpinnings of G4G Intervention

### 6.5.2 G4G intervention structure and components

Participants selected for G4G participation, received the 10-week intervention, consisting of  $1 \times 60$  minutes, specifically tailored, and research-informed LGF session per week (10 sessions in total between March and May 2017). The G4G intervention includes three major components, as guided by the associated theoretical constructs, specifically the (1) participant component, (2) coach education component, and (3) parent/guardian/community component (see Table 6.1).

The programme was structured using a consistent format each week and included 6 rotatory LGF skills stations (i.e. adopting a stations-based sport pedagogy approach) (Butler & Griffin, 2010). In each of the 6 stations (10-minutes each), coaches were instructed to focus on different skills, specific to LGF, and these skills were informed by the weekly joint planning sessions between the lead researcher and the G4G club coordinators. Due to a large number of G4G participants, coaches had access to two full-size Gaelic Athletic Association (G.A.A.) pitches to set up the 6 stations and deliver the weekly G4G programme sessions (see Figure 6.2). In terms of targeting specific intervention components within G4G, research guided baseline findings (Farmer et al., 2017, Farmer et al., 2018) informed the design and the development of this proof of concept feasibility trial, and in particular, highlighted the need for targeting low levels of PA and FMS proficiency, through addressing participants' self-efficacy levels, inclusive of fun, and socially-stimulating PA environments.

## 6.5.2.1 G4G participant components

## 1) Developing FMS

Using the existing G4G framework (6 x 10-minute rotatory LGF stations), participants in the revised G4G intervention were exposed to a variety of FMS, and LGF sport-specific skills, through a Teaching Games for Understanding (TGFU) instructional model (Sheppard, 2014), within a rotatory station-based approach (see Figure 6.2). The previous implementation of a station-based, and TGFU approach has been shown to empower young children, deepen their knowledge of game tactics, help improve skills, and contribute to enjoyment levels (Butler & Griffen, 2010). As only 2% of participants were proficient in FMS, across a range of 7 object

control and locomotor skills at baseline (Farmer et al., 2018), as part of the G4G intervention, one of the six stations each week focused specifically on the deliberate practice of key FMS, as a strategy for increasing motor competence among girls. The FMS activities were implemented through fun games, with appropriate strategies to stimulate development and for the provision of adequate time for participants to perform both locomotor and object control FMS-related activities (Robinson et al., 2015).

To further encourage the development of FMS proficiency, and foster independence and autonomy in practicing sports-specific skills, participants received weekly G4G skill cards upon completion of each session. Each sport specific G4G skill card (10 in total) every week comprised of a different skill focus within the game of LGF (e.g. kick, catch, hand pass), and pedagogically, these were presented with a visual, and a brief description of the key coaching cue points.

A sub-sample of participants (n = 52) took part in an innovative LGF and FMS dance for 30 minutes, before the commencement of the G4G pitch sessions, from weeks three to eight. The rationale for the inclusion of the G4G FMS dance was to provide an additional opportunity for participants to practice basic FMS (e.g. coordination, skipping), combined with sport specific LGF skills into a fun movement sequence of dance, using a prevalent song in the charts as an auditory stimulus. The Canadian Sport for Life (Sport for Life Society, 2016) states that participation in rhythmic activities, such as dance, aids in the ease and efficiency of performing many FMS, by enhancing the fluidity of movement patterns.

The G4G dance, therefore, provided an appropriate medium to increase girls' PA and apply the SDT through intrinsically motivating and building girls' perceived autonomy, competence, and relatedness (Jago et al., 2013). Participants performed the G4G dance for parents, coaches and the invited LGF player 'role models' (Morgan et al., 2017) at the presentation evening on week 8 of the programme, as part of a culminating festival of sport within G4G (Appendix I and J).

## 2) Promotion of a Social and Inclusive Environment

Participants were exposed to a specific 'Team Challenges' station each week (see Figure 1), to promote fun, friendship, and inclusivity. Previously published G4G baseline findings reported that the biggest participation barrier to PA was 'not being as good as others at it' (Farmer et al., 2018). Additionally, the influence of friends

and the importance of having someone to participate as a key emergent theme (Farmer et al., 2018) among primary school-aged children, and these are key factors in young female's participation in PA (Carlin et al., 2015). As such, the 'Team Challenges' station was incorporated to promote a sense of inclusiveness, friendship, and belonging, while providing feedback to each of the participants within a fun noncompetitive environment. The need for relatedness (SDT) and feeling connected to others (peers) can help cultivate the development of intrinsic motivation (Crespo et al., 2013). Additionally, the 'Team Challenges' station encouraged the social aspect of PA participation with friends and promoted encouragement with each other through participation in these types of activities (Maturo & Cunningham, 2013).

## 6.5.2.2 G4G coach components

## 1) Coach Education

The lead researcher conducted an initial 2-hour Coach Education Workshop session, prior to the start of the G4G intervention. This followed with 8 x 1-hour Coach Education sessions, specifically with the lead researcher engaging face-to-face with the G4G club coordinators (N=2) and coaches (N=15) each week, approximately three to four days prior to each G4G intervention session. The provision of Continuous Professional Development (CPD), similar to the Robbins et al., (2013) 'Girls on the Move' intervention protocol was adhered to in terms of discussing issues, and reinforcing G4G policies and procedures. Additionally, G4G club coordinators and coaches received an on-demand 'WhatsApp' instant messaging service for additional support, where necessary, while the programme was ongoing. This instant messaging service included coaching session ideas for consideration within the G4G intervention or communication correspondence regarding the weekly training session set-up structure.

The coach education sessions were theoretical (SDT and elements of SEM) in their design, increasing levels of self-determination for coaches, through the provision of opportunities to support girls' through the three basic psychological needs of autonomy, competency, and relatedness (Deci & Ryan, 2002). The sessions also aimed to develop the G4G coordinator and coaches' self-efficacy to coach with

confidence and to support the participants to engage in more PA participation. A coinstruction model between the lead researcher and the G4G club coaches was utilised throughout the entire intervention, whereby G4G coaches independently instructed sessions with the support of the lead researcher. This type of experiential learning is a process through which the learner (i.e., the coach) can construct knowledge, skill, and value directly from an experience within the environment, and this CPD strategy has been proven to be a successful approach (Marlow & McLain, 2011).

As part of the weekly coach education workshops, G4G coordinators and coaches were provided with the opportunity to suggest additions or develop new strategies to engage the G4G participants. Coaches were then given a reflective task as part of these coach education workshops, utilising a similar protocol as 'The Girls Peer Activity (G-PACT) Project for adolescent girls (Owen et al., 2018). Coaches were prompted to consider how active they thought the G4G participants were during the previous week's coaching session, how the participants spent their time during training, and potential modifications the coaches could make to some of their drills to increase opportunities for PA participation (Guagliano et al., 2014).

The content for the following week was also discussed, alongside how best to deliver the sessions, which allowed the lead researcher to maintain consistency in the delivery and content coverage of the G4G intervention in the sports club (Owen et al., 2018). At the end of the 10-week intervention, the lead researcher assisted the G4G coordinator and coaches with the development of a club sports policy and action plan, which specifically encouraged the coaches and coordinators to continue using the G4G programme autonomously (i.e. without the lead researcher influence), and integrating the key components into current practice within the club setting. The coach education component targeted all G4G coordinators (N=2) and coaches (N=15) to act as change agents and supportive role models for PA promotion in the community-sports club environment (Belton et al., 2018).

## 2) G4G Coaching Manual

While the CPD sessions were designed using the research-informed protocol and existing national programme curricula from the LGFA, the lead researcher also tailored the intervention towards the inclusion of information on different coaching

methods/styles, content knowledge, and theoretical underpinnings (as guided by baseline findings (Farmer et al., 2017, 2018). To support coaches throughout the intervention, every coach received a comprehensive G4G Resource Pack, which was aligned to the evidence-informed SAAFE (Lubans et al., 2017) and SHARP principles (Powell et al., 2016).

The SAAFE principles and practical strategies (Lubans et al., 2017) have been designed to enable practitioners to deliver engaging PA sessions to youth, in a manner that maximises PA participation, and promotes physical literacy by enhancing affective, cognitive, motivational, and movement skill outcomes. G4G coordinators and coaches were encouraged to (i) be Supportive in their teaching, (ii) maximise participants' opportunities to be physically Active, (iii) create an Autonomous learning environment by including elements of choice and providing a rationale for activities, (iv) design and deliver session experiences that are Fair by allowing all participants to experience success regardless of their physical abilities, and (v) provide an Enjoyable experience by focusing on fun and variety.

Additionally, the 'SHARP Principles', were a set of weekly sport pedagogy principles [adapted from (Powell et al., 2016)], which formed a core element during the G4G intervention, and involved increasing opportunities for young girls to be physically active through the integration of key pedagogical components: e.g. Stretching whilst moving; High Repetitions of motor skills; Accessibility through differentiation; Reducing sitting & standing, and Promoting on-field physical activity (see Table 6.1).

The G4G resource pack consisted of the following: (1) Ten visual G4G session plans (emphasizing Teaching Games for Understanding, SAAFE, and SHARP principles); (2) A user manual containing pertinent coach education topics (such as 'Effective Coaching', 'Differentiation, and 'Including all Children'); (3) A sample range of FMS and LGF games; 4) Overall signposting information to useful online/print resources. This G4G resource pack was used to ensure continuity and consistency across the 10-week intervention.

## 6.5.2.3 G4G parental and guardian component

Given the continued influence parents and guardians play, specifically as children transition into adolescence (Belton et al., 2019), the incorporation of familial support was accounted for in the G4G intervention. Previously published baseline findings further highlighted the need to actively engage parents and guardians in young girls' PA and sporting experiences (Farmer et al., 2018). Furthermore, parental and guardian support services (in alignment with SDT) were implemented into the revised G4G intervention through the following strategies;

- *Technological resources* the lead researcher set up a 'WhatsApp' instant messaging and 'Facebook' group for all interested parents. The purpose of this messaging support service was to provide a communication link for encouraging parents to engage in regular text messaging regarding updates on the G4G programme (see Table 6.1). Parents utilised this platform to discuss the G4G programme, carpooling to training, attending games, and other matters (relatedness).
- Parent/Guardian Evening On week 4 of the G4G programme, the lead researcher designed and delivered a 45 minute 'G4G Parent/Guardian Workshop' in the community sports setting, prior to the commencement of week 4's coaching session for children. The rationale behind this workshop was to inform and educate parents/guardians on the important role they play in positively impacting their daughter's sporting experience. Other items surrounding the importance of FMS and club registration were also discussed. The lead researcher also encouraged parents and guardians to actively get involved with their daughter through the LGF skills at home (promote youth 'ownership'), specifically via the G4G Skill Cards.



Figure 6. 2: Gaelic4Girls weekly session station-based approach pitch layout

SDT	G4G Participant	G4G Coordinators/Coaches	Parental/Guardian Component
Components			
Autonomy	• Received weekly Ladies Gaelic Football (LGF) Skill Cards - encouraged to practice each week.	• Although instruction was provided in terms of following the SHARP and SAAFE Principals, coaches were actively engaged in the weekly session planning (coaches input of previous/existing knowledge/experience).	• Autonomous in positive role- modeling for their daughters (promoting an autonomous supportive relationship).
Competence	<ul> <li>FMS/LGF Dance Sessions (30 minutes from weeks 3 to 8)</li> <li>G4G Weekly LGF Skill Cards</li> <li>Specific FMS Station (10 minutes)</li> </ul>	<ul> <li>Coaches were in control of their behaviour, contributed to choosing the content of the session, and actively engaged in the planning stage of the sessions.</li> <li>Coaches competence developed through joint planning (feedback sessions) and the SHARP Principles.</li> </ul>	• Understanding PA/Sport participation barriers for young girls.
Relatedness	<ul> <li>Specific Team Challenge Station (10 minutes)</li> <li>Opportunity to practice FMS and LGF skills in a fun, social, and enjoyable environment</li> </ul>	<ul> <li>Coaches sense of belonging, intervention supported by lead researcher and G4G coordinators providing an instant support network for coaches involved.</li> <li>The weekly CPD joint planning meetings assisted in providing social belonging and support for the G4G Coordinators and the G4G Coaches.</li> </ul>	• Support/guidance system via the 'WhatsApp messaging group and the G4G Facebook page'.
SEM Components	G4G Participant	G4G Coordinators/Coaches	Parental/Guardian Component

## Table 6. 1: Gaelic4Girls (G4G) intervention components and theoretical constructs

Individual-level	<ul> <li>Increasing girl's knowledge of the importance of stretching/sport participation.</li> <li>Developing girl's knowledge and skills of FMS combined with skills of LGF through a fun, enjoyable dance routine, weekly LGF Skill Cards and SHARP and SAAFE Principles.</li> </ul>	<ul> <li>Increasing coaches' awareness of children's PA levels and attitudes to sport.</li> <li>Developing coaches' knowledge and skills of FMS/LGF skills/coaching/SHARP and SAAFE Principles through weekly CPD joint planning sessions with the lead researcher.</li> </ul>	• Increasing parental/guardian awareness of developing knowledge on children's PA/FMS levels. Importance of parents as active role models, through the sharing of weekly videos/information on G4G Facebook page and Parental Workshop on week 4.
Interpersonal level	• Ongoing support from G4G coaches and coordinators (positive reinforcement and praise, as guided by the SHARP and SAAFE Principles).	<ul> <li>Ongoing support from the lead researcher and the G4G coordinator</li> <li>Ongoing support for the G4G coordinator through regular emails, 'WhatsApp' messaging, and meetings (action plan and progress were reviewed).</li> </ul>	• Ongoing support for the parents and guardians through regular reminder messages, 'WhatsApp' messaging support service, and Facebook page.
Organisational level		<ul> <li>Ongoing support from the G4G Coordinator/lead researcher.</li> <li>Development of a club sports policy and action plan with the G4G</li> <li>Coordinator integrating SHARP and SAAFE Principles.</li> </ul>	

SDT: self-determination theory; SEM: social-ecological model; LGFA: ladies Gaelic football association; PA: physical activity; FMS: fundamental movement skills; CPD: continuous professional development

## **6.6 OUTCOMES**

#### 6.6.1 Data collection

Pre- and post-intervention assessments were conducted by trained research assistants at the community sports club (FMS assessment) and local primary school settings (questionnaire and FGs). Pre FMS testing took place on week 1 of the intervention, prior to the G4G's commencement, while post-testing took place upon completion of week 10 of the intervention. To enhance the quality of the data across all collection sites, the research assistants (all pre-service physical education teachers) were formally trained in the standardised measurement procedures and protocols associated with FMS, self-report questionnaires, and focus group (FG) interviews. Each research assistant attended a 2-hour robust field training workshop and was given detailed manuals, checklists, and scripts to read when informing the participants about the measures.

Objective measurements, such as FMS were conducted in the community sports club setting hall, with a ratio of one researcher to five child participants (1:5). Subjective pre- and post-self-report PA measurements took place in a supervised classroom or computer lab, and the ratio of the researcher to child participants was 1:10. All questionnaires were completed online through the tool 'Survey Monkey'.

FG's with child participants, coaches, parents, and guardians occurred in a school classroom, under supervision, and lasted between 45 to 50 min each on average. All FGs were conducted after the 10-week G4G intervention by the leading researcher, who has training and experience facilitating FGs. Participants were reminded that they could withdraw from the interviews at any stage and that all recordings would remain confidential. The FG interviews were recorded by dictaphone and were transcribed verbatim. Throughout this data collection, participants were assigned identification numbers for anonymity purposes.

## 6.6.1.1 Physical activity self-report assessment

MVPA was assessed using a modified version of the Physical Activity Questionnaire for Older Children (PAQ-C) (Kowalski et al., 2004). Studies have established the reliability and validity of the 7-day recall of children (Moore et al., 2007). The PAQ-C for this study included 15 physical activities, 10 leisure/free-time activities, activities in school (Physical Education), transport activities (walking to and from

school), and other activities. The participants were told to recall what activities they had engaged in the previous seven days and how many times and the number of minutes they participated in each of these activities for. Habitual PA was also assessed using two questions from the Physician-based Assessment and Counseling for Exercise (PACE) questionnaire: how many days in the last week (PACE 1) and a usual week (PACE 2) does the subject do at least 60 min of physical activity.

### 6.6.1.2 Fundamental movement skills

The FMS proficiency of participants across seven-movement skills was assessed in conjunction with the behavioural components from three established instruments, namely the Test of Gross Motor Development (TGMD) (Ulrich, 1985), Test of Gross Motor Development-2 (TGMD-2) (Ulrich, 2000), and the Get Skilled Get Active resource (Ryde, 2000). Each of these instruments and their associated protocols has established validity and reliability in children and are designed to give an objective measurement of gross motor skill proficiency. The seven-item test included culturally relevant skills to LGF: three locomotor skills (run, skip, and vertical jump), one stability (balance) skill, and three object skills (stationary dribble, catch, kick), which combine to give an overall maximum raw score of 60.

Participants performed the skill on three occasions, including one familiarisation practice, and two performance trials, as reported in previous Irish movement skill data assessments (Farmer et al., 2017). Participant performance, along with the execution of the required skill, were recorded using digital video cameras (2× Canon type Legria FS21 cameras; Canon Inc., Tokyo, Japan) and 2 x Apple I-Pads to allow for greater measurement scrutiny and accuracy of measurement precision during analysis. Once data collection was completed, the principal investigators were required to reach a minimum of 95% inter-observer agreement for scoring all seven FMS.

## 6.6.1.3 Psychological correlates of PA

Questions on psychological correlates of PA (self-efficacy, physical self-worth, physical performance self-concept, PA attitudes, perceived motivators and barriers to PA participation, subjective norms, behavioural control, and intentions) were all taken, with scales intact, from the Fife Active survey (Rowe & Murtagh, 2012). Participants' self-efficacy was assessed using a modified version of the Children's

Physical Activity Self-Efficacy Survey, an 8-item scale developed by Sherwood et al., (2004). Physical self-worth was assessed using a global physical self-worth subscale from the Child and Youth Physical Self-Perception Profile (taken from Physical Activity in Scottish School-age children [PASS]) (Inchley et al., 2008). Participants' physical performance self-concept was assessed using the Athletic Competence subscale, taken from the Self-Perception Profile for Children (Sherwood et al., 2004). Participants' perceived PA motivators and barriers were assessed using a modified version of the Physical Activity Outcome Expectancies Scale (Taylor et al, 2004) and PASS, respectively (Inchley et al., 2008). PA attitudes, subjective norms, behavioural control, and intentions were assessed using the scale from Rhodes & Smith's (2006) questionnaire.

## 6.6.1.4 Focus Group Interviews

Three FG interviews with child participants and one FG interview with coaches, parents/guardians (combined) at post-intervention were undertaken to assess programme implementation. Specifically, FG interviews were used to elicit child participant and coach/parent (1) experiences/perspectives on the G4G programme; (2) PA beliefs, feelings, motivators, and barriers; (3) improvements/future directions of the programme and LGF. A semi-structured interview guide, using questions designed by the research team, was developed. Each of the FG interviews was conducted by two researchers: a facilitator and a note-taker.

## 6.6.2 Data analysis

This paper reports the study protocol and methodological procedures used in the design, development, and implementation of the 10-week G4G community sportsbased intervention only. For the purpose of this intervention protocol and component design manuscript, intervention evaluation comparisons are not reported. Such intervention comparisons are reported in chapter 7 to-follow'.

## **6.7 DISCUSSION**

In this paper, the authors describe the rationale and study protocol for the G4G intervention, which is an innovative, multi-component, community sports-based

coaching intervention. This is the first study of its kind in Ireland highlighting novel coaching strategies for increased PA levels, improvements in FMS movement mastery, and heightened psychosocial wellbeing, as a means to counteract the worryingly low levels of PA and FMS proficiency of pre-adolescent girls (Farmer et al., 2017; Kelly et al., 2018).

A recent systematic review and meta-analysis of PA community sports-based interventions among girls (Pearson et al., 2015) found that greater intervention effectiveness appears to result from multi-component interventions, theory-based, (Van Sluijs et al., 2007) focused on girls only (Camacho-Minano et al., 2011), target both PA and sedentary behaviour (Pearson et al., 2015), and include a non-formal coach education element (Koh et al., 2014). Multi-component interventions including several methods, namely; (1) combinations of support components (i.e. coaches, family, friends, etc.), (2) individual components (i.e., specifically tailored programmes for individuals/groups), (3) choice components (different options to facilitate behaviour change), and (4) educational (i.e., coach education) and environmental components have proven successful in targeting and improving youth PA participation (Pearson et al., 2015). The social interaction, fun challenges, motor skill focus, non-competitive nature, and personally relevant PA opportunities provided in G4G aligns with previous studies (Beni et al., 2017; Springer, 2013), reporting similar themes identified as central influences to young people's meaningful experiences and engagement in PA and sport.

Recommendations for school- and community-based PA programmes from various countries have FMS development as an integral aspect of physical education, school, and community sport (Morgan et al., 2013). Recent evidence supports a positive association between competence in FMS and PA in young people (Bremer & Cairney, 2018; Hulteen et al., 2018). The continuous inclusion of the SHARP and SAAFE coaching principles has the potential to provide coaches with key FMS elements to focus on in both the planning stages and in the delivery stages of the future G4G sessions. Coaches, therefore, have the opportunity to influence their athletes' PA and FMS levels, utilising the SHARP (Powell et al., 2016), and SAAFE principles as specific coaching strategies to monitor the intensity of PA during these G4G sessions (Guagliano et al., 2014). Findings from previous community sports-based youth interventions (Lubans et al., 2017) provide evidence that FMS coach

support strategies may be an effective way to improve FMS competencies and mastery.

An important element of G4G is the coach education component (Robbins et al., 2013; Owen et al., 2018). The literature widely acknowledges that youth PA interventions through the provision of non-formal coach education, such as CPD opportunities (training, resource manual, and opportunities to work with coaches and instructors) can positively support programme implementation, and childhood engagement in PA (Casey et al., 2014; Koh et al., 2014). There are currently no evidence-based studies regarding coach education and the provision of coach CPD opportunities within the existing LGFA organisational context in Ireland. Due to the amateur nature of Ireland's national game (G.A.A.), many coaches, parents, and guardians are voluntarily coaching, and this may lead to a lack of confidence for effective coach pedagogy (Gearity, 2012). Coaches within an OYS environment have the potential to be ideal candidates to promote PA (Guagliano et al., 2014), and are considered role models for PA, promoting a fun, friendly, and supportive team environment that provides children with sport-specific development. The G4G coach component comprising of weekly CPD learning opportunities (including the SHARP and SAAFE Principles) has attempted to address the need through the provision of a valuable framework for sustainable coaching practice, which may be used to facilitate positive change in OYS settings.

The friendship and social support structure within G4G (friends, family, and coaches) appears to be a sensible approach going forward for the G4G programme, as previous studies have reported that youth who feel socially connected to their coach, or other children in a sports team (Farmer et al., 2018; Carlin et al., 2015) tend to have greater satisfaction and positive emotions in the context of youth sport. Social agents, such as the G4G coaches and coordinators in this study, can structure the PA environment to maximise feelings of competence and personal mastery, with children more likely to enjoy the activity, maintain interest in involvement, and commitment to the activity (Granero-Gallegos et al., 2017)

Limitations of this study include the relatively small sample size of the intervention club, along with the non-randomised design, limiting the generalisability of the findings. The study design was limited to one subjective method for the assessment of participant's MVPA; a future recommendation would be to use more objective measurements of PA, such as accelerometers, or wearable

technologies. The strength of this study is the enhanced ecological validity, stemming from the naturalistic community sports-based G4G intervention setting. Another strength of the study is the use of simple, innovative, and feasible intervention strategies, based on a sound conceptual framework, which can be adapted in other similar OYS settings.

In order to develop and implement targeted strategies to increase team sport participation and, consequently, PA levels among pre-adolescent girls, there is a need to understand trends in, and influences on, pre-adolescent girls' participation in a team sport, including knowledge surrounding coaching practice and coach education. Although previous systematic reviews have examined the effectiveness of interventions designed to increase PA in general (Metcalf et al., 2012), to date there is no known systematic review investigating the effectiveness of multi-component PA interventions for pre-adolescent girls that aim to promote participation in team sport through developmental coaching pathways. There is a growing need to address youth PA in the community, as to date, the knowledge translation from youth PA research to practice in the community is weak (Peters et al., 2013). In a national context, Coaching Ireland (through the national Sport Ireland representative body) aims to raise the standard, status, and profile of coaching by implementing quality coach and tutor education supports (Chambers & Gregg, 2016). There is, however, a current lack of literature within the LGFA community in terms of quality coaching provision for athletes and PA participation opportunities.

## **6.8** CONCLUSION

The current trial of this protocol study presents, as far as the authors are aware, the first intervention to be conducted in an Irish OYS context, designed to increase participant's MVPA, FMS proficiency, and psychological wellbeing among 8-12-year-old girls through the professional development of coach education practices. This G4G (Gaelic4Girls) feasibility study will provide evidence regarding the implementation of a multi-component, community-sports based coaching intervention, and can be used as a starting point to inform the development of theory-based coaching interventions targeting PA promotion for pre-adolescent girls in Ireland. Examining the future efficacy of the G4G programme, as part of a larger trial would seem prudent in terms of the gathering sustainable, longitudinal evidence.

Acknowledgements: The authors wish to acknowledge the help of the undergraduate students from the Sports Studies and Physical Education undergraduate degree programme who were involved in the data collection. Sincere thanks also to the participants, parents, and coaches from the community sports club involved in this pilot intervention phase of the study. The authors would also like to express gratitude to the National Ladies Gaelic Football Association (LGFA) in Ireland for their continuous support and assistance.

## **Declaration of interest statement**

The authors declare that there are no conflicts of interest within this manuscript publication. The current research is funded through the Government of Ireland Irish Research Council Postgraduate Scholarship Scheme (October 2018-October 2019).

## 6.9 LINK TO CHAPTER 7

- As outlined in chapter 6, the G4G community sports-based intervention for Irish pre-adolescent girls was formulated from baseline findings (2016-2017) and guided by the appropriate intervention literature for female youth PA promotion. Chapter 6 provided a descriptive overview of the theoretical underpinnings and components of the G4G intervention.
- In chapter 7, the reader will examine the primary outcome measures for the 10-week G4G research-informed intervention, by comparing PA, FMS and psychological results between the 1) G4G research-informed intervention club, 2) a club utilising the existing G4G programme run by the LGFA, and 3) a control club without any programme conditions. The results will be illustrated through critical discussion and future directions for G4G research will be outlined.

## CHAPTER 7

## **'GAELIC4GIRLS' – THE EFFECTIVENESS OF A 10-WEEK MULTI-COMPONENT COMMUNITY-SPORTS BASED PHYSICAL ACTIVITY INTERVENTION FOR 8-12-YEAR-OLD GIRLS**



# Chapter 7: 'Gaelic4Girls' – the effectiveness of a 10-week multi-component community-sports based physical activity intervention for 8-12-year-old girls

## Preamble

The following article reports the effectiveness of the 'Gaelic4Girls' (G4G) intervention study; a 10 week specifically designed multi-component PA and FMS community sports-based coaching intervention, informed by contextually relevant data (chapters 4 and 5). In chapter 7, the reader will examine the primary outcome measures for the 10-week G4G intervention by comparing PA, FMS, and psychological results between the specifically tailored, research-informed G4G intervention group (1), a second treatment intervention group (2) and control group (3). The results will be illustrated through critical discussion and future directions for the G4G research during 2018 will be outlined. This is a published manuscript which has been peer-reviewed and published in the International Journal of Environmental Research and Public Health online [September 22nd, 2020]. The following is the citation for this article:

Farmer, O., Cahill, K., & O'Brien, W. (2020). Gaelic4Girls—the effectiveness of a 10-week multicomponent community sports-based physical activity intervention for 8 to 12-year-old girls. International Journal of Environmental Research and Public Health, *17*(18), 6928. https://doi.org/10.3390/ijerph17186928

## Statement of authorship;

I hereby declare that I, Orlagh Farmer, am the principal author of this article. The following statements outline my contributions to the work:

- I was the principal investigator responsible for all the data collected (2018) in this chapter. I organised all stages of data collection consisting of timetabling undergraduate students, contacting all primary schools and the LGFA club involved (principals, teachers, coaches, and students), equipment preparation, data storage, data cleaning and data inputting.
- Organised the rigorous training and protocol workshop for field staff assistants which was an important component of this baseline study.

- Made substantial contributions to the conception and design of the work; the acquisition, analysis and interpretation of data for the work.
- Wrote the first draft of the full article and contributed significantly to the subsequent redrafts until submission.
- Final approval of the version to be published, and agreement to be accountable for all aspects of the work in ensuring that questions related to the accuracy or integrity of any part of the work are appropriately investigated and resolve

## 7.1 PURPOSE OF CHAPTER

## 7.1.1 Rationale

In the last decade, there has been a gradual increase in the number of randomised controlled trials (RCT's) that have evaluated the impact of PA-based interventions on young girls (Voskuil et al., 2017; Jago et al., 2015; Sutherland et al., 2016), however, minimal evaluations of pre-adolescent girls within a community sports-based setting exist. Therefore, this study aims to assess the effectiveness of a multi-component, community sports-based intervention for increasing girl's PA levels, FMS proficiency, and psychological wellbeing.

## 7.1.2 Contribution to the field

The current intervention evaluation trial of this G4G exploratory study presents, as far as the authors are aware, the first three-armed intervention effectiveness comparative trial to be conducted in an Irish OYS context, designed to increase participants MVPA, FMS proficiency, and psychological wellbeing among 8- to 12year-old girls. This study strengthens the evidence base of the available OYS interventions for pre-adolescent girls, providing justification for the development of an RCT to assess the true effectiveness of G4G on PA levels, FMS proficiency and psychological wellbeing.

## 7.2 ABSTRACT

**Background:** This study aims to assess the effectiveness of a multicomponent, community sports-based intervention for increasing girl's physical activity (PA) levels, fundamental movement skill (FMS) proficiency, and psychological wellbeing.

**Methods:** One hundred and twenty female-only participants (mean age =  $10.75 \pm 1.44$  years), aged 8 to 12 years old from three Ladies Gaelic Football (LGF) community sports clubs (rural and sub-urban) were allocated to one of three conditions; 1) Intervention Group 1 (n = 43) received a novel, specifically tailored, research-informed 'Gaelic4Girls' (G4G) intervention, 2) Intervention Group 2 (n = 44) used the traditionally delivered, national G4G programme, as run by the Ladies Gaelic Football (LGF) Association of Ireland, and 3) Control Group 3 (n = 33) received no G4G intervention (group 1 or 2) conditions and were expected to carry out their usual LGF community sports activities. Primary outcome measurements (at both pre-and-10-week follow up) examining the effectiveness of the G4G intervention included 1) PA, 2) FMS and 3) Psychological correlates (enjoyment levels, self-efficacy, peer and parental support). A focus group (FG) interview (n = 6) was also conducted with the child participants from the intervention group 1 only, at the follow-up phase.

**Results:** Following a two (pre to post) by three (intervention group 1/intervention group 2/control group 3) mixed-model ANOVA, it was highlighted that intervention group 1 significantly increased in PA, FMS proficiency and the psychological correlates of PA. FG findings reported positive findings for intervention group 1 participants, specifically the heightened psychological wellbeing for girls, and the establishment of emerging friendships between peers.

**Discussion:** The findings demonstrate that the 10-week specifically tailored, research-informed G4G intervention is a feasible and efficacious programme, leading to a positive effect on the physical and psychological wellbeing of pre-adolescent Irish girls, as relative to the traditionally delivered national G4G comparative programme, and control group conditions.

## 7.3 INTRODUCTION

It is widely reported that girls are less physically active than boys throughout childhood (Sallis et al., 2016), and the age-related decline in physical activity (PA) participation, particularly from early adolescence onwards, is steeper for girls when compared to boys (Owen et al., 2017). Pre-adolescence (11–12 years of age) is a critical period of change in the PA participation levels of girls (Biddle, 2014).), and finding ways to help girls to become more physically active at this age is important for short and long-term health. The gender disparity in PA participation has highlighted the need to develop and evaluate interventions, specifically targeting atrisk youth (Camacho-Minano et al., 2011).

Club-based participation in organised youth sport (OYS) during childhood and adolescence contributes considerably to leisure-time PA for health-enhancing benefits (Howie et al., 2018), and has the potential to increase overall PA levels in young people (Golle et al., 2014; Pfeiffer & Wierenga, 2019), specifically among girls (Marques et al., 2016). OYS participation is also associated with important psychological benefits, which include increased wellbeing, self-efficacy, and positive peer/social relationships (Felfe et al., 2016). Evidence also highlights that a positive relationship exists between participation in OYS and fundamental movement skill (FMS) proficiency in children (Fransen, 2014; Henrique et al., 2016), allowing children to cumulatively acquire transferable movement skills that give them a sense of movement competence and confidence (Rosewater, 2009). A Portuguese cross-sectional study (n=973) reported that primary school children participating in OYS are less sedentary and participate in more frequent bouts of moderate to vigorous physical activity (MVPA) (Marques et al., 2016; Herbert et al., 2015), when compared to those who do not participate in organised sports. Although OYS does not necessarily prevent the decline in PA participation during adolescence (Bélanger et al., 2009), research does continue to illustrate that children who maintain their involvement in sport are also more likely to participate in PA during adolescence (Basterfield et al., 2016; Shull et al., 2019), and into adulthood (Bélanger et al., 2015; Hirvensalo & Lintunen, 2011).

Encouragingly, sport occupies a prominent place in Ireland's national cultural identity, which is reflected in the high prevalence (58-80%) of 10 to 18-year olds

who participate in some form of OYS, on at least one occasion in the week (Woods et al., 2018). The Global Advocacy for Physical Activity (GAPA) the Advocacy Council of the International Society for Physical Activity and Health (ISPAH) (2011) has identified participation in sports as one of the seven worldwide "investments that work" for improving youth PA levels, and for these reasons, community-sports based settings are recognised as key health settings in promoting knowledge of PA (Beets et al., 2016), and healthy lifestyle behaviours (Kokko, 2014). These youth sport participation outcomes have obvious short-term benefits, however, this domain may also help to develop 'physical literacy' (PL), and thereby support the continuum of lifelong PA. PL captures the essence of the basic skills children and adolescents should attain in order to be physically active and participate in sports (Meghann Lloyd et al., 2014). Whitehead (2010) further defines PL as "the motivation, confidence, physical competence, knowledge and understanding to maintain physical activity throughout the life course" (p. 18). Based on Lloyd et al. (2010) work, PL represents the successful interaction of four inter-related core domains: (a) physical fitness (cardiovascular fitness, muscular strength and endurance, flexibility, and coordination); (b) fundamental motor skills (e.g., catching and throwing a ball); (c) physical activity behaviours, and (d) psychological/cognitive factors (attitudes, knowledge, and feelings).

Numerous studies have shown the importance of these aforementioned PL variables as basic requirements for children and adolescents' health (Poitras et al., 2016). Furthermore, it is important to recognise that the value of the PL domains provides a powerful lens for PA, in relation to motor skill outcomes, environmental context, and broader psychological outcomes, including social and affective learning processes (Cairney et al., 2019). Despite these widely endorsed PL benefits for OYS, there are a lack of effective intervention strategies to promote sustainable PA participation in female youth (Anokye et al., 2018), particularly within a sporting context (Whitley et al., 2019). In the last decade, there has been a gradual increase in the number of randomised controlled trials (RCT's) that have evaluated the impact of PA-based interventions on young girls (Voskuil et al., 2017; Robbins et al., 2019), however, minimal evaluations of pre-adolescent girls within a community sports-based setting exist. Of those RCT's which have been undertaken, results have found small effects when PA was measured objectively (Okely et al., 2011). The need to

promote higher habitual MVPA engagement amongst children (particularly girls) in the youth sport context has been underlined as a prudent strategy (Fenton et al., 2017) particularly as knowledge translation from youth PA research to practice in the community is weak (Kelly et al., 2012).

In regard to specific PA programmes for female youth, it is recommended to incorporate psychological skill instructions that aim to enhance self-esteem, positive body image, positive attitudes towards PA, and motivation to participate in PA (R.D DeBate et al., 2009; Futris et al., 2013; Rauscher et al., 2013). Programmes should aim to also provide social environments that offer a variety of activities that girls may consider enjoyable (Howie et al., 2018; Martin et al., 2009).

The primary aim of this three-armed non-randomised controlled trial (NRCT) was to assess whether 8 to 12-year-old pre-adolescent girls who attended a multicomponent research-informed tailored 10-week 'Gaelic4Girls' (G4G) community sports-based intervention, could: i) increase their overall PA levels, ii) FMS proficiency, and iii) psychological wellbeing when compared to an intervention group 2 and a control group 3. A secondary aim was to gather intervention group 1 participant perspectives, specifically those from the research-informed tailored 10week G4G intervention (at follow-up) on the following: i) programme acceptability (structure, content and delivery); ii) identify potential barriers and facilitators to intervention uptake and continued PA participation. The current paper presents, as far as we are aware, the first NRCT conducted in an Irish 'Gaelic' Games, OYS context. It is hypothesised that the intervention group 1 would demonstrate significant improvements in all primary outcome variables when compared to those within the intervention group 2 and the control group 3 conditions.

## 7.4 MATERIALS AND METHODS

## 7.4.1 Study design

This was a quasi-experimental, non-randomised controlled before-and-after design, with a mixed-methods approach. This study adopted a three-arm NRCT design, with a specifically tailored, research-informed G4G intervention group (1), a second treatment intervention group (2) and control group (3). The study followed the transparent reporting of evaluations with nonrandomised designs (TREND)

statement for reporting (Reeves & Gaus, 2004). One hundred and twenty femaleonly participants (M age =  $10.75 \pm 1.44$  years), aged 8 to 12 years old from three LGF community sports clubs (rural and sub-urban) in Ireland were recruited to partake into one of three study conditions; intervention group 1 (n = 44), the club that received the research-led G4G intervention; intervention group 2 (n = 43), the club that used the existing 'traditional' national G4G programme structure; and the control group 3 (n = 33), the club that received no G4G intervention condition, and was expected to carry out their usual LGF community sports activities. Assessment measures were conducted at pre-intervention [March 2018] and were repeated at post-intervention [June 2018] phases. Ethical approval for this study was obtained from the Social Research Ethics Committee of the researchers' institution (University College Cork) in March 2016.

## 7.4.2 Recruitment, setting and participants

Convenience sampling was used to recruit participants from the 18 girls' only primary schools (rural and urban), from three local LGF clubs in the Cork and Kerry regions (Munster Province, Ireland). All schools were in close proximity to the local designated LGF club and the research team. To be considered eligible for this study, participants (girls only) were formally enrolled in between the years of second to sixth class of primary school (aged 8- to 12- years old) and had expressed an intention to attend the programme for its full duration. Informed parental consent and child assent were the requirements for eligible participation in this study. Information sheets and consent forms were administered to students who expressed an interest in taking part in G4G. Informed assent for participation was granted by all participants, and written consent was obtained from their parent(s)/ guardian(s), prior to the physical and psychological data collection measurements. For coaches to be eligible to partake in the programme, recognised coaching credentials from the national LGFA, and previous experience of coaching girls within the existing community sports-based setting was required; all participants were free to withdraw from the research at any stage.

Prior to the commencement of the research informed G4G intervention, the leading researcher emailed the school Principal from the five surrounding primary schools, calling for expressions of interest for the lead researcher to visit the school

and inform the girls about the G4G programme (intervention group 1). Subsequent to the granted approval from the five school Principals, the lead researcher, along with the two G4G coordinators (head club coaches from the selected LGF intervention club) visited the five primary schools, where a full outline of the G4G intervention and the associated data collection measurements were provided as part of the intervention group 1 condition. Club coordinators from the intervention group 2 and the control group 3 locations were also contacted by the lead researcher via phone call, and a full brief of the associated data collection measurements were again outlined. One hundred and sixty fifty participants were invited to participate in this study, with a total of 137 providing full consent (intervention group 1: n=50; intervention group 2: n=49; control group 3: n=38). In total, 120 participants had fully available data at both pre and post data collection time points (intervention group 1: n=44; intervention group 2: n=43: control group 3: n=33).

## Gaelic4Girls Research-Informed and Tailored Intervention Group 1

The G4G intervention, underpinned by the self-determination theory (SDT) (Deci & Ryan, 2008), and elements of the social-ecological model (SEM) (Stokols, 1996) (see Table 7.1) is a multi-component PA and FMS community sports-based intervention. Participants selected for the research informed G4G programme received the 10-week intervention, consisting of  $1 \times 60$  minutes, specifically tailored LGF session per week (10 sessions in total between March and May 2018). This G4G intervention includes three major components, as guided by the associated theoretical constructs; the (1) participant component, (2) coach education component, and (3) parent/guardian/community component (Chapter 6).

#### Gaelic4Girls Nationally Delivered and Existing G4G Intervention Group 2

Participants allocated to the intervention group 2 (n = 43) received the 10-week 'traditional' existing G4G programme, as run by the LGFA. In comparison to intervention group 1, this programme consisted of 1 × 60-minute introductory LGF sessions per week (10 sessions in total between March and May 2017), without research-informed content, coach education professional development workshops, or

any additional digital or hardcopy resources. Similar, to intervention group 1, the 'traditional' G4G programme included three major components, (1) participant component, (2) coach education component, and (3) parent/guardian/community component.

## Control Group 3

Participants selected for the control group 3 received no G4G programme, resources or coach education workshops. The participants trained as they would normally in their local LGF club setting over a 10-week period.
Table 7.1 Overview of the research informed G4G intervention group	1 programme components with associated pedagogical	
considerations		

Part	What	Pedagogical Considerations
Participant PA FMS Psychological	<ul> <li>6 x 10-minute rotatory LGF skills stations (i.e. provision of specific time for locomotor and object control skill execution and development)</li> <li>Sport-specific G4G skill cards (10 in total, e.g. kick, catch, hand pass) of key coaching cue points.</li> <li>G4G innovative LGF and FMS dance for a duration of 30 minutes indoor, prior to commencement of G4G outdoor pitch sessions, from weeks 3 to 8 (extra 300 minutes of FMS specific execution).</li> <li>Specific 'Team Challenges' station to promote fun, friendship and inclusivity.</li> </ul>	<ul> <li>TGfU instructional model (Sheppard, 2014)</li> <li>SHARP Principles (Powell et al., 2016)</li> <li>SAAFE Principles (Lubans et al., 2017).</li> </ul>
<u>Coach</u> Coach Education Workshops Resources Support Structures	<ul> <li>Initial 2-hour Coach Education Workshop session, prior to the start of the G4G intervention with the lead researcher.</li> <li>8 x 1-hour Coach Education sessions, specifically with the lead researcher engaging face-to-face with the G4G club coordinators and coaches each week before sessions</li> <li>G4G Coaching Manual &amp; Resource Pack/Electronic video clips of activities with key coaching points, via booklet and on-demand 'WhatsApp' instant messaging service. This instant messaging service was also used for communication correspondence regarding the weekly training session set-up structure.</li> <li>G4G Coordinator and Coach Reflective Task provided with the opportunity to suggest additions or develop new strategies to engage the G4G participants on a weekly basis.</li> </ul>	<ul> <li>SHARP Principles (Powell et al., 2016)</li> <li>SAAFE Principles (Lubans et al., 2017).</li> <li>SDT Interpersonal, Autonomy, Competence and Relatedness</li> </ul>
Parental Support & Engagement	<ul> <li>Technological resources – the lead researcher set up a 'WhatsApp' instant messaging and 'Facebook' group for all interested parents</li> <li>On week 4 of the G4G programme, the lead researcher designed and delivered a 45 minute 'G4G Parent/Guardian Workshop' in the community sports setting.</li> <li>Parent and guardians were encouraged to actively get involved with their daughter through the LGF skills at home, specifically via the G4G Skill Cards.</li> </ul>	<ul> <li>Interpersonal (communication/relatedness with other parents</li> <li>Encourage parents to promote 'youth ownership'</li> </ul>

#### 7.4.3 Outcomes

#### 7.4.3.1 Data collection

Pre- and post-intervention assessments were conducted by trained research assistants at the community sports club and local primary school settings. Pre-testing took place prior to week 1 of the study, while post-testing took place upon completion of week 10 of the study. To enhance the quality of the data across all collection sites, the research assistants (all pre-service physical education teachers) were formally trained in the standardised measurement procedures and protocols associated with FMS, self-report questionnaires, and focus group (FG) interviews. Specifically, each research assistant attended a 3-hour robust field training workshop and was given detailed manuals, checklists and scripts to read, when informing the participants about the measures.

Objective measurements, such as FMS were conducted in the community sports club setting hall, with a ratio of one researcher to five child participants (1:5). Subjective pre and post-self-report PA measurements took place in a supervised classroom or computer lab, and the ratio of a researcher to child participants was 1:10. All questionnaires were completed online through the tool 'Survey Monkey'.

The FG was conducted after the 10-week G4G intervention by the leading researcher, who has training and experience facilitating FGs. Participants were reminded that they could withdraw from the interview at any stage and that the recordings would remain confidential. The FG interview was recorded by dictaphone and was transcribed verbatim. Throughout this data collection, participants were assigned identification numbers for anonymity purposes.

#### 7.4.3.2 Physical activity self-report assessment

Moderate to vigorous PA (MVPA) was assessed using a modified version of the Physical Activity Questionnaire for Older Children (PAQ-C) (Kowalski et al., 2004). Studies have established the reliability and validity of the 7-day recall of children (Moore et al., 2007). The PAQ-C for this study included 15 physical activities, 10 leisure/free-time activities, activities in school (Physical Education), transport activities (walking to and from school), and other activities. The participants were told to recall what activities they had engaged in the previous seven days and how many times and the number of minutes they participated in each of these activities

for. Habitual PA was also assessed using two questions from the Physician-based Assessment and Counseling for Exercise (PACE) questionnaire: how many days in the last week (PACE 1) and a usual week (PACE 2) does the subject do at least 60 min of PA.

#### 7.4.3.3 Fundamental movement skills

The FMS proficiency of participants across seven-movement skills was assessed in conjunction with the behavioural components from three established instruments, namely the Test of Gross Motor Development (TGMD) (Ulrich, 1985), Test of Gross Motor Development-2 (TGMD-2) (Ulrich, 2000), and the Get Skilled Get Active resource (NSW Department of Education, 2000). Each of these instruments and their associated protocols has established validity and reliability in children and are designed to give an objective measurement of gross motor skill proficiency. The selected seven FMS included culturally relevant skills to LGF: three locomotor skills (run, skip, and vertical jump), one stability (balance) skill, and three object control skills (stationary dribble, catch, kick), which combine to give an overall maximum raw score of 58.

Participants performed the skill on three occasions, including one familiarisation practice, and two performance trials, as reported in previous Irish movement skill data assessments (Farmer et al., 2017). Participant performance, along with the execution of the required skill, were recorded using digital video cameras (2× Canon type Legria FS21 cameras; Canon Inc., Tokyo, Japan) and 2 x Apple I-Pads to allow for greater measurement scrutiny and accuracy of measurement precision during analysis. Once data collection was completed, the principal investigators were required to reach a minimum of 95% inter-observer agreement for scoring all seven FMS.

#### 7.4.3.4 Psychological correlates of PA

A variety of psychological outcomes were assessed using existing questionnaires that have demonstrated reliability and validity for use with this age group. Table 7.2 presents information regarding all psychological correlates for the PA instruments used in this present study, along with the internal consistency values (Cronbach's alpha value) and relevant references: namely, self-efficacy, perceived selfconfidence, PA enjoyment, PA attitudes, perceived peer and parental social support.

These internal consistency values were greater than .70, representing acceptable internal consistency values (Nunnally, 1978).

#### 7.4.3.5 Focus group Interview

One FG interview was conducted at post-intervention with child participants from the research informed G4G programme (intervention group 1) to assess programme implementation. The primary purpose of the structured discussion with participants from the research informed G4G programme was to determine the applicability and sustainability of the G4G intervention, specifically in terms of assessing enjoyment, knowledge acquired, and suggestions for future developments. Six participants were randomly selected from the intervention group 1, following the completion of the self-report, anthropometric, and FMS data collection. The FG occurred in a meeting room in the clubhouse, under researcher supervision, and lasted 45–50 minutes approximately. A semi-structured interview guide, using questions designed by the research team, was developed. The FG interview was conducted by two researchers: a facilitator (lead researcher) and a note-taker.

#### 7.4.4 Data analysis

The combined PA, FMS and psychological correlate dataset was analysed using SPSS version 20.5 for Windows (IBM, Armonk, NY). Participants with incomplete data for a given variable were excluded from the analysis. Descriptive statistics and frequencies for all PA, FMS and psychological variables were employed to describe participants' characteristics. Comparability of the three groups at pre-test was ascertained using a One-way ANOVA. The pre-test to post-test variable changes was evaluated using repeated-measures analysis of variance (ANOVA) to determine time and time-by-group differences. All statistical assumptions were tested for normality, outliers, and homogeneity of variance. Specifically, a two (pre to post) by three (intervention group 1, intervention group 2 and control group 3) mixed model analyses of variance (i.e. between-within subjects) was conducted to analyse any main effects and time x group interactions for PA, FMS proficiency, and the psychological variables. Levene's test was used to determine whether variances were equal between each group.

To reduce the risk of Type I error, a Bonferroni adjustment to the alpha level was made by dividing p < .05 by 10 (number of all variable comparisons); thus, the alpha level was set to p < .005. In instances where significant interaction effects were found, post hoc comparisons were carried out using the Tukey honest significance difference (HSD) to determine where the differences occurred. Effect sizes of significant differences were evaluated using partial eta-squared ( $\eta$ 2). To calculate effect sizes, Cohen d values (1988) were applied, with d = 0.2 representing a small effect size, 0.5 representing a medium effect size, and 0.8 representing a large effect size.

The FG data transcript was thematically analysed (Braun & Clarke, 2006) by the lead author, involving coding, categorisation, and summarising (Elo & Kyngäs, 2008). The FG was coded, and, under each heading, the coded comments were clustered into categories based on similar content. Then, a thematic analysis was conducted to produce an in-depth description and understanding of the transcript phrases from the transcript were highlighted and grouped, from which themes and subthemes emerged (Neergaard et al., 2009; Pawlowski et al., 2014). Any discrepancies in the coding or reduction process were discussed until consensus was reached. Finally, both members of the primary research team selected indicative quotes representing the core wider views. Field notes and email correspondence were also recorded, filed and included within analyses. The FG audio recording was transcribed verbatim. The leading author double-checked the transcript against the original recording for robustness and accuracy prior to analysis.

Variable	Instrument	Scale	Max score	Reference (taken from)	α
SE	Modified version of the Children's Physical Activity Self-Efficacy Survey (PASES)	11-items (4-likert)	44	Saunders et al., 1997, adapted by Sherwood et al.,2004	.73
PSC	Physical self-confidence scale	15-items	150	McGrane et al., 2017	.93
PA Enjoyment	Modified version of Choices questionnaire	10-items (5-likert)	50	Motl et al., 2001	.80
PA Attitudes	The Fife Active Survey Scale	4-item (4-likert)	16	Rowe & Murtagh, 2012	.82
Peer/Social Support	Modified version of Choices Questionnaire	5-item (5-likert)	25	Birnbaum et al., 2005	.74
Family Support	Modified version of Choices Questionnaire	5-item (5-likert)	25	Saunders et al., 1997	.76

Table 7.2. G4G psychological correlates of PA	assessment instruments and scales	, with Alpha (	Cronbach Coefficients
---	-----------------------------------	----------------	-----------------------

Abbreviations:  $PA = physical activity; \alpha = Cronbach's Alpha Coefficient; SE = self-efficacy; PSC = perceived self-confidence$ 

#### 7.5 RESULTS

#### 7.5.1 Descriptive statistics

All three intervention and control groups (N=120) were similar for most pre-test characteristics, with a one-way ANOVA revealing no significant differences between groups for any of the primary outcome variables. Participants mean scores for all primary outcome variables pre and post-test (PA, FMS and psychological), as split by group condition are shown in Table 7.3.

Variables	Intervention G1		Interver	tion G2	Control G3	
	Pre	Post	Pre	Post	Pre	Post
N	43	43	44	44	31	31
	M SD	M SD	M SD	M SD	M SD	M SD
PA (mins)	$325.81\pm98.76$	$365.58\pm64.56$	$306.81\pm84.57$	$300.68\pm68.32$	$308.71 \pm 85.97$	$290.32\pm 66.36$
LOM	$18.67\pm2.90$	$19.91\pm2.70$	$19.80\pm2.33$	$19.37\pm2.42$	$18.48\pm2.71$	$17.68\pm2.48$
OC	$17.12\pm2.72$	$17.70\pm2.58$	$17.41\pm2.22$	$17.43\pm2.24$	$16.61 \pm 2.22$	$15.71\pm1.64$
<b>Overall</b> FMS	$44.21\pm5.13$	$46.07\pm4.43$	$45.14\pm4.07$	$44.59\pm4.21$	$42.21\pm4.49$	$41.00\pm3.79$
SE	$33.79\pm4.02$	$35.60\pm4.74$	$32.45\pm4.38$	$31.98 \pm 4.86$	$33.13\pm4.19$	$31.03\pm6.16$
PSC	$118.00\pm23.24$	$121.30\pm23.28$	$116.12\pm28.17$	$115.12 \pm 27.64$	$123.84\pm26.01$	$118.10\pm24.84$
Enjoyment	$43.00\pm3.52$	$45.93\pm3.48$	$44.48\pm3.82$	$44.30\pm3.43$	$42.70\pm3.99$	$41.87\pm3.89$
PA Attitudes	$48.19\pm 6.37$	$49.33\pm5.06$	$46.07\pm6.29$	$45.39\pm5.71$	$46.81\pm5.64$	$45.42\pm5.28$
Peer/SS	$17.79\pm3.18$	$19.26\pm3.32$	$17.68\pm3.22$	$17.00\pm3.16$	$18.87\pm3.90$	$17.90\pm3.58$
Family S	$18.00\pm3.35$	$19.49 \pm 3.53$	$17.40\pm3.90$	$17.71\pm3.64$	$18.87\pm4.36$	$18.16\pm4.48$

Table 7.3. Descriptive statistics [means and standard deviations (M±SD)] of PA, FMS and psychological measurements, stratified by group and time-period

Abbreviations: PA =physical activity; M = mean; SD = standard deviation; LOM = locomotor; OC = object-control; FMS = fundamental movement

skills; SE = self-efficacy; PSC: perceived self-confidence; SS = social support

#### 7.5.2 Physical activity

A mixed between-within subject's analysis of variance was conducted to assess the impact of the three different groups (intervention group 1, intervention group 2, and the control group 3) on participants' self-reported PA, across two time periods (preand-post-test). There was a significant interaction effect between programme type (group) and time for PA, Wilks' Lambda = .83, F (2, 115) = 11.77, p <.0001, partial eta squared = .170. Post-hoc tests (adjusted Bonferroni) revealed that there was a significant PA difference between the intervention group 1 (mean change = 39.7, SD = 81.66, p=.003) and the intervention group 2 (Mean change = -6.13, SD = 76.45, p < .002), from pre- to post- time periods. Similarly, there was a significant PA difference between the intervention group 1 (mean change = 39.7, SD = 81.66, p=.003) and the control group 3 (Mean change = -2.1, SD = 5.18, p = .004), from pre- to post- time periods. Conversely, there were no significant PA differences between the intervention group 2 and the control group 3 from pre- to post- periods. In terms of the direction of the results, the findings suggested a small PA difference (Cohen d = .03) between the three groups; only the intervention group 1 showed a PA increase from pre- to post- periods. For the intervention group 2 and control group 3, however, a reduction in PA was observed across the two time periods. Mean PA differences from pre- to post- in terms of self-reported weekly moderateto-vigorous physical activity (MVPA) minutes among the groups (n = 120) can be seen in Figure 7.1.



Figure 7.1. Comparison of the mean self-reported minutes of MVPA over time (pre to post) by group

#### 7.5.3 Fundamental movement skills

Following the two by three mixed model ANOVA, there were significant increases in overall FMS proficiency, from pre-to post-time periods for those in the intervention group 1, when compared to the intervention group 2, and the control group 3 (p = .003). Specifically, results showed a small- to moderate significant main effect for the three groups (F (2,115) = 6.16, p < 0.003,  $\eta^2$  = .09). There was a significant interaction effect between programme type (group) and time, Wilks' Lambda = .68, F (2, 115) = 27.71, p < .0001,  $\eta^2$  = .325. Follow-up post-hoc indicated that participants in the intervention group 1 (mean change = 1.86, SD = 4.78) reported significantly greater increases (p = .005) in overall FMS proficiency, when compared to those in the control group 3 (Mean change = -1.21, SD = 4.14) only.

For locomotor skills, when comparing the three programme (group) types (between-subjects effect), there was a significant interaction effect observed (F (2, 115) = 21.98, p = .000, partial eta squared = .28), suggesting small- to moderate differences only. Post-hoc tests revealed (adjusted Bonferroni) that only the intervention group 1 showed a small increase in locomotor skill proficiency over

time. Both the intervention group 2 and the control group 3 showed a small reduction in locomotor skill proficiency over time.

For object control skills, when comparing the three types of intervention/control groups (between-subjects effect), there was a small- to moderate main effect observed (F (2, 115) = 3.46, p = .035, partial eta squared = .057) There was a small but non-significant (p =.060) increase in object control proficiency scores for participants in both the intervention group 1 (mean change = 0.55) and intervention group 2 (mean change = 0.02). Similar to locomotor skills, a small but non-significant (p=.100) reduction in scores for those in the control group 3 was shown (mean change = 0.90).



Figure 7.2. Mean locomotor, object-control, and overall FMS proficiency differences among groups over time (pre to post stages; n = 120)

#### 7.5.4 Psychological correlates of PA

There was no significant main effect for the three groups across all psychological variables (see Table 7. 4). However, there were significant interaction effects for time within and between groups across all the psychological variables. Follow-up post-hoc tests revealed statistically significant differences for SE, PA enjoyment, and attitudes towards PA between the three groups, particularly for the intervention

group 1, when compared to their control group 3 counterparts. Post-hoc comparisons (adjusted Bonferroni) revealed statistically significant differences in mean scores between the three groups, with small- to moderate effect sizes observed (using Cohen's d value in Table 7.4). Table 7. 4 presents the statistically significant differences for all significant psychological correlates of PA.

Variable	Ν	G1 <i>(MC)</i>	G2 (MC)	G3 <i>(MC)</i>	Main interaction effects	d	Post hoc MC	Differences
SE	118	1.81	-0.47	-2.1	F(2,115) = 15.18, p = .000	.209	G1 & G2	<i>p</i> < .002
							G1 & G3	<i>p</i> < .003
PSC	114	3.30	-1.00	-5.74	F(2,114) = 8.81, p = .000	.134	G1 & G3	<i>p</i> < .002
							G2 & G3	<i>p</i> < .001
PA Enjoyment	118	2.93	-0.18	-0.83	F(2,118) = 13.82, p = .000	.107	G1 & G3	<i>p</i> < .003
Attitudes towards PA	115	1.51	-0.48	-0.09	F(2,115) = 12.64, p = .000	.180	G1 & G2	<i>p</i> < .003
Peer Social Support	115	1.47	-0.68	-0.97	F(2,115) = 37.17, p = .000	.393	G1 & G2	<i>p</i> < .003
							G1 & G3	<i>p</i> < .001
Family Support	113	1.49	0.31	-0.17	F(2,115) = 10.16, p = .000	.152	G1 & G2	<i>p</i> < .002

Table 7.4. Mean change (MC) differences and significant interaction effects (within time and between-group) for the psychological variables

Abbreviations: MC = mean change; G1 = intervention group 1; G2 = intervention group 2; G3 = control group 3; d = Cohen's d; SE: self-efficacy; PSC: perceived self-confidence, PA: physical activity

#### 7.5.5 Focus group

Qualitative results (post-test) from children concerning the G4G intervention's (group 1) acceptability in terms of structure, content/perception of learning, delivery, potential intervention barriers and changes/future directions of the G4G programme for continued PA participation, are summarised below in Table 7.5. Table 7. 5 provides an overview of the main intervention components, with exemplary quotes from children. The themes emerging from the FG were i) delivery pattern, ii) engagement and enjoyment, iii) perception of learning, iv) the confidence levels of the child and v) sustainability of the intervention.

#### **Intervention design**

#### Structure

1) Taster sessions

Most girls expressed the opportunity to experience an introductory 'sample session' prior to signing up for the G4G programme 'so that you don't have to commit from the very start'. The girls reported that this might give girls, who may not be interested in sport, a chance to get a feel for LGF and as a result would possibly be more likely to attend.

2) Station-based approach

Based on their experience of the 10-minute station-based approach every week, most children felt the novel activities challenged them and were enjoyable and welcomed the station-based approach. One participant cited *'I like the way we moved stations and learned different things'*. All participants expressed a desire for longer station time which further alluded to their enjoyment of the station-based approach.

3) Timing/duration of G4G

Some participants indicated that one hour in the evening was 'long enough' and an appropriate time for delivering the intervention. Others indicated that an hour and a half would be better suited for the programme allowing for more time at the stations. Participants also suggested the G4G programme should take place in the summer as the weather is better. One participant expressed 'I think the summer is better because I wouldn't want to be going out in the cold'. Two participants suggested running the programme in Autumn instead, as this would be a feasible option to guarantee intervention attendance, as family holidays may pose as a barrier during the summer.

The girls provided a range of responses (e.g. one hour, one hour and a half during the Summer and Autumn/Winter) but when asked for a show of hands, a large majority preferred the session to be delivered for one hour in the summer evenings.

#### Content

#### 1) Fun, friends and fitness

Most of the intervention group 1 children did appear to associate the G4G activities/programme with positive learning, having fun with friends and they felt fit, happy and confident following the activities. One participant expressed 'we got fitter and did new things', while another participant highlighted the friendship aspect as an appealing part of the programme 'It was nice to meet girls from other schools there'. Most children felt the novel activities challenged them and were enjoyable (particularly the G4G dance). 'I liked the way it (G4G Dance) was fun and we learned new skills'.

#### 2) Confidence in skills

The children associated the G4G activities with learning, specifically having felt they increased their perceived confidence levels in skill execution. Specifically, a few participants expressed that their fear of catching the ball reduced at post-intervention, one participant cited '*when I was playing matches, and everyone was trying to get the ball and I sort of hid and now I feel I can catch the ball – before I didn't do that'*. Other participants stated that their '*soloing and kicking skills*' significantly improved over the 10-week G4G intervention. Additionally, one participant referred to the G4G Skill Chart as a motivator to practise skills at home '*I thought the G4G Skill Chart really helped me and motivated me to practice'*.

3) Health messages

Participants identified a range of health benefits associated with partaking in the programme. One participant alluded that the programme and football was an escape to 'get all the things out of your mind'. Another participant expressed that the G4G programme 'made me happy'.

#### 4) Other appealing activities

The 'other' appealing activities the participants referred to in terms of enjoyment were the specific Team Challenge and FMS stations as part of the station-based approach that *'mightn't necessarily be a football game'*. Participants liked the idea of

*'not playing football the whole time'* during the weekly 60-minute LGF sessions and highlighted how the activities were fun and different.

#### Delivery

1) Female role models

The children expressed a strong desire for female role models to coach/join in with the activities expressing, that they would further motivate and engage the G4G participants. One participant cited '*If they could teach us at a young age, then we might like to be like them when we are older*'. Additionally, participants referred to the visit of the elite athlete role-model '*Cork players*' on week 8 and how this garnered the interest of all participants. One participant stated '*I liked when the Cork girls (senior elite athletes) came down, we got to meet them, and they signed our jerseys*'. This theme re-surfaced when discussing the changes/future direction of the G4G programme for continued PA participation.

#### 2) Variety of coaches

The variety of coaches and the changing nature of the 10-minute station-based approach for the weekly G4G sessions was particularly liked among participants. A number of participants felt it provided an opportunity to learn *'different games and skills'* and maintain interest during the session. One participant cited *'I like the way you moved on to a new coach after every 10 minutes so you wouldn't get bored of the same person'*.

#### Barriers

Additionally, a range of barriers were identified that were perceived to affect intervention uptake and continued PA participation (Table 7.5). Barriers included; i) lack of confidence, ii) friends (social acceptance), iii) body image, iv) fear of injury, v) competitive nature.

i) Lack of confidence

Participants identified skill-related lack of confidence among girls as a potential barrier for G4G intervention uptake or continued sport participation. One participant expressed that some girls *'might doubt themselves when trying to kick the ball and keep soloing'*. Another participant indicated that the fear of failure when performing skills may cause frustration, which may also impede continued sport participation *'If they can't do it (skills) they might get frustrated because they can't do it'*.

#### ii) Friends (social acceptance)

Participants also alluded to the fact that social acceptance among friends may act as a barrier to G4G intervention and continued sport participation. Specifically, social comparison *'with friends thinking you're not good at football and you're not good at stuff*'. Other participants referred to social acceptance and that *'some girls are really shy' and* drop out of sport because they don't *'really have many friends'*. Another participant explained that bullying may also be a contributing factor to not partaking in sport or dropping out *'maybe girls get bullied by some people and end up dropping out'*.

#### iii) Body image

Body image was identified among some participants as a potential barrier to G4G intervention uptake and continuation in sport participation. When posed the question what might stop girls from taking part in sport, some participants cited '*I think some of my friends who don't play football wouldn't play because they don't want to get mucky, get their hair messy or sweaty'*.

iv) Fear of injury

Participants also referred to the fear of injury as a potential barrier highlighting that players may be fearful of the ball itself with one participant citing '*Like you could be really scared of the ball coming to you*'.

v) Too much competitiveness

The notion of being 'too competitive' was alluded to by some participants as an impediment for sports participation and continuation. Participants indicated that the 'competitiveness' increased as they transition from primary to secondary school. For example, one participant stated, 'When you go to secondary school, you move up to the U14's, and that's a bit more competitive than U12's but maybe girls don't enjoy it anymore then'.

#### Sustainability of the intervention

The children also provided some interesting suggestions for changes to the intervention, including i) longer station time/fewer rotations, ii) more game/match time, iii) playing the G4G 'Blitz' earlier in the programme, iv) more role model and teacher involvement and v) seasonal timing and duration of the programme. Exemplar participant quotes are presented in Table 7.5.

	Key Themes	Quotes	Theoretical lens
Structure	Taster 'Introductory' Sessions	'Maybe you could sign up for G4G after the 1 <sup>st</sup> session if you want – maybe do a sample session first so you don't have to commit from the very start'	Intra, Inter, Organisational
		'Maybe you could do an introductory day in school'	
		'In school, girls might be more comfortable with their school friends'	Relatedness
		'Maybe you could tell the teacher about it (G4G) and we could do football for PE'	
	Station-based approach	'I like the way you move stations'	Organisational
		'I like the way we moved stations and learned different things'	
		'The stations were fun but too short'	
	Timing/duration of G4G	'I thought the hour was long enough'	Organisational
		'I think it should be an hour and a half'	
		'I thought the hour was perfect'	
		'I think it shouldn't be on in the summer'	
		'Maybe if it was on in the Autumn/Winter instead of the summer because I missed the Blitz I was on my holidays; people are busier during the summer'	
		'I think the summer is better because I wouldn't want to be going out in the cold'	
Content/	Fun, Friends and Fitness	'I really liked the way it was fun, and we learned new skills'	Intra, Inter
Perception of		'It was good to make new friends'	Relatedness
learning		'It was nice to meet girls from other schools there'	Competence
		'Everyone on the team is friends. Yesterday there was a football match and the new girls were there and now I know their names'	

#### Table 7.5. G4G research-informed intervention group 1 focus group thematic analysis findings (N=6)

'We got fitter and did new things'

Delivery

Confidence in skill execution	'I felt I improved my soloing. I wasn't the best at soloing but after the programme, I feel I'm after getting a bit better'	Intra
	'When I catch the ball before I was sort of afraid, but now I'm not afraid and the programme helped	Autonomy
	me'	Competence
	'My kicking and hand-passing got way better – now I can kick it like really really far'	
	'When I was playing matches, and everyone was trying to get the ball and I sort of hid and now I feel I can catch the ball – before I didn't do that'	
	'I couldn't solo the ball at first but when I did the programme, I started to become better'	
	'I thought the G4G Skill Chart really helped me and motivated me to practice'	
'Other' appealing	'I like the way there was matches and football but also fun games like that would keep you fit – it	Intra, Inter
activities/resources	mightn't necessarily be a football game so that you're not playing football the whole time'	
	'I liked the jerseys' 'I thought the G4G Skill Chart really helped me to practice the skills and motivated me to practice'	
	'The feel of the ball was soft – it was a lot lighter; it was easier to kick it and it went up a lot higher than the other balls' (SpotonSports football)	Organisational
	'It (G4G Dance) was great fun, I like the way we had the ball included in the dance, and we did the skills and stuff inside the dance' 'It's a good way to practice skills'	
Female Role Models	'I liked when the Cork girls (senior elite athletes) came down, we got to meet them, and they signed our jerseys' 'We met new Cork players that we mightn't have known'	
	'If they could teach us at a young age, then we might like to be like them when we are older'	
	'They'd help us keep going with football'	
Different Coaches	'I like the way you moved on to a new coach after every 10 minutes so you wouldn't get bored of the same person'	Inter

		'I liked moving around, usually your stuck with one coach, so I liked moving around cause <i>(sic)</i> different coaches had different games and skills and I liked that'	Relatedness
		'I liked the way we had different coaches and that we weren't just with the one coach the whole time'	
Programme	Lack of confidence	'Some girls might doubt themselves when trying to kick the ball and keep soloing'	Intra
barriers		'If they can't do it (skills) they might get frustrated because they can't do it'	Competence
	Friends (social acceptance)	'Maybe people or friends saying 'you're not good at football and you're not good at stuff'	Inter, Relatedness
		'Some girls are really shy; I know a girl who dropped out because she didn't really have many friends'	
		'Maybe girls get bullied by some people and end up dropping out'	
	Body image/stereotyping	'I think some of my friends who don't play football wouldn't play because they don't want to get mucky, get their hair messy or sweaty'	Societal/cultural perspective
	Fear of injury	'Like you could be really scared of the ball coming to you'	Intra, Autonomy
	Too much competitiveness	'When you go to secondary school, you move up to the U14's, and that's a bit more competitive than U12's but maybe girls don't enjoy it anymore then'	Organisational
Changes/		'we didn't have enough time at the stations, maybe 15 minutes instead or having less rotations'	Organisational
future directions		'maybe a few short stations and a few long ones too'	Inter/
of G4G Programme		'longer time to play the matches'	Relatedness
		'more role-models to help out and motivate us'	Autonomy
		'not to have the Blitz at the very end when the programme is finished, maybe on the 6 <sup>th</sup> week or something'	
		'maybe if like sometimes we could choose the games to play at the different stations- that would be fun'	

#### 7.6 DISCUSSION

The main purpose of the 'Gaelic4Girls' NRCT was to assess the effectiveness of a multi-component, community sports-based PA intervention for increasing girls PA levels, FMS proficiency, and psychological wellbeing, as well as to investigate participants programme experiences and perspectives for future direction. It is hypothesised that the intervention group would demonstrate significant improvements in all primary outcome measures, in excess of those within the comparison and control groups.

The PA data showed small, but significant increases in weekly minutes of self-reported MVPA, from pre-to post-time periods for those in intervention group 1, unlike those in intervention group 2 or the control group 3. This positive PA observation may be due to the fact that the coaches in the intervention group one were guided by research-informed implementation coaching principles, and were exposed to evidence-based frameworks, which were designed to guide the planning, delivery, and evaluation of the organised weekly G4G sessions in the community sports-based club. Two specific pedagogical coaching principles included the Lubans et al (2017) 'SAAFE' (Supportive, Active, Autonomous, Fair, and Enjoyable) delivery principles, and the Powell et al (2016) 'SHARP' principles, which involved the following key pedagogical aspects: Stretching whilst moving; high repetition of motor skills; accessibility through differentiation; reducing sitting and standing and promoting in-class/session PA.

One important factor was the introduction of dynamic stretches for intervention group one, which the coaches then integrated into the warm-up elements of their sessions for the duration of the research informed G4G programme, replacing their traditional static stretching routines (Bukowsky et al., 2014). Coaches in the intervention group one, therefore, had specific pedagogical guidance which sought to influence participants' PA levels, alongside focusing an increase in PA intensity during the weekly coaching sessions (Gugliano, 2014). The SHARP pedagogical principles (Powell et al., 2016) have recently proved successful in supporting teachers to increase 7- to 9-year-old children's MVPA by 30%, in primary PE lessons.

Additionally, it is plausible that the increase in self-reported minutes of MVPA among intervention participants in group 1 may be due to the Teaching

Games for Understanding (TGfU) (Sheppard, 2014) nature of the G4G station-based sessions. Participants in the intervention group were exposed to a variety of FMS, and LGF sport-specific skills, through a TGFU instructional model (Sheppard, 2014), within a rotatory station-based approach. Each 10-minute station allocated a significant amount of time to learning the skills of LGF via modified and small-sided games, which directly reinforced heightened periods for PA and minimising opportunities for physical inactivity among participants (Slingerland et al., 2014). As per the FG data collected from the child participants, the nature of the small-sided team games, the student enjoyment in the session stations, and the additional freedom utilising the TGfU approach may explain the improvement of self-reported PA levels. Previous research studies utilising the TGfU approach have proven successful in terms of increasing PA among children and adolescents in both school and community sports-based contexts (Harvey et al., 2016; Renshaw et al., 2016). Researchers have determined that the TGfU approach, alongside other pedagogical instructional models in teaching and coaching, can be effective in increasing the short-term (Perlman, 2012; Schneider et al., 2008) and long-term (Fu et al., 2016; Sallis et al., 1997) PA levels of children and adolescents. Previous pedagogical strategy interventions have produced a mean PA increase of 6.27% in children's MVPA during physical education lessons, with intervention groups spending 14% more time in the desired PA intensity when compared to control groups (Lonsdale et al., 2013).

The present results revealed that youth in the intervention group also reported significantly greater increased scores in their overall FMS proficiency over time when compared to both groups. Significant differences over-time were also found in overall locomotor skills proficiency between intervention group 1 and intervention group 2 and between the intervention group 1 and control group. Small, but non-significant increased differences were also found in the object-control subset, specifically between the intervention 1 group and control group. Based on the fact that the intervention group were organised and instructed by coaches who were trained and upskilled by the lead researcher, who had significant experience in physical education teaching and sports coaching. The small increase in overall FMS performance among the intervention group may have been as a result of the consistent exposure to the 10- minute weekly FMS coaching station, where basic

motor skill activities were implemented through fun games. Appropriate coaching strategies to stimulate FMS development were provided to participants to perform both locomotor and object control related activities (Robinson et al., 2015). Coaches in the research-informed intervention group, therefore, had the opportunity to influence the girls' PA and FMS levels, utilising the SHARP (Powell et al., 2016), and SAAFE principles as specific coaching strategies to monitor the intensity of PA during these G4G sessions (Guagliano et al., 2014).

Increases in the intervention group's overall FMS proficiency may also have been due to the fact that children received performance-related feedback on FMS and sport-specific skills. In order to support coaches throughout the intervention, every coach received a comprehensive G4G Resource Pack, which was aligned to the evidence informed SAAFE (Lubans et al., 2017) and SHARP principles (Powell et al., 2016). The initial 2-hour coach education workshop session (professional development prior to the start of the G4G intervention) alongside the 8 x 1-hour coach education sessions with the lead researcher may have had an impact. Specifically, these professional development opportunities for coaches targeted faceto-face engagement with the intervention groups coaches, whereby there were consistent visual demonstrations on the specific FMS performance criteria and coaching cues. Additionally, the G4G resource pack and electronic video resources may have assisted the coaches in understanding FMS, as these resources further reinforced the key coaching skill performance criteria.

The provision of Continuous Professional Development (CPD), similar to the Robbins et al., (2013) 'Girls on the Move' intervention protocol was adhered to in terms of discussing issues, and reinforcing G4G policies and procedures, which may have been a contributing factor to the coaches understanding and delivery of FMS. A similar approach was utilised by Cohen et al. (2015), with 460 children (54.1% girls; age  $8.5 \pm 0.6$  years) in the Supporting Children's Outcomes Using Rewards, Exercise and Skills (SCORES) intervention, where teachers learnt about FMS and were instructed to provide students with specific feedback (Cohen et al., 2015). The literature widely acknowledges that youth PA interventions through the provision of non-formal coach education, such as CPD opportunities (training, resource manual, and opportunities to work with coaches and instructors) can positively support programme implementation, and childhood engagement in PA (Casey et al., 2014; Koh et al., 2014).

In addition to this, the inclusion of the G4G FMS dance provided an additional opportunity for participants in intervention group 1 to practice basic FMS (e.g. coordination, skipping), combined with the sport-specific LGF skills into a fun dance sequence, using a prevalent song in the charts as an auditory stimulus. Participants in intervention group 1 spent an extra 30 minutes of dance practice (from weeks 1 to 8) each week before the scheduled G4G on-pitch session. The culminating FMS and sport-skill dance routine was performed to coaches and parents on week 8 of the programme. Participation in rhythmic activities, such as dance, has shown to aid in the ease and efficiency of performing many FMS, by enhancing the fluidity of movement patterns (Sport for Life Society, 2016). For example, findings from an 8-week intervention study (Lykesas et al., 2014) among 7 to 9-year-old primary school children in Greece confirmed the crucial role of the dance programme for the development of kinetic education. Creative dance is a pedagogical approach of exploration and experimentation, it teaches children to use their creative abilities that lead them effortlessly in the development of movement abilities, and it also increases the availability and willingness of children to engage in motor activities (Cheung, 2010). In the present study, the dosage of FMS specific time for the intervention group participants [including the weekly 10-minute FMS station (100 minutes) and FMS dance practice and performance (300 minutes)] totalled 400 minutes throughout the 10-week intervention duration. Therefore, the research-informed intervention group 1 were exposed to a higher dosage of FMS activities, and it seems likely that they would improve their overall FMS proficiency to a higher level than those in intervention group 2 or the control group 3.

Further to this, Bryant et al. (2016) found that a 1 day per week PE intervention, focusing on FMS, resulted in the improvement of motor skills in children (mean age  $\pm$  SD = 8.3  $\pm$  0.4 years) within the intervention group (n=82) relative to children in the control group (n=83). Most recently, Costello & Warne (2020) showed that a four-week fundamental motor skill intervention effectively improves motor skills in 8 to 10-year-old Irish primary school children (n=100). Findings from previous community sports-based youth interventions provide evidence that FMS supporting strategies for coaches, including the implementation of a station-based format, and a TGfU pedagogical approach may be an effective way to improve 1) FMS competencies for children; 2) deepen children's knowledge

of game tactics; and 3) heighten overall enjoyment levels for children (Butler & Griffen, 2010; Bryant et al., 2016).

The present results revealed that participants in the intervention group also reported small, but significantly greater scores in some of the psychological **c**orrelates (see Table 4), including their SE and perceived PSC over time when compared to other two groups. Coaches in the intervention group were encouraged to utilise purposeful praise, positive reinforcement and performance-related feedback for the duration of the 10 weeks, which was highlighted as part of the CPD offering. Bandura (1997) posited that SE, as a product of a complex of self-persuasion, relies on the cognitive processing of diverse sources of efficacy information, including mastery experiences, vicarious experiences, and verbal persuasion. Performancerelated feedback given by coaches has the potential to have a profound effect on players' performance, as well as on their perception of the motivational climate (Staley & Moore, 2016). It would seem likely that participants in the intervention group were more susceptible (as observed) to heightening their SE and perceived PSC levels, than those in the comparison and control groups.

Significant increases in participants PA enjoyment and attitudes was evident among the intervention group from pre- to post-intervention, in comparison to the two other groups. One explanation for this may have been the integration of the weekly 10-minute 'Team Challenges' fun station. This intervention component promoted the sense of inclusiveness, friendship and belonging to participants, while at this station, coaches also provided feedback to each of the participants in a fun non-competitive environment. Additionally, numerous participants cited that the 'G4G Dance' was 'the most enjoyable' part of the programme. The need for relatedness (SDT) and feeling connected to others (peers) can help cultivate the development of intrinsic motivation (Crespo et al., 2013), for increased PA participation (Bortoli et al., 2018; Timo Jaakkola et al., 2017; Vitali et al., 2019). Previous studies have acknowledged that the experiences of enjoyment are a critical factor in motivating young individuals to continue participating in PA (Brazendale et al., 2015; Cairney et al., 2012). It is anticipated that those who enjoy the benefits of physical activities have a favourable and positive attitude toward PA participation (Chadwick, 2011). This was evident in the current study with the intervention group 1 showing significant gains in PA attitudes in comparison to the intervention group 2 and the control group 3 conditions. Within the PA context, enjoyment represents a

positive attitude toward PA practice (Grasten et al., 2012) and constitutes one of the most important correlates for PA participation (Garn et al., 2019).

Significant increases in participants perceived peer and familial social support was evident among the intervention group 1 from pre- to post-intervention, in comparison to the two other groups. Social support was a key component of the research-informed intervention group 1 for participants (via positive reinforcement and praise). Peer influence within the G4G research-informed intervention may have been influenced through 1) modelling of PA (Team Challenge station), and (2) coparticipation (TGfU stations) with participants. Within the research informed G4G intervention group, parents undertook the following i) co-participated in the LGF skills with their daughter (G4G Skill Cards), ii) attended a 'Parent's Evening' educational workshop on week 4 (led by the lead researcher), and iii) were provided with a communicative support structure platform (G4G Facebook page and What's App group) for the duration of the 10-weeks. The co-participation (e.g. performing physical activities together) and modelling communicative support structures may have led to increased parental social support and involvement with their daughter's engagement in the programme. Seabra et al. (2012) study, examining correlates of PA in schoolchildren aged 8-10 years, found perceived acceptance by peers in sports, and parental encouragement was positively related to PA in girls (n = 683).

The focus group findings elicited children's views on the G4G intervention acceptability (structure, content and delivery, data), with qualitative findings revealing the importance of incorporating multiple viewpoints for programme sustainability. Intervention participants associated the G4G activities/programme with positive learning, having fun with friends, and they felt fit, happy and confident (in terms of skill execution) following the programme completion. Most intervention participants expressed that the G4G dance was one of their highlights about the programme, which provided an appropriate medium to increase girls' PA and effectively apply the SDT (Jago et al., 2013). The 'other' appealing and '*not necessarily football*' activities (i.e. the FMS and Team Challenge stations) encouraged the social element of PA participation and may have played a contributing role to increase the intervention participants perceived peer support (Maturo & Cunningham, 2013). Most children felt that the activities challenged them and were enjoyable (particularly the G4G dance), where they acknowledged liking the novelty of the activities, and the station-based approach. Participants, however,

did request that more station time was needed, particularly to play matches and games. These findings are consistent with Jago et al. (2009) and Stearns et al. (2019), in which the authors both identified fun and friends as key motivators for PA participation among young girls. Research postulates that children like doing PA with friends rather than alone (Sanders et al., 2014) and describe their friends as influencing their PA in various ways, such as modelling, co-participation, and encouragement (Jago et al., 2009; Bukowski et al., 2011). Seabra et al. (2012) study examining correlates of PA in schoolchildren aged 8-10 years, found perceived acceptance by peers and parental encouragement were both positively and significantly related to PA in girls (n= 683). Research has shown that the presence of peers being physically active has resulted in an increase in PA among children and adolescents from the ages of 3 to 15 years old (Lau et al., 2016; Salvy et al., 2012).

To ensure the programme is enjoyable going forward, the girls outlined that a key feature of the intervention structure is the provision of taster sessions, whereby introductory LGF sessions are provided in schools and clubs, prior to the programme rollout. Girls believed that the taster sessions, alongside the G4G programme itself, would be better with the inclusion of female 'role-models. Providing primary school girls with active role models and links to existing community activities could positively influence girls PA in combat to the age-related PA decline in adolescence (Morgan et al., 2019). This was reflected in the present study, when social agents, such as the G4G coaches/coordinators, peers and role-models structured the PA environment to maximise feelings of competence and personal mastery, with children more likely to enjoy the activity, maintain interest in involvement, and commitment to the activity (Granero-Gallegos et al., 2017).

Additionally, some participants expressed interest in being involved in the PA station choices. Providing girls with such PA ownership is consistent with multiple principles of SDT, including intrinsic motivation, relatedness needs, autonomy support and autonomous motivation (Ryan & Deci, 2000). It has been noted in research that involving children directly in activities which they have choice and control are often the most enjoyable (Brooks & Magnusson, 2007). This is an important consideration for future PA promotion programmes, as a lack of choice can be one of the main reasons that girls disengage from PA (Mitchell et al., 2013).

Limitations of the current study include the relatively small sample size of the intervention club, along with the non-randomised controlled research design,

which limits the generalisability of the findings. The qualitative nature of the FG interviews conducted by the lead researcher who was actively involved on site in the intervention may have led to socially desirable answers, meaning the G4G participants give more favourable responses to the question. The novelty of participating in intervention group 1 or intervention group 2 may have demonstrated a heightened positive attitude towards PA and its correlates. These intervention conditions may have increased the participant/coach engagement with the intervention and influenced the findings in unknown ways. The study design was limited to a subjective method for the assessment of participants PA; a future recommendation would be to use more objective measurements of PA, such as accelerometers, or wearable technologies.

A strength of this study is the enhanced ecological validity, stemming from the naturalistic community sports based G4G intervention setting. Finally, a further strength of the study is the use of simple, innovative and feasible intervention strategies, based on a sound conceptual framework, which can be adapted in other similar OYS settings.

#### 7.7 CONCLUSION

The results of this study indicate that the 10-week G4G multi-component community sports-based intervention led to improvements in i) self-reported PA levels, ii) overall FMS proficiency, and iii) several psychological correlates of PA including; perceived SE, perceived PSC, PA enjoyment, attitudes toward PA, and perceived peer and parental support. The current intervention evaluation trial of this G4G protocol study presents, as far as the authors are aware, the first three-armed intervention effectiveness study to be conducted in an Irish OYS context, designed to increase participants MVPA, FMS proficiency, and psychological wellbeing among 8-12-year-old girls. This G4G NRCT has provided sufficient evidence regarding the implementation of a multi-component, community-sports based intervention, and can be used as a starting point to inform the development of theory-based interventions targeting PA promotion for pre-adolescent girls in Ireland. Examining the future efficacy of the G4G programme, as part of a larger trial would seem prudent in terms of the gathering sustainable, longitudinal evidence.

Acknowledgements: The authors wish to acknowledge the help of the undergraduate students from the Sports Studies and Physical Education undergraduate degree programme who were involved in the data collection. Sincere thanks also to the participants, parents, and coaches from the community sports club involved in this pilot intervention phase of the study. The authors would also like to express gratitude to the National Ladies Gaelic Football Association (LGFA) in Ireland for their continuous support and assistance.

#### **Declaration of interest statement**

The authors declare that there are no conflicts of interest within this manuscript publication. The current research is funded through the Government of Ireland Irish Research Council Postgraduate Scholarship Scheme (October 2018-October 2019).

### CHAPTER 8

### DISCUSSION, CONCLUSION, AND FUTURE DIRECTIONS OF 'GAELIC4GIRLS



#### **8.1 INTRODUCTION**

The G4G research study was an exploration into the design, development, implementation, and evaluation of a community sports-based PA intervention for Irish 8 to 12-year-old girls. Specifically, the purpose of this thesis was to investigate whether a multi-component intervention would have the potential to increase PA and FMS levels, and improve psychological correlates of PA in 8- to 12-year-old girls. This chapter aims to provide an overview of each study, by discussing and highlighting the contributions these findings make to the existing evidence base. The chapter will further identify the strengths and limitations of the thesis, and outline final recommendations for future research, policy and practice.

#### **8.2 THESIS OVERVIEW**

The G4G research is a response to the recently documented low levels (self-report and accelerometer) of childhood PA participation (Aubert et al., 2018; Guthold et al., 2019; Tremblay et al., 2016) worldwide. In an Irish context, these low levels of PA participation are consistent with the literature, with only 17% of children aged 9- to 11- years old, and 10% of adolescents aged 12- to 17- years old reporting to meet the PA guidelines for health (Woods et al., 2018). Further to these low levels of PA participation, global and Irish specific evidence has shown a consistent gender-based disparity in youth PA participation, whereby girls are significantly less active than boys (Pearce et al., 2012; Silva et al., 2013; Telford et al., 2016; Van Hecke et al., 2016). In a first step to address these alarming low levels of youth PA participation among young girls, a feasible, cost-efficient, and sustainable exploratory intervention (G4G) trial was carried out in Ireland between 2017 and 2018.

Much research for youth PA promotion suggests that multi-component interventions are effective (Camacho-Miñano et al., 2011; Pearson et al., 2015; Russ et al. 2015), with community sports-based settings providing an opportunity to increase avenues for PA sustainability among youth, particularly vulnerable at-risk girls (Finch & Donaldson, 2010; Hebert et al., 2015; Sims et al., 2015). The structure of the G4G intervention was guided by the literature evidence gathered from previously successful community sport-based interventions amongst young people (Eime et al., 2018; Eime et al., 2015; Robbins et al., 2019).

Chapter 4, the first study in this thesis, was a particularly novel examination of pre-adolescent actual FMS proficiency, perceived motor confidence and competence, and PA levels from baseline data. With many studies now reporting positive associations between actual and perceived FMS and PA in young people (Barnett et al., 2015; Britton et al., 2020; McGrane et al., 2016; Stellino & Sinclair, 2013; Vedul-Kjelsås et al., 2015), chapter 4 was a distinctive study examining this relationship among pre-adolescent girls, prior to the G4G intervention. Although findings indicated no association between PA and self-concept on overall FMS proficiency level, perceived PSC was a significant predictor of the young girls' overall PA levels. Nonetheless, a recent systematic review (De Meester et al., 2020) has reported that the strength of the association between actual motor competence and perceived motor competence/physical self-perception in youth is low to moderate, with data demonstrating that the strength of association does not differ by age, sex, developmental status, or alignment between measurement instruments (De Meester et al., 2020).

Chapter 5, the second study in this thesis, contextualised the development of the G4G intervention for Irish pre-adolescent youth. Within this chapter, the evidence and background to the development of the G4G intervention is given. Findings from the baseline data informed the development of the intervention included low levels of pre-adolescent girls PA, insufficient levels of FMS proficiency, and substantial differences in self-efficacy, attitude towards PA, perceptions of health, and barriers to participation between the classified active and inactive participants. While these findings align with other recent research (Belton et al., 2014; Bolger et al., 2018; Connolly et al., 2020; Mikaelsson et al., 2020; Somerset & Hoare, 2018), this current study reinforced the need for a specifically tailored pre-adolescent PA intervention. From these specific baseline findings, the G4G pilot intervention was constructed based on the SDT and SEM theoretical frameworks (Deci & Ryan, 2000; Stokols, 1996).

Chapter 6 provided a detailed insight into the theory-based design protocol of the G4G intervention study, namely the G4G participant, coach, parent/guardian, and intervention components. This chapter laid the foundations for the overall evaluation of G4G, as described in Chapter 7. Chapter 6 introduced the reader to the originality of the G4G intervention; with a specific focus on the theoretical and pedagogical

coaching strategies with the aim of increasing PA, FMS, and psychological correlates among 8 to 12-year-old girls within the club-based environment. Similar theoretical and coaching pedagogical strategies (e.g. SHARP and SAAFE principles, TGfU) have been successfully utilised in previous PA and sport studies among young girls (Casey, et al., 2014; Shannon et al., 2018; Tibbitts et al., 2019). By documenting each pillar associated with G4G, the reader was provided with an overview of this multi-component intervention.

Chapter 7 assessed the effectiveness of the 10-week multi-component community-sports based G4G intervention for 8- to 12-year-old girls at pre- and post-test stages. The three-armed experimental study involved one intervention group who received the research informed G4G intervention over the course of 10weeks (March-June 2018), a second intervention group who used the existing 'traditional' national G4G programme structure, and one control group who received their usual LGF training sessions for the same period without any researcher input or influence. It was encouraging to observe through the thesis results that over the 10weeks of the intervention, there was an increase in the intervention group's selfreported daily MVPA. Furthermore, a significant improvement in overall FMS proficiency for those exposed to the G4G intervention, relative to their control counterparts was detected. Research has found that community sports-based interventions with coaches adequately trained in player-centred instruction can significantly improve the PA levels (DeBate & Bleck, 2016) and FMS competency (Lander et al., 2017) of early years adolescent girls.

This preliminary G4G trial provides support for the simultaneous integration of PA, FMS, and psychological wellbeing through a theoretical and pedagogical approach during weekly LGF training sessions amongst Irish 8 to 12-year-old female youth. Yet, as is acknowledged and discussed in chapter 7, the relatively small sample size within the three clubs must be considered a limiting factor when interpreting these findings. In addition, it is recognised that while preliminary research findings may be valid in the Irish pre-adolescent LGF sporting setting, without further intervention examination, these results must be interpreted with caution. For these reasons, a definitive randomised controlled trial (RCT), with a larger sample is recommended to confirm results.

#### **8.3** CONTRIBUTION OF THESIS TO THE LITERATURE

The studies within this thesis are very much original and nuanced within youth PA and community organised sport participatory research. This thesis makes several contributions to the literature and throughout this chapter, the following will be discussed: 1) integration of pedagogical coaching frameworks as an avenue for increased PA and motor skill opportunities, 2) emphasis on a psychological value and needs-driven approach for intervention design, and 3) the impact of social support by significant others (peers, parental and role model) on 8 to 12-year-old girls' PA and sport participation.

## **8.3.1** Pedagogical coaching frameworks: an avenue for increased physical activity and motor competence in 8 to 12-year-old Irish girls

An important finding across studies in this thesis is the girls' consistent failure to meet the minimum daily PA recommendations for health, with a significant majority also not achieving the FMS proficiency expected for their age (Chapter 4, 5 and 7). This evidence is in agreement with previous representative samples of Irish primary school children (Behan et al., 2019; Bolger et al., 2018; Woods et al., 2018), and consistent with global data on estimates of female youth PA (Demetrio et al., 2019; Owen et al., 2017; Bardid et al., 2016; Bryant et al., 2014;), providing further arguments to find strategies for enhancing girls' PA levels and motor skill proficiency levels in organised sports. For this reason, pedagogical sports coaching approaches were included in the NRCT in Chapter 7, which were underpinned by theory along with participant suggestions from the baseline studies (Chapter 4 and 5).

As highlighted in Chapter 2, a recent systematic review and meta-analysis of PA community sports-based interventions among girls (Pearson et al., 2015) found that greater intervention effectiveness appear to result from interventions that include a non-formal coach education element (Koh et al., 2014). The combination of the two specific 'SAAFE' (Lubans et al., 2017) and SHARP (Powell et al., 2016) pedagogical coaching principles as part of the G4G programme (Chapters 6 and 7) proved successful in heightening the participants self-reported perceived PA and enjoyment levels. To date, very few published interventions give clear details on

how specific theory-based pedagogical coaching frameworks were used to guide the design of community sports-based interventions.

Additionally, participants in the intervention group (as opposed to the control groups) were exposed to a variety of FMS, and LGF sport-specific skills, through a TGfU instructional model (Sheppard, 2014), using a rotatory station-based approach. Each 10-minute station allocated a significant amount of time to learning the skills of LGF via modified and small-sided games (Slingerland et al., 2014). The integration and implementation of non-formal CPD coach education opportunities (training, FMS supporting strategies, resource manual, and opportunities to work with coaches and instructors), and a TGfU pedagogical approach in previous community sportsbased youth interventions may be an effective way to improve 1) FMS competencies for children; 2) deepen children's knowledge of game tactics; and 3) heighten overall enjoyment levels for children, and childhood engagement in PA (Sierra-Díaz et al., 2019; Casey et al., 2014; Bryant et al., 2016).

Successful interventions that provided training with fun activities and appropriate pedagogical curricula were encouraged by Cregan, 2020; Hardy et al., 2012; Mazur et al., 2020; Owen et al., 2018; Van Beurden et al., 2003. For example, González-Valero et al. (2020) recent 'Rhythmic Gymnastics' intervention study among 104 11- and 12-years old girls (11.66  $\pm$  0.47), reported more optimal levels of physical self-concept and PA enjoyment post-intervention (17 sessions). González-Valero et al. (2020) coaching strategies, particularly the task (designing activities), grouping (working groups), evaluation (assessment criteria) and time (pace of learning and teaching) align with the pedagogical approaches used in the G4G intervention. Similarly, in the Bristol Girls Dance Project (BGDP), the SDT-focused element of the training explored the practical application of the theory to dance sessions and proved successful in improving girls (N = 281) reported PA enjoyment levels (Sebire et al., 2016). The BGDP 20-week school-based intervention, consisting of two 75 min after-school dance sessions. The SDT-focused element of the instructor training and session plan manual explored the practical application of the theory to dance sessions. Similar to the G4G study, the instructors were provided the opportunity to use autonomy-supportive styles of instruction, seek clarification and obtain feedback from study researchers (Edwards et al., 2016).

There are currently no evidence-based studies regarding coach education and the provision of coach CPD opportunities to increase PA and FMS opportunities within the existing LGFA organisational context in Ireland, thus contributing to new knowledge within a Gaelic Games context. The G4G intervention provided nonformal learning methods within coach education (such as continuous professional development (CPD and weekly workshops) (Maclean & Lorimer, 2016) and addressed specific topics that are more authentic, meaningful, and contextualised to the coaches within the LGF sporting culture (Mallett et al., 2009).

# 8.3.2 Psychological correlates of PA – knowing your target audience for intervention design

#### 8.3.2.1 'Girls just wanna have fun': capitalising on PA enjoyment

A key finding within this thesis is the 'fun factor' associated with 8 to 12-year-old girls' sport participation. The most frequently cited reason among pre-adolescent Irish girls within this thesis (Chapter 5 and 7) indicates that fun, enjoyment, and social satisfaction are the strong motivators, and lack of fun is a primary reason for sport attrition/drop-out. Lack of enjoyment was the most frequently cited reason among 8 to 12-year-old girls for dropping out of sport, expressing feelings of being bored and not fulfilled; along with having a dislike for the teacher/coach and a lack of support/encouragement from coach/teacher. These findings are in line with current research on female youth revealing that having fun through feelings of acceptance, achievement, involvement and social interactions are significant motivators for female youth PA and OYS participation (Corr & Murtagh, 2020; Enright & O' Sullivan, 2010; Farmer et al., 2018; Pawlowski et al., 2014). Additionally, the absence of fun is noted as a common reason for youth sport attrition globally (Visek et al., 2015; Witt & Dangi, 2018). In chapter 5, findings indicate that the 'competitive' nature and the idea of winning was not as critical and could even spoil the fun and, in some cases, may be the cause of discontinued engagement. These findings align with other studies where the 'competitive' nature posed as a barrier in female youth PA and OYS (Corepal et al., 2018; Farmer et al., 2018; Sport Northern Ireland, 2016). Interestingly, team-based competitions have
shown potential to better encourage PA for girls as a motivator not to leave the team down (Corepal et al., 2018).

In chapter 7, most children felt the novel activities challenged them and were enjoyable (particularly the G4G dance). Participants referred to the 'matches' and 'other' non-football activities, such as the specific FMS and Team Challenge stations, including the G4G Dance, as appealing and enjoyable aspects of the programme. The use of game-centred approaches and Sport Education (SE) instructional models (including TGfU) have been found to contribute to students having fun and finding value in organised youth sport (Fry et al., 2010; Georgakis & Light, 2009). Collectively, these findings support the use of a modified games approach (i.e. inclusion of Team Challenges and G4G Dance) and SE models (TGfU) as ways to promote fun through meaningful experiences in youth sport for young girls (Beni et al., 2017).

At present, fun remains a relatively elusive concept (Visek et al., 2015), with limited efforts made to characterise and quantify fun in youth sport, and there is little or no consensus of its meaning in the literature (Bengoechea et al., 2004; Visek et al., 2018). For instance, relatively few studies have attempted to identify specific factors that comprise fun in youth sports (Visek et al., 2018; Witt & Dangi, 2018; 2018; Martin, 2014,) or identify what is done to promote these positive movement experiences. Additionally, there is a lack of longitudinal studies on the evolution of enjoyment in organised sport and factors that are associated with the enjoyment of sport from a longitudinal perspective (Lagestad & Sorensen, 2018). Most recently, Visek et al.'s (2018) 'FUN MAPS' concept provides evidence-based blueprints for the fun integration theory (FIT); this appears to be the only stakeholder-derived and fully conceptualised theoretical framework for understanding fun in youth sport today. On a smaller scale to Visek et al.'s (2018) FUN MAPS, the G4G intervention study (Chapter 7) in Ireland made efforts to characterise and identify quantify fun from a pre-adolescent girl perspective, bridging a gap of standard practice for promoting fun in Gaelic Games youth programmes.

# 8.3.2.2 'I won't because I think I can't': Building confidence and competence in 8 to 12-year-old girls

The G4G intervention study results revealed that participants in the researchinformed intervention group 1 (Chapter 7) also reported small, but significantly greater scores in some of the psychological correlates, including their SE and perceived PSC over time when compared to the other two groups. Additionally, qualitative findings revealed the G4G participants in the research-informed group 1 associated the G4G activities with learning, having felt they increased their perceived confidence levels in skill execution. Specifically, two participants expressed that their fear of catching the ball reduced at post-intervention, while other participants stated that their soloing and kicking skills significantly improved over the 10-week G4G intervention.

FG findings in Chapter 5 reported a lack of confidence in skill execution linked with peer social comparison, as potential barriers for continued intervention uptake or continued sport participation, which are important findings for G4G intervention sustainability. Similarly, in Chapter 7, participants identified skillrelated lack of confidence among girls as a potential barrier for G4G intervention uptake or continued sport participation, which is in alignment with previous research specific to young girl's sport participation (Basterfield et al., 2016; Somerset & Hoare, 2018). In sport and exercise psychology, the physical self-concept has been identified as a psychological component through which motor skill level in childhood reverberates into physically active and healthy lifestyles later in adolescence (Babic et al., 2014). Furthermore, the aforementioned G4G findings provide a comprehensive understanding of the issues surrounding perceived PSC and motivation in LGF among 8 to 12-year-old girls.

## 8.3.3 The impact of social support from 'significant others' on 8 to 12-year-old girls' PA: peers, parents, and positive role-models

#### 8.3.3.1 Peer and parental social support

The G4G intervention programme is unique due to the integration of a combination of social support structures from key stakeholders, namely peers, parents, coaches, and role model LGF players, all contributing to the perceived psychological

enjoyment within the intervention study (Chapter 7). Both quantitative and qualitative findings in this thesis reveal the social support structures around the G4G participants were perceived as enabling or inhibiting factors for continued sport participation (Chapter 5 and 7). Specifically, FG findings in Chapter 5 illustrate that positive outside encouragement/support from family, friends, and coaches act as models of motivation, encouraging PA engagement among the female participants. Self-report and FG findings within this thesis (Chapter 5 and 7), however, do indicate that social pressures were reported as participatory barriers, and these findings are in line with previous research on female youth (Perry et al., 2011; Verloigne et al., 2016). The G4G social support structures facilitating (positive encouragement/praise) or impeding engagement (negative coach relationship) among 8- to 12- year-old female youth are deemed important findings in the context of this G4G study and are consistent with recent studies (Battaglia et al., 2017; Fenton et al., 2017; Laird et al., 2018).

Research postulates that coach, parent and peer support play a significant role in shaping youth sport experiences both from a positive (athlete motivation levels) and negative (drop-out) perspective (Howie et al., 2020; Sheridan et al., 2014). The social support structures implemented in G4G mirrored the youth sport triangle or triad (Lisinskienė & Šukys, 2014), which is a conceptualisation of the total youth sport experience of the G4G participant. The athletic triangle (participant, coach, and parent components) can influence the young athlete's decision to participate in sport, to remain in the sport, and to pursue a sporting career (Lisinskiene et al., 2019). The research informed G4G intervention is made up of the three intervention components, the coach, the parent(s) or guardian(s), and the athlete, and active participation of all three members is important and necessary for the positive development of the intervention's success. Positive relationships with teammates and friends are associated with favourable self-perceptions, emotions, selfdetermined motivation, and moral development (Smith & McDonough, 2008).

In Chapter 7, intervention participants associated the G4G activities/programme with positive learning and having fun with friends. Participants also alluded to the fact that social acceptance among friends may act as a barrier to G4G intervention and continued sport participation. As outlined in Chapter 2, it has been well-established through a detailed literature review, that peers and parents play

many and important roles in organised youth sports. In addition, children who have supportive friends and peers have also been found to be more physically active (Laird et al., 2018; Fitzgerald et al., 2012).

Increasing the PA levels of pre-adolescent girls is challenging, with most efforts to date focusing on school-based settings in older adolescent populations. The potential combined positive influence of key stakeholders, including parental and peer support structures, through co-participation within a youth sporting environment is a possible avenue for increased PA among young girls. The rationale for a co-participatory approach to sport participation and promotion is due to the family and peers being considered as important influencers for shaping children's PA (Timperio et al., 2013). The influence of parents on young girls' PA level in the G4G intervention programme occurred through modelling and co-participation (Madsen et al., 2009), but especially through supporting them, through encouragement or through the provision of logistical support (Beets et al., 2010; Sebire et al., 2013). Similarly, a co-participatory approach among peers was evident in the G4G research-informed intervention via the station-based activities (Team Challenge station). Parents and peers PA through the modelling of PA or sport (Demetriou et al., 2019); and co-participation with children (Cleland et al., 2011), along with peer support (Laird et al., 2018) can play an extremely important role in shaping the PA behaviours of young girls.

8.3.3.2 Role-models as significant influencers for increased participation in OYS The impact of other role-models, specifically OYS coaches and 'elite' LGFA players as potential enabling and inhibiting factors for continued sport participation is another important finding revealed within this thesis. As an enabling factor, the qualitative findings in Chapters 5 and 7 highlighted the important motivational role 'elite' LGF players have on 8 to 12-year-old Irish girls within the LGF context. In Chapter 7, the G4G participants from the research-informed intervention group recognised the positive impact of the interaction between the G4G participants when the 'elite' athletes visited the revised intervention club. It has been suggested that role models can inspire young people to become involved in, or maintain involvement in, sport and PA (Adriaanse & Crosswhite, 2008). The World Health

Organisation (WHO) specifically recommends the use of role models within local communities to increase PA among females (Bailey et al., 2005). To this end, the G4G findings provide a further understanding of the positive influence LGF female players as role models can have on Irish 8 to 12-year-old girls.

Researchers have highlighted that youth sport participants are most likely to experience positive developmental outcomes when interactions are characterised by positive and informational feedback, appropriate role modelling, and autonomy-supportive engagement styles (Atkins et al., 2015). Research is needed to test the effectiveness of social support interventions in youth sport (Balish et al., 2014). To date, few studies (Conroy & Coatsworth, 2007; Lavallee et al., 2019) have used social support theoretically informed intervention-based approaches to assessing the effectiveness of a specific intervention or training programme for providing effective support to youth athletes. Investigating girls' perspectives and experiences of social support provided by significant others for PA behaviour would improve our understanding of how social support influences PA behaviour, which in turn could inform the development of PA intervention strategies aimed at engaging inactive girls in PA (Laird et al., 2018).

#### **8.4 STRENGTHS AND LIMITATIONS**

The strengths of this thesis include the formative and pedagogical approach to intervention design, guided by appropriate theory. The detailed account of how the theoretical frameworks were used in intervention design along with the use of simple, innovative, and feasible intervention strategies, acknowledges the importance of comprehensive multi-component interventions in OYS settings. A novel and innovative intervention was designed and developed for 8 to 12-year-old girls in the Irish LGF sporting context, as guided by a thorough examination of the evidence-based literature, and by the baseline studies (Chapter 4 and 5) to inform decisions on the G4G research.

The mixed-methods approach within the studies is also a strength, as the qualitative data provided important context and insight into the quantitative data collected (Choy, 2014). The qualitative data in the FGs were collected as a method of involving participant opinions to inform the design of a revised community

sports-based intervention, that otherwise may not have been detected through quantitative data alone. Additionally, as mentioned in Chapter 3, qualitative data aided in the exploration of PA enablers and barriers, which was necessary to ensure appropriate intervention functions and pedagogical coaching techniques were being considered in the revised G4G intervention. The use of practical objective and subjective measurement tools, specifically the PAQ-C, TGMD-2, GSGA, and numerous psychological instruments were reported to have acceptable validity, reliability, and practicality for use with children and adolescents (Benitez-Porres et al., 2016).

Furthermore, the study is conducted in a real-life 'Gaelic Games' sport setting, where the usual participants, coaches, and parents, under expert guidance from the research group, carried out the implementation of the intervention components. This also means that any findings of the trial are more generalisable because they are performed in the setting in which they are most likely to be implemented. This complies with the need for more effectiveness trials, defined as interventions delivered under real-world conditions (Glasgow et al., 2003).

In addition to the individual study limitations described in previous chapters (4 to 7), the following overall thesis limitations should be noted. An important limitation that must be acknowledged is the small sample size used within the studies, as described in this thesis. These small sample sizes result in difficultly generalising results to a wider population, as well as issues with assessing effectiveness, due to a lack of power. The lack of follow-up retention assessment is an important limitation which could have provided a greater understanding on the sustainable impact of the G4G intervention over time. A disadvantage of the 'real life' community sports-based setting is that it may lack breadth; gathering very detailed information means being unable to gather data from a very large number of people or groups. In Chapter 7 for the exploratory trial, in terms of matching criteria, the three clubs were selected for inclusion based on geographical location; in terms of sample size, the control club was not an exact match to the intervention. A followup assessment, however, was not feasible in the timeframe of this PhD project. It must be noted that all participants volunteered to participate in the G4G intervention study (Chapter 7), and therefore had an interest in increasing their PA levels and may not be representative of the true Irish population.

Additionally, the PA measurement outcome variable was limited to a subjective method for the assessment of participants' PA; a future recommendation would be to use more objective measurements of PA, such as accelerometers, or wearable technologies. The FMS inter-rater reliability process used within the studies may have given rise to subjectivity among the raters according to different interpretations of each skill evaluation criteria. The younger girls (8-9 years) within the studies may have produced less reliable responses in the questionnaire than the older girls, as cognitive ability between the 8 to 12-year-old girls differed. Furthermore, these limitations should be addressed in a full RCT, as discussed in the future research recommendations section.

#### **8.5 RESEARCH, POLICY, AND PRACTICE IMPLICATIONS**

#### 8.5.1 Future directions of G4G

The revised G4G intervention study, as evaluated in chapter 7, was a quasiexperimental non- randomised controlled trial. Following the MRC (2000) framework (as discussed in Chapter 2) for complex interventions, this research study has undertaken the theory (pre-clinical), modelling (phase 1), and exploratory trial (phase 2) phases of investigation in the evaluation of G4G. Following from the findings of the exploratory trial, it is important that the G4G study progressively expands to the definitive randomised controlled trial (RCT) (Phase 3). This next phase will compare the fully-defined G4G intervention to an appropriate control alternative using a protocol that is theoretically defensible, reproducible and adequately controlled with appropriate statistical power (Medical Research Council, 2000). Following the research presented in this thesis, the following are this researcher's recommendations for future directions:

• A fully powered randomised controlled trial of the G4G programme should be conducted to assess the effectiveness of the programme to increase the PA, FMS proficiency levels and psychological wellbeing of 8 to 12-year-old girls. A future trial should also aim to include a larger sample of participants, from clubs situated in different areas (rural and urban) throughout the varying regions of the country to reach a wider population (four provinces in Ireland), and include pre-post and follow-up testing.

- Evaluate G4G parental and coach perceptions of the G4G programme (intervention design/acceptability), eliciting their perspectives on the programme through qualitative analysis. Including active participation of all three G4G user groups is central to a phased approach to complex intervention design and is deemed essential to their success (Davison et al., 2013).
- Assess the impact of peer, parental and LGF player role-model support structures on G4G participants PA and psychological wellbeing across the duration of the programme for the duration of the programme, using both intervention and control groups.

Along with research recommendations, the results of this thesis present several recommendations for both policy and practice. As researchers, we must look to systematically build a culture that fosters young girls' involvement in sport and PA. This requires all those interested and involved in sport and PA provision for young people to look at current policies and practices and mount a cohesive challenge to existing practices.

#### **8.5.2 Implications for policy**

The importance of sport and PA in the Republic of Ireland is reflected by recent changes in policy, specifically related to the development of the National Physical Activity Plan (NPAP) 'Get Ireland Active!' (Department of Health, 2016) and the National Sports Policy (Department of Transport, Tourism, and Sport, 2018). The following recommendations, relating to implications for policy and practice, are emanating from the findings of this thesis:

Ireland's NPAP consists of some key guiding principles, including creating increased opportunities for people to be active every day and to create a supportive environment for PA participation (Department of Health 2016). The overarching target of the plan is to 'increase the proportion of the population across each life stage undertaking regular physical activity by 1%

per annum across the lifetime of Healthy Ireland' (p.13). The G4G findings within this thesis align with three of the seven Action Areas, specifically Action Area 2 – dealing with Children and Young People, Action Area 6 – Sport and PA in the community, and Action Area 7, Research Monitoring and Evaluation. Aligned with Action Area 2 of the NPAP, dealing with Children and Young People, Sport Ireland and Sport Northern Ireland are leading out on the development of a physical literacy (PL) position statement, and this project was commissioned for inception in 2018. The findings of this thesis on the G4G research programme are part of this PL study, and will contribute to developing Ireland's first-ever PL consensus. Developing a strong consensus around PL throughout the sport, physical education (PE) and PA systems are regarded as a key potential enabler in efforts to achieve the desired policy objectives around participation, including addressing the drop-out rates from PA and sport participation of young girls in Ireland (Department of Health, 2018).

- Sport Ireland, with the support of the Dormant Account Fund (DAF), have developed, in conjunction with the Department of Children and Youth Affairs, Hub na Óg, and the National Governing Bodies (NGBs) of Sport, a toolkit to give volunteers and members of staff in sporting organisations a clear understanding of what they need to do to involve their young members in decision-making from club level to NGB level. The value needs-driven approach of G4G programme and findings could inform the continued development of the toolkit, by encouraging sporting organisations to find out what is important to their young members in order to support their involvement in decision-making and to consult with them in more depth on any topic. The studies within this thesis, particularly Chapter 5 and Chapter 7 illustrate the importance of listening to the pre-adolescent voice, particularly when providing them with opportunities to be active.
- One target area of the Sport Ireland policy on Women in Sport (reflective of the National Sports Policy 2018-2027 and National Strategy for Women and Girls 2017-2020) is 'Active Participation'. In order to reduce the drop-out rate from PA and sport in young girls, the policy aims to promote the establishment of a 'girls in sport toolkit' which better assists coaches in

supporting young girls who are at risk of drop-out. Findings within this thesis, specifically the importance of highlighting the psychological and social support engaging and retaining young girl's in community sports-based settings has been consistently documented as important for PA participation for girls. The research-informed G4G intervention and future iterations of G4G can contribute to these actions in supporting the establishment of the Sport Ireland 'girls in sport toolkit'.

• Another objective of the Women in Sports Policy is to invest in participation programmes which lead to sustained engagement in sport by women and girls, specifically to address transitions and dropout of PA and sport participation. Aligned with Action Area 6 of the NPAP (Sport and PA in the community), there is scope for the revised research-informed G4G programme, in conjunction with the LGFA, to be adopted for implementation at local sports partnerships (LSP) across the country. If LSPs adopted the programmes such as G4G as part of their Women in Sport initiative, then the programme could potentially be delivered to more clubs and schools with a wider range of participants across the country to improve PA and health behaviours as well as increase sport participation. This is important as G4G has impacted over 500 LGF clubs to date and its reach continues to grow.

#### 8.5.3 Implications for practice

## 8.5.3.1 A mentor-based approach building role models within the G4G and LGF community

The findings underline the importance of the role of female role models and the necessity for organisations to build strong links between the G4G programme and the senior county players. The findings from this study contend that inviting established elite LGF players as part of the programme (week 8) to distribute G4G certificates of completion, pass down their skill-based knowledge, and share their playing experience could play a crucially important function in the improvement of 8 to 12-year-old girls psychological wellbeing and enjoyment of LGF. The possibility of including existing county LGF players as a fourth component to the G4G programme may prove successful in positively influencing young girls.

The current 20x20 campaign, presented by the Federation of Irish Sport and supported by Healthy Ireland is championed by Ireland's NGBs and LSPs, is calling on the people of Ireland and all those involved in Irish sport and PA to get behind female sport in a concerted effort to increase media coverage, boost attendances and ultimately, grow involvement in female sport and PA by 20% by the end of 2020. International strategies and research reports, which aim to improve female participation, acknowledge the importance of female role models as a significant factor (Lawler et al., 2020; Dunn, 2015). The G4G thesis findings complement the current 20x20 campaign by further emphasising the need to incorporate female role-model support structures within the existing school and community youth sports-based programmes in Ireland.

# 8.5.3.2 The school-community link – educating and empowering teachers and students

Findings from this thesis may inform further development of the existing LGFA TEACH Programme – a coaching course designed for Primary and Post Primary school teachers interested in coaching LGF, through their PE classes or school teams. The certified course aims to provide training and support to teachers, who impart valuable coaching services to many young female players. The suggestion from G4G participants within this thesis of providing 'taster' G4G sessions within the school environment prior to signing up for G4G may prove successful for future G4G participant recruitment strategies.

Additionally, findings from this thesis may inform the continued development of the existing Transition Year Active School (TY/AS) LGFA Programme available in secondary schools across Ireland TY/AS programme. Specifically, the G4G findings could enhance the TY students understanding of 8 to 12-year-old girls' motivators and barriers to sport participation as part of the 'Basics of Coaching' training day, as outlined in the TY/AS Curriculum. As the transfer for participants from school-based sport to community opportunities is imperative (Eime & Payne, 2009), there is potential for the TEACH and TY/AS programmes to link in with the research-informed G4G programme as a method of introducing young girls to taster' LGF sessions, ensuring the opportunities are tailored to their

needs/interests. Adopting a student-centred approach ensures that all youth receive a broad, accessible and inclusive programme of extra-curricular activities. The studies within this thesis, particularly Chapter 4 and Chapter 7 illustrate the importance of listening to the pre-adolescent voice, particularly when providing them with opportunities to be active. Stronger links between sports clubs and schools can aid the integration of schools into local community life and strengthen local social cohesion. Sports clubs, in turn, can acquire regular structured access to young people in schools and thus enhance their ability to recruit and develop new generations of players and club members (Fahey et al., 2005).

### 8.5.3.3 Enhancing LGFA coach education practices for increased PA levels, movement confidence, competence, and psychological wellbeing

The innovative G4G research-informed resources designed specifically as part of this thesis for the G4G intervention (Chapters 6 and 7), namely the electronic FMS and LGF coaching video clips, the Coaching Manual, and G4G Skill Chart can be shared with existing LGFA coaches, reaching hundreds of coaches and young LGF players around Ireland. The findings of this study highlight the need for consistent (weekly) non-formal CPD coach development support for the duration of the G4G programme. Coaches at the youth level should be encouraged to ground their practice around an athlete-centred model underpinned by elements of the SDT and SEM principles, with the aim of increasing 8 to 12-year-old girls LGF participation levels and psychological wellbeing through research-informed pedagogical coaching approaches. When applied, this approach to coaching is considered a powerful tool in empowering young athletes to learn and become intrinsically motivated, which, ultimately, results in enhanced performance, a thriving, supportive team environment, as well as the development of individual personal assets (Fraserthomas et al., 2010). Such an approach over time may lead to the formation of a specific holistic LGFA policy as part of their vision for increasing player participation and coach educational development.

### **8.6 CONCLUDING THOUGHTS**

The thesis aimed to design, develop, implement, and evaluate the efficacy of a targeted LGF community sports-based intervention, 'G4G', specifically designed to increase PA levels, FMS proficiency and psychological wellbeing of 8-12-year-old female Irish youth. The series of papers presented demonstrate that the integration of specific PA, FMS and psychological wellbeing pedagogical strategies, underpinned by theoretical frameworks, in a 10-week LGF community setting is effective and feasible. To address further gaps in the literature, the baseline studies (Chapter 4 and 5) conducted in three post-primary schools aimed to address current PA levels, movement skill-competencies and psychological correlates of PA among 8 to 12year-old girls to further inform the design of the G4G intervention study. The qualitative synthesis (Chapter 5 and 7) highlighted young girls' perceptions of PA, specifically noting the potential motivators and barriers of girls' participation in PA and sport (Chapter 5), as well as the G4G intervention perceptions and acceptability (Chapter 7). The unique and original component of the G4G intervention has been the integration of PA, FMS and psychological wellbeing in a pedagogically sound manner, by using appropriate theory in intervention design, and incorporating the participant's voice in a community sports-based climate for 8- to 12-year-old female youth. The findings demonstrate that the 10-week specifically tailored, researchinformed G4G intervention is a feasible and efficacious programme and had a positive effect on the physical and psychological wellbeing of 8 to 12-year-old Irish girls.

While small sample sizes limit the generalisability of the results, the results of all four studies of this thesis add significant contributions to the currently sparse evidence base relating to community sports-based interventions for young girls and provide justification for the integration of a theoretically sound multi-component PA promotion intervention this age-cohort within an OYS setting. To better understand the implications of these results, future research should aim to expand on the findings of this thesis using the recommendations for practice, strengthening the evidence base for theory-based PA interventions for 8 to 12-year-old girls. Additionally, the results of this review identify recommendations for future interventions at a local, a national and an international level. For example;

- At a local level, club coaches, teachers, and those involved in LGF session delivery (e.g., schools, local LGF clubs, and local activity providers) should include young people in designing interventions and facilities to ensure they are meeting the needs of this age group and providing the right opportunities for this age cohort of girls to be active. Findings from this thesis suggest the need to tailor PA, PE and sports programmes, specifically to increase selfefficacy and enjoyment of PA among girls (Butt et al., 2011; Dishman et al., 2005; James et al., 2018).
- At a national and international level, PA and sport participation initiatives, such as the G4G programme can help improve uptake, sustainability and overall success of future projects by considering the following;
  - Physical activities should be varied and fun meeting the needs of the agegroup.
  - The role of the parent-guardian should be considered. Involving parents/guardians is a key ingredient for successful implementation.
  - The importance of coach education should be encouraged.
  - Individual, physical, and psychological characteristics should be accounted for, including specific FMS deliberate practice and outcomes such as self-concept, social behaviours, and self-efficacy.
  - Identify and promote positive role models. A variety of positive sporting role models can have a powerful effect on children's attitudes to PA.
  - Barriers should be identified early and a plan to overcome them developed.
  - Focus on fun and friendship; Informal, girls-only physical activity sessions, which focus on fun and enjoyment with friends, are very successful.

It was encouraging to observe that many of the conclusions in studies by Pearson et al. (2015) and Allison et al. (2017) are comparable to the future development of G4G. To move forward, there is a need for the provision of quality sport participation data to supply the evidence to inform well-structured programmes and policies to meet young girls' and community needs. This is not possible if sport participation remains hidden in the broader PA context, or if it is not investigated in more detail (Allison et al., 2017). Examining the future efficacy of the G4G programme, as part of a larger trial would seem prudent in terms of gathering sustainable, longitudinal evidence.

In summary, the development of the revised G4G intervention has been insightful in terms of critically reviewing the literature around the area. The overall research process for designing, developing, implementing, and evaluating this exploratory study has undoubtedly been challenging, laborious, yet ultimately an enriching and rewarding learning experience. On a personal note, during the course of my 4-year PhD research study, despite the complex unpredictable nature of the research process, I have increased my intellectual freedom, self-development, and ability to think critically and creatively in intervention design and scientific evaluation for grass-root community sports-based initiatives. Whilst specialising in the field of PA promotion in Gaelic Games, specifically for female children, I am excited to continue working closely with local community groups (GAA, LGFA etc.), coaches, parents, teachers and children – I will strive to continue educating, empowering and making a difference to the lives of young girls.

'Children are not things to be moulded but are people to be unfolded'

Jess Lair

'Do not follow where the path may lead. Go instead where there is no path and leave a trail'

Ralph Waldo Emerson

### References

- Aadland, E., Andersen, L. B., Anderssen, S. A., & Resaland, G. K. (2018). A comparison of 10 accelerometer non-wear time criteria and logbooks in children. *BMC Public Health*, 18(1). https://doi.org/10.1186/s12889-018-5212-4
- Adamo, K. B., Prince, S. A., Tricco, A. C., Connor-Gorber, S., & Tremblay, M. (2009). A comparison of indirect versus direct measures for assessing physical activity in the pediatric population: *A systematic review. International Journal* of Pediatric Obesity, 4(1), 2–27. https://doi.org/10.1080/17477160802315010
- Addy, C. L., Trilk, J. L., Dowda, M., Byun, W., & Pate, R. R. (2014). Assessing Preschool Children's Physical Activity: How Many Days of Accelerometry Measurement. *Pediatric Exercise Science*, 26(1), 103–109. https://doi.org/10.1123/pes.2013-0021
- Adriaanse, J. A., & Crosswhite, J. J. (2008). David or Mia? The influence of gender on adolescent girls' choice of sport role models. *Women's Studies International Forum*, 31(5), 383–389. https://doi.org/10.1016/j.wsif.2008.08.008
- Aggio, D., Fairclough, S., Knowles, Z., & Graves, L. (2016). Validity and reliability of a modified english version of the physical activity questionnaire for adolescents. *Archives of Public Health*, 74(3). https://doi.org/10.1186/s13690-016-0115-2
- Ainsworth, B. E., Haskell, W. L., Herrmann, S. D., Meckes, N., Bassett, D. R., Jr, Tudor-Locke, C., Greer, J. L., Vezina, J., Whitt-Glover, M. C., & Leon, A. S. (2011). Compendium of Physical Activities: a second update of codes and MET values. *Medicine and Science in Sports and Exercise*, 43(8), 1575–1581. https://doi.org/10.1249/MSS.0b013e31821ece12

Ainsworth, B. E., Haskell, W. L., Whitt, M. C., Irwin, M. L., Swartz, A. M., Strath, S. J., O'Brien, W. L., Bassett, D. R., Jr, Schmitz, K. H., Emplaincourt, P. O., Jacobs, D. R., Jr, & Leon, A. S. (2000). Compendium of physical activities: an update of activity codes and MET intensities. *Medicine and Science in Sports and Exercise*, *32*(9 Suppl), S498–S504. https://doi.org/10.1097/00005768-200009001-00009

Ajzen, I. (1988). Attitudes, personality and behaviour. Dorsey Press.

- Ajzen, I., & Madden, T. (1986). Prediction of goal-directed behaviour: Attitudes, intentions and perceived behavioural control. *Journal of Experimental Social Psychology*, 22(5), 453–474. https://doi.org/10.1016/0022-1031(86)90045-4
- Allison, R., Bird, E. L., & McClean, S. (2017). Is Team Sport the Key to Getting Everybody Active, Every Day? A Systematic Review of Physical Activity Interventions Aimed at Increasing Girls ' Participation in Team Sport. *Public Health*, 4(2), 202–220. https://doi.org/10.3934/publichealth.2017.2.202
- Alvesson, M., & Sköldberg, K. (2000). *Reflexive Methodology: New Vistas for Qualitative Research*. SAGE.
- Andersen, L. B., Mota, J., & Di Pietro, L. (2016). Update on the global pandemic of physical inactivity. *The Lancet*, 388(10051), 1255–1256. https://doi.org/10.1016/S0140-6736(16)30960-6
- Anokye, N., Mansfield, L., Kay, T., Sanghera, S., Lewin, A., & Fox-Rushby, J. (2018). The effectiveness and cost-effectiveness of a complex community sport intervention to increase physical activity: An interrupted time series design. *BMJ Open*, 8(12), e024132. https://doi.org/10.1136/bmjopen-2018-024132

- Ary, D., Jacobs, L. ., Razavieh, A., & Sorensen, C. . (2010). Introduction to research in education (8th ed.). Hult Rinchart & Wiston.
- Atkins, M., Johnson, D., Force, E., & Petrie, T. (2013). "Do I still want to play?"
  Parents' and peers' influences on girls' continuation in sport. *Journal of Sport Behavior*, 36(4), 329–345.
  http://connection.ebscohost.com/c/articles/91815436/do-still-want-play-parentspeers-influences-girls-continuation-sport
- Aubert, S., Barnes, J. D., Abdeta, C., Nader, P. A., Adeniyi, A. F., Aguilar-Farias, N., Tenesaca, D. S. A., Bhawra, J., Brazo-Sayavera, J., Cardon, G., Chang, C. K., Delisle Nyström, C., Demetriou, Y., Draper, C. E., Edwards, L., Emeljanovas, A., Gába, A., Galaviz, K. I., González, S. A., ... Tremblay, M. S. (2018). Global Matrix 3.0 physical activity Report Card grades for children and youth: Results and analysis from 49 countries. *Journal of Physical Activity and Health*, *15*, S251–S273. https://doi.org/10.1123/jpah.2018-0472
- Australian Department of Health. (2019). *Guidelines for Healthy Growth & Development for Children & Young People (5 to 17 Years)*. Department of Health. www.health.gov.au
- Babic, M. J., Morgan, P. J., Plotnikoff, R. C., Lonsdale, C., White, R. L., & Lubans,
  D. R. (2014). Physical Activity and Physical Self-Concept in Youth: Systematic
  Review and Meta-Analysis. *Sports Med*, 44, 1589–1601.
  https://doi.org/10.1007/s40279-014-0229-z
- Bäckmand, H., Kujala, U., Sarna, S., & Kaprio, J. (2010). Former athletes' healthRelated lifestyle behaviours and self-rated health in late adulthood. *International Journal of Sports Medicine*, *31*(10), 751–758.

https://doi.org/10.1055/s-0030-1255109

- Bailey, R. ., Wellard, I., & Dismore, H. (2005). Girls' Participation in Physical Activities and Sports: Benefits, Patterns, Influences and Ways Forward. http://www.icsspe.org/documente//Girls.pdf
- Bakhtiar, S. (2014). Fundamental motor skill among 6-year-old children in Padang,
  West Sumatera, Indonesia. Asian Social Science, 10(5), 155–158.
  https://doi.org/10.5539/ass.v10n5p155
- Balish, S. M., McLaren, C., Rainham, D., & Blanchard, C. (2014). Correlates of youth sport attrition: A review and future directions. *Psychology of Sport and Exercise*, 15(4), 429–439. https://doi.org/10.1016/j.psychsport.2014.04.003
- Balyi, I., Way, R., & Higgs, C. (2013). Long-Term Athlete Development. Human Kinetics. https://doi.org/10.1519/jsc.00000000003321
- Bandura, A. (1997). Self-efficacy: The exercise of control. W H Freeman/Times Books/ Henry Holt & Co.
- Bardid, F., De Meester, A., Tallir, I., Cardon, G., Lenoir, M., & Haerens, L. (2016).
  Configurations of actual and perceived motor competence among children:
  Associations with motivation for sports and global self-worth. *Human Movement Science*, 50, 1–9. https://doi.org/10.1016/j.humov.2016.09.001
- Barnett, L., Hinkley, T., Okely, A. D., & Salmon, J. (2013). Child, family and environmental correlates of children's motor skill proficiency. *Journal of Science and Medicine in Sport*, 16(4), 332–336. https://doi.org/10.1016/j.jsams.2012.08.011

Barnett, L. M, Ridgers, N. D., Zask, A., & Salmon, J. (2015). Face validity and

reliability of a pictorial instrument for assessing fundamental movement skill perceived competence in young children. *Journal of Science and Medicine in Sport*, *18*(1), 98–102. https://doi.org/10.1016/j.jsams.2013.12.004

Barnett, L., Stodden, D., Cohen, K. E., Smith, J. J., Lubans, D. R., Lenoir, M.,
Iivonen, S., Miller, A. D., Laukkanen, A., Dudley, D., Lander, N. J., Brown, H.,
& Morgan, P. J. (2016). Fundamental movement skills: An important focus. *Journal of Teaching in Physical Education*, 35(3), 219–225.
https://doi.org/10.1123/jtpe.2014-0209

- Barnett, Lisa M., Ridgers, N. D., & Salmon, J. (2015). Associations between young children's perceived and actual ball skill competence and physical activity. *Journal of Science and Medicine in Sport*, 18(2), 167–171. https://doi.org/10.1016/j.jsams.2014.03.001
- Barnett, Lisa M, Beurden, E. Van, Morgan, P. J., Brooks, L. O., Beard, J. R.,
  Barnett, L. M., Beurden, E. Van, Morgan, P. J., & Lyndon, O. (2010). Gender
  Differences in Motor Skill Proficiency From Childhood to Adolescence Gender
  Differences in Motor Skill Proficiency From Childhood to Adolescence : A
  Longitudinal Study. *Research Quarterly for Exercise and Sport*, *81*(2), 162–
  170. https://doi.org/10.1080/02701367.2010.10599663
- Barnett, Lisa M, Lai, S., Hardy, L. L., & Cliff, D. P. (2016). Correlates of gross motor competence in children and adolescents : A systematic review and metaanalysis. *Sports Medicine*, 46(11), 1663–1688. https://doi.org/DOI 10.1007/s40279-016-0495-z
- Barnett, Lisa M, Morgan, P. J., Beurden, E. Van, & Beard, J. R. (2008). Perceived sports competence mediates the relationship between childhood motor skill

proficiency and adolescent physical activity and fitness: a longitudinal assessment. *International Journal of Behavioral Nutrition and Physical Activity*, *5*(40). https://doi.org/10.1186/1479-5868-5-40

- Basman, A. J. (2019). Assessment criteria of fundamental movement skills for various age groups: A systematic review. *Journal of Physical Education and Sport*, 19(1), 722–732. https://doi.org/10.7752/jpes.2019.01104
- Basterfield, L., Gardner, L., Reilly, J. K., Pearce, M. S., Parkinson, K. N., Adamson,
  A. J., Reilly, J. J., & Vella, S. A. (2016). Can't play, won't play: longitudinal changes in perceived barriers to participation in sports clubs across the child– adolescent transition. *BMJ Open Sport & Exercise Medicine*, 2(1), e000079. https://doi.org/10.1136/bmjsem-2015-000079
- Basterfield, L., Reilly, J. K., Pearce, M. S., Parkinson, K. N., Adamson, A. J., Reilly, J. J., & Vella, S. A. (2015). Longitudinal associations between sports participation, body composition and physical activity from childhood to adolescence. *Journal of Science and Medicine in Sport*, *18*, 178–182. https://doi.org/10.1016/j.jsams.2014.03.005
- Battaglia, A. V., Kerr, G., & Stirling, A. E. (2017). Youth Athletes' Interpretations of Punitive Coaching Practices. *Journal of Applied Sport Psychology*, 29(3), 337–352. https://doi.org/10.1080/10413200.2016.1271370
- Bauman, A. E., Reis, R. S., Sallis, J. F., Wells, J. C., Loos, R. J. F., & Martin, B. W. (2012). Correlates of physical activity: Why are some people physically active and others not? *The Lancet*, 380, 258–271. https://doi.org/10.1016/S0140-6736(12)60735-1

Beaulac, J., Kristjansson, E., & Calhoun, M. (2011). 'Bigger than hip-hop?' Impact

of a community- based physical activity program on youth living in a disadvantaged neighborhood in Canada. *Journal of Youth Studies*, *14*(8), 961–974. https://doi.org/10.1080/13676261.2011.616488

- Beets, M. W., Cardinal, B. J., & Alderman, B. L. (2010). Parental social support and the physical activity-related behaviors of youth: A review. *Health Education* and Behavior, 37(5), 621–644. https://doi.org/10.1177/1090198110363884
- Beets, M. W., Okely, A., Weaver, R. G., Webster, C., Lubans, D., Brusseau, T., Carson, R., & Cliff, D. P. (2016). The theory of expanded, extended, and enhanced opportunities for youth physical activity promotion. *International Journal of Behavioral Nutrition and Physical Activity*, *13*(120). https://doi.org/10.1186/s12966-016-0442-2
- Behan, S., Belton, S., Peers, C., O'Connor, N. E., & Issartel, J. (2019). Moving
  Well-Being Well: Investigating the maturation of fundamental movement skill
  proficiency across sex in Irish children aged five to twelve. *Journal of Sports Sciences*, 37(22), 2604–2612. https://doi.org/10.1080/02640414.2019.1651144
- Bélanger, M., Casey, M., Cormier, M., Laflamme Filion, A., Martin, G., Aubut, S.,
  Chouinard, P., Savoie, S.-P., & Beauchamp, J. (2011). Maintenance and decline of physical activity during adolescence: insights from a qualitative study. *International Journal of Behavioral Nutrition and Physical Activity*, 8(1), 117.
  https://doi.org/10.1186/1479-5868-8-117
- Bélanger, M., Gray-Donald, K., O'Loughlin, J., Paradis, G., Hutcheon, J.,
  Maximova, K., & Hanley, J. (2009). Participation in organised sports does not slow declines in physical activity during adolescence. *International Journal of Behavioral Nutrition and Physical Activity*, 6(22). https://doi.org/10.1186/1479-

- Bélanger, M., Sabiston, C. M., Barnett, T. A., O'Loughlin, E., Ward, S., Contreras, G., & O'Loughlin, J. (2015). Number of years of participation in some, but not all, types of physical activity during adolescence predicts level of physical activity in adulthood: Results from a 13-year study. *International Journal of Behavioral Nutrition and Physical Activity*, *12*(76). https://doi.org/10.1186/s12966-015-0237-x
- Belton, S., McCarren, A., McGrane, B., Powell, D., & Issartel, J. (2019). The youthphysical activity towards health (Y-PATH) intervention: Results of a 24 month cluster randomised controlled trial. *PLoS ONE*, *14*(9), e0221684. https://doi.org/10.1371/journal.pone.0221684
- Belton, S., O' Brien, W., Meegan, S., Woods, C., & Issartel, J. (2014). Youth-Physical Activity Towards Health: evidence and background to the development of the Y-PATH physical activity intervention for adolescents. *BMC*, *14*(122). https://doi.org/doi:10.1186/1471-2458-14-122
- Belton, S., O'Brien, W., McGann, J., & Issartel, J. (2018). Bright spots, physical activity investments that work: Workplace Challenge. *British Journal of Sports Medicine*, 0(0), 1026–1028. https://doi.org/10.1136/bjsports-2017-097716
- Beltran-Valls, M. R., Janssen, X., Farooq, A., Adamson, A. J., Pearce, M. S., Reilly, J. K., Basterfield, L., & Reilly, J. J. (2019). Longitudinal changes in vigorous intensity physical activity from childhood to adolescence: Gateshead
  Millennium Study. *Journal of Science and Medicine in Sport*, 22, 450–455. https://doi.org/10.1016/j.jsams.2018.10.010

Bender, A., & Litfin, J. (2015). The Effects of Gender Grouping on Female

Students' Participation in Physical Education. In *Retrieved from Sophia, the St. Catherine University repository website.* https://sophia.stkate.edu/maed/96%0AThis

- Beni, S., Fletcher, T., & Ní Chróinín, D. (2017). Meaningful Experiences in Physical Education and Youth Sport: A Review of the Literature. *Quest*, 69(3), 291–312. https://doi.org/10.1080/00336297.2016.1224192
- Benitez-Porres, J., Lopez-Fernandez, I., Raya, J. F., Carnero, S. A., Alvero-Cruz, J.
  R., & Carnero, E. A. (2016). Reliability and Validity of the PAQ-C
  Questionnaire to Assess Physical Activity in Children. *Journal of School Health*, 86, 677–685. https://doi.org/10.1111/josh.12418
- Bentley, G. F., Goodred, J. K., Jago, R., Sebire, S. J., Lucas, P. J., Fox, K. R., Stewart-Brown, S., & Turner, K. M. (2012). Parents' views on child physical activity and their implications for physical activity parenting interventions: A qualitative study. *BMC Pediatrics*, *12*(180). https://doi.org/10.1186/1471-2431-12-180
- Bervoets, L., Van Noten, C., Van Roosbroeck, S., Hansen, D., Van Hoorenbeeck, K., Verheyen, E., Van Hal, G., & Vankerckhoven, V. (2014). Reliability and Validity of the Dutch Physical Activity Questionnaires for Children (PAQ-C) and Adolescents (PAQ-A). *Archives of Public Health*, *72*(47). https://doi.org/10.1186/2049-3258-72-47
- Biddle, S. J. H, Braithwaite, R., & Pearson, N. (2014). The effectiveness of interventions to increase physical activity among young girls: A meta-analysis. *Preventive Medicine*, *62*, 119–131. https://doi.org/10.1016/j.ypmed.2014.02.009

- Biddle, S. J. H, Ciaccioni, S., Thomas, G., & Vergeer, I. (2019). Physical activity and mental health in children and adolescents: An updated review of reviews and an analysis of causality. *Psychology of Sport and Exercise*, 42, 146–155. https://doi.org/10.1016/j.psychsport.2018.08.011
- Biddle, Stuart J.H., Atkin, A. J., Cavill, N., & Foster, C. (2011). Correlates of physical activity in youth: A review of quantitative systematic reviews. *International Review of Sport and Exercise Psychology*, 4(1), 25–49.
  https://doi.org/10.1080/1750984X.2010.548528
- Bolger, L. A., Bolger, L. E., O'Neill, C., Coughlan, E., Lacey, S., O'Brien, W., &
  Burns, C. (2019). Fundamental Movement Skill Proficiency and Health Among
  a Cohort of Irish Primary School Children. *Research Quarterly for Exercise and Sport*, 90(1), 24–35. https://doi.org/10.1080/02701367.2018.1563271
- Bolger, L., Bolger, L., O'Neill, C., Coughlan, E., O'Brien, W., Lacey, S., & Burns,
  C. (2018). Age and sex differences in fundamental movement skills among a
  cohort of Irish school children. *Journal of Motor Learning and Development*, 6,
  81–100. https://doi.org/10.1123/jmld.2017-0003
- Borbee, A., Camiré, M., Boutet, M., Whitley, M. A., & Massey, W. V. (2019).
  Sport-based youth development interventions in the United States: a systematic review. *BMC Public Health*, *19*(1). https://doi.org/10.1186/s12889-019-6387-z
- Bortoli, L., Vitali, F., Di Battista, R., Ruiz, M. C., & Robazza, C. (2018). Initial validation of the psychobiosocial states in physical education (PBS-SPE) scale. *Frontiers in Psychology*, 9(2446), 1–15. https://doi.org/10.3389/fpsyg.2018.02446
- Bradley, R. H., Mcritchie, S., Houts, R. M., Nader, P., & Brien, M. O. (2011).

Parenting and the decline of physical activity from age 9 to 15. *Internatinal Journal of Behavioural Nutrition and Physical Activity*, 8(33). https://doi.org/10.1186/1479-5868-8-33

- Braun, V., & Clarke, V. (2006). Using thematic analysis in psychology. *Qualitative Research in Psychology*, 3(2), 77–101. https://doi.org/10.1191/ 1478088706qp063oa
- Brazendale, K., Graves, B. S., Penhollow, T., Whitehurst, M., Pittinger, E., &
  Randel, A. B. (2015). Children's Enjoyment and Perceived Competence in
  Physical Education and Physical Activity Participation Outside of School. *Emotional & Behavioral Disorders in Youth*, 7, 65–69.
  https://doi.org/10.1177/1356336X010071005
- Bremer, E., & Cairney, J. (2018). Fundamental Movement Skills and Health-Related
   Outcomes: A Narrative Review of Longitudinal and Intervention Studies
   Targeting Typically Developing Children. *American Journal of Lifestyle Medicine*, 12(2), 148–159. https://doi.org/10.1177/1559827616640196
- Bristow, A., & Saunders, M. (2014). Heightening awareness of research philosophy:
  the development of a reflexive tool for use with students. *British Academy of Management (BAM) Conference Proceedings 2014.*

Britton, U., Issartel, J., Symonds, J., & Belton, S. (2020). What keeps them physically active? Predicting physical activity, motor competence, healthrelated fitness, and perceived competence in irish adolescents after the transition from primary to second-level school. *International Journal of Environmental Research and Public Health*, 17(2874). https://doi.org/10.3390/ijerph17082874 Brodersen, N. H., Steptoe, A., Williamson, S., & Wardle, J. (2005).
Sociodemographic, developmental, environmental, and psychological correlates of physical activity and sedentary behavior at age 11 to 12. *Annals of Behavioral Medicine*, 29(1), 2–11.
https://doi.org/10.1207/s15324796abm2901 2

Brooks, F., & Magnusson, J. (2007). Physical activity as leisure: The meaning of physical activity for the health and well-being of adolescent women. *Health Care for Women International*, 28(1), 69–87.
https://doi.org/10.1080/07399330601003499

- Bryant, E., Duncan, M., Birch, S., & James, R. (2016). Can Fundamental Movement
  Skill Mastery Be Increased via a Six Week Physical Activity Intervention to
  Have Positive Effects on Physical Activity and Physical Self-Perception?
  Sports, 4(10). https://doi.org/10.3390/sports4010010
- Bryant, E. S., Duncan, M. J., & Birch, S. L. (2014). Fundamental movement skills and weight status in British primary school children. *European Journal of Sport Science*, 14(7), 730–736. https://doi.org/10.1080/17461391.2013.870232
- Budd, E. (2016). The Role of Physical Activity Enjoyment in the Pathways from Social and Physical Environments to Physical Activity of Early Adolescent Girls. In *ProQuest LLC*. Washington University.

Budd, E. L., McQueen, A., Eyler, A. A., Haire-Joshu, D., Auslander, W. ., &
Brownson, R. . (2018). The role of physical activity enjoyment in the pathways
from the social and physical environments to physical activity of early
adolescent girls. *Preventive Medicine*, *111*, 6–13.
https://doi.org/10.1016/j.ypmed.2018.02.015

Bukowsky, M., Faigenbaum, A. D., & Myer, G. D. (2014). Fundamental Integrative Training (FIT) for Physical Education. *Journal of Physical Education, Recreation & Dance*, 85(6), 23–30.
https://doi.org/10.1080/07303084.2014.926842

Burns, R. D., Brusseau, T. A., & Hannon, J. C. (2017). Effect of Comprehensive School Physical Activity Programming on Cardio-metabolic Health Markers in Children from Low-Income Schools. *Journal of Physical Activity and Health*, 1–20. https://doi.org/10.1123/jpah.2016-0691

- Burns, R. D., Fu, Y., Fang, Y., Hannon, J. C., & Brusseau, T. A. (2017). Effect of a 12-Week Physical Activity Program on Gross Motor Skills in Children. *Perceptual and Motor Skills*, 124(6), 1121–1133. https://doi.org/10.1177/0031512517720566
- Butler, J., & Griffen, L. (2010). *More teaching games for understanding: Moving globally*. Human Kinetics.
- Butt, J., Weinberg, R. S., Breckon, J. D., & Claytor, R. P. (2011). Adolescent physical activity participation and motivational determinants across gender, age, and race. *Journal of Physical Activity and Health*, 8, 1074–1083. https://doi.org/10.1123/jpah.8.8.1074
- Butterfield, S., Angell, R., & Mason, C. A. (2012). Age and Sex Differences in Object Control Skills by Children Ages 5 to 14. *Perceptual and Motor Skills*, *114*(1), 261–274. https://doi.org/10.2466/10.11.25.PMS.114.1.261-274
- Bull, F. C., Al-Ansari, S. S., Biddle, S., Borodulin, K., Buman, M. P., Cardon, G., Carty, C., Chaput, J. P., Chastin, S., Chou, R., Dempsey, P. C., Dipietro, L., Ekelund, U., Firth, J., Friedenreich, C. M., Garcia, L., Gichu, M., Jago, R.,

Katzmarzyk, P. T., ... Willumsen, J. F. (2020). World Health Organization 2020 guidelines on physical activity and sedentary behaviour. British Journal of Sports Medicine, 54, 1451–1462. https://doi.org/10.1136/bjsports-2020-102955

- Cain, K. L., Sallis, J. F., Conway, T. L., Dyck, D. Van, & Calhoon, L. (2013). Using Accelerometers in Youth Physical Activity Studies: A Review of Methods. *Journal of Physical Activity and Health*, 10(3), 437–450. https://doi.org/10.1123/jpah.10.3.437
- Cairney, J, Dudley, D., Kwan, M., Bulten, R., & Kriellaars, D. (2019). Physical Literacy, Physical Activity and Health: Toward an Evidence-Informed Conceptual Model. *Sports Medicine*, 49, 371–383. https://doi.org/https://doi.org/10.1007/s40279-019-01063-3
- Cairney, John, Kwan, M. Y. W., Velduizen, S., Hay, J., Bray, S. R., & Faught, B. E. (2012). Gender, perceived competence and the enjoyment of physical education in children : a longitudinal examination. *International Journal of Behavioral Nutrition and Physical Activity*, 9(1), 26. https://doi.org/10.1186/1479-5868-9-26
- Camacho-Minano, M. J., Lavoi, N. M., & Barr-Anderson, D. J. (2011). Interventions to promote physical activity among young and adolescent girls: a systematic review. *Health Education Research*, *26*(6), 1025–1049. https://doi.org/10.1093/her/cyr040
- Camiré, M., & Trudel, P. (2014). Helping youth sport coaches integrate psychological skills in their coaching practice. *Qualitative Research in Sport, Exercise and Health*, 6(4), 617–634. https://doi.org/10.1080/2159676X.2013.841281

- Carlin, A., Murphy, M. H., & Gallagher, A. M. (2015). Current influences and approaches to promote future physical activity in 11–13 year olds: a focus group study. *BMC Public Health*, 15(1270). https://doi.org/10.1186/s12889-015-2601-9
- Casey, M. M., Eime, R. M., Payne, W. R., & Harvey, J. T. (2009). Using a Socioecological Approach to Examine Participation in Sport and Physical Activity Among Rural Adolescent Girls. *Qualitative Health Research*, 19(7), 881–893. https://doi.org/10.1177/1049732309338198
- Casey, M. M., Harvey, J. T., Telford, A., Eime, R. M., Mooney, A., & Payne, W. R. (2014). Effectiveness of a school-community linked program on physical activity levels and health-related quality of life for adolescent girls. *BMC Public Health*, 14(649). https://doi.org/10.1186/1471-2458-14-649
- Casey, M. M., Telford, A., Mooney, A., Harvey, J. T., Eime, R. M., & Payne, W. R. (2014). Linking secondary school physical education with community sport and recreation for girls: a process evaluation. *BMC Public Health*, *14*(1), 1039. https://doi.org/10.1186/1471-2458-14-1039
- Caspersen, C. J., Powell, K. E., & Christenson, G. M. (1985). Physical activity, exercise, and physical fitness: definitions and distinctions for health-related research. *Public Health Reports* (Washington, D.C. : 1974), *100*(2), 126–131.

Cattuzzo, M. T., dos Santos Henrique, R., Ré, A. H. N., de Oliveira, I. S., Melo, B. M., de Sousa Moura, M., de Araújo, R. C., & Stodden, D. (2016). Motor competence and health related physical fitness in youth: A systematic review. *Journal of Science and Medicine in Sport*, *19*, 123–129. https://doi.org/10.1016/j.jsams.2014.12.004

- Chaddock-heyman, L., Erickson, K. I., Voss, M. W., Knecht, A. M., Pontifex, M. B., Castelli, D. M., Hillman, C. H., & Kramer, A. F. (2013). The effects of physical activity on functional MRI activation associated with cognitive control in children : a randomized controlled intervention. *Frontiers in Human Neuroscience*, 7(72). https://doi.org/10.3389/fnhum.2013.00072
- Chadwick, S. (2011). An Examination of Physical Activity Participation, Sedentary Behaviour, Health, Correlates of Physical Activity and Physical Activity Enjoyment among Irish Adolescents. Dublin City University.
- Chambers, F., & Gregg, R. (2016). Coaching and coach education. *International Sport Coaching Journal*, *3*, 65–74. https://doi.org/10.4324/9780203131862
- Chan, C. H. S., Ha, A. S. C., Ng, J. Y. Y., & Lubans, D. R. (2019). The A + FMS cluster randomized controlled trial: An assessment-based intervention on fundamental movement skills and psychosocial outcomes in primary schoolchildren. *Journal of Science and Medicine in Sport*, 22, 935–940. https://doi.org/10.1016/j.jsams.2019.05.002
- Chen, H., Sun, H., & Dai, J. (2017). Peer support and adolescents' physical activity: The mediating roles of self-efficacy and enjoyment. *Journal of Pediatric Psychology*, 42(5), 569–577. https://doi.org/10.1093/jpepsy/jsw103
- Chen, S., Kim, Y., & Gao, Z. (2014). The contributing role of physical education in youth 's daily physical activity and sedentary behavior. 1–7.
- Cheung, R. H. P. (2010). Designing movement activities to develop children's creativity in early childhood education. *Early Child Development and Care*, *180*(3), 377–385. https://doi.org/10.1080/03004430801931196

- Chinapaw, M. J. M., De Niet, M., Verloigne, M., De Bourdeaudhuij, I., Brug, J., & Altenburg, T. M. (2014). From sedentary time to sedentary patterns: Accelerometer data reduction decisions in youth. *PLoS ONE*, 9(11). https://doi.org/10.1371/journal.pone.0111205
- Choy, L. T. (2014). The Strengths and Weaknesses of Research Methodology:
  Comparison and Complimentary between Qualitative and Quantitative
  Approaches. *IOSR Journal of Humanities and Social Science*, *19*(4), 99–104.
  https://doi.org/10.9790/0837-194399104
- Chuan, C. C., Yusof, A., Soon, C. C., & Abudullah, M. C. (2014). Application of theory of planned behavior to predict recreational sports activities participation of students in Malaysia. *Journal of Physical Education and Sport*, 14(2), 172– 177. https://doi.org/10.7752/jpes.2014.02027
- Clark, J. E., & Metcalfe, J. S. (2002). The Mountain of Motor Development: a Metaphor. In *Motor Development: Research and Reviews* (Vol. 2). NASPE Publications.
- Clarke, J. ., & Metcalfe, J. . (2002). The mountain of motor development. Motor Dev. Res. Rev. 2002, 2, 163 190. *Motor Development*, 2, 163–190.
- Cliff, D. P., Okely, A. D., Morgan, P. J., Jones, R. A., Steele, J. R., & Baur, L. A. (2012). Proficiency Deficiency : Mastery of Fundamental Movement Skills and Skill Components in Overweight and Obese Children. *Pediatric Obesity*, 20, 1024–1033. https://doi.org/10.1038/oby.2011.241
- Coen, S. E., Rosenberg, M. W., & Davidson, J. (2018). "It's gym, like g-y-m not J-im": Exploring the role of place in the gendering of physical activity. *Social Science and Medicine (1982), 196*, 29–36.

https://doi.org/10.1016/j.socscimed.2017.10.036

- Cohen. (1988). *Statisical power analysis for behavioural sciences* (2nd ed). Lawerence Erlbaum Associates.
- Cohen, K., Morgan, P. ., Plotnikoff, R. ., Barnett, L. ., & Lubans, D. . (2015).
  Physical activity and skills intervention: SCORES cluster randomized controlled trial. *Medicine and Science in Sports and Exercise*, 47(4), 765–774. https://doi.org/0.1249/MSS.000000000000452
- Cohen, Morgan, P. J., Plotnikoff, R. C., Callister, R., & Lubans, D. R. (2014).
  Fundamental movement skills and physical activity among children living in low-income communities : a cross-sectional study. *Internatinal Journal of Behavioural Nutrition and Physical Activity*, 11(49).
  https://doi.org/doi:10.1186/1479-5868-11-49
- Cole, T. J., & Lobstein, T. (2012). Extended international (IOTF) body mass index cut-offs for thinness, overweight and obesity. *Pediatric Obesity*, 7, 284–294. https://doi.org/10.1111/j.2047-6310.2012.Y00064.x
- Coleman, L., Cox, L., & Roker, D. (2008). Girls and young women's participation in physical activity: Psychological and social influences. *Health Education Research*, 23(4), 633–647. https://doi.org/10.1093/her/cym040
- Connolly, S., Carlin, A., Johnston, A., Woods, C., Powell, C., Belton, S., O'brien, W., Saunders, J., Duff, C., Farmer, O., & Murphy, M. (2020). Physical activity, sport and physical education in northern ireland school children: A crosssectional study. *International Journal of Environmental Research and Public Health*, 17(6849). https://doi.org/10.3390/ijerph17186849

- Conroy, D. E., & Coatsworth, J. D. (2007). Assessing Autonomy-Supportive Coaching Strategies in Youth Sport. *Psychological Sport Exercise*, 8(5), 671– 684. https://doi.org/10.1016/j.psychsport.2006.12.001
- Cooper, A. R., Goodman, A., Page, A. S., Sherar, L. B., Esliger, D. W., van Sluijs,
  E. M. ., Andersen, L. B., Anderssen, S., Cardon, G., Davey, R., Froberg, K.,
  Hallal, P., Janz, K. F., Kordas, K., Kreimler, S., Pate, R. R., Puder, J. J., Reilly,
  J. J., Salmon, J., ... Ekelund, U. (2015). Objectively measured physical activity
  and sedentary time in youth: The International children's accelerometry
  database (ICAD). *International Journal of Behavioral Nutrition and Physical Activity*, *12*, 113. https://doi.org/10.1186/s12966-015-0274-5
- Coppinger, T., Lacey, S., O'Neill, C., & Burns, C. (2016). "Project Spraoi": A randomized control trial to improve nutrition and physical activity in school children. *Contemporary Clinical Trials Communications*, *3*, 94–101. https://doi.org/10.1016/j.conctc.2016.04.007
- Corder, K., Ekelund, U., Steele, R. M., Wareham, N. J., & Brage, S. (2008).
  Assessment of physical activity in youth. *Journal of Applied Physiology*, *105*, 977–987. https://doi.org/10.1152/japplphysiol.00094.2008
- Corepal, R., Best, P., O'Neill, R., Tully, M. A., Edwards, M., Jago, R., Miller, S. J., Kee, F., & Hunter, R. F. (2018). Exploring the use of a gamified intervention for encouraging physical activity in adolescents: A qualitative longitudinal study in Northern Ireland. *BMJ Open*, e019663. https://doi.org/10.1136/bmjopen-2017-019663
- Corr, M. (2019). Physical Activity Promotion Strategies for Adolescent Girls : Exploring Experiences, Co-Design and Intervention Development. Mary

Immaculate College.

Corr, M., & Murtagh, E. (2020). 'No one ever asked us': a feasibility study assessing the co-creation of a physical activity programme with adolescent girls. *Global Health Promotion*, 27(3), 34–43. https://doi.org/10.1177/1757975919853784

Costello, K., & Warne, J. (2020). A four-week fundamental motor skill intervention improves motor skills in eight to 10-year-old Irish primary school children. *Cogent Social Sciences*, 6(1724065).
https://doi.org/10.1080/23311886.2020.1724065

- Côté, J., & Gilbert, W. (2009). An Integrative Definition of Coaching Effectiveness and Expertise. *International Journal of Sports Science & Coaching*, 4(3), 307– 323. https://doi.org/10.1260/174795409789623892
- Côté, J., & Hancock, D. J. (2016). Evidence-based policies for youth sport programmes. *International Journal of Sport Policy*, 8(1), 51–65. https://doi.org/10.1080/19406940.2014.919338
- Cousson-Gélie, F., Carayol, M., Fregeac, B., Mora, L., Jeanleboeuf, F., Coste, O., Pereira, B., & Gourlan, M. (2019). The "great live and move challenge": A program to promote physical activity among children aged 7-11 years. Design and implementation of a cluster-randomized controlled trial. *BMC Public Health*, *19*(367). https://doi.org/10.1186/s12889-019-6648-x
- Crane, J. R., & Temple, V. A. (2014). A systematic review of dropout from organized sport among children and youth. *European Physical Education Review*, 21(1). https://doi.org/10.1177/1356336X14555294

Crespo, N. C., Corder, K., Marshall, S., Norman, G. J., Patrick, K., Sallis, J. F., &
Elder, J. P. (2013). An examination of multilevel factors that may explain gender differences in children's physical activity. *Journal of Physical Activity and Health*, *10*, 982–992. http://www.scopus.com/inward/record.url?eid=2s2.0-84900261480&partnerID=40&md5=0ddc3daf02ae731631081eef0069fdca

- Creswell, J. . (2014). *Qualitative Inquiry & Research Design: Choosing among Five Approaches* (4th ed.). SAGE.
- Creswell, J. ., & Plano Clark, V. . (2011). *Designing and Conducting Mixed Methods Research* (2nd ed.). SAGE.
- Creswell, J. W. (2013). *Research design: Qualitative, quantitative, and mixed methods approaches*. Sage Publications, Incorporated.
- Crocker, P. R., Eklund, R. C., & Kowalski, K. C. (2000). Children's physical activity and physical self-perceptions. *Journal of Sports Sciences*, 18(6), 383–394. https://doi.org/10.1080/02640410050074313
- Dacey, M. (2009). Link motivation and the stages of change to encourage exercise (International Council on Aging: Functional U). *Functional Exercise and Activity for Healthy Aging*, 7(1), 1–17.
- Dalene, K. E., Anderssen, S. A., Andersen, L. B., Steene-Johannessen, J., Ekelund,
  U., Hansen, B. H., & Kolle, E. (2018). Cross-sectional and prospective
  associations between sleep, screen time, active school travel, sports/exercise
  participation and physical activity in children and adolescents. *BMC Public Health*, 18(1). https://doi.org/10.1186/s12889-018-5610-7
- Darabi, F., Kaveh, M. H., Majlessi, F., Khalaj Abadi Farahani, F., Yaseri, M., & Shojaeizadeh, D. (2017). Effect of theory-based intervention to promote

physical activity among adolescent girls: a randomized control trial. *Electronic Physician*, *9*(4), 4238–4247. https://doi.org/10.19082/4238

- Davies, D. S. C., Atherton, F., McBride, M., & Calderwood, C. (2019). UK Chief Medical Officers' Physical Activity Guidelines. In *Department of Health and Social Care*. https://www.gov.uk/government/publications/physical-activityguidelines-uk-chief-medical-officers-report
- Davison, K. ., Jurkowski, J. ., Li, K., Kranz, S., & Lawson, H. . (2013). A childhood obesity intervention developed by families for families: results from a pilot study. *Internatinal Journal of Behavioural Nutrition and Physical Activity*, *10*(3). https://doi.org/10.1186/1479-5868-10-3
- de Moraes, A. C. F., Guerra, P. H., & Menezes, P. R. (2013). The worldwide prevalence of insufficient physical activity in adolescents; a systematic review. *Nutrición Hospitalaria*, 28(3), 575–584. https://doi.org/10.3305/nh.2013.28.3.6398
- DeBate, R.D;, & Bleck, J. . (2016). Changes in developmental assets and physical activity frequency among 3rd–5th grade girls participating in a girl-focused sport-based positive youth development program. *International Journal of Health Promotion and Education*, *54*(2), 95–112. https://doi.org/International Journal of Health Promotion and Education
- DeBate, R.D, Pettee Gabriel, K., Zwald, M., Huberty, J., & Zhang, Y. (2009).
  Changes in psychosocial factors and physical activity frequency among third-to eighthgrade girls who participated in a developmentally focused youth sport program: a preliminary study. *Journal of School Health*, *79*(10), 474–484.
  https://doi.org/10.1111/j.1746-1561.2009.00437.x

- Deci, E. L., & Ryan, R. M. (2000). The "what" and "why" of goal pursuits: Human needs and the self-determination of behavior. *Psychological Inquiry*, 11(4), 227–268. https://doi.org/10.1207/S15327965PLI1104\_01
- Deci, E. L., & Ryan, R. M. (2002). Overview of self-determination theory: An organismic- dialectical perspective. In E. L. Deci & R. M. Ryan (Eds.), Handbook of self-determination research. University of Rochester Press.
- Deci, E. L., & Ryan, R. M. (2008). Facilitating optimal motivation and psychological well-being across life's domains. *Canadian Psychology*, 49(1), 14–23. https://doi.org/10.1037/0708-5591.49.1.14
- DeMeester, A., Maes, J., Stodden, D., Cardon, G., Goodway, J., Lenoir, M., & Haerens, L. (2016). Identifying profiles of actual and perceived motor competence among adolescents: associations with motivation, physical activity, and sports participation. *Journal of Sports Sciences*, *34*(21), 2027–2037. https://doi.org/10.1080/02640414.2016.1149608
- Demetriou, Y., & Bachner, J. (2019). A school-based intervention based on selfdetermination theory to promote girls ' physical activity: study protocol of the CReActivity cluster randomised controlled trial. 19(519). https://doi.org/10.1186/s12889-019-6817-y (2019)
- Demetriou, Y., Vondung, C., Bucksch, J., Schlund, A., Schulze, C., Knapp, G., Coen, S. E., Puil, L., Phillips, S. P., & Reimers, A. K. (2019). Interventions on children's and adolescents' physical activity and sedentary behaviour: Protocol for a systematic review from a sex/gender perspective. *Systematic Reviews*, 8(65). https://doi.org/10.1186/s13643-019-0963-2

Department of Health. (2016). Get Ireland Active! The National Physical Activity

Plan for Ireland.

- Department of Health. (2018). National Physical Activity Plan Implementation Summary 2018.
- Department of Health & Human Services. (2018). 2018 Physical Activity Guidelines Advisory Committee. In *Physical Activity Guidelines Advisory Committee Scientific Report*. https://doi.org/10.1111/j.1753-4887.2008.00136.x
- Department of Transport Tourism and Sport. (2018). *National Sports Policy 2018 2027*. www.dttas.gov.ie
- DeVillis, R. . (2012). Scale development: Theory and applications (Vol. 26). Sage Publications.
- Ding, D., Lawson, K. D., Kolbe-Alexander, T. L., Finkelstein, E. A., Katzmarzyk, P. T., van Mechelen, W., & Pratt, M. (2016). The economic burden of physical inactivity: a global analysis of major non-communicable diseases. *The Lancet*, *388*(10051), 1311–1324. https://doi.org/10.1016/S0140-6736(16)30383-X
- Dishman, R. K., Mciver, K. L., Dowda, M., Saunders, R. P., & Pate, R. R. (2016).
  Motivation and Behavioral Regulation of Physical Activity in Middle-School
  Students. *Medicine & Science in Sports & Exercise*, 47(9), 1913–1921.
  https://doi.org/10.1249/MSS.00000000000616
- Dishman, R. K., Motl, R. W., Saunders, R., Felton, G., Ward, D. S., Dowda, M., & Pate, R. R. (2005). Enjoyment mediates effects of a school-based physicalactivity intervention. *Medicine and Science in Sports and Exercise*, 37(3), 478– 487. https://doi.org/10.1249/01.MSS.0000155391.62733.A7

Dobbins, M., Husson, H., Decorby, K., & Larocca, R. (2013). School-based physical

activity programs for promoting physical activity and fitness in children and adolescents aged 6 to 18. *Cochraine Database of Systematic Reviews*, *2*, 1–262. https://doi.org/10.1002/14651858.CD007651.pub2.www.cochranelibrary.com

- Drenowatz, C., Greier, K., Ruedl, G., & Kopp, M. (2019). Association between club sports participation and physical fitness across 6-to 14-year-old austrian youth. *International Journal of Environmental Research and Public Health*, 16(3392). https://doi.org/10.3390/ijerph16183392
- Drowatzky, J. (1996). *Ethical Decision Making in Physical Activity Research*. Human Kinetics.
- Dudley, D. A. (2015). A Conceptual Model of Observed Physical Literacy. *The Physical Educator*, 72, 236–260. https://doi.org/10.18666/TPE-2015-V72-I5-6020
- Dumith, S. C., Gigante, D. P., Domingues, M. R., & Harold, W. K. I. (2011). Physical activity change during adolescence : a systematic review and a pooled analysis. *International Journal of Epidemiology*, 40, 685–698. https://doi.org/10.1093/ije/dyq272
- Duncan, M. J., Bryant, E., & Stodden, D. (2017). Low fundamental movement skill proficiency is associated with high BMI and body fatness in girls but not boys aged 6–11 years old. *Journal of Sports Sciences*, 35(21), 2135–2141. https://doi.org/10.1080/02640414.2016.1258483
- Duncan, M. J., Roscoe, C. M. P., Faghy, M., Tallis, J., & Eyre, E. L. J. (2019).
  Estimating physical activity in children aged 8-11 years using accelerometry:
  Contributions from fundamental movement skills and different accelerometer
  placements. *Frontiers in Physiology*, 10(2), 42.

https://doi.org/10.3389/fphys.2019.00242

- Duncan, S. C., Strycker, L. A., & Chaumeton, N. R. (2015). Personal, Family, and Peer Correlates of General and Sport Physical Activity among African American, Latino, and White Girls. *Journal of Health Disparities Research and Practice*, 8(2), 12–28.
  http://www.ncbi.nlm.nih.gov/pubmed/26185734%0Ahttp://www.pubmedcentra 1.nih.gov/articlerender.fcgi?artid=PMC4500153
- Eather, N., Bull, A., Young, M. D., Barnes, A. T., Pollock, E. R., & Morgan, P. J. (2018). Fundamental movement skills: Where do girls fall short? A novel investigation of object-control skill execution in primary-school aged girls. *Preventive Medicine Reports*, *11*, 191–195. https://doi.org/10.1016/j.pmedr.2018.06.005
- Eather, N., Morgan, P. J., & Lubans, D. R. (2013). Social support from teachers mediates physical activity behavior change in children participating in the Fit-4-Fun intervention. *The International Journal of Behavioral Nutrition and Physical Activity*, 10(1), 68. https://doi.org/10.1186/1479-5868-10-68
- Eddolls, W. T. B., McNarry, M. A., Stratton, G., & Mackintosh, K. A. (2016).
  Parental influences on children's physical self-perceptions, body composition, and physical activity levels. *The Lancet*, 388(S45).
  https://doi.org/10.1016/S0140-6736(16)32281-4
- Edwards, M. J., May, T., Kesten, J. M., Banfield, K., Bird, E. L., Powell, J. E.,
  Sebire, S. J., & Jago, R. (2016). Lessons learnt from the Bristol Girls Dance
  Project cluster RCT : implications for designing and implementing after-school
  physical activity interventions. *BMJ Open*, *6*, e010036.

https://doi.org/10.1136/bmjopen-2015-010036

- Eime, R., Harvey, J., & Charity, M. (2018). Girls' transition from participation in a modified sport program to club sport competition a study of longitudinal patterns and correlates. *BMC Public Health*, *18*(718). https://doi.org/10.1186/s12889-018-5609-0
- Eime, R., Harvey, J., Charity, M., & Westerbeek, H. (2020). Longitudinal Trends in Sport Participation and Retention of Women and Girls. *Frontiers in Sports and Active Living*, 2(39). https://doi.org/10.3389/fspor.2020.00039
- Eime, R. M., Harvey, J. T., & Charity, M. J. (2019). Sport drop-out during adolescence: is it real, or an artefact of sampling behaviour? *International Journal of Sport Policy and Politics*, 11(4), 715–726. https://doi.org/10.1080/19406940.2019.1630468
- Eime, R. M., Harvey, J. T., Charity, M. J., & Payne, W. R. (2016). Population levels of sport participation: implications for sport policy. *BMC Public Health*, *16*(752). https://doi.org/10.1186/s12889-016-3463-5
- Eime, R. M., Casey, M. M., Harvey, J. T., Charity, M. J., Young, J. A., & Payne, W.
  R. (2015). Participation in modified sports programs: A longitudinal study of children's transition to club sport competition. *BMC Public Health*, *15*(649). https://doi.org/10.1186/s12889-015-2012-y
- Eime, R. M., Harvey, J. T., Sawyer, N. A., Craike, M. J., Symons, C. M., & Payne,
  W. R. (2016). Changes in sport and physical activity participation for
  adolescent females: a longitudinal study. *BMC Public Health*, *16*(533).
  https://doi.org/10.1186/s12889-016-3203-x

- Eime, R. M, Harvey, J. T., Charity, M. J., Casey, M. M., Westerbeek, H., & Payne,
  W. R. (2016). Age profiles of sport participants. *BMC Sports Science, Medicine* and Rehabilitation, 8(6). https://doi.org/10.1186/s13102-016-0031-3
- Eime, R. M, Harvey, J. T., Sawyer, N. A., Craike, M. J., Symons, C. M., Polman, R.
  C. ., & Payne, W. R. (2013). Understanding the contexts of adolescent female participation in sport and physical activity. *Research Quarterly for Exercise and Sport*, 84(2), 157–166. https://doi.org/10.1080/02701367.2013.784846
- Eime, R. M, Young, J. A., Harvey, J. T., Charity, M. J., & Payne, W. R. (2013). A systematic review of the psychological and social benefits of participation in sport for children and adolescents : informing development of a conceptual model of health through sport. *Internatinal Journal of Behavioural Nutrition and Physical Activity*, *10*(98). https://doi.org/10.1186/1479-5868-10-98
- Elo, S., & Kyngäs, H. (2008). The qualitative content analysis process. *Journal of Advanced Nursing Health*, 62, 107–115. https://doi.org/10.1111/j. 1365-2648.2007.04569.x
- Engel, A. C., Broderick, C. R., van Doorn, N., Hardy, L. L., & Parmenter, B. J.
  (2018). Exploring the Relationship Between Fundamental Motor Skill
  Interventions and Physical Activity Levels in Children: A Systematic Review and Meta-analysis. *Sports Medicine*, 48(8), 1845–1857.
  https://doi.org/10.1007/s40279-018-0923-3
- Erdim, L., Ergün, A., & Kuğuoğlu, S. (2019). Reliability and validity of the turkish version of the physical activity questionnaire for older children (Paq-c). *Turkish Journal of Medical Sciences*, 49(1), 162–169. https://doi.org/10.3906/sag-1806-212

Erickson, F. (1986). Qualitative methods in research on teaching. In M. Wittrock (Ed.), Handbook of research on teaching: A project of the American
Educational Research Association (3rd ed.). American Educational Research
Association.

Fahey, T., Delaney, L., & Gannon, B. (2005). School children and sport in Ireland. In *Economic and Social Research Institute (ERSI)*. http://ideas.repec.org/b/esr/resser/bmi182.html; http://www.esri.ie/publications/search\_for\_a\_publication/searc
h results/view/index.xml?id=1997

- Farmer, O., Belton, S., & O'Brien, W. (2017). The Relationship between Actual Fundamental Motor Skill Proficiency, Perceived Motor Skill Confidence and Competence, and Physical Activity in 8–12-Year-Old Irish Female Youth. *Sports*, 5(74). https://doi.org/10.3390/sports5040074
- Farmer, O., Duffy, D., Cahill, K., Lester, D., Belton, S., & O'Brien, W. (2018).
  Enhancing the evidence base for irish female youth participation in physical activity—the development of the Gaelic4Girls program. *Women in Sport and Physical Activity Journal*, 26(2), 111–123. https://doi.org/10.1123/wspaj.2017-0046
- Farooq, M. A., Parkinson, K. N., Adamson, A. J., Pearce, M. S., Reilly, J. K., Hughes, A. R., Janssen, X., Basterfield, L., & Reilly, J. J. (2018). Timing of the decline in physical activity in childhood and adolescence: Gateshead Millennium Cohort Study. *British Journal of Sports Medicine*, 52(15), 1002– 1006. https://doi.org/10.1136/bjsports-2016-096933

Farrokhi, A., Zareh, Z. M., Karimi, A. L., Kazemnejad, A., & Ilbeigi, S. (2014).

Reliability and validity of test of gross motor development-2 (Ulrich, 2000) among 3-10 aged children of Tehran City. *Journal of Physical Education and Sport Management*, 5(2), 18–28. https://doi.org/10.5897/JPESM12.003

- Federation of Irish Sport. (2013). The Role of Sport in Ireland Today: Federation of Irish Sport. Annual Review 2013. http://irishsport.ie/wpress/wpcontent/uploads/2013/03/Federation-of-Irish-Sport-Annual-Review-2013.pdf
- Felfe, C., Lechner, M., & Steinmayr, A. (2016). Sports and Child Development. PLoS ONE, 11(5), e0151729. https://doi.org/10.1371/journal.pone.0151729
- Feltz, D. L. (1988). Self-Confidence and Sports Performance. Exercise and Sport Sciences Reviews, 16, 423–457.
- Fenton, S. A. M., Duda, J. L., Appleton, P. R., & Barrett, T. G. (2017). Empowering youth sport environments: Implications for daily moderate-to-vigorous physical activity and adiposity. *Journal of Sport and Health Science*, *6*, 423–433. https://doi.org/10.1016/j.jshs.2016.03.006
- Fenton, S. A. M., Duda, J. L., & Barrett, T. (2016). Optimising physical activity engagement during youth sport: a self-determination theory approach. *Journal* of Sports Sciences, 34(19), 1874–1884. https://doi.org/10.1080/02640414.2016.1142104
- Field, S. C., & Temple, V. A. (2017). The Relationship between Fundamental Motor Skill Proficiency and Participation in Organized Sports and Active Recreation in Middle Childhood. *Sports*, 5(43), 1–10. https://doi.org/10.3390/sports5020043

Finch, C. F., & Donaldson, A. (2010). A sports setting matrix for understanding the

implementation context for community sport. *British Journal of Sports Medicine*, 44, 973–978. https://doi.org/10.1136/bjsm.2008.056069

- Fitzgerald, A., Fitzgerald, N., & Aherne, C. (2012). Do peers matter ? A review of peer and / or friends ' in fl uence on physical activity among American adolescents. *Journal of Adolescence*, 35, 941–958. https://doi.org/10.1016/j.adolescence.2012.01.002
- Fletcher, A., Jamal, F., Moore, G., Evans, R. E., Murphy, S., & Bonell, C. (2016).
  Realist complex intervention science: Applying realist principles across all phases of the Medical Research Council framework for developing and evaluating complex interventions. *Evaluation*, 22(3), 286–303.
  https://doi.org/10.1177/1356389016652743
- Flick, U. (2007). Designing qualitative research. SAGE.
- Fortier, M. S., Duda, J. L., Guerin, E., & Teixeira, P. J. (2012). Promoting physical activity: development and testing of self-determination theory-based interventions. *The International Journal of Behavioral Nutrition and Physical Activity*, 9(20). https://doi.org/10.1186/1479-5868-9-20
- Foweather, L. (2010). Fundamental movement skill competence among 10-11 year old children: Year 2 PEPASS Physical Activity Project The PEPASS Project. In *Wigan Council* (Issue August).
- Fransen, J. (2014). Pathways to successful sports involvement in children and adolescents: from motor competence development to a specialisationdiversification trade-off. Ghent University. Faculty of Medicine and Health Sciences, Ghent, Belgium. Ghent University Belgium.

- Fraser-thomas, J., Côté, J., & MacDonald, D. J. (2010). Community Size in Youth Sport Settings: Examining Developmental Assets and Sport Withdrawal. *Revue PhénEPS / PHEnex Journal*, 2(2). http://hdl.handle.net/1974/14376
- Freedson, P., Bowles, H. R., Troiano, R., & Haskell, W. (2012). Assessment of physical activity using wearable monitors: recommendations for monitor calibration and use in the field. *Medicine & Science in Sports & Exercise*, 44(Supp1), S1–S4. https://doi.org/10.1249/MSS.0b013e3182399b7e
- Fröberg, A., Lindroos, A. K., Ekblom, Ö., & Nyberg, G. (2020). Organised physical activity during leisure time is associated with more objectively measured physical activity among Swedish adolescents. *Acta Paediatrica, International Journal of Paediatrics*, 1–10. https://doi.org/10.1111/apa.15187
- Fu, Y., Gao, Z., Hannon, J. ., Burns, R. ., & Brusseau, T. . (2016). Effect of the SPARK program on physical activity, cardiorespiratory endurance, and motivation in middle-school students. *Journal of Physical Activity and Health*, *13*(5), 534–542. https://doi.org/10.1123/jpah.2015-0351
- Futris, T. G., Sutton, T. E., & Richardson, E. W. (2013). An Evaluation of the Relationship Smarts Plus Program on Adolescents in Georgia. *Journal of Human Sciences and Extension*, 1(2).
- Fyfe-Johnson, A. L., Saelens, B. E., Christakis, D. A., & Tandon, P. S. (2019).
  Physical Activity and Parental Attitudes and Beliefs of Children Attending a
  Nature Preschool. *The International Journal of Early Childhood Environmental Education*, 6(3), 3.
- Galatti, L. R., Côté, J., Reverdito, R. S., Allan, V., Seoane, A. M., & Paes, R. R.(2016). Fostering elite athlete development and recreational sport participation:

A successful club environment. *Motricidade*, *12*(3), 20–31. https://doi.org/10.6063/motricidade.6099

- Gallahue, D. ., Ozmun, J. ., & Goodway, J. . (2012). Understanding motor development: Infants, children, adolescents and adults (7th ed). McGraw Hill.
- Gallahue, D. L., & Ozmun, J. C. (2006). Understanding motor develop- ment: Infants, children, adolescents, adults (6th ed.). McGraw Hill.
- Gardner, L. A., Magee, C. A., & Vella, S. A. (2017). Enjoyment and behavioral intention predict organized youth sport participation and dropout. *Journal of Physical Activity and Health*, 14(11), 861–865. https://doi.org/10.1123/jpah.2016-0572
- Garn, A. C., Moore, E. W., Centeio, E. E., Kulik, N., Somers, C., & McCaughtry, N. (2019). Reciprocal effects model of children's physical activity, physical self-concept, and enjoyment. *Psychology of Sport and Exercise*, 45, 101568. https://doi.org/10.1016/j.psychsport.2019.101568
- Gavin, A., Keane, E., Callaghan, M., Kelly, C., Molcho, M., & Nic Gabhainn, S.
  (2015). *The Irish Health Behaviour in School-aged Children (HBSC) Study* 2014. https://doi.org/10.1037/e531492013-001
- Gearity, B. T. (2012). Poor teaching by the coach: A phenomenological description from athletes' experience of poor coaching. *Physical Education and Sport Pedagogy*, 17(1), 79–96. https://doi.org/10.1080/17408989.2010.548061
- Gee, J. P. (1999). An Introduction to Discourse Analysis: Theory and Method. Routledge.

Geidne, S., Quennerstedt, M., & Eriksson, C. (2013). The youth sports club as a

health-promoting setting: An integrative review of research. *Scandinavian Journal of Public Health*, *41*, 269–283. https://doi.org/10.1177/1403494812473204

- Gibson, J. . (2012). Interviews and Focus Groups With Children: Methods That Match Children's Developing Competencies. *Journal of Family Theory and Review*, 4(2), 148–159. https://doi.org/10.1111/j.1756-2589.2012.00119.x
- Giles-corti, B., & Donovan, R. (2002). The relative influence of individual, social and physical environment determinants of physical activity. *Social Science and Medicine*, 1–20. https://doi.org/10.1016/S0277-9536(01)00150-2
- Ginis, K. A. M., Nigg, C. R., & Smith, A. L. (2013). Peer-delivered physical activity interventions: An overlooked opportunity for physical activity promotion. *Translational Behavioral Medicine*, 3(4), 434–443. https://doi.org/10.1007/s13142-013-0215-2
- Glaser, B. ., & Strauss, A. . (1967). *The Discovery of Grounded Theory: Strategies* for Qualitative Research. Aldine.
- Glesne, C. (1999). *Becoming qualitative researchers: An introduction* (2nd ed.). Don Mills.
- Global Advocacy for Physical Activity (GAPA) the Advocacy Council of the
  International Society for Physical Activity and Health (ISPAH). (2011). Global
  Advocacy for Physical Activity (GAPA) the Advocacy Council of the
  International Society for Physical Activity and Health (ISPAH). In NCD
  Prevention: Investments that Work for Physical Activity (Issue February).
  internal-pdf://213.115.39.205/investmentswork.pdf%0Ahttp://www.paha.org.uk/Resource/toronto-charter-for-physical-

activity-a-global-call-for-action

- Gobbi, E., Elliot, C., Varnier, M., & Carraro, A. (2016). Psychometric properties of the physical activity questionnaire for older children in Italy: Testing the validity among a general and clinical pediatric population. *PLoS ONE*, *11*(5), 1– 12. https://doi.org/10.1371/journal.pone.0156354
- Goles, T., & Hirschheim, R. (2000). The paradigm is dead, the paradigm is dead . . . long live the paradigm: the legacy of Burrell and Morgan. *The International Journal of Management Science*, 28, 249–268. https://doi.org/10.1016/S0305-0483(99)00042-0
- Golle, K., Granacher, U., Hoffmann, M., Wick, D., & Muehlbauer, T. (2014). Effect of living area and sports club participation on physical fitness in children: A 4 year longitudinal study. *BMC Public Health*, 40(449). https://doi.org/10.1186/1471-2458-14-499
- González-Valero, G., Zurita-Ortega, F., Ubago-Jiménez, J. L., & Puertas-Molero, P. (2020). Motivation, Self-Concept and Discipline in Young Adolescents Who Practice Rhythmic Gymnastics. An Intervention. *Children*, 7(135). https://doi.org/10.3390/children7090135
- Goodway, J. D., Ozmun, J. C., & Gallahue, D. L. (2020). Understanding Motor Development: Infants, Children, Adolescents, Adults (Eight). Jones & Bartlett Learning.
- Gouldner, A. ., & House, F. . (1971). The Coming Crisis of Western Sociology. The ANNALS of the American Academy of Political and Social Science, 394(1), 184–184. https://doi.org/10.1177/000271627139400179

- Granero-Gallegos, A., Gómez-López, M., Rodríguez-Suárez, N., Arturo Abraldes, J., Alesi, M., & Bianco, A. (2017). Importance of the motivational climate in goal, enjoyment, and the causes of success in handball players. *Frontiers in Psychology*, 8(2081). https://doi.org/10.3389/fpsyg.2017.02081
- Gråstén, A., Liukkonen, J., Jaakkola, T., & Tammelin, T. (2014). Finnish Report Card 2014 on Physical Activity for Children and Youth Finnish Report Card 2014 on Physical Activity for Children and Youth.
- Grasten, A., Timo, J., Jarmo, L., Anthony, W., & Sami, Y.-P. (2012). Prediction of enjoyment in school physical education: EBSCOhost. *Journal of Sports Science* and Medicine, 11(2), 260–270.
- Gray, C. E., Larouche, R., Barnes, J. D., Colley, R. C., Bonne, J. C., Arthur, M.,
  Cameron, C., Chaput, J. ., Faulkner, G., Janssen, I., Kolen, A. ., Manske, S. .,
  Salmon, A., Spence, J. ., Timmons, B. ., & Tremblay, M. S. (2014). Are We
  Driving Our Kids to Unhealthy Habits ? Results of the Active Healthy Kids
  Canada 2013 Report Card on Physical Activity for Children and Youth. *International Journal of Environmental Research and Public Health*, *11*, 6009–6020. https://doi.org/10.3390/ijerph110606009
- Griffiths, M., & Armour, K. (2014). Volunteer sports coaches as community assets?
  A realist review of the research evidence. *International Journal of Sport Policy*, 6(3), 307–326. https://doi.org/10.1080/19406940.2013.824496
- Grydeland, M., Hansen, B. H., Ried-Larsen, M., Kolle, E., & Anderssen, S. A.
  (2014). Comparison of three generations of ActiGraph activity monitors under free-living conditions: Do they provide comparable assessments of overall physical activity in 9-year old children? *BMC Sports Science, Medicine and*

*Rehabilitation*, 6(1=26). https://doi.org/10.1186/2052-1847-6-26

- Guagliano, J. M., Lonsdale, C., Kolt, G. S., & Rosenkranz, R. R. (2014). Increasing girls' physical activity during an organised youth sport basketball program: a randomised controlled trial protocol. *BMC Public Health*, 14(383). https://doi.org/10.1186/1471-2458-14-383
- Guagliano, J., Rosenkranze, R., & Kolt, G. (2012). Girls' Physical Activity Levels during Organized Sports in Australia. *Medicine & Science in Sports & Exercise*, 116–122. https://doi.org/10.1249/MSS.0b013e31826a0a73
- Guthold, R., Cowan, M. J., Autenrieth, C. S., Kann, L., & Riley, L. M. (2010).
  Physical Activity and Sedentary Behavior Among Schoolchildren: A 34Country Comparison. *The Journal of Pediatrics*, *157*(1), 43–49.
  https://doi.org/10.1016/j.jpeds.2010.01.019
- Guthold, R., Stevens, G. A., Riley, L. M., & Bull, F. C. (2018). Worldwide trends in insufficient physical activity from 2001 to 2016: a pooled analysis of 358 population-based surveys with 1.9 million participants. *The Lancet Global Health*, 6(10), e1077–e1086. https://doi.org/10.1016/S2214-109X(18)30357-7
- Guthold, R., Stevens, G. A., Riley, L. M., & Bull, F. C. (2019). Global trends in insufficient physical activity among adolescents: a pooled analysis of 298 population-based surveys with 1.6 million participants. *The Lancet Child and Adolescent Health*, *4*, 23–35. https://doi.org/10.1016/S2352-4642(19)30323-2
- Haggis, C., Sims-gould, J., Winters, M., Gutteridge, K., & Mckay, H. A. (2013).
  Sustained impact of community-based physical activity interventions: key elements for success. *BMC Public Health*, *13*(892).
  https://doi.org/10.1186/1471-2458-13-892

- Haileamlak, A. (2019). Physical Inactivity: The Major Risk Factor for Non-Communicable Diseases. *Ethiopian Journal of Health Sciences*, 29(1), 810. https://doi.org/10.4314/ejhs.v29i1.1
- Hallal, P. C., Andersen, L. B., Bull, F. C., Guthold, R., Haskell, W., & Ekelund, U. (2012). Physical Activity 1 Global physical activity levels : surveillance progress, pitfalls, and prospects. *The Lancet*, *380*(1), 247–257. https://doi.org/10.1016/S0140-6736(12)60646-1
- Hands, B. P. (2012). How fundamental are fundamental movement skills? *Active and Healthy Magazine*, *19*(1), 14–17.
- Haradhan, M. (2017): Two Criteria for Good Measurements in Research: Validity and Reliability. Annals of Spiru Haret University, 17(4), 56-82
- Hardcastle, S. J., Hancox, J., Hattar, A., Maxwell-Smith, C., Thøgersen-Ntoumani,
  C., & Hagger, M. S. (2015). Motivating the unmotivated: how can health
  behavior be changed in those unwilling to change? *Frontiers in Psychology*,
  6(835). https://doi.org/10.3389/fpsyg.2015.00835
- Hardy, L.L, Barnett, L., Espinel, P., & Okely, A. . (2013). Thirteen-year trends in child and adolescent fundamental movement skills: 1997–2010. *Jurnal Teknologi*, 45(10), 1965–1970. https://doi.org/10.1249/MSS.0b013e318295a9fc
- Hardy, L. L., Mihrshahi, S., Drayton, B. A., & Bauman, A. (2016). NSW School Physical Activity and Nutrition Survey (SPANS) 2015: Full Report.
- Hardy, L. L, King, L., Farrell, L., Macniven, R., & Howlett, S. (2010). Fundamental movement skills among Australian preschool children. *Journal of Science and Medicine in Sport*, 13, 503–508. https://doi.org/10.1016/j.jsams.2009.05.010

- Hardy, L. L, Reinten-Reynolds, T., Espinel, P., Zask, A., & Okely, A. D. (2012).
  Prevalence and correlates of low fundamental movement skill competency in children. *Pediatrics*, *130*, e390-8. https://doi.org/10.1542/peds.2012-0345
- Harrington, D., Belton, S., & Murphy, M. (2016). The 2016 Ireland North and South Report Card on Physical Activity for Children and Youth: Are our Kids Moving With The Times.
- Harrington, D. M., Belton, S., Coppinger, T., Cullen, M., Donnelly, A., Dowd, K., Keating, T., Layte, R., Murphy, M., Murphy, N., Murtagh, E., & Woods, C. (2014). Results From Ireland's 2014 Report Card on Physical Activity in Children and Youth. *Journal of Physical Activity and Health*, *11*(S1), S63-68. https://doi.org/10.1123/jpah.2014-0166
- Harter, S. (1982). The Perceived Competence Scale for Children. *Child* Development, 53(1), 87. https://doi.org/10.2307/1129640
- Harter, S. (1999). *The Construction of the Self: A Developmental Perspective*.Guilford Press.
- Harter, S., & Pike, R. (1984). The Pictorial Scale of Perceived Competence and Social Acceptance for Young Children. *Child Development*, 55(6), 1969–1982. https://doi.org/10.2307/1129772
- Harvey, S., Song, Y., S. Harvey, Y. S., Baek, J.-H., & Van der Mars, H. (2016). Two sides of the same coin: Student physical activity levels during a game-centred soccer unit. *European Physical Education Review*, 22(4), 411–429. https://doi.org/10.1177/1356336X15614783
- Hearst, M. O., Patnode, C. D., Sirard, J. R., Farbakhsh, K., & Lytle, L. a. (2012).

Multilevel predictors of adolescent physical activity: a longitudinal analysis. *The International Journal of Behavioral Nutrition and Physical Activity*, *9*(8). https://doi.org/10.1186/1479-5868-9-8

- Hebert, J. J., Møller, N. C., Andersen, L. B., & Wedderkopp, N. (2015). Organized sport participation is associated with higher levels of overall health-related physical activity in children (CHAMPS study-DK). *PLoS ONE*, *10*(8), e0134621. https://doi.org/10.1371/journal.pone.0134621
- Helmerhorst, H. J. F., Brage, S., Warren, J., Besson, H., & Ekelund, U. (2012). A systematic review of reliability and objective criterion-related validity of physical activity questionnaires. *International Journal of Behavioral Nutrition* and Physical Activity, 9(103). https://doi.org/10.1186/1479-5868-9-103
- Hendrix, C. G., Prins, M. R., & Dekkers, H. (2014). Developmental coordination disorder and overweight and obesity in children: a systematic review. *Obesity Reviews*, 15, 408–423. https://doi.org/10.1111/obr.12137
- Henrique, R. S., Ré, A. H. N., Stodden, D. F., Fransen, J., Campos, C. M. C., Queiroz, D. R., & Cattuzzo, M. T. (2016). Association between sports participation, motor competence and weight status: A longitudinal study. *Journal of Science and Medicine in Sport*, 19(10), 825–829. https://doi.org/10.1016/j.jsams.2015.12.512
- Hidding, L. M., Chinapaw, M. J. M., van Poppel, M. N. M., Mokkink, L. B., &
  Altenburg, T. M. (2018). An Updated Systematic Review of Childhood
  Physical Activity Questionnaires. *Sports Medicine*, 48(12), 2797–2842.
  https://doi.org/10.1007/s40279-018-0987-0

Hirvensalo, M., & Lintunen, T. (2011). Life-course perspective for physical activity

and sports participation. *European Review of Aging and Physical Activity*, 8, 13–22. https://doi.org/10.1007/s11556-010-0076-3

Hoare, E., Stavreski, B., Jennings, G., & Kingwell, B. (2017). Exploring Motivation and Barriers to Physical Activity among Active and Inactive Australian Adults. *Sports*, 5(4), 47. https://doi.org/10.3390/sports5030047

Hoffman, J. R., Kang, J., Faigenbaum, A. D., & Ratamess, N. A. (2005).
Recreational Sports Participation is Associated with Enhanced Physical Fitness in Children. *Research in Sports Med*, *13*(2), 149–161.
https://doi.org/10.1080/15438620590956179

- Holfelder, B., & Schott, N. (2014). Relationship of fundamental movement skills and physical activity in children and adolescents: A systematic review. *Psychology of Sport and Exercise*, 15, 382–391.
  https://doi.org/10.1016/j.psychsport.2014.03.005
- Holt, N. L., Neely, K. C., Slater, L. G., Camiré, M., Côté, J., Fraser-thomas, J.,
  Macdonald, D., Strachan, L., Katherine, A., Holt, N. L., Neely, K. C., Slater, L.
  G., Camiré, M., Côté, J., Fraser-thomas, J., Macdonald, D., Strachan, L., &
  Tamminen, K. A. (2016). A grounded theory of positive youth development
  through sport based on results from a qualitative. *International Review of Sport and Exercise Pscyhology*, 10(1), 1–49.

https://doi.org/10.1080/1750984X.2016.1180704

Horstman, M., Aldiss, S., Richardson, A., & Gibson, F. (2008). Methodological Issues When Using the Draw and Write Technique With Children Aged 6 to 12 Years. *Qualitative Health Research*, 18(7), 1001–1011. https://doi.org/10.1177/1049732308318230

- Howe, C., Clevenger, K., Leslie, R., & Ragan, M. (2018). Comparison of
  Accelerometer-Based Cut-Points for Children's Physical Activity: Counts vs.
  Steps. *MDPI Children*, 5(105). https://doi.org/10.3390/children5080105
- Howie, E. K., Daniels, B. T., & Guagliano, J. M. (2018). Promoting Physical Activity Through Youth Sports Programs: It's Social. *American Journal of Lifestyle Medicine*, 14(1), 78–88. https://doi.org/10.1177/1559827618754842
- Hulteen, R. M., Morgan, P. J., Barnett, L. M., Stodden, D. F., & Lubans, D. R.
  (2018). Development of Foundational Movement Skills: A Conceptual Model for Physical Activity Across the Lifespan. *Sports Medicine*, 48(7), 1533–1540. https://doi.org/10.1007/s40279-018-0892-6
- Iivonen, S., & Sääkslahti, A. K. (2014). Preschool children's fundamental motor skills : a review of significant determinants. *Early Child Development and Care*, 184(7), 1107–1126. https://doi.org/10.1080/03004430.2013.837897
- Inchley, J. C., Kirby, J. L. M., & Currie, C. (2008). Physical Activity in Scottish Schoolchildren (PASS) Project: physical activity among adolescents in Scotland: final report of the PASS study.
- Inchley, J., Currie, D., Budisavljevic, S., Torsheim, T., Jåstad, A., Cosma, A., & et al. (2020). Spotlight on adolescent health and well-being. Findings from the 2017/2018 Health Behaviour in School-aged Children (HBSC) survey in Europe and Canada. International report. Volume 1. Key findings (Vol. 1). http://apps.who.int/iris.%0ASales,
- Inchley, J., Kirby, J., & Currie, C. (2011). Longitudinal changes in physical selfperceptions and associations with physical activity during adolescence. *Pediatric Exercise Science*, 23, 237–249.

http://www.ncbi.nlm.nih.gov/pubmed/21633136

- Inchley, J., Currie, D., Jewell, J., Breda, J., & Barnekow, V. (2017). Adolescent obesity and related behaviours: trends and inequalities in the WHO European Region 2002-2014. *World Health Organisation*, 10–98. https://doi.org/10.4103/0972 - 5229.178178
- Innerd, A. L., Azevedo, L. B., & Batterham, A. M. (2019). The effect of a curriculum-based physical activity intervention on accelerometer-assessed physical activity in schoolchildren: A non-randomised mixed methods controlled before-and-after study. *PLoS ONE*, *14*(12), e0225997. https://doi.org/10.1371/journal.pone.0225997
- International Society for the Advancement of Kinanthropometry. (2001). International Society for the Advancement of Kinanthropometry (ISAK). International Society for the Advancement of Kinanthropometry.
- Ishii, K., Shibata, A., Adachi, M., Nonoue, K., & Oka, K. (2015). Gender and grade differences in objectively measured physical activity and sedentary behavior patterns among Japanese children and adolescents: a cross-sectional study. *BMC Public Health*, 15(1254), 1–9. https://doi.org/10.1186/s12889-015-2607-3
- Issartel, J., McGrane, B., Fletcher, R., O'Brien, W., Powell, D., & Belton, S. (2017). A cross-validation study of the TGMD-2: The case of an adolescent population. *Journal of Science and Medicine in Sport*, 20(5), 475–479. https://doi.org/10.1016/j.jsams.2016.09.013
- Jaakkola, T, Hillman, C., Kalaja, S., & Liukkonen, J. (2015). The associations among fundamental movement skills, self-reported physical activity and academic performance during junior high school in Finland. *Journal of Sports*

Sciences, 33(16), 1719–1729. https://doi.org/10.1080/02640414.2015.1004640

- Jaakkola, T., & Washington, T. (2013). The relationship between fundamental movement skills and self-reported physical activity during Finnish junior high school. *Physical Education and Sport Pedagogy*, 18(5), 492–505. https://doi.org/10.1080/17408989.2012.690386
- Jaakkola, T., Yli-Piipari, S., Barkoukis, V., & Liukkonen, J. (2017). Relationships among perceived motivational climate, motivational regulations, enjoyment, and PA participation among Finnish physical education students. *International Journal of Sport and Exercise Psychology*, 15(3), 273–290. https://doi.org/10.1080/1612197X.2015.1100209
- Jago, R., Brockman, R., Fox, K. R., Cartwright, K., Page, A. S., & Thompson, J. L. (2009). Friendship groups and physical activity: Qualitative findings on how physical activity is initiated and maintained among 10-11 year old children. *International Journal of Behavioral Nutrition and Physical Activity*, 6(4). https://doi.org/10.1186/1479-5868-6-4
- Jago, R., Edwards, M. J., Sebire, S. J., Cooper, A. R., Powell, J. E., Bird, E. L., Simon, J., & Blair, P. S. (2013). Bristol girls dance project (BGDP): protocol for a cluster randomised controlled trial of an after-school dance programme to increase physical activity among 11 – 12 year old girls. *BMC Public Health*, *13*(1003). https://doi.org/10.1186/1471-2458-13-1003
- Jago, R., Edwards, M. J., Sebire, S. J., Tomkinson, K., Bird, E. L., Banfield, K.,
  May, T., Kesten, J. M., Cooper, A. R., Powell, J. E., & Blair, P. S. (2015).
  Effect and cost of an after-school dance programme on the physical activity of 11 12 year old girls: The Bristol Girls Dance Project, a school-based cluster

randomised controlled trial. *International Journal of Behavioral Nutrition and Physical Activity*, *12*(128). https://doi.org/10.1186/s12966-015-0289-y

- Jago, R., Solomon-Moore, E., Macdonald-Wallis, C., Sebire, S. J., Thompson, J. L., & Lawlor, D. A. (2017). Change in children's physical activity and sedentary time between Year 1 and Year 4 of primary school in the B-PROACT1V cohort. *International Journal of Behavioral Nutrition and Physical Activity*, 14(33). https://doi.org/10.1186/s12966-017-0492-0
- James, M., Todd, C., Scott, S., Stratton, G., McCoubrey, S., Christian, D., Halcox, J., Audrey, S., Ellins, E., Anderson, S., Copp, I., & Brophy, S. (2018). Teenage recommendations to improve physical activity for their age group: a qualitative study. *BMC Public Health*, 18(372). https://doi.org/10.1186/s12889-018-5274-3
- Janssen, I., & Leblanc, A. G. (2010). Systematic review of the health benefits of physical activity and fitness in school-aged children and youth. *Internatinal Journal of Behavioural Nutrition and Physical Activity*, 7(40). https://doi.org/doi: 10.1186/1479-5868-7-40
- Jekauc, D., Reimers, A. K., Wagner, M. O., & Woll, A. (2013). Physical activity in sports clubs of children and adolescents in Germany: Results from a nationwide representative survey. *Journal of Public Health (Germany)*, 21(6), 505–513. https://doi.org/10.1007/s10389-013-0579-2
- Joffe, H., & Yardley, L. (2003). Content and thematic analysis. *Research Methods* for Clinical and Health Psychology, 56–68.
- Jones, D., Innerd, A., Giles, E. L., & Azevedo, L. B. (2020). Association between fundamental motor skills and physical activity in the early years: A systematic review and meta-analysis. *Journal of Sport and Health Science*, 00, 1–11.

https://doi.org/10.1016/j.jshs.2020.03.001

- Jukic, I., Prnjak, K., Zoellner, A., Tufano, J. J., Sekulic, D., & Salaj, S. (2019). The Importance of Fundamental Motor Skills in Identifying Differences in Performance Levels of U10 Soccer Players. *Sports*, 7(7), 178. https://doi.org/10.3390/sports7070178
- Jurak, G., Cooper, A., Leskošek, B., & Kovač, M. (2013). Long-term effects of 4year longitudinal school-based physical activity intervention on the physical fitness of children and youth during 7-year follow-up assessment. *Central European Journal of Public Health*, 21(4), 190–195. https://doi.org/10.21101/cejph.a3823
- Kahlmeier, S., Wijnhoven, T. M. A., Alpiger, P., Schweizer, C., Breda, J., & Martin,
  B. W. (2015). National physical activity recommendations: Systematic
  overview and analysis of the situation in European countries. *BMC Public Health*, 15(133). https://doi.org/10.1186/s12889-015-1412-3
- Kalman, M., Inchley, J., Sigmundova, D., Iannotti, R. J., Tynjälä, J. A., Hamrik, Z., Haug, E., & Bucksch, J. (2015). Secular trends in moderate-to-vigorous physical activity in 32 countries from 2002 to 2010: A cross-national perspective. *European Journal of Public Health*, 25(Suppl 2), 37–40. https://doi.org/10.1093/eurpub/ckv024
- Karas, M., Bai, J., Strączkiewicz, M., Harezlak, J., Glynn, N. W., Harris, T.,
  Zipunnikov, V., Crainiceanu, C., & Urbanek, J. K. (2018). Accelerometry Data
  in Health Research: Challenges and Opportunities: Review and Examples. *Statistics in Biosciences*, *11*(2), 210–237. https://doi.org/10.1007/s12561-0189227-2

- Kaushik, V., & Walsh, C. A. (2019). Pragmatism as a Research Paradigm and Its Implications for Social Work Research. *Social Sciences*, 255(9). https://doi.org/10.3390/socsci8090255
- Keegan, R. J., Spray, C. M., & Harwood, C. (2010). The Motivational Atmosphere in Youth Sport: Coach, Parent, and Peer Influences on Motivation in Specializing Sport Participants. *Journal of Applied Social Psychology*, 22(1), 87–105. https://doi.org/10.1080/10413200903421267
- Kelly, L., O'Connor, S., Harrison, A. J., & Ní Chéilleachair, N. J. (2019). Does fundamental movement skill proficiency vary by sex, class group or weight status? Evidence from an Irish primary school setting. *Journal of Sports Sciences*, 37, 1055–1063. https://doi.org/10.1080/02640414.2018.1543833
- Kelly, P., Fitzsimons, C., & Baker, G. (2016). Should we reframe how we think about physical activity and sedentary behaviour measurement? Validity and reliability reconsidered. *International Journal of Behavioral Nutrition and Physical Activity*, *13*(32). https://doi.org/10.1186/s12966-016-0351-4
- Kelly, P., Matthews, A., & Foster, C. (2012). Young and Physically Active: a blueprint for making physical activity appealing to youth. https://doi.org/ISBN 978 92 890 0285 1

King, N., & Horrocks, C. (2010). Interviews in qualitative research. SAGE.

Kirby, J., Levin, K. A., & Inchley, J. (2013). Socio-Environmental influences on physical activity among young people: A qualitative study. *Health Education Research*, 28(6), 954–969. https://doi.org/10.1093/her/cyt085

Kitzinger, J. (1995). Qualitative Research: Introducing focus groups. British Medical

Journal, 311(7000), 299-302. https://doi.org/10.1136/bmj.311.7000.299

- Kjønniksen, L., Torsheim, T., & Wold, B. (2008). Tracking of leisure-time physical activity during adolescence and young adulthood: a 10-year longitudinal study. *Internatinal Journal of Behavioural Nutrition and Physical Activity*, *5*(69). https://doi.org/10.1186/1479-5868-5-69
- Koh, K. T., Ong, S. W., & Camiré, M. (2014). Implementation of a values training program in physical education and sport: perspectives from teachers, coaches, students, and athletes. *Physical Education and Sport Pedagogy*, 295–312. https://doi.org/10.1080/17408989.2014.990369
- Kokko, S. (2014). Sports clubs as settings for health promotion: Fundamentals and an overview to research. *Scandinavian Journal of Public Health*, 42(15), 60–65. https://doi.org/10.1177/1403494814545105
- Költő, A., Gavin, A., Molcho, M., Kelly, C., Walker, L., & Nic Gabhainn, S. (2020). *The Irish Health Behaviour in School-aged Children (HBSC) Study 2018*.
  https://doi.org/: 10.13025/zvwm-rp65
- Koorts, H., & Gillison, F. (2015). Mixed method evaluation of a community-based physical activity program using the RE-AIM framework: Practical application in a real-world setting Energy balance-related behaviors. *BMC Public Health*, *15*(1102). https://doi.org/10.1186/s12889-015-2466-y
- Kordi, R., Nourian, R., Ghayour, M., Kordi, M., & Younesian, A. (2012).
  Development and evaluation of a basic physical and sports activity program for preschool children in nursery schools in Iran: An interventional study. *Iranian Journal of Pediatrics*, 22(3), 357–363.

Kowalski, K. ., Crocker, P. R. ., & Donen, R. . (2004). *Physical Activity Questionnaire for Older Children (PAQ-C) and Adolescents (PAQ-A) Manual.* 

- Krueger, R., & Casey, M. . (2000). Focus groups: A practical guide for applied research (3rd ed.). London, UK: Sage Publications. (3rd ed.). Sage Publications.
- Kuen, W., & Mok, V. (2013). A Case Study on Application of the Theory of Planned Behaviour: Predicting Physical Activity of Adolescents in Hong Kong. *Journal* of Community Medicine & Health Education, 3(5), 100231. https://doi.org/10.4172/2161-0711.1000231
- Lagestad, P., & Sorensen, A. (2018). Longitudinal changes in sports enjoyment among adolescents. *Journal of Physical Education and Sport*, 18(1), 89–97. https://doi.org/10.7752/jpes.2018.01011
- Laird, Y., Fawkner, S., Kelly, P., McNamee, L., & Niven, A. (2016). The role of social support on physical activity behaviour in adolescent girls: A systematic review and meta-analysis. *International Journal of Behavioral Nutrition and Physical Activity*, 13(79). https://doi.org/10.1186/s12966-016-0405-7
- Laird, Y., Fawkner, S., & Niven, A. (2018). A grounded theory of how social support influences physical activity in adolescent girls. *International Journal of Qualitative Studies on Health and Well-Being*, 13(1), 1435099.
  https://doi.org/10.1080/17482631.2018.1435099
- Lancaster, G. A. (2015). Pilot and feasibility studies come of age! *Pilot and Feasibility Studies*, *I*(1). https://doi.org/10.1186/2055-5784-1-1

Lander, Natalie J, Hanna, L., Brown, H., Telford, A., Morgan, P. J., Salmon, J., &

Barnett, L. M. (2017). Physical Education Teachers 'Perspectives and Experiences When Teaching FMS to Early Adolescent Girls. *Journal of Teaching in Physical Education*, *36*, 113–118. https://doi.org/10.1123/jtpe.2015-0201

- Lander, Natalie Jayne, Barnett, L. M., Brown, H., & Telford, A. (2015). Physical education teacher training in fundamental movement skills makes a difference to instruction and assessment practices. *Journal of Teaching in Physical Education*, 34(3), 548–556. https://doi.org/10.1123/jtpe.2014-0043
- Laroche, M., Roussel, P., Cury, F., & Boiche, J. (2019). Understanding the dynamics of physical activity practice in the health context through Regulatory Focus and Self-Determination theories. *PLoS ONE*, *14*(8), e0216760. https://doi.org/10.1371/journal.pone.0216760
- Larson, R., Wilson, S., Brown, B., Furstenberg, F., & Verma, S. (2002). Changes in Adolescents' Interpersonal Experiences: Are They Being Prepared for Adult Relationships in the Twenty-First Century? *Journal of Research on Adolescence*, 12(1), 31–68.
- Lau, E. Y., Faulkner, G., Qian, W., & Leatherdale, S. T. (2016). Longitudinal associations of parental and peer influences with physical activity during adolescence: findings from the COMPASS study. *Health Promotion and Chronic Disease Prevention in Canada*, 36(11), 235–242.
  https://doi.org/10.24095/hpcdp.36.11.01
- Lavallee, D., Sheridan, D., Coffee, P., & Daly, P. (2019). Social Support Intervention to Reduce Intentions to Drop-out from Youth Sport: The GAA Super Games Centre. *Psychosocial Intervention*, 28(1), 11–17.

https://doi.org/10.5093/pi2018a15

- Lawler, M. (2016). Adolescents ' Patterns of Physical Activity and Psychosocial Factors Associated with Maintenance, Drop Out and Uptake of Physical Activity over Twelve Months. University of Dublin, Trinity College.
- Layte, R., & McCrory, C. (2011). Growing Up in Ireland National Longitudinal Study of Children: Overweight and Obesity Among 9 year olds Report No. 2. In *Growing Up in Ireland*.
- Lee, J., Pope, Z., & Gao, Z. (2018). The Role of Youth Sports in Promoting Children's Physical Activity and Preventing Pediatric Obesity: A Systematic Review. *Behavioral Medicine*, 44(1), 62–76. https://doi.org/10.1080/08964289.2016.1193462
- LeGear, M., Greyling, L., Sloan, E., Bell, R. I., Williams, B., Naylor, P., & Temple,
  V. A. (2012). A window of opportunity? Motor skills and perceptions of
  competence of children in Kindergarten. *International Journal of Behavioral Nutrition and Physical Activity*, 9(29). https://doi.org/10.1186/1479-5868-9-29
- Legg, E., Wells, M. S., & Barile, J. P. (2015). Factors Related to Sense of Community in Youth Sport Parents. *Journal of Park and Recreation Administration*, 33(2), 73–86. http://eds-1a-1ebscohost-1com-1hanquest.stcproxy.han.nl/eds/pdfviewer/pdfviewer?vid=5&sid=26b44c79bad9-4ab0-9eb4-de4dd6495980@sessionmgr4003&hid=4111
- Lester, D., McGrane, B., Belton, S., Duncan, M., Chambers, F., & O'Brien, W. (2017). The Age-Related Association of Movement in Irish Adolescent Youth. *Sports*, 5, 77. https://doi.org/10.3390/sports5040077

Levin, K. A. (2006). Study design III: Cross-sectional studies. *Evidence-Based Dentistry*, 7(1), 24–25. https://doi.org/10.1038/sj.ebd.6400375

- Liangruenrom, N., Craike, M., Biddle, S. J. H., Suttikasem, K., & Pedisic, Z. (2019).
  Correlates of physical activity and sedentary behaviour in the Thai population:
  A systematic review. *BMC Public Health*, *19*(414).
  https://doi.org/10.1186/s12889-019-6708-2
- Lin, S.-J., & Yang, S.-C. (2015). The Development of Fundamental Movement Skills by Children Aged Six to Nine. Universal Journal of Educational Research, 3(12), 1024–1027. https://doi.org/10.13189/ujer.2015.031211

Lincoln, Y., & Guba, E. G. (1985). Naturalistic inquiry. SAGE.

- Lisinskiene, A., Lochbaum, M., May, E., & Huml, M. (2019). Quantifying the coach–athlete–parent (C–A–P) relationship in youth sport: Initial development of the positive and negative processes in the C–A–P questionnaire (PNPCAP). *International Journal of Environmental Research and Public Health*, *16*(4140). https://doi.org/10.3390/ijerph16214140
- Lisinskienė, A., & Šukys, S. (2014). The athlete triangle: Coach, athlete and parents as an educational system. *Global Journal of Sociology*, *4*(2), 46–51. www.awer-?-center/gjs
- Liszewska, N., Scholz, U., Radtke, T., Horodyska, K., Liszewski, M., & Luszczynska, A. (2018). Association between children's physical activity and parental practices enhancing children's physical activity: The moderating effects of children's BMI z-score. *Frontiers in Psychology*, 8(2359). https://doi.org/10.3389/fpsyg.2017.02359

- Litchfield, C., & Elliott, S. (2020). Maximising enjoyment to sustain girls' sport participation: a unique case study of Netball in Australia. *Qualitative Research in Sport, Exercise and Health*. https://doi.org/10.1080/2159676X.2020.1778063
- Lloyd, M, Colley, R. ., & Tremblay, M. S. (2010). Advancing the debate on 'fitness testing' for children: Perhaps we're riding the wrong animal. *Pediatric Exercise Science*, 22(2), 176–182. https://doi.org/10.1123/pes.22.2.176.
- Lloyd, Meghann, Saunders, T. J., Bremer, E., & Tremblay, M. S. (2014). Long-Term Importance of Fundamental Motor Skills : A 20-Year Follow-Up Study. *Adapted Physical Activity Quarterly*, *31*, 67–78.
  https://doi.org/10.1123/apaq.2013-0048
- Logan, K., Cuff, S., & Council on Sports Medicine and Fitness (2019). Organized
   Sports for Children, Preadolescents, and Adolescents. *Pediatrics*, e20190997.
   Advance online publication. https://doi.org/10.1542/peds.2019-0997
- Logan, S., Robinson, L. E., Getchell, N., Logan, S. W., Webster, E. K., Getchell, N., Pfeiffer, K. A., & Robinson, L. E. (2015). Relationship Between Motor Competence and Physical Activity: A Systematic Review Competence and Physical Activity During Childhood and Adolescence A Systematic Review. *Kinesiology Review*, 4(10.1123/kr.2013-0012), 416–426. https://www.researchgate.net/publication/293213254%0ARelationship

Lonsdale, C., Rosenkranz, R. ., Peralta, L. ., Bennie, A., Fahey, P., & Lubans, D. .
(2013). A systematic review and meta-analysis of interventions designed to increase moderate-to-vigorous physical activity in school physical education lessons. *Preventive Medicine*, 56(2), 152–161.
https://doi.org/10.1016/j.ypmed.2012.12.004

- López Sánchez, G. F., González Víllora, S., & Díaz Suárez, A. (2016). Level of habitual physical activity in children and adolescents from the Region of Murcia (Spain). *SpringerPlus*, 5(386). https://doi.org/10.1186/s40064-016-2033-8
- Loprinzi, P. D., Davis, R. E., & Fu, Y. (2015). Early motor skill competence as a mediator of child and adult physical activity. *Preventive Medicine Reports*, *2*, 833–838. https://doi.org/10.1016/j.pmedr.2015.09.015
- Lubans, D. R., Lonsdale, C., Cohen, K., Eather, N., Beauchamp, M. R., Morgan, P. J., Sylvester, B. D., & Smith, J. J. (2017). Framework for the design and delivery of organized physical activity sessions for children and adolescents:
  Rationale and description of the "SAAFE" teaching principles. *International Journal of Behavioral Nutrition and Physical Activity*, *14*(24).
  https://doi.org/10.1186/s12966-017-0479-x
- Lubans, D. R., Morgan, P. J., Cliff, D. P., Barnett, L. M., & Okely, A. D. (2010).
  Fundamental Movement Skills in Children and Adolescents Review of Associated Health Benefits. *Sports Medicine*, 40(12), 1019–1035.
  https://doi.org/0112-1642/10/0012
- Lunn, P., Kelly, E., & Fitzpatrick, N. (2013). Keeping Them in the Game: Taking Up and Dropping Out of Sport and Exercise in Ireland. In *The Economic and Social Research Institute* (Issue 33).
  http://www.irishsportscouncil.ie/research/keeping-them-in-the-game-2013-

/keeping-them-in-the-game.pdf

Lykesas, G., Tsapakidou, A., & Tsompanaki, E. (2014). Creative Dance as a Means of Growth and Development of Fundamental Motor Skills for Children in First Grades of Primary Schools in Greece. *Asian Journal of Humanities and Social Studies*, *2*(1), 211–218. http://ajouronline.com/index.php?journal=AJHSS&page=article&op=view&pat h%5B%5D=982

Lynch, B. A., Kaufman, T. K., Rajjo, T. I., Mohammed, K., Kumar, S., Murad, M. H., Gentile, N. E., Koepp, G. A., McCrady-Spitzer, S. K., & Levine, J. A. (2019). Accuracy of Accelerometers for Measuring Physical Activity and Levels of Sedentary Behavior in Children: A Systematic Review. *Journal of Primary Care and Community Health*, *10*, 1–8. https://doi.org/10.1177/2150132719874252

Macdonald-Wallis, K., Jago, R., Page, A. S., Brockman, R., & Thompson, J. L. (2011). School-based friendship networks and children's physical activity: A spatial analytical approach. *Social Science and Medicine*, 73(1), 6–12. https://doi.org/10.1016/j.socscimed.2011.04.018

- MacDonald, D. J., Côté, J., Eys, M., & Deakin, J. (2011). The role of enjoyment and motivational climate in relation to the personal development of team sport athletes. *Sport Psychologist*, 25(1), 32–46. https://doi.org/10.1123/tsp.25.1.32
- Maclean, J., & Lorimer, R. (2016). Are coach education programmes the most effective method for coach development? *International Journal of Coaching Science*, 10(2), 1–31.

https://www.researchgate.net/publication/307512572\_Are\_coach\_education\_pr ogrammes\_the\_most\_effective\_method\_for\_coach\_development

MacPhail, A., & Kirk, D. (2006). Young people's socialisation into sport: Experiencing the specialising phase. *Leisure Studies*, *25*(1), 57–74. https://doi.org/10.1080/02614360500116290

- Maddison, R., Marsh, S., Hinckson, E., Duncan, S., Mandic, S., Taylor, R., & Smith,
  M. (2016). Results from New Zealand's 2016 report card on physical activity
  for children and youth. *Journal of Physical Activity and Health*, *13*(2), 225–230. https://doi.org/10.1123/jpah.2016-0323
- Madill, A., Jordan, A., & Shirley, C. (2000). Objectivity and reliability in qualitative analysis: Realist, contextualist and radical constructionist epistemologies. *British Journal of Psychology*, 91(1), 1–20.
  https://doi.org/10.1348/000712600161646
- Mäkelä, S., Aaltonen, S., Korhonen, T., Rose, R. J., & Jaakko, K. (2017). Diversity of leisure-time sport activities in adolescence as a predictor of leisure-time physical activity in adulthood. *Scandinavian Journal of Medicine & Science in Sport*, 27(12), 1902–1912. https://doi.org/10.1111/sms.12837
- Malina, R. (2012). Movement Proficiency in Childhood: Implications for Physical Activity and Youth Sport. *Kinesiologia Slovenica*, *18*(3), 19.
- Mallett, C. J., Trudel, P., Lyle, J., & Rynne, S. B. (2009). Formal vs. Informal Coach Education. *International Journal of Sports Science & Coaching*, 4(3), 325–364. https://doi.org/10.1260/174795409789623883
- Mandic, S., Bengoechea, E., Stevens, E., de la Barra, S., & Skidmore, P. (2012).
  Getting kids active by participating in sport and doing It more often: focusing on what matters. *International Journal of Behavioral Nutrition and Physical Activity*, 9(86). https://doi.org/10.1186/1479-5868-9-86

Mann, C. J. (2003). Observational research methods. Research design II: cohort,
cross sectional, and case-control studies. *Emergency Medicine Journal*, 20, 54–60.

- Mansfield, L., Anokye, N., Fox-rushby, J., & Kay, T. (2015). The Health and Sport Engagement (HASE) Intervention and Evaluation Project: protocol for the design , outcome, process and economic evaluation of a complex community sport intervention to increase levels of physical activity. *BMJ Open*, *5*, e009276. https://doi.org/10.1136/bmjopen-2015-009276
- Mansfield, L., Kay, T., Anokye, N., & Fox-Rushby, J. (2018). A qualitative investigation of the role of sport coaches in designing and delivering a complex community sport intervention for increasing physical activity and improving health. *BMC Public Health*, *18*(1196). https://doi.org/10.1186/s12889-018-6089-y
- Marks, J., De La Haye, K., Barnett, L. M., & Allender, S. (2015). Friendship network characteristics are associated with physical activity and sedentary behavior in early adolescence. *PLoS ONE*, *10*(12), e0145344. https://doi.org/10.1371/journal.pone.0145344
- Marlow, M., & McLain, B. (2011). Assessing the impacts of experiential learning on teacher classroom practice. *Research in Higher Education Journal*, 14, 1–15. https://doi.org/10.1089/dia.2011.0083
- Marques, A., Ekelund, U., & Sardinha, L. B. (2016). Associations between organized sports participation and objectively measured physical activity, sedentary time and weight status in youth. *Journal of Science and Medicine in Sport*, 19, 154–157. https://doi.org/10.1016/j.jsams.2015.02.007

Martin, B. W., Mäder, U., Stamm, H., & Braun-Fahrländer, C. (2009). Physical

activity and health – what are the recommendations and where do we find the Swiss population? *Schweizerische Zeitschrift Für Sportmedizin Und Sporttraumatologie*, *57*(2), 37–43. https://doi.org/10.5167/uzh-29108

- Martin, J. J., Waldron, J. J., McCabe, A., & Choi, Y. S. (2009). The impact of "Girls on the Run" on Self-Concept and Fat Attitudes. *Journal of Clinical Sport Psychology*, *3*, 127–138. https://doi.org/10.1123/jcsp.3.2.127
- Maturo, C. C., & Cunningham, S. A. (2013). Influence of friends on children's physical activity: A review. *American Journal of Public Health*, 103(7), 23–38. https://doi.org/10.2105/AJPH.2013.301366
- Mazur, J., Dzielska, A., Kleszczewska, D., Oblacińska, A., & Fijałkowska, A.
  (2020). Changes in physical activity of adolescent girls in the context of their perception of the Healthy Me programme. *European Journal of Public Health*, 30(3), 433–437. https://doi.org/10.1093/eurpub/ckz242
- McEachan, R. R. C., Conner, M., Taylor, N. J., & Lawton, R. J. (2011). Prospective prediction of health-related behaviours with the theory of planned behaviour: A meta-analysis. *Health Psychology Review*, 5(2), 97–144. https://doi.org/10.1080/17437199.2010.521684
- McGrane, B., Belton, S. J., Powell, D., Woods, C. B., & Issartel, J. (2016). Physical self-confidence levels of adolescents: Scale reliability and validity. *Journal of Science and Medicine in Sport*, 19, 563–567. https://doi.org/10.1016/j.jsams.2015.07.004
- McGrane, B., Belton, S., Powell, D., & Issartel, J. (2016). The relationship between fundamental movement skill proficiency and physical self-confidence among adolescents. *Journal of Sports Sciences*, *35*(17), 1709–1714.

https://doi.org/10.1080/02640414.2016.1235280

Medical Research Council. (2000). A Framework for Development and Evaluation of RCTs for Complex Interventions to Improve Health. In *Medical Research Council*.

http://www.mrc.ac.uk/Utilities/Documentrecord/index.htm?d=MRC003372

- Mehtälä, M. A., Sääkslahti, A., Inkinen, M., & Poskiparta, M. E. (2014). A socioecological approach to physical activity interventions in childcare: a systematic review. *International Journal of Behavioral Nutrition and Physical Activity*, 11(1), 22. https://doi.org/10.1186/1479-5868-11-22
- Mendonça, G., Cheng, L. A., Mélo, E. N., & De Farias Júnior, J. C. (2014). Physical activity and social support in adolescents: A systematic review. *Health Education Research*, 29(5), 822–839. https://doi.org/10.1093/her/cyu017
- Merkel, D. L. (2013). Youth sport: positive and negative impact on young athletes. *Journal of Sports Medicine*, *4*, 151–160.
- Messing, S., Rütten, A., Abu-Omar, K., Ungerer-Röhrich, U., Goodwin, L., Burlacu,
  I., & Gediga, G. (2019). How can physical activity be promoted among children and adolescents? A systematic review of reviews across settings. *Frontiers in Public Health*, 7(55). https://doi.org/10.3389/fpubh.2019.00055
- Metcalf, B., Henley, W., & Wilkin, T. (2012). Effectiveness of intervention on physical activity of children: Systematic review and meta-analysis of controlled trials with objectively measured outcomes (EarlyBird 54). *British Medical Journal*, 345, e5888. https://doi.org/10.1136/bmj.e5888

Michie, S., Ashford, S., Sniehotta, F., Dombrowski, S., Bishop, a, & French, D.

(2011). A refined taxonomy of behaviour change techniques to help people change their physical activity and healthy eating behaviours: the CALO-RE taxonomy. *Psychology & Health*, 0446(April 2012), 37–41. https://doi.org/10.1080/08870446.2010.540664

Mikaelsson, K., Rutberg, S., Lindqvist, A. K., & Michaelson, P. (2020). Physically inactive adolescents' experiences of engaging in physical activity. *European Journal of Physiotherapy*, 22(4), 191–196. https://doi.org/10.1080/21679169.2019.1567808

- Mitchell, B., McLennan, S., Latimer, K., Graham, D., Gilmore, J., & Rush, E. (2013). Improvement of fundamental movement skills through support and mentorship of class room teachers. *Obesity Research & Clinical Practice*, 7, e230–e234. https://doi.org/10.1016/j.orcp.2011.11.002
- Mitchell, F., Gray, S., & Inchley, J. (2013). "This choice thing really works..."
  Changes in experiences and engagement of adolescent girls in physical education classes, during a school-based physical activity programme. *Physical Education and Sport Pedagogy*, 1–22.
  https://doi.org/10.1080/17408989.2013.837433
- Mitchell, F., Inchley, J., Fleming, J., & Currie, C. (2015). A socio-ecological approach to understanding adolescent girls' engagement and experiences in the PE environment: a case study design. *Graduate Journal of Sport, Exercise and Physical Education Research*, *3*, 44–62.
- Moeijes, J., van Busschbach, J. T., Bosscher, R. J., & Twisk, J. W. R. (2018). Sports participation and health-related quality of life: a longitudinal observational study in children. *BMC Public Health*, *18*(702). https://doi.org/10.1007/s11136-

- Moeijes, J., van Busschbach, J. T., Bosscher, R. J., & Twisk, J. W. R. (2019). Sports participation and health-related quality of life: a longitudinal observational study in children. *Quality of Life Research*, 28(9), 2453–2469. https://doi.org/10.1007/s11136-019-02219-4
- Mohamadian, H., & Ghannaee Arani, M. (2014). Factors predicting the physical activity behavior of female adolescents: A test of the health promotion model. *Journal of Preventive Medicine and Public Health*, 47(1), 64–71. https://doi.org/10.3961/jpmph.2014.47.1.64
- Moore, J. B., Hanes, J. C., Barbeau, P., Gutin, B., Treviño, R. P., & Yin, Z. (2007).
  Validation of the physical activity questionnaire for older children in children of different races. *Pediatric Exercise Science*, *19*(1), 6–19.
  https://doi.org/10.1123/pes.19.1.6
- Morano, M., Bortoli, L., Ruiz, M. C., Vitali, F., & Robazza, C. (2019). Self-efficacy and enjoyment of physical activity in children: Factorial validity of two pictorial scales. *PeerJ*, 7, e7402. https://doi.org/10.7717/peerj.7402
- Morgan, D. . (2014). Integrating Qualitative and Quantitative Methods : A Pragmatic Approach. SAGE. https://doi.org/10.4135/9781544304533
- Morgan, K., Jones, R. L., Gilbourme, D., & Llewellyn, D. (2013). Innovative approaches in 4 coach education pedagogy. In P. Potrac, W. Gilbert, & J. Denison (Eds.), Routledge 5 handbook of sports coaching (P. Potrac, W. Gilbert, & J. Denison (eds.)). Routledge.

Morgan, Kelly, Van Godwin, J., Darwent, K., & Fildes, A. (2019). Formative

research to develop a school-based, community-linked physical activity role model programme for girls: CHoosing Active Role Models to INspire Girls (CHARMING). *BMC Public Health*, *19*(437). https://doi.org/10.1186/s12889-019-6741-1

- Morgan, P., Barnett, L. M., Cliff, D. P., Okely, A. D., Scott, H. A., Cohen, K. E., & Lubans, D. R. (2013). Fundamental Movement Skill Interventions in Youth: A Systematic Review and Meta-analysis. *Pediatrics*, *132*(5), e1361–e1383. https://doi.org/10.1542/peds.2013-1167
- Mukherjee, S., Ting Jamie, L. C., & Fong, L. H. (2017). Fundamental Motor Skill Proficiency of 6- to 9-Year-Old Singaporean Children. *Perceptual and Motor Skills*, 124(3), 584–600. https://doi.org/10.1177/0031512517703005
- Murillo Pardo, B., García, E., Julián, J. A., & Generelo, E. (2014). Empowering adolescents to be physically active: Three-year results of the Sigue la Huella intervention. *Preventive Medicine*, 66, 6–11. https://doi.org/10.1016/j.ypmed.2014.04.023
- Mutz, M., & Albrecht, P. (2017). Parents' Social Status and Children's Daily Physical Activity: The Role of Familial Socialization and Support. *Journal of Child and Family Studies*, *26*(11), 3026–3035. https://doi.org/10.1007/s10826-017-0808-3
- Neergaard, M. A., Olesen, F., Andersen, R. S., & Sondergaard, J. (2009). Qualitative description-the poor cousin of health research? *BMC Medical Research Methodology*, 9(52). https://doi.org/10.1186/1471-2288-9-52
- Nelson, T., Stovitz, S., Thomas, M., LaVoi, N., Bauer, K., & Neumark-Sztainer. (2011). Do youth sports prevent pediatric obesity? A systematic review and

commentary. *Curr Sports Med Rep*, *10*(6), 612–624. https://doi.org/10.1249/JSR.0b013e318237bf74

- Neuman, W. . (2000). Social research methods Qualitative and quantitative approaches (4th ed.). Allyn and Bacon.
- New South Wales Department of Education and Training. (2000). *Get Skilled: Get Active; A K-6 resource to support the teaching of fundamental movement skills.* NSW Department of Education and Training Curriculum Support Directorate.
- Nilsen, A. K. O., Anderssen, S. A., Johannessen, K., Aadland, K. N., Ylvisaaker, E., Loftesnes, J. M., & Aadland, E. (2020). Bi-directional prospective associations between objectively measured physical activity and fundamental motor skills in children: A two-year follow-up. *International Journal of Behavioral Nutrition and Physical Activity*, 17(1). https://doi.org/10.1186/s12966-019-0902-6
- Noonan, R. J., Boddy, L. M., Fairclough, S. J., & Knowles, Z. R. (2016). Write, draw, show, and tell: A child-centred dual methodology to explore perceptions of out-of-school physical activity. *BMC Public Health*, *16*(326). https://doi.org/10.1186/s12889-016-3005-1
- Nunnally, J. C., & Bernstein, I. H. (1994). *Psychometric theory* (3rd ed.). McGraw Hill.
- O' Brien, W., Belton, S., & Issartel, J. (2016a). Fundamental movement skill proficiency amongst adolescent youth. *Physical Education and Sport Pedagogy*, 21(6), 557–571. https://doi.org/10.1080/17408989.2015.1017451
- O' Brien, W., Belton, S., & Issartel, J. (2016b). The relationship between adolescents' physical activity, fundamental movement skills and weight status.

Journal of Sports Sciences, 34(12), 1159–1167. https://doi.org/10.1080/02640414.2015.1096017

- O' Connor, S., Whyte, E. F., Gibbons, B., Feeney, O., Luc, S., & Ní Chéilleachair,
  N. J. (2018). Fundamental movement skill proficiency in juvenile Gaelic
  games. *Sport Sciences for Health*, *14*, 161–172. https://doi.org/10.1007/s11332-017-0421-2
- O' Keefe, S., & Smyth, P. . (1999). Fundamental Motor Skill Development, Irish Scientist Year Book. http://www.irishscientist.ie/p187a.ht
- O'Brien, W., Duncan, M. J., Farmer, O., & Lester, D. (2017). Do Irish adolescents have adequate functional movement skill and confidence? *Journal of Motor Learning and Development*, 6(s2), S301–S319. https://doi.org/10.1123/jmld.2016-0067
- O'Brien, W., Duncan, M. J., Farmer, O., & Lester, D. (2018). Do Irish adolescents have adequate functional movement skill and confidence? *Journal of Motor Learning and Development*, 6(s2), S301–S319. https://doi.org/10.1123/jmld.2016-0067
- O'Cathain, A., Croot, L., Duncan, E., Rousseau, N., Sworn, K., Turner, K. M., Yardley, L., & Hoddinott, P. (2019). Guidance on how to develop complex interventions to improve health and healthcare. *BMJ Open*, 9(8), e029954. https://doi.org/10.1136/bmjopen-2019-029954
- Occhino, J., Mallett, C., Rynne, S., & Carlisle, K. (2014). Autonomy-Supportive
  Pedagogical Approach to Sports Coaching : Research , Challenges and
  Opportunities. *International Journal of Sports Science & Coaching*, 9(2), 401–416. https://doi.org/10.1260/1747-9541.9.2.401

- Office for Human Research Protections. (2001). Office for Human Research Protections.
- Okely, A. D., Lubans, D. R., Morgan, P. J., Cotton, W., Peralta, L., Miller, J., Batterham, M., & Janssen, X. (2017). Promoting physical activity among adolescent girls: The Girls in Sport group randomized trial. *International Journal of Behavioral Nutrition and Physical Activity*, 14(81). https://doi.org/10.1186/s12966-017-0535-6
- Okely, A, Booth, M., & Chey, T. (2004). Relationships between body composition and fundamental movement skills among children and adolescents. *Res Q Exerc Sport*, 75(3), 238–247. https://doi.org/10.1080/02701367.2004.10609157
- Okely, A., Cotton, W., Lubans, D., Morgan, P., Puglisi, L., Miller, J., Wright, J.,
  Batterham, M. J., Peralta, L. R., & Perry, J. (2011). A school-based intervention to promote physical activity among adolescent girls: Rationale, design , and baseline data from the Girls in Sport group randomised controlled trial. *BMC Public Health*, *11*(658). https://doi.org/10.1186/1471-2458-11-658
- Okely, T., & Booth, M. L. (2004). Mastery of fundamental motor skills among New South Wales school students: Prevalence and sociodemographic distribution. *Journal of Science and Medicine in Sport*, 7(3), 358–372. https://doi.org/10.1016/S1440-2440(04)80031-8
- Olds, T., Dollman, J., & Maher, C. (2009). Adolescent sport in Australia : Who, when, where and what ? *ACHPER Healthy Lifestyles Journal*, *56*(1).
- Orme, M., Wijndaele, K., Sharp, S. J., Westgate, K., Ekelund, U., & Brage, S. (2014). Combined influence of epoch length, cut-point and bout duration on accelerometry-derived physical activity. *International Journal of Behavioral*

*Nutrition and Physical Activity*, *11*(34). https://doi.org/10.1186/1479-5868-11-34

- Ortega, F. B., Konstabel, K., Pasquali, E., Ruiz, J. R., Hurtig-Wennlöf, A., Mäestu, J., Löf, M., Harro, J., Bellocco, R., Labayen, I., Veidebaum, T., & Sjöström, M. (2013). Objectively Measured Physical Activity and Sedentary Time during Childhood, Adolescence and Young Adulthood: A Cohort Study. *PLoS ONE*, 8(4), e60871. https://doi.org/10.1371/journal.pone.0060871
- Owen, M. B., Curry, W. B., Kerner, C., Newson, L., & Fairclough, S. J. (2017). The effectiveness of school-based physical activity interventions for adolescent girls: A systematic review and meta-analysis. *Preventive Medicine*, 105, 237– 249. https://doi.org/10.1016/j.ypmed.2017.09.018
- Owen, M., Kerner, C., Taylor, S., Noonan, R., Newson, L., Kosteli, M.-C., Curry, W., & Fairclough, S. (2018). The Feasibility of a Novel School Peer-Led Mentoring Model to Improve the Physical Activity Levels and Sedentary Time of Adolescent Girls: The Girls Peer Activity (G-PACT) Project. *Children*, 5(67). https://doi.org/10.3390/children5060067
- Patay, M. E., Patton, K., Parker, M., Fahey, K., & Sinclair, C. (2015). Understanding Motivators and Barriers to Physical Activity. *The Physical Educator*, 72, 496– 517.
- Patnaik, E. (2013). Reflexivity: Situating the Researcher in Qualitative Research. *Humanities and Social Science Studies*, 2(2), 98–106. https://www.researchgate.net/publication/263916084
- Patton, M. Q. (2002). *Qualitative research and evaluation methods* (3rd ed.). Sage Publications.

- Pawlowski, C. S., Tjørnhøj-Thomsen, T., Schipperijn, J., & Troelsen, J. (2014). Barriers for recess physical activity: a gender specific qualitative focus group exploration. *BMC Public Health*, 14(639). https://doi.org/10.1186/1471-2458-14-639
- Pearce, A., Kirk, C., Cummins, S., Collins, M., Elliman, D., Connolly, A. ., & Law, C. (2009). Gaining children's perspectives: A multiple method approach to explore environmental influences on healthy eating and physical activity. *Health and Place*, *15*(2), 614–621. https://doi.org/10.1016/j.healthplace.2008.10.007.
- Pearce, M. S., Basterfield, L., Mann, K. D., Parkinson, K. N., Adamson, A. J.,
  Reilly, J. J., Millennium, G., & Core, S. (2012). Early Predictors of Objectively
  Measured Physical Activity and Sedentary Behaviour in 8 10 Year Old
  Children : The Gateshead Millennium Study. *PLoS ONE*, 7(6), e37975.
  https://doi.org/10.1371/journal.pone.0037975
- Pearson, N., Braithwaite, R., & Biddle, S. J. H. (2015). The effectiveness of interventions to increase physical activity among adolescent girls: A metaanalysis. *Academic Pediatrics*, 15, 9–18. https://doi.org/10.1016/j.acap.2014.08.009
- Perlman, D. (2012). The influence of the Sport Education Model on amotivated students' in-class physical activity. *European Physical Education Review*, 18(3), 335–345. https://doi.org/10.1177/1356336X12450795
- Perry, C. K., Saelens, B. E., & Thompson, B. (2011). Intrapersonal, behavioral, and environmental factors associated with meeting recommended physical activity among rural Latino youth. *Pediatric Exercise Sciencexercise Science*, 23(4),

521-536.

http://www.ncbi.nlm.nih.gov/pmc/articles/PMC3677811/pdf/nihms470462.pdf

- Peters, D. H., Tran, N. T., & Adam, T. (2013). Implementation Research in Health: a practical guide. In *Alliance for Health Policy and Systems Research and World Health Organisation*. https://doi.org/ISBN 978 92 4 150621 2
- Petersen, T. L., Møller, L. B., Brønd, J. C., Jepsen, R., & Grøntved, A. (2020). Association between parent and child physical activity: a systematic review. *International Journal of Behavioral Nutrition and Physical Activity*, 17(1), 67. https://doi.org/10.1186/s12966-020-00966-z
- Pfeiffer, K. A., & Wierenga, M. J. (2019). Promoting Physical Activity Through Youth Sport. *Kinesiology Review*, 8(3), 204–210. https://doi.org/10.1123/kr.2019-0033
- Pienaar, A. E., Barhorst, R., & Twisk, J. W. R. (2013). Child : Relationships between academic performance , SES school type and perceptual-motor skills in first grade South African learners : NW-CHILD study. *Child: Care, Health and Development*, 40(3), 370–378. https://doi.org/10.1111/cch.12059
- Pitney, W. ., & Parker, J. (2009). Qualitative research in physical activity and the health professions. Human Kinetics.
- Planinšec, J., Fošnarič, S., & Pišot, R. (2004). Physical self-concept and physical exercise in children. *Studia Psychologica*, 46(2), 89–95.
- Player, K. (2010a). The Interaction of Motivational Climate and Self-Determination in Examining Commitment Level in Female Youth Sport. Graduate School of Clemson University.

- Player, K. (2010b). The interation of motivatioanl climate and self-determination in examing commitment level in female youth sport (Issue August).
- Poitras, V. J., Carson, V., Lee, E. Y., Hewitt, L., Jennings, C., Hunter, S., Kuzik, N., Stearns, J. A., Unrau, S. P., Gray, C., Adamo, K. B., Janssen, I., Okely, A. D., Spence, J. C., Timmons, B. W., Sampson, M., & Tremblay, M. S. (2016).
  Systematic review of the relationships between objectively measured physical activity and health indicators in school-aged children and youth. *Applied Physiology, Nutrition, and Metabolism, 41*, S197–S239.
  https://doi.org/10.1186/s12889-017-4860-0
- Polit, D. F., & Beck, C. T. (2013). Essentials of Nursing Research Seventh Edition Appraising Evidence for Nursing Practice. In *Journal of Chemical Information* and Modeling (Vol. 53). https://doi.org/10.1017/CBO9781107415324.004
- Porter, S. (2016). Children and Adolescents Self-Determined Motivation for Leisure and Organized Physical Activity Behaviours. In *ProQuest LLC*. Northcentral University.
- Post, P., & Palacios, R. (2019). Aggie Play: A Gender-Relevant Physical Activity Program for Girls. *Journal of Sport and Exercise Psychology*, 41(4), 194–205. https://doi.org/10.1123/jsep.2018-0299

Pouliou, T., Sera, F., Griffiths, L., Joshi, H., Geraci, M., Cortina-Borja, M., & Law,
C. (2015). Environmental influences on children's physical activity. *Journal of Epidemiology and Community Health*, 69(1), 77–85.
https://doi.org/10.1136/jech-2014-204287

Powell, E., Wood, L. A., & Nevill, A. M. (2016). Increasing physical activity levels in primary school physical education: The SHARP Principles Model. Preventive Medicine Reports, 3, 7–13. https://doi.org/10.1016/j.pmedr.2015.11.007

- Prochaska, J. J., Sallis, J. F., & Long, B. (2001). A physical activity screening measure for use with adolescents in primary care. *Archives of Pediatrics & Adolescent Medicine*, 155, 554–559. https://doi.org/10.1001/archpedi.155.5.554
- Quested, E., Ntoumanis, N., Viladrich, C., Ommundsen, Y., Hoye, A. Van, Mercé, J., Hall, H. K., Zourbanos, N., Duda, J. L., & Taylor, P. (2013). Intentions to drop-out of youth soccer: A test of the basic needs theory among European youth from five countries. *International Journal of Sport and Exercise Psychology*, *11*(4), 395–407. https://doi.org/10.1080/1612197X.2013.830431
- Rainer, P., & Jarvis, S. (2015). Fundamental movement skills and their relationship with measures of health-related physical fitness of primary school children prior to secondary school transition: a Welsh perspective. *Education*, 48(1), 3–13. https://doi.org/10.1080/03004279.2019.1573264
- Rauscher, L., Kauer, K., & Wilson, B. D. M. (2013). The Healthy Body Paradox: Organizational and Interactional Influences on Preadolescent Girls' Body Image in Los Angeles. *Gender and Society*, 27(2), 208–230. https://doi.org/10.1177/0891243212472054
- Reeves, B. C., & Gaus, W. (2004). Guidelines for reporting non-randomised studies. *Research in Complementary and Classical Natural Medicine*, 11(supp 1), 46– 52. https://doi.org/10.1159/000080576
- Reimers, A. K., Schmidt, S. C. E., Yolanda Demetriou, Marzi, I., & Woll, A. (2019).
  Parental and peer support and modelling in relation to domain-specific physical activity participation in boys and girls from Germany. *PLoS ONE*, *14*(10),

e0223928. https://doi.org/10.1371/journal.pone.0223928

- Ren, Z., Hu, L., Yu, J. J., Yu, Q., Chen, S., Ma, Y., Lin, J., Yang, L., Li, X., & Zou,
  L. (2020). The Influence of Social Support on Physical Activity in Chinese
  Adolescents: The Mediating Role of Exercise Self-Efficacy. *Children*, 7(3), 23.
  https://doi.org/10.3390/children7030023
- Renshaw, D., Araújo, C., Button, J., Chow, Y., Davids, K., & Moy, B. (2016). Why the Constraints-Led Approach is not Teaching Games for Understanding: a clarification. *Physical Education & Sport Pedagogy*, 21(5), 459–480. https://doi.org/10.1080/17408989.2015.1095870
- Rhodes, R. E., & Smith, N. E. I. (2006). Personality correlates of physical activity: A review and meta-analysis. *British Journal of Sports Medicine*, 40(12), 958–965. https://doi.org/10.1136/bjsm.2006.028860
- Ridgers, N. D., & Fairclough, S. (2011). Assessing free-living physical activity using accelerometry: Practical issues for researchers and practitioners. *European Journal of Sport Science*, *11*(3), 205–213. https://doi.org/10.1080/17461391.2010.501116
- Robbins, L. B., Ling, J., Sharma, D. B., Dalimonte-Merckling, D. M., Voskuil, V.
  R., Resnicow, K., Kaciroti, N., & Pfeiffer, K. A. (2019). Intervention Effects of "Girls on the Move" on Increasing Physical Activity: A Group Randomized Trial. *Annals of Behavioral Medicine*, *53*, 493–500. https://doi.org/10.1093/abm/kay054
- Robbins, L. B., Pfeiffer, K. A., Vermeesch, A., Resnicow, K., You, Z., An, L., &Wesolek, S. M. (2013). "Girls on the Move" intervention protocol for increasing physical activity among low-active underserved urban girls: a group

randomized trial. *BMC Public Health*, *13*(474). https://doi.org/10.1186/1471-2458-13-474

- Robinson, L. E. (2010). The relationship between perceived physical competence and fundamental motor skills in preschool children. *Child: Care, Health and Development*, 37(4), 589–596. https://doi.org/10.1111/j.1365-2214.2010.01187.x
- Robinson, L. E., Stodden, D. F., Barnett, L. M., Lopes, V. P., Logan, W., Rodrigues,
  L. P., & D'Hondt, E. (2015). Motor Competence and its Effect on Positive
  Developmental Trajectories of Health. *Sports Medicine*, 45(7), 9–24.
  https://doi.org/10.1007/s40279-015-0351-6
- Roesch, S. C., Norman, G. J., Merz, E. L., & Patrick, K. (2013). Longitudinal measurement invariance of psychosocial measures in physical activity research:
  An application to adolescent data. *Journal of Applied Social Psychology*, 43(4), 721–729. https://doi.org/10.1111/jasp.12000
- Roscoe, C. M. P., James, R. S., & Duncan, M. J. (2019). Accelerometer-based physical activity levels, fundamental movement skills and weight status in British preschool children from a deprived area. *European Journal of Pediatrics*, *178*(7), 1043–1052. https://doi.org/10.1007/s00431-019-03390-z
- Rosewater, A. (2009). Learning to Play and Playing to Learn: Organized Sports and Educational Outcome. *Political Science*, *75*(1), 50–56.
- Rowe, D., & Murtagh, S. (2012). The FifeActive Study. Physical activity participation in Fife school children: Research report of current levels, modes, contexts, preferences and determinants. In *Fife Council*. https://www.researchgate.net/publication/259398343

- Rudd, J. R., Barnett, L. M., Farrow, D., Berry, J., Borkoles, E., & Polman, R. (2016).
  Effectiveness of a 16 week gymnastics curriculum at developing movement
  competence in children. *Journal of Science and Medicine in Sport*, 20(2), 164–
  169. https://doi.org/10.1016/j.jsams.2016.06.013
- Saint-Maurice, P. F., Welk, G. J., Beyler, N. K., Bartee, R. T., & Heelan, K. a. (2014). Calibration of self-report tools for physical activity research: the Physical Activity Questionnaire (PAQ). *BMC Public Health*, 14, 461. https://doi.org/10.1186/1471-2458-14-461
- Sallis, J. ., Bull. F, Guthold, R., Heath, G. ., Inoue, S., Kelly, P., Oyeyemi, A. L., Perez, L. G., Richards, J., & Hallal, P. C. (2016). Progress in physical activity over the Olympic quadrennium. In *The Lancet Physical Activity Series 2 Executive Committee* (Vol. 388, Issue 10051). https://doi.org/10.1016/S0140-6736(16)30581-5
- Sallis, J. F. (1991). Self-report measures of children's physical activity. *Journal of School Health*, *61*, 215–219.
- Sallis, J. F., Cervero, R. B., Ascher, W., Henderson, K. A., Kraft, M. K., & Kerr, J. (2006). An ecological approach to creating active living communities. *Annual Review of Public Health*, 27, 297–322. https://doi.org/10.1146/annurev.publhealth.27.021405.102100

Sallis, J. F., McKenzie, T. L., Alcaraz, J. E., Kolody, B., Faucette, N., & Hovell, M. F. (1997). The Effects of a 2-year Physical Education Program (SPARK) on Physical Activity and Fitness in Elementary School Students. Sports, Play and Active Recreation for Kids. *American Journal of Public Health*, 87(8), 1328–1334. https://doi.org/10.2105/AJPH.87.8.1328

Salvy, S., Haye, K. De, Bowker, J. C., & Hermans, R. C. J. (2012). Influence of Peers and Friends on Children's and Adolescents' Eating and Activity Behaviors. *Physiology and Behavior*, *106*(3), 369–378. https://doi.org/10.1016/j.physbeh.2012.03.022.Influence

Sanders, G., Peacock, C., Williamson, M., Wilson, K., Carnes, A., & Barkley, J. (2014). The effect of friendship groups on children's physical activity: An experimental study. *Journal of Behavioral Health*, 3(2), 95–100. https://doi.org/10.5455/jbh.20140424022157

- Santos, F., Corte-Real, N., Regueiras, L., Dias, C., Martinek, T. J., & Fonseca, A. (2019). Coaching effectiveness within competitive youth football: youth football coaches' and athletes' perceptions and practices. *Sports Coaching Review*, 8(2), 172–193. https://doi.org/10.1080/21640629.2018.1459356
- Saunders, M., Lewis, P., & Thornhill, A. (2009). Understanding research philosophies and approaches to theory development. In *Understanding research philosophies and approaches* (pp. 122–161). Pearson Education. https://www.researchgate.net/publication/309102603\_Understanding\_research\_ philosophies\_and\_approaches
- Sawka, K. J., McCormack, G. R., Nettel-Aguirre, A., Hawe, P., & Doyle-Baker, P. K. (2013). Friendship networks and physical activity and sedentary behavior among youth: a systematized review. *Internatinal Journal of Behavioural Nutrition and Physical Activity*, *10*(130). https://doi.org/10.1186/1479-5868-10-130
- Schmidt, S. C. E., Schneider, J., Reimers, A. K., Niessner, C., & Woll, A. (2019). Exploratory determined correlates of physical activity in children and

adolescents: The MoMo study. *International Journal of Environmental Research and Public Health*, *16*(415). https://doi.org/10.3390/ijerph16030415

- Schneider, M., Dunton, G. F., & Cooper, D. M. (2008). Physical Activity and Physical Self-Concept among Sedentary Adolescent Females; An Intervention Study. *Psychology of Sport and Exercise*, 9(1), 1–14. https://doi.org/10.1016/j.psychsport.2007.01.003
- Schranz, N., Glennon, V., Evans, J., Gomersall, S., Hardy, L., Hesketh, K. D.,
  Lubans, D., Ridgers, N. D., Straker, L., Stylianou, M., Tomkinson, G. R., Vella,
  S., Ziviani, J., & Olds, T. (2018). Results from Australia's 2018 Report Card on
  Physical Activity for Children and Youth. *Journal of Physical Activity & Health*, 15(S2), S315–S317. https://doi.org/10.1123/jpah.2018-0418
- Schweizer, M. ., Braun, B. ., & Milstone, A. . (2016). Research Methods in Healthcare Epidemiology and Antimicrobial Stewardship – Quasi-Experimental Designs. *Physiology & Behavior*, *37*(10), 1135–1140. https://doi.org/10.1017/ice.2016.117.
- Seabra, A. C., Seabra, A. F., Mendonça, D. M., Brustad, R., Maia, J. A., Fonseca, A. M., & Malina, R. M. (2012). Psychosocial correlates of physical activity in school children aged 8-10 years. *European Journal of Public Health*, 23(5), 794–798. https://doi.org/10.1093/eurpub/cks149
- Sebire, S. J., Jago, R., Fox, K. R., Edwards, M. J., & Thompson, J. L. (2013). Testing a self-determination theory model of children's physical activity motivation: a cross-sectional study. *Internatinal Journal of Behavioural Nutrition and Physical Activity*, 10(111). http://www.ijbnpa.org/content/10/1/111

- Sebire, S. J., Kesten, J. M., Edwards, M. J., May, T., Banfield, K., Tomkinson, K., Blair, P. S., Bird, E. L., Powell, J. E., & Jago, R. (2016). Using selfdetermination theory to promote adolescent girls' physical activity: Exploring the theoretical fidelity of the Bristol Girls Dance Project. *Psychology of Sport and Exercise*, 24, 100–110. https://doi.org/10.1016/j.psychsport.2016.01.009
- Sedgwick, P. (2014). Cross sectional studies: advantages and disadvantages. *BMJ* (Online), 348(g2276). https://doi.org/10.1136/bmj.g2276
- Shadish, W. R. (2011). Randomized controlled studies and alternative designs in outcome studies: Challenges and opportunities. *Research on Social Work Practice*, 21(6), 636–643. https://doi.org/10.1177/1049731511403324
- Shannon, S., Brennan, D., Hanna, D., Younger, Z., Hassan, J., & Breslin, G. (2018).
  The Effect of a School-Based Intervention on Physical Activity and Well-Being: a Non-Randomised Controlled Trial with Children of Low Socio-Economic Status. *Sports Medicine - Open*, 4(16).
  https://doi.org/10.1186/s40798-018-0129-0
- Shen, B., Centeio, E., Garn, A., Martin, J., Kulik, N., Somers, C., & McCaughtry, N. (2018). Parental social support, perceived competence and enjoyment in school physical activity. *Journal of Sport and Health Science*, 7(3), 346–352. https://doi.org/10.1016/j.jshs.2016.01.003
- Sheppard, J. (2014). Personal and Social Responsibility Through Game Play: Utilizing the Teaching Games for Understanding Instructional Model. University of Toronto.
- Sheridan, D., Coffee, P., & Lavallee, D. (2014). A systematic review of social support in youth sport. *International Review of Sport and Exercise Psychology*,

7, 198-228. https://doi.org/10.1080/1750984X.2014.931999

- Sherwood, N. E., Taylor, W. C., Treuth, M., Klesges, L. M., Baranowski, T., Zhou,
  A., Pratt, C., Mcclanahan, B., Robinson, T., Pruitt, L., & Miller, W. (2004).
  Measurement characteristics of activity-related psychosocial measures in 8- to
  10-year-old African-American girls in the Girls health Enrichment Multisite
  Study (GEMS). *Preventive Medicine*, *38*, S60–S68.
  https://doi.org/10.1016/j.ypmed.2003.12.030
- Shull, E. R., Dowda, M., Saunders, R. P., McIver, K., & Pate, R. R. (2019). Sport participation, physical activity and sedentary behavior in the transition from middle school to high school. *Journal of Science and Medicine in Sport*, 23(4), 385–389. https://doi.org/10.1016/j.jsams.2019.10.017
- Silva, G., Andersen, L. B., Aires, L., Mota, J., Oliveira, J., Silva, G., Andersen, L.
  B., Aires, L., Mota, J., Oliveira, J., Silva, G., Andersen, L. B. O., Aires, L.,
  Mota, J., & Ribeiro, J. C. (2013). Associations between sports participation ,
  levels of moderate to vigorous physical activity and cardiorespiratory fitness in
  childrenand adolescents. *Journal of Sports Sciences*, *31*(12), 1359–1367.
  https://doi.org/10.1080/02640414.2013.781666
- Sims, J., Scarborough, P., & Foster, C. (2015). The Effectiveness of Interventions on Sustained Childhood Physical Activity : A Systematic Review and Meta-Analysis of Controlled Studies. *PLoS ONE*, 10(7). https://doi.org/10.1371/journal.pone.0132935
- Slater, A., & Tiggemann, M. (2010). "Uncool to do sport": A focus group study of adolescent girls' reasons for withdrawing from physical activity. *Psychology of Sport and Exercise*, 11(6), 619–626.

https://doi.org/10.1016/j.psychsport.2010.07.006

- Slingerland, M., Haerens, L., Cardon, G., & Borghouts, L. (2014). Diferences in perceived competence and physical activity levels during single-gender modifed basketball game play in middle school physical education. *European Physical Education Review*, 20(1), 20–35. https://doi.org/0.1177/1356336X13496000
- Slykerman, S., Ridgers, N. D., Stevenson, C., & Barnett, L. M. (2016). How important is young children's actual and perceived movement skill competence to their physical activity? *Journal of Science and Medicine in Sport*, 19(6), 488–492. https://doi.org/10.1016/j.jsams.2015.07.002
- Smith, A. L., & McDonough, M. H. (2008). Peers. In A.L. Smith & S.J.H. Biddle (Eds.), Youth physical activity and sedentary behavior. Human Kinetics.
- Smyth, E. (2016). Arts and Cultural Participation among Children and Young people: Insights from the Growing Up in Ireland Study. The Economic and Social Research Insititute.
- Somerset, S., & Hoare, D. J. (2018). Barriers to voluntary participation in sport for children: A systematic review. *BMC Pediatrics*, 18(47). https://doi.org/10.1186/s12887-018-1014-1
- Sommer, L. (2011). The Theory Of Planned Behaviour And The Impact Of Past Behaviour. International Business & Economics Research Journal (IBER), 10(1), 91–110. https://doi.org/10.19030/iber.v10i1.930
- Spessato, B. C., Gabbard, C., Valentini, N., Rudisill, M., Coiro, B., Gabbard, C., Valentini, N., & Rudisill, M. (2013). Gender differences in Brazilian children's fundamental movement skill performance. *Early Child Development and Care*,

183(7), 916–923. https://doi.org/10.1080/03004430.2012.689761

- Sport for Life Society. (2016). Canadian Sport for Life: Long-Term Athlete Development Framework Paper 2.1. In *Sport for Life Society*. https://proxy.lib.ohiostate.edu/login?url=http://search.ebscohost.com/login.aspx?direct=true&db=s3h &AN=23787650&site=ehost-live
- Sport Northern Ireland. (2016). *Exploring attitudes to and experiences of sport and physical activity among children and young people in Northern Ireland. A mixed methods project Evidence from the 2016 Young Life and Times (YLT) and Kids Life and Times (KLT) surveys.* http://www.sportni.net/sportni/wpcontent/uploads/2017/03/Exploring-attitudes-to-and-experiences-of-sport-andphysical-activity-among-children-and-young-people-in-Northern-Ireland.pdf
- Springer, J. B. (2013). "I Am Very, Very Proud of Myself": Improving Youth Activity Levels Using Self-Determination Theory in Program Development. *Frontiers in Public Health*, 1(46). https://doi.org/10.3389/fpubh.2013.00046
- Staley, A., & Moore, E. W. G. (2016). Proper Use of Feedback Leads to an Optimal Motivational Climate. *Journal of Physical Education, Recreation & Dance*, 87(8), 47–52. https://doi.org/10.1080/07303084.2016.1216483
- Standage, M., Gillison, F. B., Ntoumanis, N., & Treasure, D. C. (2012). Predicting students' physical activity and health-related well-being: A prospective crossdomain investigation of motivation across school physical education and exercise settings. *Journal of Sport and Exercise Psychology*, 34, 37–60. https://doi.org/10.1123/jsep.34.1.37

Stanley, R. M., Ridley, K., & Dollman, J. (2012). Correlates of children's time-

specific physical activity: A review of the literature. *International Journal of Behavioral Nutrition and Physical Activity*, 9(50). https://doi.org/10.1186/1479-5868-9-50

- Stanley, R. M., Ridley, K., Olds, T. S., & Dollman, J. (2014). Increasing specificity of correlate research: Exploring correlates of children's lunchtime and afterschool physical activity. *PLoS ONE*, 9(5), e96460. https://doi.org/10.1371/journal.pone.0096460
- Stearns, J. A., Godley, J., Veugelers, P. J., Ekwaru, J. P., Bastian, K., Wu, B., &
  Spence, J. C. (2019). Associations of friendship and children's physical activity during and outside of school: A social network study. *SSM Population Health*, 7, 100308. https://doi.org/10.1016/j.ssmph.2018.10.008
- Stellino, M. B., & Sinclair, C. D. (2013). Psychological Predictors of Children's Recess Physical Activity Motivation and Behavior. *Research Quarterly for Exercise and Sport*, 84(2), 167–176. https://doi.org/10.1080/02701367.2013.786159
- Sterdt, E., Liersch, S., & Walter, U. (2014). Correlates of physical activity of children and adolescents : A systematic review of reviews. *Health Education Journal*, 73(1), 72–89. https://doi.org/10.1177/0017896912469578
- Stodden, D. F., Gao, Z., Goodway, J. D., & Langendorfer, S. J. (2014). Dynamic Relationships Between Motor Skill Competence and Health-Related Fitness in Youth. *Pediatric Exercise Science*, 26, 231–241. https://doi.org/10.1123/pes.2013-0027
- Stodden, D. F., Goodway, J. D., Langendorfer, S. J., Roberton, M. A., Rudisill, M.E., Garcia, C., & Garcia, L. E. (2008). A Developmental Perspective on the

Role of Motor Skill Competence in Physical Activity: An Emergent Relationship. *Motor Competence and Physical Activity*, *60*, 290–306.

- Stokols, D., Allen, J., & Bellingham, R. (1996). The social ecology of health promotion: Implications for research and practice. *American Journal of Health Promotion*, 10(4), 247–251. https://doi.org/10.4278/0890-1171-10.4.247
- Story, C. R., Knutson, D., Brown, J. B., Spears-Laniox, E., Harvey, I. S., Gizlice, Z., & Whitt-Glover, M. C. (2017). Changes in social support over time in a faith-based physical activity intervention. *Health Education Research*, 32(6), 513–523. https://doi.org/10.1093/her/cyx062
- Sullivan, P., Paquette, K. J., Holt, N. L., & Bloom, G. A. (2012). The relation of coaching context and coach education to coaching efficacy and perceived leadership behaviors in youth sport. *Sport Psychologist*, *26*(1), 122–134. https://doi.org/10.1123/tsp.26.1.122
- Sutherland, R. L., Campbell, E. M., Lubans, D. R., Morgan, P. J., Nathan, N. K.,
  Prom, M. H., Wolfenden, L., Okely, A. D., Gillham, K. E., Sc, M. S., Hollis, J.
  L., Oldmeadow, C. J., Williams, A. J., N, B. H. S., Davies, L. J., Wiese, J. S.,
  Bisquera, A., & Wiggers, J. H. (2016). The Physical Activity 4 Everyone
  Cluster Randomized Trial. *American Journal of Preventive Medicine*, *51*(2),
  195–205. https://doi.org/10.1016/j.amepre.2016.02.020

Sylvia, L. G., Bernstein, E. E., Hubbard, J. L., Keating, L., & Anderson, E. J. (2014). Practical guide to measuring physical activity. *Journal of the Academy of Nutrition and Dietetics*, *114*(2), 199–208. https://doi.org/10.1016/j.jand.2013.09.018

Tannehill, D., MacPhail, A., Walsh, J., & Woods, C. (2013). What young people say

about physical activity: the Children's Sport Participation and Physical Activity (CSPPA) study. *Sport, Education and Society*, *20*(4), 442–462. https://doi.org/10.1080/13573322.2013.784863

- Tashakkori, A., & Creswell, J. W. (2007). Editorial: The New Era of Mixed Methods. In *Journal of Mixed Methods Research* (Vol. 1, Issue 1). https://doi.org/10.1177/2345678906293042
- Taylor, P., Guzmán, J. F., & Kingston, K. (2012). Prospective study of sport dropout: A motivational analysis as a function of age and gender. *European Journal of Sport Science*, *12*(5), 431–442. https://doi.org/10.1080/17461391.2011.573002
- Taylor, W. ., Chan, W., & Cummings, S. S. (2004). Healthy growth: Project description and baseline finding. Ethnicity & Disease, 38, 78 – 87. *Ethnicity & Disease*, 38, 78–87.
- Teddlie, C., & Tashakkori, A. (2009). Foundations of Mixed Methods Research-Integrating Quantitative and Qualitative Approaches in the Social and Behavioral Sciences.pdf. SAGE.
- Telford, R. M., Telford, R. D., Olive, L. S., Cochrane, T., & Davey, R. (2016). Why Are Girls Less Physically Active than Boys? Findings from the LOOK Longitudinal Study. *PLoS ONE*, *11*(3), e0150041. https://doi.org/10.1371/journal.pone.0150041
- Tesch, R. (1990). *Qualitative research: Analysis types and software tools*. Falmer Press.

Tibbitts, B., Porter, A., Sebire, S. J., Bird, E. L., Sanderson, E., Metcalfe, C., Powell,

J. E., & Jago, R. (2019). Action 3:30R: Process evaluation of a cluster randomised feasibility study of a revised teaching assistant-led extracurricular physical activity intervention for 8 to 10 year olds. *BMC Public Health*, *19*(1111). https://doi.org/10.1186/s12889-019-7347-3

- Timperio, A. F., van Stralen, M. M., Brug, J., Bere, E., Chinapaw, M. J. M., De
  Bourdeaudhuij, I., Jan, N., Maes, L., Manios, Y., Moreno, L. A., Salmon, J., &
  te Velde, S. J. (2013). Direct and indirect associations between the family
  physical activity environment and sports participation among 10-12 year-old
  European children: Testing the EnRG framework in the ENERGY project. *International Journal of Behavioral Nutrition and Physical Activity*, *10*(15).
  https://doi.org/10.1186/1479-5868-10-15
- Tompsett, C., Sanders, R., Taylor, C., & Coblet, S. (2017). Pedagogical Approaches to and Effects of Fundamental Movement Skill Interventions on Health Outcomes: A Systematic Review. *Sports Medicine*, 49(7), 1795–1819. https://doi.org/10.1007/s40279-017-0697-
- Tong, A., Sainsbury, P., & Craig, J. (2007). Consolidated criteria for reporting qualitative research (COREQ): A 32-item checklist for interviews and focus groups. *International Journal for Quality in Health Care*, 19(6), 349–357. https://doi.org/10.1093/intqhc/mzm042
- Tremblay, M. S., Barnes, J. D., González, S. A., Katzmarzyk, P. T., Onywera, V. O.,
  Reilly, J. J., Tomkinson, G. R., Aguilar-Farias, N., Akinroye, K. K., Al-Kuwari,
  M. G., Amornsriwatanakul, A., Aubert, S., Belton, S., Gołdys, A., HerreraCuenca, M., Jeon, J. Y., Jürimäe, J., Katapally, T. R., Lambert, E. V., ... Wong,
  S. H. (2016). Global matrix 2.0: Report card grades on the physical activity of

children and youth comparing 38 countries. *Journal of Physical Activity and Health*, *13*(Suppl 2), S343–S366. https://doi.org/10.1123/jpah.2016-0594

- Trost, S. G. (2007). Measurement of Physical Activity in Children and Adolescents. *American Journal of Lifestyle Medicine*, 1(4), 299–314. https://doi.org/10.1177/1559827607301686
- Trost, S. G., Blair, S. N., & Khan, K. M. (2014). Physical inactivity remains the greatest public health problem of the 21st century: Evidence, improved methods and solutions using the "7 investments that work" as a framework. *British Journal of Sports Medicine*, 48(3), 169–170. https://doi.org/10.1136/bjsports-2013-093372
- Trudel, P., Gilbert, W., & Werthner, P. (2010). "Coach Education Effectiveness" in Sport Coaching: Professionalism and Practice (J. Lyle & C. Cushion (eds.)). Elsevier.
- Ulrich, B. D. (1987). Perceptions of physical competence, motor competence, and participation in organized sport: Their interrelationships in young children. *Research Quarterly for Exercise and Sport*, 58(1), 57–67.
  https://doi.org/10.1080/02701367.1987.10605421

Ulrich, D. . (1985). Test of gross motor development. Pro-Ed.

Ulrich, D. . (2000). Test of gross motor development-2 (2nd ed.). Pro-Ed.

Vallence, A. M., Hebert, J., Jespersen, E., Klakk, H., Rexen, C., & Wedderkopp, N. (2019). Childhood motor performance is increased by participation in organized sport: the CHAMPS Study-DK. *Scientific Reports*, 9(18920). https://doi.org/10.1038/s41598-019-54879-4 van Beurden, E., Zask, A., Barnett, L. M., & Dietrich, U. C. (2002). Fundamental movement skills - How do primary school children perform? The "Move it Groove it" program in rural Australia. *Journal of Science and Medicine in Sport* / Sports Medicine Australia, 5(3), 244–252. https://doi.org/10.1016/S1440-2440(02)80010-X

Van Der Horst, K., Paw, M. J. C. A., Twisk, J. W. R., & Van Mechelen, W. (2007).
A brief review on correlates of physical activity and sedentariness in youth. *Medicine and Science in Sports and Exercise*, 39(8), 1241–1250.
https://doi.org/10.1249/mss.0b013e318059bf35

- Van Hecke, L., Loyen, A., Verloigne, M., van der Ploeg, Hidde PLakerveld, J., Brug, J., De Bourdheaudhuij, I., Ekelund, U., Donnelly, A., Hendriksen, I., & Deforche, B. (2016). Variation in population levels of sedentary time in European children and adolescents according to cross-European studies: a systematic literature review within DEDIPAC. *The International Journal of Behavioral Nutrition and Physical Activity*, *13*(70). https://doi.org/10.1186/s12966-016-0396-4
- Van Lippevelde, W., Verloigne, M., De Bourdeaudhuij, I., Brug, J., Bjelland, M.,
  Lien, N., & Maes, L. (2012). Does parental involvement make a difference in school-based nutrition and physical activity interventions? A systematic review of randomized controlled trials. *International Journal of Public Health*, 57(4), 673–678. https://doi.org/10.1007/s00038-012-0335-3
- Van Maanen, J., Sørensen, J. B., & Mitchell, T. R. (2007). The interplay between theory and method. *Academy of Management Review*, 32(4), 1145–1154. https://doi.org/10.5465/AMR.2007.26586080

- Van Sluijs, E. M. F., & Kriemler, S. (2016). Reflections on physical activity intervention research in young people dos, don'ts, and critical thoughts. *International Journal of Behavioral Nutrition and Physical Activity*, *13*(25). https://doi.org/10.1186/s12966-016-0348-z
- Van Sluijs, E, McMinn, A. M., & Griffin, S. J. (2007). Effectiveness of interventions to promote physical activity in children and adolescents: Systematic review of controlled trials. *British Journal of Sports Medicine*, 335(703). https://doi.org/10.1136/bmj.39320.843947.BE
- Van Sluijs, EM, McMinn, A. M., Inskip, H. M., Ekelund, U., Godfrey, K. M., Harvey, N. C., & Griffin, S. J. (2013). Correlates of light and moderate-tovigorous objectively measured physical activity in fouryear-old children. *PLoS ONE*, 8(9), e74934. https://doi.org/10.1371/journal.pone.0074934
- Vanhelst, J., Vidal, F., Drumez, E., Béghin, L., Baudelet, J. B., Coopman, S., & Gottrand, F. (2019). Comparison and validation of accelerometer wear time and non-wear time algorithms for assessing physical activity levels in children and adolescents. *BMC Medical Research Methodology*, *19*(72). https://doi.org/10.1186/s12874-019-0712-1
- Vedul-Kjelsås, V., Stensdotter, A. K., Haga, M., & Sigmundsson, H. (2015).
  Physical fitness, self-perception and physical activity in children with different motor competence. *European Journal of Adapted Physical Activity*, 8(1), 45–57. https://doi.org/10.5507/euj.2015.004
- Vella, S. A., Cliff, D. P., & Okely, A. D. (2014). Socio-ecological predictors of participation and dropout in organised sports during childhood. *Internatinal Journal of Behavioural Nutrition and Physical Activity*, 11(62).

- Vella, S. A., Cliff, D. P., Okely, A. D., Scully, M. L., & Morley, B. C. (2013).
  Associations between sports participation, adiposity and obesity-related health behaviors in Australian adolescents. *Internatinal Journal of Behavioural Nutrition and Physical Activity*, *10*(113). https://doi.org/10.1186/1479-5868-10-113
- Verloigne, Maite, Cardon, G., De Craemer, M., D'Haese, S., & De Bourdeaudhuij, I. (2016). Mediating effects of self-efficacy, benefits and barriers on the association between peer and parental factors and physical activity among adolescent girls with a lower educational level. *PLoS ONE*, *11*(6). https://doi.org/10.1371/journal.pone.0157216
- Verloigne, Maïté, Lippevelde, W. Van, Maes, L., Mine, Y., Chinapaw, M., & Manios, Y. (2012). Levels of physical activity and sedentary time among 10- to 12-year-old boys and girls across 5 European countries using accelerometers : an observational study within the ENERGY-project. *Internatinal Journal of Behavioural Nutrition and Physical Activity*, *9*(34). https://doi.org/10.1186/1479-5868-9-34
- Visek, A. J., Achrati, S. M., Manning, H., McDonnell, K., Harris, B. S., & DiPietro,
  L. (2015). The Fun Integration Theory: Towards Sustaining Children and
  Adolescents Sport Participation. *Journal of Physical Activity & Health*, 12(3),
  424–433. https://doi.org/10.1123/jpah.2013-0180
- Visek, A. J., Mannix, H., Chandran, A., Cleary, S. D., & Dipietro, L. (2018).Perceived importance of the fun integration theory's factors and determinants:A comparison among players, parents, and coaches. *International Journal of*

*Sports Science & Coaching*, *13*(6), 849–862. https://doi.org/10.1177/1747954118798057

- Vitali, F., Robazza, C., Bortoli, L., Bertinato, L., Schena, F., & Lanza, M. (2019). Enhancing fitness, enjoyment, and physical self-efficacy in primary school children: A DEDIPAC naturalistic study. *PeerJ*, 7, e6436. https://doi.org/10.7717/peerj.6436
- Voorhees, C. C., Murray, D., Welk, G., Birnbaum, A., Ribisl, K.M., J., Ohnson, C. C., & Jobe, J. B. (2005). The role of peer social network factors and physical activity in adolescent girls. *American Journal of Health Behavior*, 29(2), 183–190. https://doi.org/10. 5993/AJHB.29.2.9
- Voskuil, V. R., Frambes, D. A., & Robbins, L. B. (2017). Effect of Physical Activity Interventions for Girls on Objectively Measured Outcomes: A Systematic Review of Randomized Controlled Trials. *Journal of Pediatric Health Care: Official Publication of National Association of Pediatric Nurse Associates & Practitioners*, 31(1), 75–87. https://doi.org/10.1016/j.pedhc.2016.03.003
- Wang, J. J., Baranowski, T., Lau, P. W. C., Chen, T. A., & Zhang, S. G. (2016).
  Psychological correlates of self-reported and objectively measured physical activity among chinese children—psychological correlates of PA. *International Journal of Environmental Research and Public Health*, *13*(1006).
  https://doi.org/10.3390/ijerph13101006
- Wang, J. J., Baranowski, T., Lau, W. C. P., Chen, T. A., & Pitkethly, A. J. (2016).
  Validation of the Physical Activity Questionnaire for Older Children (PAQ-C) among Chinese Children. *Biomedical and Environmental Sciences*, 29(3), 177–186. https://doi.org/10.3967/bes2016.022

- Wang, L., & Wang, L. (2015). Using Theory of Planned Behavior to Predict the Physical Activity of Children: Probing Gender Differences. *BioMed Research International*, 536904. https://doi.org/10.1155/2015/536904
- Wetton, A. R., Radley, R., Jones, A. R., & Pearce, M. S. (2013). What are the barriers which discourage 15-16 year-old girls from participating in team sports and how can we overcome them? *BioMed Research International*, 1–8. https://doi.org/10.1155/2013/738705
- Wheeler, S. (2012). The significance of family culture for sports participation. International Review for the Sociology of Sport, 47(2), 235–252. https://doi.org/10.1177/1012690211403196
- White, H., & Sabarwal, S. (2014). *Quasi-Experimental Design and Methods* (Issue 8). Methodological Briefs: Impact Evaluation 8.
- White, R. L., Parker, P. D., Lubans, D. R., Macmillan, F., Olson, R., Astell-burt, T., & Lonsdale, C. (2018). Domain-specific physical activity and affective wellbeing among adolescents: an observational study of the moderating roles of autonomous and controlled motivation. *Internatinal Journal of Behavioural Nutrition and Physical Activity*, *15*(87). https://doi.org/10.1186/s12966-018-0722-0
- Whitehead, M. (2010). The Concept of Physical Literacy. In M. Whitehead (Ed.), Physical Literacy throughout the life course. Routledge.
- Whitehead, S., & Biddle, S. (2008). Adolescent girls' perceptions of physical activity: A focus group study. *European Physical Education Review*, 14(243). https://doi.org/10.1177/1356336X08090708

- Whitley, M. A., Massey, W. V., Camiré, M., Boutet, M., & Borbee, A. (2019).
  Sport-based youth development interventions in the United States: a systematic review. *BMC Public Health*, *19*(89). https://doi.org/10.1186/s12889-019-6387-z
- Wick, K., Leeger-Aschmann, C. S., Monn, N. D., Radtke, T., Ott, L. V., Rebholz, C.
  E., Cruz, S., Gerber, N., Schmutz, E. A., Puder, J. J., Munsch, S., Kakebeeke, T.
  H., Jenni, O. G., Granacher, U., & Kriemler, S. (2017). Interventions to
  Promote Fundamental Movement Skills in Childcare and Kindergarten: A
  Systematic Review and Meta-Analysis. *Sports Medicine*, *47*, 2045–2068.
  https://doi.org/10.1007/s40279-017-0723-1
- Wiium, N., & Säfvenbom, R. (2019). Participation in organized sports and selforganized physical activity: Associations with developmental factors.
   *International Journal of Environmental Research and Public Health*, 16(4).
   https://doi.org/10.3390/ijerph16040585
- Wijtzes, A., & Verloigne, M. (2016). Results from the Belgium;s 2016 Report Card on Physical Activity for Children and Youth. *Journal of Physical Activity & Health*, 13(Suppl 2), S95–S103. https://doi.org/10.1123/jpah.2016-0306
- Wilkie, H. J., Standage, M., Gillison, F. B., Cumming, S. P., & Katzmarzyk, P. T. (2018). Correlates of intensity-specific physical activity in children aged 9-11 years: A multilevel analysis of UK data from the International Study of Childhood Obesity, Lifestyle and the Environment. *BMJ Open*, *8*, e018373. https://doi.org/10.1136/bmjopen-2017-018373
- Willis, K., Tibbitts, B., Sebire, S. J., Reid, T., MacNeill, S. J., Sanderson, E.,Hollingworth, W., Kandiyali, R., Campbell, R., Kipping, R. R., & Jago, R.(2019). Protocol for a cluster randomised controlled trial of a Peer-Led physical

Activity iNtervention for Adolescent girls (PLAN-A). *BMC Public Health*, *19*(644). https://doi.org/10.1186/s12889-019-7012-x

- Witt, P. A., & Dangi, T. (2018). Why Children/Youth Drop Out of Sports. Journal of Park and Recreation Administration, 36, 191–199. https://doi.org/10.18666/jpra-2018-v36-i3-8618
- Woods, C.B., Tannehill, D., Quinlan, A., Moyna, N., & Walsh, J. (2010). The Children's Sport Participation and Physical Activity Study (CSPPA). Research Report No 1.
- Woods, C, Moynihan, N., Quinlan, A., Tannehill, D., & Walsh, J. (2010). The Children's Sport Participation and Physical Activity Study (CSPPA Study).
   Research Report No. 1. http://www.irishsportscouncil.ie/Research/Publications/
- Woods, Catherine B, Powelll, C., Saunders, J. ., O'Brien, W., Murphy, M. ., Duff,
  C., Farmer, O., Johnston, A., Connolly, S., & Belton, S. (2018). *The Children's* Sport Participation and Physical Activity Study (CSPPA 2018).
  https://doi.org/10.13140/RG.2.2.22091.95521
- Woods, Catherine, Walsh, J., & Tannehill, D. (2010). The Children's Sport
  Participation and Physical Activity Study (CSPPA Study) Volunteer Study
  Research Report No. 2. In School of Health and Human Performance, Dublin
  City University and the Irish Sports Council.
- Yang, C. C., & Hsu, Y. L. (2010). A review of accelerometry-based wearable motion detectors for physical activity monitoring. *Sensors*, 10(8), 7772–7788. https://doi.org/10.3390/s100807772

Yao, C. A., & Rhodes, R. E. (2015). Parental correlates in child and adolescent

physical activity: A meta-analysis. In *International Journal of Behavioral Nutrition and Physical Activity* (Vol. 12, Issue 10). https://doi.org/10.1186/s12966-015-0163-y

- Yefimov, V. (2004). On Pragmatist Institutional Economics. IDEAS Working Paper Series from RePEc; Munich. Munich Personal RePEc Archive.
- Young, D., Saksvig, B. I., Wu, T. T., Zook, K., Li, X., Champaloux, S., Grieser, M., Lee, S., & Treuth, M. S. (2014). Multilevel Correlates of Physical Activity for Early, Mid, and Late Adolescent Girls. *Journal of Physical Activity & Health*, *11*(5), 950–960. https://doi.org/10.1123/jpah.2012-0192
- Youth Sport Trust. (2017). Key Findings From Girls Active Survey. In *Women in Sport*. https://www.womeninsport.org/wp-content/uploads/2017/11/Girls-Active-statistics-1.pdf?x99836
- Zhang, T., Solmon, M. A., Gao, Z., & Kosma, M. (2012). Promoting School Students' Physical Activity: A Social Ecological Perspective. *Journal of Applied Sport Psychology*, 24(1), 92–105. https://doi.org/10.1080/10413200.2011.627083
- Zuvela, F., Bozanic, A., & Miletic, D. (2011). POLYGON A new fundamental movement skills test for 8 year old children: Construction and validation. *Journal of Sports Science and Medicine*, 10(1), 157–163.
- Zwolinsky, S., Kime, N., Pringle, A., Widdop, P., & McKenna, J. (2018). Designing programmes of physical activity through sport: learning from a widening participation intervention, 'City of Football.' *BMC Public Health*, 18(1). https://doi.org/10.1186/s12889-018-6049-6
# **APPENDIX A:**

# **ETHICAL APPROVAL LETTER**



University College Cork, Ireland Coláiste na hOllscoile Corcaigh

Oifig an Leas - Uachtaráin Taighde agus Nuálaíochta Office of the Vice President for Research and Innovation

4th Floor, Block E, Food Science Building, University College Cork, College Road, Cork, Ireland.

T +353 (0)21 4903500 E vpresearch@ucc.ie www.ucc.ie

Wesley O'Brien & Orlagh Farmer Sports Studies and Physical Education Department School of Education University College Cork

18<sup>th</sup> March 2016

Dear Wesley and Orlagh,

Thank you for submitting your research (project Log 2015-005 entitled 'The design, development, implementation and evaluation of the Gaelic4Girls intervention') to SREC for ethical perusal. I am pleased to say that we see no ethical impediment to your research as proposed and we are happy to grant approval.

We wish you every success in your research.

Your sincerely,

Men my

Mike Murphy,

Chair of Social Research Ethics Committee

## **APPENDIX B:**

# **G4G INFORMATION SHEET AND CONSENT FORMS**

# **Research Project Information – 'Gaelic4Girls' Study**







Dear Parent/Guardian

My name is Orlagh Farmer, and I am a PhD student in the area of physical activity promotion for girls. I am also a member of the Cork Senior Ladies Football team and I am carrying out a study (by University College Cork) as part of the Gaeic4Girls programme in your child's local GAA club. The study aims to gather information on the physical activity levels of young girls in the club, their attitudes, perspectives on physical activity/sport & skills and implement an intervention (Gaelic4Girls) to help **improve** these levels, and then to assess whether or not activity levels have been improved. Please find overleaf an informed consent form for your child's participation in a physical activity study titled "Gaelic4Girls intervention- promotion of Physical Activity Participation in Female Youths".

In order for your child to participate in this study, please read and sign the attached form and return on the first night of Gaelic4Girls on Monday 27th March along with the signed registration form. Your child can still take part in the 12-week Gaelic4Girls programme without being involved in the study. If at any point throughout the study your child wishes to discontinue please inform the coaches/research team.

Thank you for your time.

Yours sincerely,

<u>Werley O'Brien</u> Orlagh Farmer

Dr. Wesley O'Brien. Ph.D. Principal investigator

**Project Title:** "The design, development, implementation and evaluation of the Gaelic4Girls (G4G's) Intervention- Promotion of Physical Activity through a targeted LGFA-based intervention 'Gaelic4Girls".

#### Introduction to the study:

Physical activity has been shown to be extremely beneficial to youth, however in Ireland currently 81% of primary school children are not meeting the recommended minimum of 60 minutes moderate to vigorous physical activity (MVPA) every day. In order to develop effective physical activity programmes for your child's age group, it is important that researchers understand what influences children in Ireland to become and remain active.

#### This is what will happen during the research project study: Your child will

- Complete a *physical activity questionnaire*. This questionnaire will be filled out at the training session with the help of the coaches and Ms. Orlagh Farmer (PhD Candidate UCC).
- Have their height and weight measured.
- Be recorded using a video camera in the hall to measure how well they can *-Run, Jump, Kick, Bounce, Skip, Catch & Balance*
- Your child may be asked to wear a *small device (accelerometer)* around their waist to measure how much they move in a specific length of time (7 days).Each parent/guardian may receive a text message on their mobile phone each morning from the research team reminding their children to put on the accelerometer.
- Your child may be asked to take part in a *group discussion*. These discussions will take place in the club with Ms. Orlagh Farmer and either a coach or another adult.

\*Involvement in the research is <u>completely voluntary</u>. Participants may choose to withdraw from the study at any time prior to PhD submission. This means they can withdraw before the study commences, even if participants agreed to participate; or discontinue after data collection has started. All <u>information</u> <u>gathered</u> will be treated in the <u>strictest of confidence</u>. To ensure this, your child's name will be removed from all data and replaced with an <u>ID number</u>. Only the researcher will know your child's ID number, and only the researchers will have access to the information.

#### Please read below and complete as appropriate.

-----%------

## Child to be included in the study – Informed Consent Form

Are there any medical conditions or other relevant information e.g. behavioural problems, inhalers used, allergies? Please specify below.

Yes	Please specify
No 🗌	
Signature	



gaelic4girls



## **APPENDIX C:**

# **G4G QUESTIONNAIRE**



Your RECENT physical activity

#### PHYSICAL ACTIVITY IS ANY BODY MOVEMENT

• Moderate Effort - makes your heart rate and breathing rate faster than normal. You may also sweat a little. Brisk walking and jogging are good examples.

• Vigorous Effort - makes your <u>heart rate much faster</u> and you have to<u>breathe deeper and faster</u> than normal. You will probably sweat. Playing football or tennis are good examples.

Examples: Exercise Running, dancing, etc. Sports Basketball, football, athletics, swimming, etc. General Brisk walking, washing the car, walking or cycling to school, etc.

\* 1. What is your name? (include surname)

(\*Remember your name will be changed to a code number for anonymity purposes)

\* 2. What is your date of birth? (e.g. 26/03/2017)

Day/Month/Year

\* 3. What's the name of your school?

0	1st class
0	2nd class
0	3rd class
0	4th class
0	5th class
0	6th class
5.0	<ul> <li>b) you have <u>younger</u> or <u>older</u> sibling(s) (i.e.brother/sister)</li> <li>Yes - older brother(s) or sister(s)</li> <li>Yes - younger brother(s) or sister(s)</li> <li>L have no brother or sister</li> </ul>
Oth	v (plance specify)

# \* 6. OVER THE PAST 7 DAYS... on how many days were you physically active for a total of at least 60 minutes (one hour) per day?

\*The total 60 minutes can be broken down into various stages within the day.

#### Please click on the correct number of days.







\* 7. Over a typical or usual week SCHOOL week <u>how many DAYS</u> are you physically active for a total of at least 60 minutes (one hour) per day?

\*The total 60 minutes can be broken into various stages within the day. Please click on the correct number of days.

- 0 days
- 1 day
- 2 days
- 3 days
- 4 days
- 5 days
- 6 days
- 7 days





#### PHYSICAL ACTIVITIES (not including PE Class)

#### 8. Did you do the following SPORTS ACTIVITIES in the PAST 7 DAYS?

- Please write Yes or No for each of the sports

- If you selected yes please select the number of days and minutes roughly that you spent doing that sport/activity.

	YES / NO	No of DAYS in past 7 days (1-7)	MINUTES per day
Ladies Gaelic Football	\$	•	\$
Camogie	\$	\$	\$
Gymnastics	\$	\$	\$
Dancing	\$	\$	\$
Soccer	\$	\$	\$
Basketball / Volleyball	\$	\$	\$
Rounder's / Cricket	\$	\$	\$
Hockey	\$	\$	\$
Rugby	\$	\$	\$
Martial Arts (e.g. Judo, Karate)	\$	\$	\$
Running or jogging	\$	\$	\$
Swimming lessons	\$	\$	\$
Swimming for fun	\$	\$	\$
Tennis / Badminton / squash / other racquet sports	\$	\$	\$
Golf	\$	\$	\$
Other sport	\$	\$	\$
Other (please specify)			

#### 9. How many times do you have **PE in school per week?** Please tick **ONE** box only



#### 10. And on average, how long is each PE class?(hours/minutes)



11. In this question we want to know about things you have doneIN THE LAST 7 DAYS that involve sitting down

- Please tick Yes or No for the following activities that you have done in the PAST 7 DAYS

- If YES, on how many days did you do the activity?

- On average, how many minutes did you do this activity on the days that you did it?

#### SITTING DOWN ACTIVITIES

	YES / NO	NO OF DAYS IN LAST 7 DAYS (1-7)	MINUTES PER DAY
Using Computer / Internet / IPad/ Smartphone	•	\$	\$
Sitting Playing Video Games (Play station)	\$	\$	\$
Homework, studying	•	\$	\$
Reading (not for school)	\$	\$	\$
Art & Craft (pottery, drawing, painting etc.)	\$	\$	\$
Sitting during school breaks	\$	\$	\$
Sitting and talking to friends	•	\$	\$
Listening to music	\$	\$	\$
Talking on the phone	\$	\$	\$
Television or DVD watching	\$	\$	\$
Playing a musical instrument	\$	\$	\$
Travel by car/bus to school	•	\$	\$
Other	\$	\$	\$

Other (please specify)

12. How do you usually travel to school?

Please tick one box only - for the LONGEST distance of your usual journey to school.

Walk
Cycle
Car
Bus

13. How long does your journey to school usually take? (minutes)

#### 14. How do you usually travel home from school?

Walk
Cycle
Car
Bus

15. How long does your journey home from school usually take? (minutes)



#### STATEMENTS about PHYSICAL ACTIVITY

#### Below are some statements about PHYSICAL ACTIVITY

Remember that physical activity means any activity that makes your heart beat faster and makes you get out of breath some of the time.

#### 16. Physical Activity Enjoyment Scale

When I am active .....

	Disagree a lot	Disagree a little	Neither disagree nor agree	Agree a little	Agree a lot
1. I enjoy it	0	0	$\bigcirc$	0	
2. I feel bored	0	0	0	0	
3. I dislike it	$\bigcirc$	0	0	$\bigcirc$	0
4. I feel as though I would rather be doing something else	0	0	0	0	0
5. It's no fun at all	$\bigcirc$	0	0	$\bigcirc$	
6. It feels good	0	0	0	0	0
7. It makes me depressed (sad)	0	0	0	0	0
8. My body feels good	0	0	0	0	0
9. It's very exciting	0	0	0	0	0
10. It gives me a strong feeling of success	0	0	0	0	0

\* 17. Please tick **ONE** box for <u>each</u> statement to show how true it is for you.

	Very true	Quite true	Not very true	Not very true at all
I could do physical activity even if I was <b>tired</b>		$\bigcirc$	0	
I could do physical activity even if I had other things I wanted to do		$\bigcirc$	$\bigcirc$	
I could do physical activity even if I had to <b>exercise on my own</b>		$\bigcirc$	0	
l could do physical activity even if I had <b>a</b> <b>bad day at school</b>	$\bigcirc$	$\bigcirc$	$\bigcirc$	$\bigcirc$
I could do physical activity even if I was feeling lazy		$\bigcirc$	0	
I could do physical activity even if I was <b>not very good at it</b>	$\bigcirc$	$\bigcirc$	$\bigcirc$	$\bigcirc$

18. Please tick **ONE box** for <u>each sentence</u> to show how true it is for you.

	No	Not sure	Yes
I could be physically active even if it is <b>very</b> <b>hot or cold outside</b>	$\bigcirc$	$\bigcirc$	$\bigcirc$
I have the skills I need to be physically active	$\bigcirc$	$\bigcirc$	$\bigcirc$
I can ask my best friend to be physically active with me	$\bigcirc$	$\bigcirc$	$\bigcirc$
I can ask my parents / other adults to do physically active things with me	$\bigcirc$	$\bigcirc$	$\bigcirc$
I can be physically active no matter how <b>busy</b> my day is	0	0	0

#### \* 19. Below are some statements how you see yourself physically

Please choose **ONE item per box for each sentence** to show how true it is for you (best describes you)

	Very true	Quite true	Not very true	Not very true at all
I am good enough at sports		$\bigcirc$		$\bigcirc$
I do very well at all kinds of sport	$\bigcirc$	$\bigcirc$	$\bigcirc$	$\bigcirc$
I am better than friends at sport	$\bigcirc$	$\bigcirc$	$\bigcirc$	$\bigcirc$
I like to play sports rather than watch	$\bigcirc$	$\bigcirc$	$\bigcirc$	0
I do well at any new sport	$\bigcirc$	$\bigcirc$	$\bigcirc$	$\bigcirc$
I am happy with how I am and what I can do physically		$\bigcirc$		0
I feel positive about myself physically		$\bigcirc$		$\bigcirc$
I feel confident about myself physically	$\bigcirc$	$\bigcirc$	$\bigcirc$	$\bigcirc$
I like to watch sports rather than play	$\bigcirc$	$\bigcirc$	$\bigcirc$	$\bigcirc$
I wish I could feel better about myself physically	$\bigcirc$	$\bigcirc$	0	0

#### \* 20. Below are some reasons <u>WHY</u> people do physical activity.

Please tick ONE box for each statement to show how true it is for you

#### Doing Physical Activity......

	Very true	Sort of true	Not very true	Not very true at all
is fun	$\bigcirc$	$\bigcirc$	$\bigcirc$	$\bigcirc$
Helps me make new friends	$\bigcirc$	$\bigcirc$	$\bigcirc$	$\bigcirc$
Gives me more energy			$\bigcirc$	$\bigcirc$
Teaches me about health and fitness	$\bigcirc$	$\bigcirc$	$\bigcirc$	$\bigcirc$
Keeps me from gaining weight	$\bigcirc$	$\bigcirc$	$\bigcirc$	$\bigcirc$
Makes me look better	$\bigcirc$	$\bigcirc$	$\bigcirc$	$\bigcirc$
Helps me feel better about myself	$\bigcirc$	$\bigcirc$	$\bigcirc$	$\bigcirc$
Makes me stronger			$\bigcirc$	$\bigcirc$
Helps me be healthy			$\bigcirc$	$\bigcirc$
Is good for my mental health	$\bigcirc$	$\bigcirc$	$\bigcirc$	$\bigcirc$
Helps me cope with stress	$\bigcirc$	$\bigcirc$	$\bigcirc$	$\bigcirc$

#### \* 22. ATTITUDES TO PHYSICAL ACTIVITY

This question asks you how you FEEL about physical activity.

For each statement, please tick the most appropriate response.

	Agree in a BIG way	Agree	Disagree	Disagree in a BIG way
Doing physical activity every day would be <b>fun</b>	$\bigcirc$	$\bigcirc$	$\bigcirc$	$\bigcirc$
Doing physical activity every day would be enjoyable	$\bigcirc$	$\bigcirc$	$\bigcirc$	$\bigcirc$
Doing physical activity every day would be good for me	$\bigcirc$		$\bigcirc$	$\bigcirc$
Doing physical activity every day would be important for me	$\bigcirc$		$\bigcirc$	$\bigcirc$
My <b>family</b> wants me to do physical activity every day	$\bigcirc$	$\bigcirc$	$\bigcirc$	$\bigcirc$
My <b>friends</b> want me to do physical activity every day	$\bigcirc$	$\bigcirc$	$\bigcirc$	$\bigcirc$
My teachers want me to do physical activity every day	$\bigcirc$	$\bigcirc$	$\bigcirc$	$\bigcirc$
My <b>family</b> do physical activity every day	$\bigcirc$	$\bigcirc$	$\bigcirc$	$\bigcirc$
My friends do physical activity every day	$\bigcirc$		$\bigcirc$	$\bigcirc$
My teachers do physical activity every day	$\bigcirc$		$\bigcirc$	$\bigcirc$
I could do physical activity every day if I really wanted to	$\bigcirc$	$\bigcirc$	$\bigcirc$	$\bigcirc$
I have the time to do physical activity every day if I really wanted	dt 🔿	$\bigcirc$	$\bigcirc$	$\bigcirc$
I live in a <b>place</b> which allows me to do physical activity every day if I wanted to	$\bigcirc$	$\bigcirc$	$\bigcirc$	0
I <b>plan</b> to do physical activity every day	$\bigcirc$		$\bigcirc$	$\bigcirc$

#### 23. During a TYPICAL week, how often: Please tick ONE box only

	None	Once	Sometimes	Almost every day	Every day
1. Do you <b>encourage</b> <b>your friends</b> to do physical activities or play sports?	0	0	0	0	0
2. Do your friends encourage you to do physical activites or play sports with you?	$\bigcirc$	$\bigcirc$	$\bigcirc$	$\bigcirc$	$\bigcirc$
<ol> <li>Do your friends do physical activites or play sports with you?</li> </ol>	$\bigcirc$	$\bigcirc$	$\bigcirc$	$\bigcirc$	$\bigcirc$
4. Do other kids <b>tease</b> <b>you</b> for not being good at physical activity or sports?	$\bigcirc$	$\bigcirc$	$\bigcirc$	$\bigcirc$	$\bigcirc$
5. Do friends <b>tell you</b> that you are doing well in physical activites or sports?		0	$\bigcirc$	$\bigcirc$	

#### $24. \mbox{ During a TYPICAL week, how often has someone in your house/member of your family:} (e.g.$

father, mother, sister, grandparent, or other relative)

None	Once	Sometimes	Almost every day	Every day
	$\bigcirc$	$\bigcirc$	0	
$\bigcirc$	$\bigcirc$	$\bigcirc$	$\bigcirc$	$\bigcirc$
	0	0	0	
$\bigcirc$	$\bigcirc$	$\bigcirc$	$\bigcirc$	$\bigcirc$
	0	0	0	
	None	None         Once           O         O           O         O           O         O           O         O           O         O           O         O           O         O           O         O           O         O           O         O	None         Sometimes           Once         Sometimes           Once         Once           Once         Once	NoneSometimesAlmost every dayOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOOO

\* 25.

The scale below (0-10) to tell how confident you are tocorrectly perform the following skills:



#### How good can I.....

	1	2	3	4	5	6	7	8	9	10
Run in a straight line					0	0	0			0
Skip in a straight line	$\bigcirc$									
Leap over a bean bag	$\bigcirc$	0	0	0	0	0	0	0	0	0
Gallop in a straight line	0	0	0	0	0	0	0	0	0	0

	1	2	3	4	5	6	7	8	9	10
Side shuffle (slide) in straight line	0	0	0	0	0	0	0	0	0	0
Jump in the air for height from standing still	0	0	0	0	0	0	0	0	0	0
Jump for distance from standing still	0	0	0	0	0	0	0	0	0	0
Throw a tennis ball overarm	0	0	0	0	0	0	0	0	0	0
Catch a tennis ball using two hands					0	0	0			0
<b>Kick</b> a ball placed in front of you on the ground	0	0	0	0	$\bigcirc$	$\bigcirc$	$\bigcirc$	0	0	0
Strike a non-moving ball placed in front of you at hip height with a bat	0	0	0	0	0	0	0	0	0	0
Bounce a ball with your hand four times in a row while standing	0	0	0	0	0	0	0	0	0	0
Balance on one foot	0	0	0	0	0	0	0	0	0	0
Hop three times with your right and left foot					0	0	0			0
Roll a ball underarm	0	0	0	0	0	0	0	0	0	0



#### **PREFERRED TYPES of physical activity**

\*\*Remember that physical activity means any activity that makes your heart beat faster and makes you get out of breath some of the time

# The questions on this pages are about the types of physical activity that YOU WOULD LIKE TO DO.

26. Would you like to do MORE team games if you could? Examples: basketball, football, hockey, rounder's, rugby, volleyball

	Vec
	163

No No

#### 27. WHAT is your FAVOURITE sport/physical activity ? (CHOOSE ONE)

#### 28. WHERE would you like to do sport/team games? (Click on all that apply)

School
 Club (without school)

Leisure Centre

Home

The outdoors

Swimming pool

\* 29. WHEN would you like to do team games? (Click on all that apply)



30. Do you play Ladies Gaelic Football with a sports club or in school?





31. Would you like to play Ladies Gaelic Football with a sports team in your community?

- Yes
- No

I already play with a sports club

APPENDIX D: FUNDAMENTAL MOVEMENT SKILL PERFORMANCE Criteria

(TGMD, TGMD-2, GET SKILLED GET ACTIVE)

# G4G FMS Skill Component Performance Criteria (7 Skills)

#### LOCOMOTOR SKILLS (X3)

#### Run [TGMD-2]

	Performance Criteria	Trial 1	Trial 2	Score
1.	Arms move in opposition to legs, elbows bent			
2.	Brief period where both feet are off the ground			
3.	Narrow foot placement landing on heel or toe (i.e., not flat footed)			
4.	Non-support leg bent approximately 90 degrees (i.e. close to the buttocks)			

#### Skip [TGMD]

	Performance Criteria	Trial 1	Trial 2	Score
1.	A rhythmical repetition of the step-hop on alternate			
	feet			
2.	Foot of non-support leg carried near surface during the			
	hop phase			
3.	Arms alternately moving in opposition to legs at about			
	the waist level			

## Vertical jump [TGMD]

	Performance Criteria	Trial 1	Trial 2	Score
1.	Eyes focused forward or upward throughout the jump			
2.	Crouch with knees bent. Arms behind the body			
3.	Forceful forward and upward swing of arms			
4.	Legs straighten in air			
5.	Land on balls of feet. Bend knees to absorb land			
6.	Controlled landing with < = 1 (no more than one) step in any direction			

### **OBJECT-CONTROL SKILLS (X3)**

Catch	TGMD-2]	Performance	Trial 1	Trial 2	Score
Criteria	a				
1.	Preparation phase where hands are in	n front of the			
	body and elbows are flexed				
2.	Arms extend while reaching for the b	all as it arrives			
3.	Ball is caught by hands only (2 hands)				

<b>Statio</b>	nary dribble [TGMD-2]	Performance	Trial 1	Trial 2	Score
Criteri	a				
1.	Contacts ball with one hand at abou	it the belt level			
2.	2. Pushes ball with fingertips (not a slap)				
3.	Ball contacts surface in front of or to foot on the preferred side	o the outside of			
4.	Maintains control of ball for <u>four co</u> without having to move the feet to	<u>nsecutive</u> bounces retrieve it.			

Kick [T	GMD-2] Performanc	e Trial 1	Trial 2	Score
Criteri	a			
1.	Rapid continuous approach to the ball			
2.	An elongated stride or leap immediately prior to bal	I		
3.	Non-kicking foot placed even with or slightly in back the ball	cof		
4.	Kicks ball with instep of preferred foot (shoelaces) o toe	ır		

# STABILITY SKILL (X1)

Static Balance [GET SKILLED GET ACTIVE] Per	formance	Trial 1	Trial 2	Score
Criteria				
1. Support leg still, foot flat on the ground				
2. Non-support leg bent, not touching the su	pport leg			
3. Head stable, eyes focused forward				
4. Trunk stable and upright				
5. No excessive arm movements				

## **APPENDIX E:**

# G4G DATA COLLECTION PROTOCOL MANUAL

# **Gaelic4Girls Data Collection**

# **Protocol Manual**

## Part 1: Fundamental Movement Skills

G4G Skills: Kick, catch, run, skip, bounce, balance, and vertical jump

Instruments: TGMD-2, TGMD, Get Skilled Get Active

#### **Equipment/resources:**

Cameras/iPad, tripods, cones, duct tape, soft soccer football, tennis ball, basketball, folder with master excel sheet, participant code stickers/list, and highlighter.

It is important to note that participants perform the skill 3 times (1 trial run and 2 actual times).

#### **Skill Component Performance Criteria**

## Run [TGMD-2]

(1) Arms move in opposition to legs, elbows bent

(2) Brief period where both feet are off the ground

(3) Narrow foot placement landing on heel or toe

(4) Non-support leg bent approximately 90 degrees

Skill	Materials	Directions	Performance Criteria
1 Run 60 feet of clear space. Place two cones 50 f		Place two cones 50 feet apart. Make	1. Arms move in opposition to legs, elbows bent
1. Null	and two cones	sure there is at least 8 to 10 feet of	2. Brief period where both feet are off the ground
	space beyond the second cone for a safe stopping distance. Tell the child		<ol> <li>Narrow foot placement landing on heel or toe (i.e. not flat footed)</li> </ol>
tr O	to run as fast as he or she can from one cone to the other when you say "Go." Repeat a second trial.	<ol> <li>Nonsupport leg bent approximately 90 degrees (i.e., close to buttocks)</li> </ol>	

#### Skill illustration



Script for Run: Run as fast as you can and stop at this cone!

## <mark>Skip [TGMD]</mark>

- (1) A rhythmical repetition of the step-hop on alternate feet
- (2) Foot of non-support leg carried near surface during the hop phase
- (3) Arms alternately moving in opposition to legs at about the waist level

## **Skill illustration**



**Script for Skip:** *Watch me skip ("Step, hop, step, hop") in a straight line from cone A to cone B* 

# Vertical jump

- (1) Eyes focused forward or upward throughout the jump
- (2) Crouch with knees bent. Arms behind the body
- (3) Forceful forward and upward swing of arms
- (4) Legs straighten in air
- (5) Land on balls of feet. Bend knees to absorb land
- (6) Controlled landing with no more than one step in any direction

## **Skill illustration**



Script for Vertical Jump: Watch me jump upwards as high as I can and land.

## Catch [TGMD-2]

(1) Preparation phase where hands are in front of the body and elbows are flexed

(2) Arms extend while reaching for the ball as it arrives

(3) Ball is caught by hands only

Skill	Materials	Directions	Performance Criteria
3. Catch	A 4-inch plastic ball, 15 feet of clear space, and tape	Mark off two lines 15 feet apart. The child stands on one line and the tosser on the other. Toss the ball underhand directly to the child with a slight arc aiming for his or her chest. Tell the child to catch the ball with both hands. Only count those tosses that are between the child's shoulders and belt. Repeat a second trial.	1. Preparation phase where hands are in front of the body and elbows are flexed
			<ol><li>Arms extend while reaching for the ball as it arrives</li></ol>
			3. Ball is caught by hands only

## **Skill illustration**



**Script for Catch:** Watch me catch the ball. Then I am going to throw the ball to you, and you are going to catch it.

## Stationary dribble [TGMD-2]

- (1) Contacts ball with one hand at about the belt level
- (2) Pushes ball with fingertips (not a slap)

(3) Ball contacts surface in front of or to the outside of foot on the preferred side

(4) Maintains control of ball for <u>four consecutive bounces</u> without having to move the feet to retrieve it.

Skill	Materials	Directions	Performance Criteria
2. Stationary Dribble	An 8- to 10-inch playground ball for children ages 3 to 5; a basketball for children ages 6 to 10; and a flat, hard surface	Tell the child to dribble the ball four times without moving his or her feet, using one hand, and then stop by catching the ball. Repeat a second trial.	1. Contacts ball with one hand at about belt level
			<ol> <li>Pushes ball with fingertips (not a siap)</li> <li>Ball contacts surface in front of or to the outside of foot on the preferred side</li> </ol>
			<ol> <li>Maintains control of ball for four consecutive bounces without having to move the feet to retrieve it</li> </ol>

## **Skill illustration**



Script for Dribble: Bounce the ball 4 times and catch!

# Kick [TGMD-2]

(1) Rapid continuous approach to the ball

(2) An elongated stride or leap immediately prior to ball contact

(3) Non-kicking foot placed even with or slightly in back of the ball

(4) Kicks ball with instep of preferred foot (shoelaces) or toe

Skill	Materials	Directions	Performance Criteria
4. Kick	An 8- to 10-inch plastic, playground, or soccer ball; a beanbag; 30 feet of clear space; and tape	Mark off one line 30 feet away from a wall and another line 20 feet from the wall. Place the ball on top of the bean- bag on the line nearest the wall. Tell the child to stand on the other line. Tell the child to run up and kick the ball hard toward the wall. Repeat a second trial.	1. Rapid continuous approach to the ball
			<ol> <li>An elongated stride or leap immediately prior to ball contact</li> </ol>
			<ol> <li>Nonkicking foot placed even with or slightly in back of the ball</li> </ol>
			<ol> <li>Kicks ball with instep of preferred foot (shoe- laces) or toe</li> </ol>



Script for Kick: Watch me run up and kick the ball!

## Static Balance

- (1) Support leg still, foot flat on the ground.
- (2) Non-support leg bent, not touching the support leg.
- (3) Head stable, eyes focused forward.
- (4) Trunk stable and upright.
- (5) No excessive arm movements.

## **Skill illustration**





Script for Balance: *Watch me balance on one foot for at least 4 seconds.* 

## **Scoring Directions**

The examiner enters a "1" for successful completion, or a "0 "for failed attempt, for each of the performance criterion listed for the skill. There are no partial marks. If the SP refuses to do any of the tests, the examiner should make a note "Did not Participate" on the sheet for that skill.

# Part 2: Height and Weight (BMI) Protocol

#### Height:

#### **Equipment:**

- Standard collapsible portable stadiometer
- Data collection sheets

#### Set-up:

- Construct stadiometer following instructions.
- Place it on a level, flat, hard surface with the stabilising bar against a vertical surface such as a wall or door.

#### Instructions to participant:

- 1. Remove shoes.
- 2. If the hairstyle affects their height, ask them to adjust it for the test.
- 3. Stand with heels and toes together on the base plate.
- 4. Arms loosely by their side.
- 5. Back straight against the vertical measuring rods.
- 6. Look straight ahead.
- 7. Take a deep breath and stand as straight as possible without their heels lifting off the ground.

Note: These can be difficult instruction for children/youth to follow – make sure the head is not tilted or the shoulders raised, breath normally. Check posture before measuring (body is in a straight line - mid-axillary line parallel to stadiometer).

## Record in metres to nearest 1/10<sup>th</sup> of a centimetre

## Weight:

#### Equipment:

• Standard portable calibrated scales.

• Data collection sheets.

#### Set-up:

- Ensure scales are pre-calibrated with a known weight.
- Place scales on a hard, level surface.
- To promote confidentiality of screening results, and reduce anxiety, all students should be facing away from the scales.
- Students BMI screening results recorded are strictly confidential and should not be discussed with anyone.

#### Instructions to participant:

- 1. Wear only light garments.
- 2. Remove items such as money from pockets.
- 3. Remove shoes.
- 4. Stand on the scales, with both feet fully on the weighing platform, heels towards the back edge, and their arms loosely by their side.
- 5. Remain as still as possible with their head facing forward.
- 6. Step down from the scale.

Record in kilograms to nearest 1/2 gram

Height and weight will be used to calculate Body Mass Index (BMI)

 $BMI = Weight (kg) / Height (m^2)$ 

# Part 3: Questionnaire Protocol

### PART A: Introduce selves

Hi guys, I'm \_\_\_\_ and this is \_\_\_ and \_. We're from UCC and we're completing a study with 8-12-year old girls to find out how active people your age are, why you like sport, what would stop you from playing etc.

# We will be giving you a numbered sticker. Please wear this sticker for the duration of the testing.

## PART B: Introduce questionnaire

1. The questionnaire will be introduced with the following instructions:

**a.** We have a questionnaire that we'd like you to complete on the computers/sheets. It takes around 45 minutes to do and is designed to gather information on sport and physical activity levels of 8-12-year old girls. Your responses are for research purposes only and will be treated in the strictest confidence.

**b.** If you are unsure about how to answer any questions please put your hand up and one of us (research assistants) will come to you. Do not ask the person next to you (as they may not know the answer). [All facilitators/research assistants will have name badges, so that the participants can call them for assistance if necessary]

**c.** While many of the questions may appear quite similar, there are subtle differences between them, and you should treat each one as a separate question.

**d.** The best approach is to answer each question fairly quickly and focus on each item separately.

e. It is important to answer ALL the questions.

**f.** Your answers are strictly confidential so try to answer all questions as honestly as you can. You will be assigned an ID code so it will be totally anonymous.

**g.** This is not a test; there is no pass/fail.

**2**. While the students are filling in the 1st page ask the teacher how far a kilometre is from the school to a local landmark (i.e., the local park, or shop).

**3.** If at any stage a student cannot continue filling in the questionnaire, as assent cannot be assumed. Please assign another task, the teacher may help with this.

# PART C: Read out definition of physical activity

Before we start the questionnaire, I'm just going to read out this definition of physical activity so that everyone is clear:

### 'Physical activity is any body movement.

It can be done at different levels of effort:

**Moderate Effort** makes your heart rate and breathing rate faster than normal. You may also sweat a little. Brisk walking and jogging are good examples.

**Vigorous Effort** makes your heart rate much faster and you have to breath deeper and faster than normal. You will probably sweat. Playing football or tennis are good examples.

Physical activity includes:

Exercises - Weight training, aerobics, jogging, dancing, etc.

Sports - Hurling, football, athletics, swimming, etc.

General - Brisk walking, washing the car, walking or cycling to school, etc.

For some of the questions, you will be asked to add up all the time you spend in physical activity each day. **Only include activities of either MODERATE or VIGOROUS effort'** 

## PART D: Remainder of the questionnaire – explain question types

There are a few different types of question in the questionnaire

- i. There are questions which ask you to select a box and others where you must click the circle depending on how much you agree or disagree/true of you or not.
- ii. A few contain text boxes for you to type your own answer. There's a slight delay on the buttons so type slowly and wait a second between each letter.
- iii. Some of the questions may ask you more personal information (e.g. your feelings, etc.).

**4.** After question 5, read out the definition of physical activity in the box on page 1. It is very important that the participants are clear on this definition.

**5**. Tell the students how far a kilometre is in relation to their school and a local landmark (questions 13 and 15).

**6**. Ask students to go to 'Statements about Physical Activity Section (do not fill out now):

**a.** When you are answering these questions you have to think about yourself and what applies to you.

**b.** Then you must decide, for each question, whether you agree a lot, sort of agree or disagree a lot with the physical activity statements.

**c.** Then you must decide whether you the statements are very true of you, quite true, or not very true at all and tick the one box that applies.

**d.** Then you must decide how often the statements relate to you (once, often, sometimes etc.) and tick the one box that applies.

e. Please tick only one box for each question.

7. Please go ahead and fill in the questionnaire. If you are unsure about any questions just put up your hand and we will come to you. And remember there is no right or wrong answers. When you are finished please raise your hand [and we will collect it off you].

**8**. Once a questionnaire is finished a researcher must check through it to make sure all questions have been answered and if not, get the student to fill it in unless they totally object to filling in a certain section (hardcopy).

9. If students refuse to take part in the study, take note of why and how many.

**10.** Ensure all questionnaires are collected and/or cleared from computers once finished.

## Part 4: Focus Group Interview Protocol

#### <u>Purpose</u>

- The purpose of focus group discussions (FGDs) is to obtain information about a group's beliefs and attitudes on a particular issue or problem.
- FGDs differ from individual interviews in that the discussion allows for interaction among all the members of the group.
- FGDs differ from surveys in that they permit participants to give detailed opinions on a topic.

#### **Description**
- Five to eight G4G participants will take part in each of the FGDs ranging in age from 8-12-years from 1<sup>st</sup> 6<sup>th</sup> class.
- The FGD will be led by the lead researcher. It is very useful to have another person (notetaker) present who takes notes but does not participate in the discussion.
- FGDs should be held in a location which provides as much privacy as possible and where participants are comfortable. A classroom or community hall can work well.
- Group should be set up so that all participants have eye contact with each other and can hear each other speak. Sitting in a circle on chairs is usually best.

#### **Ethical Considerations**

- Every individual has the right to refuse to participate in a focus group, or to stop her participation at any time. The focus group facilitator/lead researcher must respect this right.
- It is important to conduct focus groups in a manner that is comfortable for all participants, so they are able to speak openly and honestly.
- No identifying information should be kept in the notes or final report.

#### Focus Group Discussion Guide

- The FGD facilitator uses a discussion guide which describes the topics to be covered.
- The discussion guide may contain examples of follow-up or probe questions for the facilitators.
- In a FGD, the facilitator covers all of the topic areas in the discussion guide, but does not necessarily follow a particular sequence. The discussion should flow as naturally as possible and some topics may be raised by group members. The facilitator should follow the lead of the group members, probing relevant topics they raise during the discussion.
- The discussion guide should be pre-tested with a group of participants similar to the ones you will be talking to later.

#### Characteristics of a Facilitator

- A child-friendly demeanour, including patience, warmth, humour, and flexibility, is an ideal characteristic of a moderator (Kennedy et al., 2001).
- Facilitator should speak the same language as the participants. For children, concrete questions phrased in simple language should be used throughout the focus group
- Particularly, younger children may have difficulty understanding excessively general or vague questions. Questions starting with "what" or "how" are preferred to questions starting with "why" or to questions requiring "yes" or "no" answers. Using prompts such as "What does everyone else think?" "Do others have different thoughts?" "Tell me more" or "How interesting" can improve the flow of a discussion, especially with shy individuals (Lund et al., 2016).
- Expressions such as "Great!" "Terrific!" or "Cool!" should be avoided because they may discourage the child from telling the parts of the story that are less cool.
- Facilitator should have the ability to respect the dignity and confidentiality of respondents.
- Facilitator should have good skills in listening in a non-judgmental and unbiased way.

#### HOW TO CONDUCT A FOCUS GROUP DISCUSSION (FGD)

- Greet the participants and thank them for their time.
- Introduce yourself (facilitator) and the note-taker.
- Communicate duration of the FGD (typically 40-50 minutes).
- Highlight ethical considerations.
- Cover all topic areas in the discussion guide (may not necessarily follow a particular sequence).
- Ensure participation of all present.
- Probe only when necessary and do not try to 'put words in people's mouth'.
- If you are recording the discussion, obtain permission from the participants first and make sure your equipment works (always carry a back-up device).

- Reporting Immediately after each FGD, the facilitator and notetaker should meet to review the main themes of the discussion.
- FGD notes should be typed and made fair, removing all identifying information such as participants' names. Noting exact quotes is important as they can be powerful elements of the report.

**APPENDIX F:** 

BASELINE STUDY FOCUS GROUP QUESTIONS AND DRAWINGS (CHAPTER 5)







# G4Gs Focus Group Questions – Exploring student's perspectives on physical activity/sport

#### 12/05/16

'Gaelic4Girls' Project

\_\_\_\_\_

#### Protocol

- Each of the 6 focus groups (2 per school) is designed to last for 40/45 min max.
- Lead facilitator (Orlagh) will control the focus groups (FGs) discussing the questions with the participants.
- Assistant facilitator (3<sup>rd</sup> Year PE student) will listen to the discussion and take notes on each section (*Reflective Field Notes* on site). They will then provide a summary at the end of each section to ensure the correct information has been gathered by students and ask are there any further comments they would like to add.
- Each FG will be recorded with a Dictaphone AND an IPhone (to later revisit the data)
- 6 Students in total will be selected randomly from 4<sup>th</sup> 5<sup>th</sup> and 6<sup>th</sup> class.
- Focus Groups will take place in a separate classroom under supervision of a schoolteacher. Students will sit together in a circle with table and chairs.
- Lead facilitator will introduce FG and give a brief overview of the Research Project in a child friendly manner and will remind participants that it is safe environment and students should feel free and safe to say what they want to say. Lead facilitator will also remind students that there are no right or wrong answers, their names won't be identified, and they have the right to withdraw from the FG at any stage.

### Section 1 – Health-Related Physical Activity

-----

- Introductory Activity (Icebreaker) Students will be asked to draw a picture of what being healthy means to them? / 'Draw a picture of yourself at the weekends surrounded by all of your favourite physical activities and objects in this box'. (Provide colours/sheets)
- Create a ranked list of your favourite activities/sports that you play from 1-5

Q1. What do you think makes you healthy?

Q2. What do you think physical activity/exercise is?

**Q3**. Do you know how much physical activity per day you should be doing? Prompt: How many minutes

Q4. How do you think physical activity affects your health?

Explain what physical activity is/definition

Note-taker will summarise key words/themes/comments from this section & facilitator will ask do the group have further comments they would like to make?

-----

#### Section 2 (Motivators/Barriers/Preferred Activities)

-----

Q1. What are your favourite types of physical activity/sports?

Q2. What physical activities/sports do you take part in?

**Q3**. Where do you take part in these activities? Prompt: PE, after school club, local club (environmental)

Q5. What kind of things do you do when you play outside?

Q6. Who do you take part with?

**Q7.** When in the week do you do these activities Prompt: PE, after school, weekend Focus on Ladies Football is mentioned.

**Q8.** Why do you take part in physical activity/sport? Prompt: fun, fitness, make friends

**Q9.** Who or what helps you to take part in physical activity/sport? Prompt: family, friends, teachers, can afford it, facilities

Q10. Why don't you like taking part in physical activity/sport?

**Q11**. Is there anything stopping you from becoming more active than you already are? Prompt: lack of facilities, if you had more money, more support from others e.g. family, friends

Reinforce physical activity explanation here

Note-taker will summarise key words/themes/comments from this section & facilitator will ask do the group have further comments they would like to make?

```
_____
```

#### Section 3 (Enhancers)

-----

**Q1**. Is there anything that would help you to take part in more physical activity/sports?

Q2. Is there anything that would help you to enjoy physical activity/sport more?

**Q.3** Do you think some people have more opportunities to play sports than other people? Why might that be the case?

Q.4 Do some people give up sports? Why, in your opinion does this happen?

Q.5 What do you think we could do to get more people playing sports?

Q.6 If you could play any sports in the world what would they be and why?

Note-taker will summarise key words/themes/comments from this section & facilitator will ask do the group have further comments they would like to make?

-----

#### Section 4 (Ladies Gaelic Football)

- **Q1.** What do you know about LGF?
- **Q2.** Do you play LGF? (club/school)
- Q3. Do you like playing it? What's your favourite part?
- Q4. What don't you like about LGF?
- Q5. What would encourage you to play LGF?
- Q6. What would stop you from playing LGF?
- **Q7.** What do you think we could do to get more girls playing LGF?

Note-taker will summarise key words/themes/comments from this section & facilitator will ask do the group have further comments they would like to make?



**Focus Group Participant Drawings** 

Swimming Golfing Running 0 tennis xyeling hurling Gaeilie Football 20000 Ø



**APPENDIX G:** 

G4G INTERVENTION STUDY FOCUS GROUP QUESTIONS AND DRAWINGS (CHAPTER 7)







\_\_\_\_\_

### Focus Group Questions – Exploring girl's perspectives on Gaelic4Girls Intervention 2018

20/06/18

# **SECTION 1**

#### **Engagement** questions

#### (participants likes, dislikes & opinions on G4G programme)

-----

**Q1.** You all took part in the 8-week Gaelic4Girls programme. What did you enjoy most about the programme? <u>What</u> was your favourite thing about the G4G programme? <u>Why</u> did you like this?

**Q2.** What didn't you enjoy about the G4G programme? <u>What/why</u> made you not to like these things?

**Q3.** Did you learn anything new during the 10-week programme? If so, what did you learn?

**Q4.** Did you find anything in the programme very challenging? If so, what did you find challenging? **(barriers/motivators)** 

**Q5.** Is there anything you would change about the programme? If so, what and why? (length of programme etc.)







# **SECTION 2**

### **Exploration Questions**

#### (Impact of G4G programme on attitudes toward Health, PA, FMS, self-esteem)

\_\_\_\_\_

**Q1.** Do you think that your health has been affected over the past 10 weeks by taking part in G4G? If so, in what way? (Prompts - Has it made you more active/fit?)

**Q2.** After taking part in the programme, what do you think about sport and ladies' football?

**Q3.** Do you think you are more confident at performing the ladies football skills now than before the programme? Why so? What do you think about the skill cards? Did they encourage you to practice skills at home? (Competence, autonomy)

Q4. If you took part in the ladies' football dance, what did you enjoy most about the dance? Do you feel that taking part in the dance has made you more confident about yourself and practising ladies football skills? (Competence, autonomy, relatedness)

**Q.5** Do you want to continue playing ladies football after the G4G programme? Why? Do you think you will be playing ladies football this time next year with a club/school? Why do you think so? (Autonomy/ PA, Sport motivators)

### **SECTION 3**

#### (PA/Sport barriers)

**Q1**. Is there anything stopping you from becoming more active than you already are? Prompt: lack of facilities, if you had more money, more support from others e.g. family, friends

**Q2.** Why do you think some girls might stop playing/drop-out of sports teams? (What would you say to somebody who is about to give up sport?

**Q3.** If your friends decided to stop playing ladies football, would you stop playing too? Explain your answer **(relatedness)** 

#### **SECTION 4**

#### Exit Questions

#### (Possible improvements/future directions)

**Q.1** Is there anything different that you would like to see in the Gaelic4Girls programme? Why would you like to see this in the programme?

Q2. What could we do to get more girls playing and enjoying ladies' football?





#### **APPENDIX H:**

#### **EDITOR ACCEPTANCE NOTIFICATION OF PUBLISHED STUDIES**





[Sports] Manuscript ID: sports-218877; doi:10.3390/sports5040074. Paper has been published.

Dear Authors,

28-Sep-2017

We are pleased to inform you that your article "The Relationship between Actual Fundamental Motor Skill Proficiency, Perceived Motor Skill Confidence and Competence, and Physical Activity in 8–12-Year-Old Irish Female Youth" has been published in Sports and is available online:

Abstract: <u>http://www.mdpi.com/2075-4663/5/4/74/</u> PDF Version: <u>http://www.mdpi.com/2075-4663/5/4/74/pdf</u> Manuscript: <u>http://www.mdpi.com/2075-4663/5/4/74/manuscript</u> (available to authors after login)

The issue release date for your article is 2017-12-22.

Thank you for choosing Sports to publish your work, we look forward to receiving further contributions from your research group in the future.

Kind regards, Milica Laca Website Editor ---MDPI AG Postfach, CH - 4020 Basel, Switzerland Office: St. Alban-Anlage 66, 4052 Basel, Switzerland Tel. +41 61 683 77 34 Fax: +41 61 302 89 18 E-mail: <u>website@mdpi.com</u> <u>http://www.mdpi.com/</u>



#### Article



# The Relationship between Actual Fundamental Motor Skill Proficiency, Perceived Motor Skill Confidence and Competence, and Physical Activity in 8–12-Year-Old Irish Female Youth

Orlagh Farmer<sup>1</sup>, Sarahjane Belton<sup>2</sup> and Wesley O'Brien<sup>1,\*</sup>

- <sup>1</sup> School of Education, Sports Studies and Physical Education Department, 2 Lucan Place, Western Road, University College Cork, Cork T12 KX72, Ireland; 111524727@umail.ucc.ie
- <sup>2</sup> School of Health and Human Performance, Dublin City University, Dublin D09 W6Y4, Ireland; sarahjane.belton@dcu.ie
- \* Correspondence: wesley.obrien@ucc.ie, Tel.: +353-21-490-2319

Received: 1 August 2017; Accepted: 21 September 2017; Published: 27 September 2017

Abstract: This study examines the relationship between actual fundamental motor skill (FMS) proficiency, perceived motor confidence and competence, and physical activity (PA) among female children (n = 160; mean age =  $10.69 \pm 1.40$  years). The Test of Gross Motor Development-2nd Edition (TGMD-2) was used to assess seven FMSs (locomotor, object-control, and stability). Motor confidence and competence were assessed using a valid skill-specific scale, and a modified version of the Self-Perception Profile for Children. PA levels were assessed using self-report (PA Questionnaire for Older Children (PAQ-C)) and classified as low, moderate, and high active. One-way and two-way ANOVAs (post-hoc honest significant difference (HSD)) and correlation coefficients were used to analyse the data. Findings indicate that the majority of youth (71.8%) were not meeting the minimum 60 min of daily PA recommended for health, and that 98.1% did not achieve the FMS proficiency expected for their age. While there were high levels of perceived physical self-confidence (PSC) reported within FMS skill-specific tasks, there was no significant correlation observed between actual FMS proficiency and perceived PSC among the cohort. Results show that low, moderately, and highly active female participants differ significantly in terms of their overall FMS (p = 0.03) and locomotor (LOC) control scores (p = 0.03). Results from a two-way between-groups analysis of variance also revealed no statistically significant interaction effect between PA grouping and physical performance self-concept (PPSC) on overall FMS proficiency levels. Results of a multiple linear regression indicate that perceived PSC is a significant predictor (beta = 0.183) of participants' overall PA levels. Data show a need for targeting low levels of PA, and low FMS proficiency in female youth, and for developing interventions aiming to enhance perceived PSC levels.

**Keywords:** fundamental movement skills; perceived physical self-confidence; physical performance self-concept; physical activity; female youth



# Women in Sport and Physical Activity Journal - Decision on Manuscript ID WSPAJ.2017-0046.R3

12-Jul-2018

Dear Ms. Farmer:

It is a pleasure to accept your manuscript entitled "Enhancing the evidence-base for Irish female youth participation in physical activity – the development of the 'Gaelic for Girls' programme." in its current form for publication in the Women in Sport and Physical Activity Journal. Thank you for attending to all of the final comments and our editorial assistant, Christina Johnson, was able to convert your figures to .tif files.

Thank you for your fine contribution. On behalf of the Editors of the Women in Sport and Physical Activity Journal, we look forward to your continued contributions to the journal.

Sincerely, Lori Gano-Overway

Editor, Women in Sport and Physical Activity Journal

lgano@bridgewater.edu



# Enhancing the Evidence Base for Irish Female Youth Participation in Physical Activity—The Development of the Gaelic4Girls Program

#### Orlagh Farmer

University College Cork

#### Donna Duffy

The University of North Carolina

#### Kevin Cahill and Diarmuid Lester

University College Cork

#### Sarahjane Belton

**Dublin City University** 

#### Wesley O'Brien

University College Cork

The purpose of the current research was to gather baseline data on female youth to inform the development of a targeted physical activity (PA) and sports-based intervention, specifically identified as "Gaelic4Girls". Cross-sectional data on PA levels, psychological correlates of PA, anthropometric characteristics, and the fundamental movement skill (FMS) proficiency of female youth (n = 331; *M* age 10.92±1.22) were collected. A subsample (n = 37) participated in focus group (FG) interviews exploring perceptions of health/sport, and identifying barriers/motivators to participation. PA levels were assessed using self-report (PA Questionnaire for Older Children) and classified as low, moderate, and high active. One- and two-way ANOVAs (post hoc Tukey honest significant difference [HSD]) were used to analyze the data. The FGs were transcribed verbatim, coded, and thematically analyzed. Findings indicated that the majority of youth (71.8%) were not meeting the minimum daily PA recommendations for health, and that 98.1% did not achieve the FMS proficiency expected for their age. Low, moderately, and highly active participants differ significantly in terms of overall FMS (p = .03), and locomotor control scores (p = .03). FG findings report fun and friendship as key PA motivators, too much competitiveness as barriers, and positive outside encouragement from family/friends/coaches as facilitators encouraging PA engagement. Findings highlight the need for targeting low levels of PA, FMS proficiency in female youth sport interventions, through addressing self-efficacy levels, inclusive of fun, and socially-stimulating PA environments.

Keywords: fundamental movement skill, physical activity motivators and barriers, psychosocial physical activity correlates, sport participation





# [IJERPH] Manuscript ID: ijerph-911992; doi: 10.3390/ijerph17186928. Paper has been published

Dear Authors,

Please note that the PDF version of your recently published manuscript has been updated by the MDPI production team. You may download the PDF and Microsoft Word versions of your paper from the article webpage:

Abstract: https://www.mdpi.com/1660-4601/17/18/6928 PDF Version: https://www.mdpi.com/1660-4601/17/18/6928/pdf Manuscript: https://www.mdpi.com/1660-4601/17/18/6928/manuscript (available to authors after login) Special Issue: https://www.mdpi.com/journal/ijerph/special\_issues/promoting\_physical\_acativity

The issue release date for your article is 2020-09-25.

Kind regards,

--

#### MDPI

Postfach, CH - 4020 Basel, Switzerland Office: St. Alban-Anlage 66, 4052 Basel, Switzerland Tel. +41 61 683 77 34 Fax: +41 61 302 89 18 E-mail: website@mdpi.com https://www.mdpi.com/



International Journal of Environmental Research and Public Health



#### Article Gaelic4Girls—The Effectiveness of a 10-Week Multicomponent Community Sports-Based Physical Activity Intervention for 8 to 12-Year-Old Girls

Orlagh Farmer <sup>1,\*</sup>, Kevin Cahill <sup>2</sup> and Wesley O'Brien <sup>1</sup>

- <sup>1</sup> Sports Studies and Physical Education Program, School of Education, 2 Lucan Place, Western Road, 0000 Cork, Ireland; Wesley.obrien@ucc.ie
- <sup>2</sup> School of Education, Postgraduate Diploma in Special Educational Needs, University College Cork, 0000 Cork, Ireland; K.Cahill@ucc.ie
- \* Correspondence: o.farmer@umail.ucc.ie; Tel.: +353-21-490-2537

Received: 10 August 2020; Accepted: 16 September 2020; Published: 22 September 2020



Abstract: Girls are less active than boys throughout childhood and adolescence, with limited research focusing on female community sports-based programs. This study aims to assess the effectiveness of a multi-component, community sports-based intervention for increasing girl's physical activity (PA) levels, fundamental movement skill (FMS) proficiency, and psychological wellbeing, as relative to a second treatment group (the traditionally delivered national comparative program), and a third control group. One hundred and twenty female-only participants (mean age =  $10.75 \pm 1.44$  years), aged 8 to 12 years old from three Ladies Gaelic Football (LGF) community sports clubs (rural and suburban) were allocated to one of three conditions: (1) Intervention Group 1 (n = 43) received a novel, specifically tailored, research-informed Gaelic4Girls (G4G) intervention; (2) Intervention Group 2 (n = 44) used the traditionally delivered, national G4G program, as run by the Ladies Gaelic Football (LGF) Association of Ireland; and (3) Control Group 3 (n = 33) received no G4G intervention (group 1 or 2) conditions and were expected to carry out their usual LGF community sports activities. Primary outcome measurements (at both pre- and 10-week follow up) examining the effectiveness of the G4G intervention included (1) PA, (2) FMS and (3) Psychological correlates (enjoyment levels, self-efficacy, peer and parental support). Following a two (pre to post) by three (intervention group 1, intervention group 2, and control group 3) mixed-model ANOVA, it was highlighted that intervention group 1 significantly increased in PA (p = 0.003), FMS proficiency (p = 0.005) and several psychological correlates of PA  $(p \le 0.005)$ . The findings demonstrate that the 10-week, specifically tailored, research-informed G4G intervention is a feasible and efficacious program, leading to a positive effect on the physical and psychological wellbeing of pre-adolescent Irish girls, relative to the traditionally delivered national G4G comparative program and control group conditions.

**Keywords:** physical activity; fundamental movement skills; psychological correlates of physical activity; pre-adolescent girls; organized youth sport; multi-component intervention

**APPENDIX I:** 

G4G COACHING MANUAL (WEEKLY SESSIONS (X3), WORKSHOP OVERVIEW, AND ELECTRONIC LGF/FMS VIDEO CLIPS)







#### REGISTRATION

• There are currently close to 70 girls registered for Gaeilc4Girls and should be expecting another 30+ girls on Thursday also.

Registration will take place in small hall/dressing room?
 Table 1 → girls who have registered (signed the forms) already (List on Excel Sheet)
 Table 2 → girls who need to register

• Once kids come into hall mentors/coaches need to take the girls into dressing rooms and onto pitch ASAP.

### EQUIPMENT

- 6 station pitch layouts (see diagram below).
- Equipment will need to be set-up prior to 6pm (would advise for all coaches/mentors to be at pitch early (meet at 5.30pm if possible), particularly on first night to get the ball rolling.
- Equipment & coaches will stay at the same station  $\rightarrow$  the kids will be moving from station to station.

#### **PITCH LAYOUT**

- 1-2 pitches 6 stations on each pitch (circa 15/16 per station)
- Centre area for G4G Coordinator
- Coaches/coordinator will need to sort girls into groups according to age on pitch
- Get all the 8-year olds to stand behind the blue cone, 9-year olds behind the red cone, 10-year olds (yellow cone) etc. easier to divide them into their groups.



SESSION 1: GAELIC4GIRLS	SAAFE & SHAARP PRINCIPLES
WARM UP	
BIBS BULLDOG	
<ul> <li>Normal bulldog, except every player has a bib hanging from shorts.</li> <li>The girls that are on (3/4) have to try and pull their bib when passing over to other side, if the taggers grab their bibs they are now on with the taggers.</li> <li>Keep playing until 2/3 left with bib.</li> <li>*NOTE-&gt; players are not allowed hold onto their bibs or tuck them in shorts. Also, non-contact game</li> </ul>	<ul> <li>ACTIVE/STRETCHING WHILE MOVING (heart-raiser game)</li> <li>Reduce time at the cone/line less talk, blow whistle sooner</li> <li>HIGH REPETITION OF MOTOR SKILLS (running, agility)</li> <li>FAIR: Call if players run outside marked boundaries and/or cheating</li> <li>ENJOYABLE: FUN &amp; POSITIVE PRAISE</li> </ul>
STATION 1 – HANDPASSING – 10 mins (balls, bibs, cones)	
PAIR PASSING	
<ul> <li>Right hand left hand (go through technique) get them moving up on their feet</li> <li>Let them practice first and then go through technique – 2 hands, 1 hand follow through. Get them to say it all together out loud – 2 hands one hand follow through – eyes on the ball)</li> <li>Challenge/focus game – how many successful hand passes and catches can each pair get in a minute on both right/left hands?</li> </ul>	<ul> <li>ACTIVE: Get them moving up on their feet</li> <li>DIFFERENTIATION: 'can you beat your right/left hand score?'</li> <li>Mix teams by ability</li> <li>AUTONOMOUS (keep own scores)</li> <li>FUN</li> </ul>

	OBSERVE, DEMONSTRATE,     FEEDBACK & POSITIVE PRAISE
POSSESSION HAND PASSING GAME	
<ul> <li>2 grids – 4 teams (5 players max per team) – 2 teams per grid</li> <li>Hand passing only (no bounce or solo – take 4 steps)</li> <li>3 passes to score – once you have passed run and touch any of the four cones and back into grid.</li> </ul>	<ul> <li>ACTIVE/MAX PARTICIPATION: Small-sided games – more ball contact</li> <li>DIFFERENTIATION: Mix teams by ability</li> <li>OBSERVE, DEMONSTRATE, FEEDBACK &amp; POSITIVE PRAISE</li> </ul>
<b>STATION 2 – SOLOING – 10 mins</b> (10 balls, cones)	
SOLO GRID	
<ul> <li>In pairs, solo around grid with your partner in same direction (hand pass to your partner after soloing).</li> <li>On whistle change direction of run around grid.</li> <li>On coaches call ('cone') players must run with the balls and touch any of the cones around the grid.</li> <li>Divide the group into two teams &amp; they run around grid in opposite directions (traffic – forces them to look up).</li> </ul>	<ul> <li>SMALL-SIDED GAME – SMALL GROUPS – MAXIMUM PARTICIPATION</li> <li>SUPPORTIVE (positive praise)</li> <li>DIFFERENTIATION (mix groups by ability)</li> <li>REDUCING STATIC SOLOING (moving around grid constantly)</li> </ul>
SOLO TAG	

<ul> <li>Mark out a big grid</li> <li>2 teams (Red &amp; Blue)</li> <li>Team Blue start with a ball each soloing around the grid.</li> <li>On whistle team Red run into the grid and try to get the ball off the blue team</li> <li>If the blue team's ball is taken/stolen they must solo their ball around the outside grid until the game is over.</li> <li>Switch teams and repeat (time the teams the next time around &amp; see which team get all the balls off other team the quickest).</li> </ul>	<ul> <li>SMALL-SIDED GAME – SMALL GROUPS – MAXIMUM PARTICIPATION</li> <li>SUPPORTIVE (positive praise)</li> <li>DIFFERENTIATION (mix groups by ability)</li> <li>REDUCING STATIC SOLOING (moving around grid constantly)</li> </ul>
<b>STATION 3 – FMS FUN GAMES– 10 mins</b> (10 balls, cones)	
RATS AND RABBITS	
<ul> <li>Put girls in pairs standing across from each other, facing each other</li> <li>One line are rats, the other line are rabbits</li> <li>Coach calls out commands (E.G. HIGH KNEES, TURN AROUND, TOUCH YOUR TOES, HIGH FIVE YOUR PARTNER, JUMPING JACKS, TOUCH KNEES ETC.)</li> <li>If coach calls RATS – all the girls that are rats must turn and run to their rat's safety line without getting caught by their partner (Rabbit).</li> <li>If a rabbit catches a rat 0 the rat, then joins the rabbit line and becomes a rabbit.</li> <li>The line with most people in it wins</li> </ul>	<ul> <li>FUN/ENJOYABLE</li> <li>ACITVE (heart-raiser)</li> <li>SAFE (space)</li> <li>REPITITION OF MOTOR SKILLS</li> <li>POSITIVE PRAISE</li> <li>KEY TEACHING POINTS</li> </ul>

 Rats Safety Line	Rats     Rabbits       X ○     X ○	
<ul> <li>Put girls in pair's facing each pairs around to give them pl</li> <li>Coach calls out different con HIGH 5 YOUR PARTNER, JUN</li> <li>On whistle girls must try sna</li> <li>Whoever snatches the cone out of 5). Repeat if you have</li> <li>To note* - If pair grab the con They are only allowed to use over cone.</li> </ul>	other with a cone between them on the ground – spread the enty of space. mmands (E.G. HIGH KNEES, TURN AROUND, TOUCH YOUR TOES, IPING JACKS, TOUCH KNEES ETC.) tch their cone before their partner gets hand on it. first gets <i>one point</i> – tell the girls to keep their own score (best time and get them to change partners! one at the same time – they can award themselves a point each! e their hands to grab the cone (no diving/feet in) and no hoovering	<ul> <li>AUTONOMOUS (keeping own score)</li> <li>FUN/ENJOYABLE</li> <li>STRETCHING WHILE MOVING</li> <li>ACITVE (heart-raiser)</li> <li>SAFE (space) and FAIR (abiding by rules)</li> <li>REPETITION OF MOTOR SKILLS/KEY TEACHING POINTS</li> <li>POSITIVE PRAISE</li> </ul>

IE HUMAN KNOT'		
Put girls in Each playe hand. Next, have Have the st a circle. Players mu communica	groups of 5/6 and get them to stand in their own circle. nust raise their right hand, reach across the circle and grab some ach player do the same with their left hand. dents untangle themselves without letting go of hands – they sh not be holding the hands of the players next to them. Aim of the e and help each other. Repeat & change groups up!	<ul> <li>FAIR</li> <li>ENJOYABLE</li> <li>ACCESS THROUGH DIFFERENTIATION</li> <li>POSITIVE PRAISE</li> <li>SAFE</li> </ul>

COMMANDO'	
Commando (Balance, Coordination) Split the class into two / three teams. The aim is to reach the opposite side of the hall first. Each team stands in a line at one side of the hall with their legs apart. The person at the back then crawls underneath all of the pairs of legs and then becomes the person at the front. Once they have reached the front of the line the child then has to shout next so the person at the back knows when to start. Slowly but surely the line starts to move forward! Tip: make the children stand really close together as it's a little easier. The first team to the opposite end of the hall are the winners.	<ul> <li>PROMOTING PA AND TEAMWORK</li> <li>ENJOYABLE</li> <li>STRETCHING WHILST MOVING</li> <li>FAIR</li> </ul>
STATION 5 – MODIFIED GAMES – 10 mins (balls, cones/poles &	bibs)
MINI GAELIC FOOTBALL GAME	
<ul> <li>2 Mini Games (only 1 bounce and solo rule)</li> <li>Set up 2 mini pitches (cones/poles as goals and mark a side-line with cones)</li> <li>4 teams of 5 max (depends on numbers) in each group (team A, B, C &amp; D)</li> <li>2 games play at same time (3 minutes)</li> <li>Team A V B and Team C V D</li> <li>After 3 minutes change teams so that each team gets to play each other</li> </ul>	<ul> <li>MAXIMUM PARTICIPATION</li> <li>DIFFERENTIATION (mix teams by ability)</li> <li>FAIR (enact rules)</li> <li>ACTIVE</li> <li>POSTIVIE PRAISE</li> <li>KEY TEACHING POINTS</li> </ul>



#### **KICK PASS THROUGH HULA HOOP**

- In groups of 5's one girl holding hula hoop & 2 on either side of hula hoop
- Practice kick passing (chest height and below chest height) through the hula hoop to player on other side (in close to make it easy for them). Once the player kicks the ball through the hoop they run to other side of hoop.
- How many successful kicks can you get through the hula hoop as a team in one minute (without dropping the ball)



- AUTONOMY
- MAXIMUM PARTICIPATION
   (ball between two)
- REDUCING STANDING AND SITTING (promote players to move on feet in between passes)
- **DIFFERENTIATION** (change pairs after certain amount of kicks)
- POSITIVE PRAISE

VARM UP	
YNAMIC CIRCLE STRETCH	
<ul> <li>All players in circle around coach, players run in towards coach while doing dynamic movements stretches;</li> <li>High knees/ Heel flicks</li> <li>Side steps/ Running backwards</li> <li>Turn and sprint outwards</li> <li>Jumping/hopping on toes – on whistle sprint outwards</li> <li>Add your own stretches and/or get the girls to choose some exercises</li> </ul>	<ul> <li>ACTIVE/STRETCHING WHILE MOVING</li> <li>HIGH REPETITION OF MOTO SKILLS (running, agility, FMS)</li> <li>AUTONOMOUS (ask girls to choose the exercises)</li> <li>ENJOYABLE: FUN &amp; POSITIVE PRAISE</li> </ul>
TATION 1 – HANDPASSING – 10 mins (balls, bibs, cones, hula hoops)	

<ul> <li>Right hand left hand (go through technique) get them moving up on their feet</li> <li>Let them practice first and then go through technique – 2 hands, 1 hand follow through. Get them to say it all together out loud – 2 hands one hand follow through – eyes on the ball)</li> <li>In groups of 5's (one girl holding hula hoop)</li> <li>Challenge/focus game – how many successful hand passes and catches can each pair get in a minute on both right/left hands THROUGH THE HOOP?</li> </ul>	<ul> <li>ACTIVE: Get them moving up on their feet</li> <li>DIFFERENTIATION: 'can you beat your right/left hand score?'</li> <li>FUN AND FAIR</li> <li>OBSERVE, DEMONSTRATE, FEEDBACK &amp; POSITIVE PRAISE</li> </ul>
<ul> <li>2 grids – 4 teams (5 players max per team) – 2 teams per grid</li> <li>Hand passing only (no bounce or solo – take 4 steps)</li> <li>3 passes to score – once you have passed run and touch any of the four cones and back into grid.</li> </ul>	<ul> <li>ACTIVE: Get them moving up on their feet</li> <li>DIFFERENTIATION: 'can you beat your right hand score with your left hand?'</li> <li>Mix teams by ability</li> <li>FUN</li> <li>OBSERVE, DEMONSTRATE, FEEDBACK &amp; POSITIVE PRAISE</li> </ul>
<b>STATION 2 – SOLOING – 10 mins</b> (balls, cones)	
---	---
SOLO GRID	
<ul> <li>In pairs, solo around grid with your partner in same direction (hand pass to your partner after soloing).</li> <li>On whistle change direction of run around grid.</li> <li>On coaches call ('cone') players must run with the balls and touch any of the cones around the grid.</li> <li>Divide the group into two teams &amp; they run around grid in opposite directions (traffic – forces them to look up).</li> </ul>	<ul> <li>SMALL-SIDED GAME – SMALL GROUPS – MAXIMUM PARTICIPATION</li> <li>SUPPORTIVE (positive praise)</li> <li>DIFFERENTIATION (mix groups by ability)</li> <li>REDUCING STATIC SOLOING (moving around grid constantly)</li> </ul>
SOLO TURN	
<ul> <li>Set up as shown below</li> <li>Solo around player on opposite side and hand pass back to group (small groups)</li> <li>Max participation- 3 players on cone</li> <li>Challenge players (e.g. go for one min without dropping the ball/left hand only etc.)</li> </ul>	<ul> <li>MAXIMUM PARTICIPATION (3 per cone)</li> <li>SUPPORTIVE (positive praise)</li> <li>DIFFERENTIATION (mix groups by ability)</li> <li>ACTIVE</li> </ul>

TATION 3 – FMS FUN GAMES– 10 mins	(cones, ladders, balls)
Ladders cones burdles bula boons footballs)	• FUN/ENIOYABLE
<ul> <li>Ladders, cones, hurdles, hula hoops, footballs)</li> <li>Place all of these around a grid and get players to practice; <ul> <li>Running through the ladders</li> <li>Jumping over the hurdles</li> <li>Throwing football up in air and catching it</li> <li>Hopping on right leg at red cones</li> <li>Hopping on left leg at green cones</li> <li>Get girls to come up with their own exercise/circuit</li> </ul> </li> </ul>	<ul> <li>FON/ENJOYABLE</li> <li>ACITVE (heart-raiser)</li> <li>SAFE (space)</li> <li>REPITITION OF MOTOR SKILL</li> <li>POSITIVE PRAISE</li> <li>KEY TEACHING POINTS</li> <li>AUTONOMY (ask girls to com up with their own exercises)</li> </ul>

<b>STATION 4 – TEAM CHALLENGES - 10 mins</b> (cones, hula hoops)	
'HULA HOOP CHALLENGE'	
<ul> <li>5 girls in circle holding hands</li> <li>Aim of game is to try and get the hula hoop around circle without breaking hands (the link) → stress to them that the team that works and communicates the best with each other will succeed – try and take focus off winning and promote teamwork</li> <li>Can then get half the group to face the other half and have race from one end to top of line to get hula hoop up without breaking chain of hands</li> </ul>	<ul> <li>FAIR</li> <li>ENJOYABLE</li> <li>ACCESS THROUGH DIFFERENTIATION</li> <li>POSITIVE PRAISE</li> <li>SAFE</li> <li>PSYCHOSOCIAL (communication, teamwork, get the girls to come up with their own conclusions)</li> </ul>
BALL BALANCE	
<ul> <li>In group of 5 place footballs between each player (see photo below) and walk towards cones marked</li> <li>If ball drops all players have to go back to the start and start again</li> <li>Stress TEAMWORK &amp; COMMUNICATION – if person at front goes too fast or person at back goes too slow the ball will drop.</li> </ul>	<ul> <li>PROMOTING TEAMWORK</li> <li>ENJOYABLE</li> <li>STRETCHING WHILST MOVING</li> <li>FAIR</li> <li>AUTONOMOUS</li> </ul>

<ul> <li>Eliminate element of competition by saying 'the team that works the best together and talks and supports each other will get to the other side without dropping the ball – just like in a match you have to work together.</li> </ul> <b>STATION 5 – MODIFIED GAMES – 10 mins</b> (balls, cones/poles & bib)	S)
3V3 BACKS AND FORWARDS GAME	
<ul> <li>3 attackers on blue starting next to goal line</li> <li>3 defenders on red as shown in diagram</li> <li>On coaches whistle 3 attackers with one ball run around blue cone and try to score a goal (1 point if score)</li> <li>AT THE SAME TIME 3 defenders run around their red cone and come out to meet the attackers and defend them</li> <li>1 point if tackle successfully and run back out past red cone with ball</li> <li>Set up two games</li> </ul>	<ul> <li>MAXIMUM PARTICIPATION</li> <li>DIFFERENTIATION (mix teams by ability)</li> <li>FAIR (enact rules)</li> <li>ACTIVE</li> <li>POSTIVIE PRAISE</li> <li>KEY TEACHING POINTS</li> </ul>

Attackers start   here with one ball     Defenders   start here	
STATION 6 – ATTACKING/DEFENDING/SHOOTING - 10 mins(baTHE NUMBERS GAME	lls, cones, hula hoops)
<ul> <li>In pairs defenders stand behind pair on red cones</li> <li>Attackers stand in front on green cone</li> <li>On coaches whistle 3 attackers with one ball run around blue cone and try to score a goal (1 point if score)</li> <li>AT THE SAME TIME 3 defenders run around their red cone and come out to meet the attackers and defend them (1 point if tackle successfully and run back out past red cone with ball)</li> </ul>	<ul> <li>MAXIMUM PARTICIPATION/ REDUCING STANDING AND SITTING (several numbers called at once)</li> <li>DIFFERENTIATION (change pairs after certain amount of calls – players get a chance to be both attackers and defenders)</li> </ul>









STATION 4 – TEAM CHALLENGES - 10 mins (cones)	
DRAGON'S TAIL	
In groups of 3 children stand in a line, one behind the other, with their arms on the person in front's shoulders. The person at the front will be catcher, and the person at the end is the tail. The tail of the dragon places a bib in the shorts or tee shirt to create a tail. In their threes the front two can try and steal the person at the backs tail. Alternatively the team of 3 can work together to steal other dragon's tails (other teams) whilst protecting their own tail. This game highlights the importance of team work!	<ul> <li>ENJOYABLE</li> <li>SAFE</li> <li>PSYCHOSOCIAL (communication, teamwork)</li> </ul>
DOMES AND DISHES	
	PROMOTING TEAMWORK
	ENJOYABLE
	STRETCHING WHILST MOVING     EAIR
	AUTONOMOUS



## **STATION 6 – TACKLING - 10 mins**

## (balls, cones, balls)

#### **CHANNEL TACKLING**

- Set up like diagram with one player in the middle at poles
- Players at either side have to try get through the poles by soloing and bouncing ball. Player in the middel tries to tackle and dispossess ball.
- Change player n the middle after 4 tackles.
- EMPHASISE the near hand tackle and tackling stance (not flat footed and square) → up on toes and ready to turn right or left.



- MAXIMUM PARTICIPATION/ REDUCING STANDING AND SITTING
- DIFFERENTIATION (change pairs after certain amount of calls – players get a chance to be both attackers and defenders)
- CONSTRUCTIVE FEEDBACK, TEACHING POINTS, POSITIVE PRAISE

# GAELIC4GIRLS COACH REFLECTIVE JOURNAL

SESSION 1					D	ATE	1			
<ol> <li>What worked well? (content/delivery/girl's reactions/structure etc.)</li> </ol>										
2. What didn't work well?										
3. What could be improved for next week's session? (organisational/content/structure etc.?)										
4. Rate your own coaching performance out of	1	2	2	3 4	15	6	7	8	9	10
<b>10</b> (0 = not confident, 10 being very confident/efficient) – circle the number most relevant)										
5. Any other comments or thoughts?										
Staged and deferred reflective practice (Huntley et al., 2014)										







## **GAELIC4GIRLS WEEKLY COACH WORKSHOP FORMAT**

1. WELCOME (5 mins)

2. REFLECTIVE PRACTICE (20 mins)

Discuss 'previous' Gaelic4Girls session (share thoughts as a group)

What went well? What didn't go well? How can we improve? What can we change?

COACH INSIGHT AND INPUT

3. SESSION CONTENT EXPLANATION & DEMONSTRATION (30 mins) equipment/electronic videos etc.

4. Q&A with COACHES, COORDINATOR and LEAD RESEARCHER (5 mins) any other questions/comments.

# SAMPLE G4G ELECTRONIC COACHING CLIPS (FMS AND LADIES FOOTBALL GAMES)

















G4G Electronic Coaching Clips



















Gaelic4Girls Dance





Gaelic4Girls 2017





#### **APPENDIX J:**

## G4G Skill Cards, Poster and Certificate

#### G4G Weekly Skill Cards



#### G4G Skill Poster/Chart



## G4G Certificate of Completion (Week 8)



## APPENDIX K: G4G PARENTS EVENING CONTENT AND PRESENTATION EVENING FLYER































GO WATCH GAMES AND SUPPORT CLUB/COUNTY TEAMS PRAISE & POSITIVE REINFORCEMENT BE A ROLE-MODEL BE A ROLE-MODEL CO-PARTICIPATION REDUCING TECHNOLOGY TIME



### **G4G Presentation Evening Flyer**

